A SIMULATION ANALYSIS OF POLICIES FOR THE NORTHERN COLOMBIA BEEF CATTLE INDUSTRY

By

Alvaro Posada

A DISSERTATION

Submitted to Michigan State University in partial fulfillment of the requirements for the degree of

DOCTOR OF PHILOSOPHY

Department of Agricultural Economics

ABSTRACT

A SIMULATION ANALYSIS OF POLICIES FOR THE NORTHERN COLOMBIA BEEF CATTLE INDUSTRY

By

Alvaro Posada

The Atlantic Coast of northern Colombia (known as the Costa) supports between 40 and 50 percent of Colombia's cattle population and, with easy access to domestic and world markets, is the most important of Colombia's five beef-producing regions. Because cattle raising is the main economic activity in the . Costa and is an extensive operation with low technical efficiency, the region has been a priority target for cattle development programs. In the mid-1960s, with the financial and technical assistance of several international agencies, the Colombian government started a cattle development program aimed at increasing beef production mainly on the Atlantic In the early 1970s this program was reinforced with Coast. a disease control program and then revised and issued as a national cattle development plan. The main instruments of this plan are credit, technical assistance, export subsidies and improved marketing and slaughtering facilities. \mathtt{Its} long-term objectives are to increase the protein supply to the Colombian population and to generate foreign exchange earnings.

Alvaro Posada

The primary purpose of this study was to develop a system simulation model to (1) analyze the effects of production incentives on the decision of farmers to adopt new production methods, and (2) estimate the effects of the expanded regional production on the income of farmers, government revenues, Colombian beef consumption and sustained level of exports. Four alternatives to traditional production were considered. Alternative 1 considered the improvement of native and artificial grasses; alternative 2 considered the improvement of artificial grasses and the substitution of artificial for native grasses; alternatives 3 and 4 added the production of forages and silage to the improvement of range lands in alternatives 1 and 2 respectively. At the present stage of the study, however, alternative 2 was the only one comprehensively tested and used as a base run for policy experimentation. The cattle system simulation model has five major components (including a cattle demography model) which (1) allocate land use according to the farmer's perceived profitabilities of cattle and crops subject to land and capital constraints; (2) calculate yield and output of cattle and crops and their respective producer and market prices; (3) provide the instrumental linkages for government revenue, export trade policies, and production campaign policies; and (4) generate the performance criteria necessary to evaluate the impacts of alternative programs on the cattle economy through time.

Alvaro Posada

The five major sets of assumptions investigated were (1) disease control in the traditional herd, (2) alternative cattle industry taxing policies, (3) alternative development credit policies, (4) alternative levels of government production campaign promotion, and (5) alternative cattle pricing and export policies. The results of the cattle policy experiments were discussed in terms of the projected time paths (from 1966 to 1985) of five of the most important performance indices incorporated in the model: (1) regional cattle population, (2) Colombian beef consumption per capita, (3) regional farm income from cattle, (4) capitalized grazing land value per hectare, and (5) annual regional government revenue from cattle. Experiments with disease control and export promotion policies each used two indices instead of the above five: regional cattle population and extraction ratio for the disease control policies and domestic market price of finished males and export margin for the export policies.

In general, the study demonstrated that (1) the projected outcomes with the government disease control campaign were greater than under precampaign practices in the traditional herd; (2) the projected outcomes with government programs easing development loan terms were in all cases greater than the base run which assumed current credit policies; (3) the projected area in improved land and the modern cattle population with government policies benefiting both the traditional and modern operations were in all cases lower than under policies benefiting only the modern

Alvaro Posada

operation; (4) the projected area in improved land with the increased land tax rate was greater than the base run which assumed current land tax rates; (5) the projected outcomes with the removal of special taxes on cattle were lower than the base run which assumed no removal of these taxes; (6) given the assumptions on farmers' decisions and accounting mechanisms in the model, availability of credit for land improvement does not seem to be a serious constraint to land modernization; and (7) the projected outcomes with a flexible exchange rate suggest that this is an effective incentive to export without involving large transfers from public revenues to exporters in the form of subsidies.

The study indicated areas where more research and regional data are needed to improve the model's performance, and discussed possible extensions that could help analyze more fully alternative policy strategies for the Costa's overall development. Finally, the study demonstrated that the system simulation approach with a computerized model of the cattle economy which incorporated information from diverse sources and accounted explicitly for the dynamic interactions and feedbacks that might occur can be a very useful methodological tool for policy analysis. То

Maria Lucia and

Clara Lucia

ACKNOWLEDGMENTS

I express my gratitude to Professor Glenn Johnson for his guidance throughout my entire graduate program and for his encouragement. Special thanks are due to Dr. Michael Abkin for his assistance and critical yet perceptive comments since the early stages of this study.

I also want to thank Drs. Thomas Manetsch, Marvin Hayenga and Leonard Kyle for their help as members of my thesis committee. I am thankful to Chris Wolf for writing the computer program; to Judy Pardee, Enid Maitland and Patti Stiffler for typing earlier drafts of the thesis; and to Addiann Hinds for her editorial assistance.

I am indebted to the Rockefeller Foundation for financially supporting my studies at Michigan State University. The U. S. Agency for International Development (under Contract AID/csd-2975) was also very helpful providing financial aid for the final stages of my research.

A special debt is due to my wife, Maria Lucia, for her love, patience and support during this period of mutual knowledge enrichment.

iii

TABLE, OF CONTENTS

		Page
LIST	OF TABLES	.viii
LIST	OF FIGURES	. x
PART	IINTRODUCTION	. 1 . 1
	CHAPTER 1SCOPE AND NATURE OF STUDY Agriculture in the Colombian Economy Need for This Study General Systems Simulation Approach as a Tool for Beef Policy Analysis Purpose and Objectives	· 3 · 3 · 5 · 7 · 14
	CHAPTER 2A GENERAL DESCRIPTION OF THE COLOMBIAN CATTLE INDUSTRY	. 15 . 15 . 18 . 28
	Cattle Industry	30 30 33 34 37 38 39
	CHAPTER 3THE REGIONAL SETTING OF THIS STUDY The Geopolitical Setting	41 42 45 45 57 64

Page

	CHAPTER 4GENERAL SPECIFICATIONS	_
	AND PROCEDURE	67
	Area of Study	67
	Farming Sectors	67
	Paraling Dectors	70
	Maloning fidebices	72
	Modern Alternatives	73
	Static Restrictions	' <u>71</u>
	·	
		•
PART	IIMODEL DESCRIPTION	- 0
	Introduction	78
	CHAPTER 5-LAND ALLOCATION AND MODERNIZATION	
	DECISIONSCROPS/CATTLE (LAMDAC)	. 80
		. 80
	Land Uses	82
	Availability of Agricultural Dana	85
	Food Crops	. 0) 87
	Cash Crops	. 07
	Pasture and Cattle	. 00
	Alternatives	• 95
	Cattle Modernization Decisions	. 96
	Profitabilities	• 97
	Promotion and Diffusion	. 104
	Transition Responses	. 105
	Cottle Transfers	. 120
		• =
	CHAPTER OAGRICULTURAL PRODUCTION	126
	CROPS/CATTLE (AGPRAC)	126
	Crop Yields	120
	Pasture Production	. 129
	Cattle Production	. 13/
	Demography	. 137
	Milk and Animals Output	. 150
	Market Model	. 155
	Disease Control	. 159
	Agricultural Accounting	. 163
	CHARTER 7 PRICE GENERATION (PG)	. 184
	Export and Market Price of Cattle	184
	Broducen Brides and Bride Averages	187
	reducer rrices and rrice workges	• ••
	ANADRED & DOLLATES FOR THE CATTREE THOUSERSY	102
	CHAPTER OFOLICIES FOR THE CATLE INDUSTRI .	102
		• 100
	rroduction Campaigns	• 722
		· 195
	Export Policies	. 190
	CHAPTER 9ACCOUNTING AND PERFORMANCE	
	CRITERIA (CRTACC)	. 197

	Dudget Accounting	107
	Budget Accounting	• 191
	Development Credit	. 200
	Commercial Credit	. 201
	Aggregated Income and Consumption	. 204
	Capital Formation and Export Incentives	. 208
	Performance Criteria	. 215
שמע	TTT. VALTDATION AND TESTING	
LUL	Trime Justica AND IESTING	221
		• 221
	· · · ·	× .
		000
	CHAPTER 1.0DATA USAGE AND MODEL TUNING	. 223
,	Model Data Requirements	. 225
	System Parameters	. 225
	Technological Coefficients	. 229
	Initial Conditions	. 230
	Tuning	231
	Consistency Checks	220
		· 207
	General validation	. 241
		- 1
	CHAPTER 11RESULTS OF SENSITIVITY ANALYSIS .	. 246
	Analysis of Results	. 247
	Methodology	. 247
	Summary	. 258
		-
	TU-MODEL APPLICATION IN DECISION MAKING	
TAUT	Introduction	261
		. 201
	CHAPTER 12POLICY EXPERIMENTS ON THE	
	NORTHERN COLOMBIA BEEF INDUSTRY	. 262
	Policy Experimentation	. 263
	Run Definitions and Organization .	. 264
	Policies Related to Disease	
	Control	. 268
	Policies Related to Cattle and	
	Land Taxog	273
	Delicier Deletec in Dependion	. 213
	Policies Related to Promotion	285
	and Development Credit	. 205
	Policies Related to Domestic	
	Supply	. 290
	Policies Related to Crop	
	Modernization	. 300
	Effects of Various Policy	-
	Combinations	. 315
	Policies Related to Export	
	Promotion	. 310
		- 221
		• 334

Page

Page

CHAPTER 13SUMMARY AND CONCLUSIONS	10 10 12
Experiments on the Costa Cattle Economy	16
Model	19
Extensions	54
APPENDIX. Computer Program	50
BTBLTOGRAPHY 4	20

LIST OF TABLES

Table		Page
I.1.	Cattle Numbers in the National Herd and Export of Live Animals, 1960-74 ('000 head)	. 19
I.2.	Costa-Land Classes and Recommended Use by Departments ('000 has.)	• 53
I.3.	CostaDistribution of Farms According to Size, 1960	. 54
I.4.	Cattle Numbers in the Costa and Non- Costa Herd, 1968 ('000 head)	. 58
I.5.	Surplus and Deficit Relationships in Beef Production Among Regions, Colombia, 1969-1972	. 60
I.6.	CostaCattle Distribution According to Size of Ranch, 1960	. 62
II.1.	Selected Coefficients and Initial Values in the Land Allocation and Modernization Decisions Component (LAMDAC)	n . 123
II.2.	Average Annual Yield and Initial Costs by Farming Sectors	. 128
II.3.	Maximum and Minimum Proportions of Cattle Sales	. 154
II.4.	Perceived Changes in Cattle Output and Costs During the Planning Horizon	. 177
II.5.	Selected Coefficients and Initial Values in the Agricultural Production Component (AGPRAC)	. 181
11.6.	Land in Crops, Average Yields and Producer Prices In the Costa, 1960	. 188
11.7.	Selected Coefficients and Initial Values in the Price Generation Component (PG)	. 191

Table	1	Page
II.8.	Selected Coefficients and Initial Values in the Accounting and Performance Criteria Component (CRTACC)	219
III.l.	Profitability Response Parameters for Traditional and Modern Grazing (Dimensionless)	227
III.2.	Cattle Production Parameters	228
III.3.	Pasture and Crop Yields (tons/ha-year)	232
III.4.	Mean Length of Cattle Production Stages (years)	233
IÌI.5.	Selected Initial Conditions (1960)	233
III.6.	Time Series Tracking	237
III.7.	Results of Consistency Checks	240
111.8.	Selected Results of Alternatives to Traditional Cattle Production1985	243
III.9.	Results of Sensitivity Tests on Selected Parameters of the Costa Model	250
IV.1.	Policy Simulation Runs	265

LIST OF FIGURES

Figure			Page
I.1.	Colombiathe five beef cattle producing regions	•	17
I.2.	Ecological zones of Northern Colombia	•	46
I.3.	Costageneral soil groups	•	49
I.4.	Hypothetical cross-section of alluvial valleys in Northern Colombia	•	50
I.5.	Agricultural regions of Northern Colombia	•	68
I.6.	A functional flow diagram of cattle production	•	71
I.7.	Building blocks of the Costa beef model	•	75
II.1.	The profitability response function	•	106
II.2.	The land "dropout" function	•	115
II.3.	The reversion response function	•	116
II.4.	Cattle production cohorts	•	138
II.5.a.	Traditional birth rate versus total digestible nutrients	•	142
II.5.b.	Modern birth rate versus total digestible nutrients	•	142
II.5.c.	Traditional death rate of growing cohort versus total digestible nutrients	•	143
II.5.a.	Modern death rate of growing cohort versus total digestible nutrients	•	143
II.5.e.	Traditional death rate of producing cohort versus total digestible nutrients	•	144

Page

Figure		F	age
II.5.f.	Modern death rate of producing cohort versus total digestible nutrients	•	1 44
II.6.	The cattle sales function	•	153
II .7 .	The on-farm resource use response function	•	172
² 11.8.	Promotion and development credit profile	•	195
III.1.	Results of "coarse" model tuning cattle population in the Costa and Colombian beef supply time series against simulated series	•	235
III.2.	Results of "coarse" model tuningmarket price of finished males and land in crops time series against simulated series	•	236
IV.1.	Cattle population in the Costa with and without a disease control program, 1965-1985	•	270
IV.2.	Extraction ratio from the Costa cattle herd with and without a disease control program, 1965-1985	•	2 7 1
IV.3.	Cattle population in the Costa under various taxing policies, 1965-1985	•	280
IV.4.	Beef consumption per capita of the Colombian population under various taxing policies, 1965-1985	•	281
IV.5.	Aggregated farm income from cattle production in the Costa under various taxing policies, 1965-1985	•	282
10.6.	Average per hectare capitalized value of grazing land in the Costa under various taxing policies, 1965-1985	•	283
IV.7.	Annual government revenues from cattle production in the Costa under various taxing policies, 1965-1985	•	284
IV.8.	Cattle population in the Costa under various promotion and credit policies, 1965-1985	•	291

Figure	Э
--------	---

	Page
ન	•

IV.9.	Beef consumption per capita of the Colombian population under various promotion and credit policies, 1965-1985
IV.10.	Aggregated farm income from cattle production in the Costa under various promotion and credit policies, 1965-1985
IV.11.	Average per hectare capitalized value of grazing land in the Costa under various promotion and credit policies, 1965-1985
IV.12.	Annual government revenues from cattle production in the Costa under various promotion and credit policies, 1965-1985
IV.13.	Cattle population in the Costa and the rest of Colombia under various domestic supply policies, 1965-1985
IV.14.	Beef consumption per capita of the Colombian population under various domestic supply policies, 1965-1985 302
IV.15.	Aggregated farm income from cattle production in the Costa under various domestic supply policies, 1965-1985 303
IV.16.	Average per hectare capitalized value of grazing land in the Costa under various domestic supply policies, 1965-1985 304
IV.17.	Annual government revenues from cattle production in the Costa under various domestic supply policies, 1965-1985305
IV.18.	Cattle population in the Costa with and without a crop modernization program, 1965-1985
IV.19.	Beef consumption per capita of the •Colombian population with and without a crop modernization program 1965- 1985

Figure		Page
`IV.20.	Aggregated farm income from cattle production in the Costa with and without a crop modernization program, 1965-1985	. 312
IV.21.	Average per hectare capitalized value of grazing land in the Costa with and without a crop modernization program, 1965-1985	. 313
ĬV.22.	Annual government revenues from cattle production in the Costa with and without a crop modernization program, 1965-1985	. 314
ÛV.23.	Cattle population in the Costa under various policy conditions, 1965- 1985	. 320
·IV.24.	Beef consumption per capita of the Colombian population under various policy conditions, 1965-1985	. 321
IV.25.	Aggregated farm income from cattle production in the Costa under various policy conditions, 1965-1985	. 322
IV.26.	Average per hectare capitalized value of grazing land in the Costa under various policy conditions, 1965- 1985	• 323
IV.27.	Annual government revenues from cattle production in the Costa under various policy conditions, 1965-1985	. 324
IV.28.	Domestic market price of finished males under various policy alternatives, 1965-1985	. 330
IV.29.	Competitive position of Colombian cattle in export markets under various policy alternatives assuming high export target and moderate and low world beef prices, 1965-1985	221
		• ```))T

x111

PART I

INTRODUCTION

The Problem

Although planning for development has been practiced in Colombia for over thirty years, it has played a small role in the preparation of economic policy, and the various plans have been conceived more as political than as economic documents.

Plans have been criticized as being essentially technocratic exercises; the general public has contributed little to plan objectives and serious intention to implement has been lacking. As a consequence, development plans have enjoyed minimal general support and have had little or no effect on changing the country's economic, social and political structures [26].

The general systems simulation technique, with its special approach to analyzing the problems of development, could be helpful in solving the issues of feasibility, credibility, and general acceptance of the planning exercise. Yet its effectiveness as a tool for development will depend on the will of Colombian authorities to provide the necessary financial and institutional support for fulfillment of the plan's goals. In Part I, Chapter 1 discusses the scope and procedure of the study and briefly outlines the "systems approach" used. Chapter 2 is a brief descriptive account of Colombian beef production, distribution and consumption that will help clarify the problems of the cattle industry. Chapter 3 describes the region studied and the characteristics of its agricultural production. Chapter 4 discusses the modeling specifications and procedure.

Part II details the Northern or Costa model. Chapters 5 through 9 cover the land allocation and modernization decisions component, the agricultural production component, the price generation component, policy entry points and the criteria and general accounting component.

Part III looks at testing and validation procedures and results. Chapter 10 discusses data needs and the processes of tuning the model to track time series of recorded behavior. The results and implications of sensitivity tests on model parameters are presented in Chapter 11.

Part IV discusses policy applications of the model, conclusions and areas for further work. Chapter 12 presents the results and analyses of runs experimenting with various cattle development policy options. Experiments include an investigation of the sensitivity of policy results to changes in certain parameter values. Finally, Chapter 13 presents summary and conclusions, and outlines areas for further work in refining, improving and extending the model.

CHAPTER 1

SCOPE AND NATURE OF STUDY

Agriculture in the Colombian Economy

Although agriculture is Colombia's main economic activity, its rate of growth during the last two decades has been lower than that of the gross domestic product (GDP). In 1969 it contributed 30 percent of GDP but its share in the national output has been declining as industrialization has proceeded. Nevertheless, agriculture continues to be the main source of employment with over half the Colombian people directly depending on it for their living.

Within the agricultural sector, livestock production occupies about 87 percent of all agricultural land, accounting for about one-third of agricultural output, or approximately 10 percent of GDP. Beef is the primary product.

But despite the agricultural industry's importance, the production of basic food crops has barely kept pace with a 3 percent population growth rate. Cattle slaughter per 1000 inhabitants has been declining since 1950.

Colombia's economic growth has been responsive to changes in the performance of the export sector and this has been dominated by agricultural exports which accounted for 78 percent of total foreign earnings in 1970. While

coffee has remained the country's major export, its share of the total value of exports has declined from 72 percent in 1960 to 61 percent in 1970. As a proportion of all agricultural exports it has declined from over 90 percent prior to 1965 to 75 percent in 1969 [41]. But Colombia's dependence on agricultural exports which have unstable world markets (e.g., coffee, bananas, sugar and cotton) has undesirable disequilibrating effects which jeopardize development efforts and create the necessity of finding new sources of foreign revenues. The development of the beef industry, for which the outlook in world markets is considered brilliant, will accomplish the aims of increasing the domestic supply of protein for an improved diet and of helping remove both the foreign exchange and instability constraints.

Yet in order to fully exploit the natural comparative advantage Colombia has for cattle raising and make it a leading industry that is competitive in world markets, a great effort has to be made to overcome the traditionalism that has characterized the industry and to supply the necessary inputs for modernization.

In recent years the Colombian government has revised its policies toward the cattle industry and reoriented them toward the attainment of increased beef production. These policies have been associated with credit and technical assistance, disease control, land ownership rights, taxation and export subsidies.

The Colombian government has given priority in the cattle development program to the Atlantic plain of Northern Colombia because of that region's favorable natural conditions, its accessibility to domestic and foreign markets, and the fact that it has the largest share of the country's cattle population. This also supports the decision to focus the present study in this geographical region.

Need for This Study

Since cattle production is not merely an important economic activity but the only practical use for millions of hectares of agricultural land not suitable for crop production because of soil conditions, climate, floodings and/or distance from markets, its performance is and will be an important factor in the success of Colombia's efforts to foster sound economic growth in agriculture.

Because of its size, probably future demands for its products and the need for improved operation, the Atlantic plain beef production system's performance has had and will have a significant impact on Colombia's agricultural economy.

Several studies have been done on various aspects of the Colombian beef production and distribution systems. General descriptions of the industry and analyses of current and proposed policies have been done by Riley [61], the Food and Agriculture Organization (FAO) [66], the Caja Agraria [6], the World Bank [42] and the Instituto Colombiano Agropecuario [31]. Production problems and projections have been recently analyzed by Henning [29], Von Oven [60], and

Atkinson [67]. Bowser [5] attempted to make production projections by regions and establish surplus and deficit areas under two systems of management. Daines [68] is attempting to incorporate the cattle subsector into a broader sector analysis of Colombian agriculture.

More specialized studies on diseases and reproduction problems have been done by the Instituto Colombiano Agropecuario (ICA) and Gomez, respectively [31, 24]. Slaughtering, marketing and opportunities for exporting have been studied by Anderson [2], Booz, Allen and Hamilton [4], Secretaria de Agricultura de Antioquia [64], and Garcia Samper [23].

Profitability studies have been done by the World Bank [42, 43], Federacion Colombiana de Ganaderos (FEDEGAN) [20], and Instituto Colombiano de la Reforma Agraria (INCORA) [34]. More recently, the Centro Internacional de Agricultura Tropical (CIAT) [9] made a survey of the cattle industry in Northern Colombia in an attempt to gather basic information and identify specific problems which are in need of further research.

These have mostly been descriptive studies, and when projections are included they are trend-like, straight forward, algebraic estimates. The credibility of these estimates has always been questioned because of their reliance on timeseries data which deserve a low degree of confidence. Yet they have served the purpose of providing background information about the cattle industry and a basis, though weak, for planning its development. But except for the Bowser, FEDEGAN,

and the CIAT studies, no attempt has been made to place the cattle industry in a regional context and assess the effect through time of alternative strategies of development on the attainment of a multiplicity of objectives such as employment, farm income, government revenues, foreign exchange earnings, and others without neglecting the interactions with other subsectors of the agricultural economy.

This dissertation is an attempt to integrate the available information into a computerized model that will provide the policy maker with a more informed basis for planning development strategies for the Colombian beef production system on the basis of the learning experience from the Northern region. The basic parameters and structural relationships estimated in this study can be utilized in the future for modeling the cattle industry in other Colombian regions, and/or for developing a broader regional model of Northern Colombia.

This study has gained from experience with other simulators of cattle population and related activities developed for Nigeria, Korea, Northeast Brazil and Venezuela [53, 62, 51, 55]. The cattle population simulator developed by the World Bank [43] also has provided an invaluable experience.

<u>General Systems Simulation Approach as a Tool for</u> Beef Policy Analysis

In recent years there has been an increasing interest in the utilization of the systems approach for analyzing

complex developmental problems. Computerized techniques have helped automate the hand calculation process and expand the range of alternatives which can be examined. Policy makers and researchers have been placing more confidence and credibility on the general system simulation technique as its methodology and approaches for development have been better understood and developed [25, 46, 47]. Public and private decision makers have been presented with an approach that attempts to build a general model to trace the consequences through time of following alternative courses of action based on at least as wide a range of kinds and sources of data and information as decision makers use without specializing in any one technique to the exclusion of techniques frequently used by decision makers. In addition, the approach carefully avoids premature application of maximization techniques in situations where decision makers realize that the multiplicity of goods sought and bads avoided has not yet been reduced to a common denominator. These characteristics make the generalized, systems-science simulation approach very similar, though more comprehensive and complex, to the descriptive and paper-and-pencil and desk-calculator projections which have maintained a high level of credibility among decision makers [47].

Researchers have been provided with a technique for analyzing the problems of development without the methodological and theoretical restrictions of more specialized techniques such as simultaneous equilibrium equations,

linear programming (LP), benefit/cost ratio analyses, internal rate of return analyses, etc. $\frac{1}{}$ And as new concepts and experiences regarding the formulation and implementation of systems approaches have evolved, many of the early objections of statisticians concerning the reliability of the estimates of parameters, criterion variables, and prescriptions by users of the systems analysis are being dissipated. The analogy between the general systems analysis simulation approach and the Bayesian approach to inference has been demonstrated by Ladipo [50], and Johnson [47, 49] has analyzed repeatedly the possibility of empirically validating and verifying the normative concepts involved in simulation models.

Econometric methods with probabilistically estimated parameters rely heavily on time-series and cross-sectional data not always available in developing countries. These techniques also tend to be specialized on linear equations and behavioral assumptions involving maximization in accordance with neoclassical theories of the firm and household; the economic forces that link the various components of an economy are assumed to be self-equilibrating as a consequence of the maximizing activities for entrepreneurs and consumers. However, the validity of these two assumptions has been challenged where the findings of these kinds of studies were to

 $\frac{1}{For}$ a more detailed discussion of various specialized techniques see Manetsch, <u>et al.</u>, [53]. For particular applications to the Nigerian cocoa industry see Chong [10].

21 116

be used for policy analysis and prescriptions [53]. These and other methodological difficulties often result in less reliable parameter estimates than available from alternative kinds and sources of data and information with less sophisticated estimation and approximation techniques.

Linear programming and benefit/cost ratios and internal rates of return are other specialized techniques which have gained considerable prominence in recent years. LP is always used in an optimizing mode in solving the problems of resource allocation and policy analysis. The other two techniques have been used more for project analysis than for analyzing alternative policies and programs. Basically, all these techniques suffer from a need for a common denominator among the goods being sought and the bads being avoided. Moreover, the approach is quite mechanistic and may not allow rigorous analysis or interaction between researchers and policy makers needed for a better understanding and improving of the system.

A consideration of the distinctive attributes of the various approaches employed for studying development alternatives in a variety of less developed settings has led to the present proposal to use the system simulation approach as a tool for development planning and policy analysis of the cattle industry in Northern Colombia.

The general systems simulation approach, following the principles of the scientific method and problem-solving research, is viewed as an iterative problem-solving process that includes problem formulation, mathematical modeling, refinement and testing of a computerized approximation of the mathematical model, and creative design and execution of simulation experiments intended to provide answers to the questions being asked by the decision makers involved. $\frac{1}{2}$

The general system simulation approach has been specially applicable for solving many of the problems of economic development which do not meet the requirements for applying the simple maximizing computations of static production, consumption and welfare economics. As decision makers seek so many different goods and avoid so many different bads in developing the economy, it is very difficult for them or anyone to find a common denominator to be maximized or minimized. Problems of ordering and imperfect knowledge about future consequences of present actions complicate the circumstances in which decisions are made. Yet reaching prescriptive conclusions to solve development problems requires development of positive and normative knowledge. The methodology used in the general system simulation approach allows the system analyst to maintain a philosophic orientation sufficiently flexible to permit analysis of questions involving both the normative and the nonnormative values always present when considering the goals of economic development.

 $[\]frac{1}{\text{For more detailed discussion of the philosophy and}}$ methodology of the general system simulation approach for problem-solving research, see Manetsch et al., [53], Abkin [1], and Rossmiller et al., [62].

There are four distinguishing features of the general system simulation approach particularly useful for the policy analysis of the Colombian beef economy. First, it is a generalized approach which makes use of a wide range of primary and secondary informatical from many sources including surveys, government records, experiments, business records, qualitative judgments and insights of subject matter experts, and descriptive work about the beef industry from various disciplines. Also a wide variety of specialized techniques are used from econometrics, linear programming, partial budget, project analysis, etc. Since the research and model-building process is iterative and flexible, new information can be incorporated easily as it becomes available, and the structure of the model modified accordingly.

Second, the system approach can incorporate many types of functional relationships into the model to closely reflect the current or potential real system. These include dynamic interactions, curvilinearities, discontinuities, time lags, probabilities and irreversibilities.

Third, the approach does not have to assume (but does not preclude) any profit or utility maximizing producers and consumers, or any self-equilibrating economic adjustments. It does not necessarily involve a unique set of optimizing solutions based upon a common objective or a predetermined singular goal, which does not exist in reality. In contrast, the approach is more guided by the problem under investigation.

Fourth, the system simulation approach provides an experimental setting for exploring the consequences of a wide range of alternative plans or management strategies that ultimately will assist policy makers in determining the best course of action. Decision makers may be shown the consequences of alternative courses of action in terms of what goods or bads will be received by or imposed upon groups of people, when, and in what quantities. After such projections are available, interaction among investigators and policy makers will lead to a better understanding of the trade-offs among the numerous goods and bads involved in the solution of the problem. Developing, extending and refining knowledge on the various goods and bads and learning about the trade-offs is a way of solving the problem of finding an interpersonally valid common denominator. As stated earlier, this problem has contested the usefulness of the approach used by some quantitative techniques of analysis in examining the problems of economic development.

Further, in the iterative process, decision makers and investigators can work interactively to solve the remaining two major theoretical difficulties found in the analysis of development when using some other problem-solving techniques. First, the sequence in which different action programs should be undertaken can be established, thus resolving the problem of how programs and projects (actions) can be ranked. Then, when consequences of alternative decision rules can be projected and studied, it is possible to develop

a basis for choosing the best rule among the alternative courses of action being considered. This solves the planners' problem of choosing a decision-making rule, especially under conditions of imperfect knowledge and uncertainty.

Purpose and Objectives

The purpose of this thesis is to develop a model to help evaluate policy decisions that might be made in expanding the production of beef in Northern Colombia through time. More specifically, the objectives are:

- To develop a credible simulator that could eventually be extended to other beef producing regions and be further integrated into a national model.
- 2) To use the simulator
 - a) to analyze the effect of new production methods on the output of cattle;
 - b) to analyze the effect of production incentives on the decision of farmers to adopt the new methods;
 - c) to estimate the effect of the expanded regional production on the income of farmers, government revenues, domestic beef consumption and sustained level of exports.

The procedures used in meeting these objectives will be discussed in Chapter 4.

CHAPTER 2

A GENERAL DESCRIPTION OF THE COLOMBIAN CATTLE INDUSTRY

The characteristics of cattle raising in Colombia described in this chapter will help in understanding the multiple problems affecting the industry.

Stock farming in Colombia is carried on in a variety of climates and ecological zones that give rise to a wide range of problems which limit beef yields and supply. The principal limiting factors are the heavy incidence of animal diseases, malnutrition, deficient marketing and slaughtering systems. Besides the technical factors, government decisions that affect the political, social, and economic environment also have a major effect on the industry's behavior.

Size and Location of Cattle Industry

Of the 114 million hectares in Colombia, only about 40 million are suitable for crop and livestock production; the remainder is under forest or is wasteland. Cropland occupies approximately 5 million hectares, which leaves 35 million hectares under livestock. Even if crop acreage increases at 7 percent per annum during this decade, which would be very rapid growth, there still would be 30 million hectares in 1980 with no practical alternative use but grazing. The nation has the choice of producing cattle on this

land or letting it go idle. Credit and managerial manpower are the only scarce inputs used by the livestock subsector; thus cattle production is not currently a competitor with crops for scarce land.

Although statistical data are not extremely reliable, Colombia supports about 20 million head of cattle, including slightly more than 17 million beef animals. The rate of increase in cattle numbers has been low, about 2 percent per annum over the period 1950 to 1969. Since 1956, the rate has been close to 3 percent [41]. Although Table I.1 shows that herd numbers increased at about 4 percent per annum over the period 1965 to 1969, ICA [31] has projected an average rate of growth of 3 percent annually for the next five years.

The majority of beef cattle are maintained in tropical zones which have been divided into five clearly differentiated producing areas (Figure I.1). $\frac{1}{}$

 The Atlantic Coast at an altitude of between 0 and 500 meters includes Cordoba, Bolivar, Atlantico, Sucre, Cesar, Magdalena, Guajira, and Northern Antioquia. Average temperature exceeds 24°C and annual rainfall varies between 250 mm and about 2000 mm. Cattle population is approximately 7.6 million head and area in pasture 9.7 million hectares.2/

 $\frac{2}{\text{Cattle population and area in pasture for regions l,}}$ 2, and 3 are taken from Caja Agraria [6]; for region 4, from Bowser, <u>ibid</u>., 684; and for region 5 from ICA [31].

<u>l</u>/Max F. Bowser, "Prerequisitos y Potencial para la Exportacion de Carne en Colombia en la Decada de 1970," <u>Agricultura Tropical</u>, XXV (Bogota, Nov., 1969), 679.



FIGURE I.1. Colombia--the five beef cattle producing regions.

- 2) The Central and Upper Magdalena Valley at an altitude of less than 1000 meters includes Central and Southern Antioquia, Eastern Caldas, Tolima, Huila, the Santanderes, Central and Western Boyaca and Cundinamarca. Average temperature exceeds 24°C and average rainfall varies between 2000 and 4000 mm. Cattle population is approximately 4.3 million head and area in pasture is 5.6 million hectares.
- 3) The Cauca Valley at an altitude of less than 1100 meters includes Valle, Cauca, and parts of Caldas, Risaralda and Quindio. Average temperature exceeds 24°C and annual rainfall varies between 1000 and 2000 mm. Cattle population is approximately 1.25 million head and area in pasture is 1.2 million hectares.
- 4) The eastern plains at an altitude between 500 and 1000 meters include Meta, Eastern Cundinamarca, Eastern Boyaca (Casanare), Arauca, Vichada and Guainia. Average temperature exceeds 24°C and annual rainfall varies between 2000 and 4000 mm. Cattle population is approximately 1.5 million head and area in pasture is 16 million hectares.
- 5) The South, at an altitude of less than 1000 meters includes South Eastern Nariño, Caqueta, Putumayo, Amazonas and Vaupes. Average temperature exceds 24°C and annual rainfall varies between 1000 mm and 4000 mm. Cattle population is approximately .69 million head and area in pasture is 3.5 million hectares.

Production and Marketing Systems

Cattle production in Colombia is an extensive operation depending almost exclusively upon pastures as a source of feed inputs. Limited amounts of feed concentrates are being used in the more intensive dairy operations, and there are a few cases of confined feeding of steers.

For the nation as a whole, the average carrying capacity is about 0.57 head per hectare, but there are wide differences in capacity among the various classes of pastures
No. of Change in Regis	tered orts
Year Cattle Inventory Exp	
1960^{a} $15,329$ 529.0 1961 $15,679$ 350.0 1962 $15,979$ 300.0 1963 $16,279$ 300.0 1964 $16,584$ 305.0 1965 $16,882$ 298.0 1966 $17,372$ 490.0 1966 $17,372$ 490.0 1966 $17,372$ 490.0 1966 $17,372$ 490.0 1968 $18,830$ 748.0 1970^{c} $19,742$ 166.0 1971 $20,334$ 592.0 1971 $20,334$ 592.0 1973 $21,573$ 579.0 1974 $22,328$ 755.0	.1 ^b / .0 .4 .8 .1 .3 .8 .7 .0 .0 .0

TABLE I.1. Cattle Numbers in the National Herd and Export of Live Animals, 1960-74. (1000 head)

<u>a</u>/ Cattle inventories between 1960 and 1969 are from World Bank Report [41].

b/ Exports between 1964 and 1971 are from Sarmiento [63].

c/ Cattle inventories between 1970 and 1974 are projections by ICA [31].

<u>d</u>/ Exports between 1972 and 1974 are low target projections by Instituto de Comercio Exterior, INCOMEX [35]. and regions. The artificial pastures which make up about "one-third to two-fifths of the total pasture area [61, 66] have the greatest carrying capacity: about 2.0 to 2.5 head per hectare under good management. These pastures have a carrying capacity 3.5 to 4.5 times greater than that of natural pastures.

The highest stocking rates are in the Costa departments: 1.35 to 1.45 head per hectare; some of the lowest rates are in the Eastern Plains (Los Llanos) where a breeding cow and her calf are carried on up to 10 hectares. In the Magdalena and Cauca Valley regions the stocking rate varies from 0.74 head to 1.00 head per hectare.

The national beef herd has been derived from "Criollo" breeds which still account for approximately 20 percent of the total. The remainder have been upgraded from "Criollo" by Cebu (mostly U. S. type Brahma) for up to three or four generations. The hybrid vigor of the first crosses and the adaptability of the Cebu to tropical conditions contributed to the popularity of this breeding practice.

The size of producing units varies widely although accurate statistics on herd size distribution do not exist. Large units exist in the Atlantic Coast, Eastern Plains and South regions while in the mountain areas of Central Colombia there are many small units with less than 10 head of cattle per farm. Sixteen departments surveyed by the Departamento Administrativo Nacional de Estadistica (DANE) [14] in 1960 had 98 percent of the cattle in herds of less than 200 head.

As various measures of productivity show, the technical efficiency of the Colombian cattle industry is low. The "extraction rate," which is the proportion of the cattle inventory extracted for domestic slaughter and exportation, is approximately 13 percent annually. This extraction rate compares with 40 percent in the United States, 29 percent in Australia and 24 percent in Argentina.

The calving rate (number of calves born as a percent of females of breeding age) is undoubtedly very low. The estimates of the national average calving rate range from 50 to 70 percent as compared to 80 to 90 percent in countries with well-developed cattle industries.

Death losses are relatively high, averaging at least 8 percent a year for all cattle. Losses are greatest among calves where mortality rates are often 20 to 30 percent or more during the first few months after the calves are born.

The average growth rate is very low; slaughter age is about 4 years, although some of the better growers now market steers at 3 years of age. This is still high compared to the average marketing age of 1.5 to 3 years common for slaughter steers in the United States. Steers from La Costa are slaughtered at about 450 kg. live weight while Los Llanos steers average 390 kg.

Yield per animal slaughtered, in relation to the live weight of the animal, is barely 50 percent, as compared with 58 to 60 percent in countries where types specially bred for the production of beef are prevalent. Beef yields per hectare

and per animal are also low when compared with other countries. Von Oven [60], reported live weight yields per hectare of 62 and ll kilograms for the Costa and Eastern Plains, respectively, as compared with 90 kilograms for the Buenos Aires province in Argentina. Live weight yields per animal unit were 83, 40 and 117 kilograms for the Costa, the Easter Plains and the Buenos Aires province, respectively.

Mortality and performance at all stages of growth are affected materially by inadequate health control measures. Major diseases or parasites which cause mortality or losses through the falling-off in production among the animals affected are endemic foot-and-mouth (types A and O), rabies, anthrax, brucellosis, septaecemia, ticks and tick-born fevers, black leg, screw worms, and a great variety of internal parasites. Although effective control measures that could be applied in Colombia exist for most of these diseases, treatments are not a common practice.

Since it first appeared in Colombia in 1950, foot-andmouth disease has caused significant losses that have been estimated by the Instituto Colombiano Agropecaurio--ICA-at Ps. 332 million annually [31]. These losses are produced by death, reduced weight, retarded maturity, reduced milk production, and culled animals because of severe health injuries. Furthermore, foot-and-mouth disease constitutes an obstacle to trade between affected and immune areas, and precludes livestock and meat exports to countries which are free from the disease or on the way to eradicating it. Brucellosis, or infectious abortion, is next to footand-mouth, among infectious and/or contagious diseases, in causing the heaviest livestock losses. Brucellosis affects about one-fifth of the cow population (1,136,000 head), and about 2 percent of the stud bulls (9,600 head) [31]. Losses in 1967 were estimated at Ps. 177.5 million and consisted of some 136,000 miscarriages, permanent sterility in about 22,700 cows, impossibility of using 9,600 sick bulls, deaths of about 5,500 cows, and permanent loss of milk in the affected cows.

Losses due to parasites are probably equal to or greater than estimated losses due to disease. In many instances an animal may be sufficiently weakened by parasites to readily succumb to identifiable diseases. The incidence of internal parasitosis is enormous, especially among calves; losses may run as high as 15 to 25 percent. External parasitosis is a disease almost entirely confined to animals in the subtropical and tropical zones, where it affects 75 percent of the stock. External parasites cause heavy losses by retarding growth, raising the mortality index and damaging the hides.

Cattle production and yields are also limited by problems of nutrition. Seasonal fodder shortages coupled with deficiencies of minerals and vitamins lead to the diminution of milk and beef yields, to retarded growth and to death in some cases. Furthermore, the reproductive functions are affected, sometimes so seriously that the

animals become infertile or fecundity is reduced; and this in turn greatly lowers the birth rate. Gomez [24] reported up to 20 percent of cows between 2 and over 10 years of age as having permanent infertility, with trophic problems (associated with nutrition) as responsible for 85 percent of the cases.

In addition to seasonal shortages, forage production is aggravated by the underdiversification of pastures and the absence of satisfactory rotation practices. Little attention is devoted to the management and care of pastures and they often deteriorate greatly.

Obsolete and even primitive practices which prevail in many stock farming activities are responsible for the majority of drawbacks and deficiencies found in cattle production. Most stock farmers are slow to adopt new techniques, and absenteeism on the part of landowners aggravates herd mismanagement and intensifies the managers' and herdsmens' tendencies to stick to traditional routine practices.

But husbandry deficiencies are not the only example of poor management. Most ranchers do not keep accounting and production records and have scanty, imperfect knowledge of supply and demand trends as well as of the market situation.

Defective management and extensive methods offer few opportunities for employment and higher salaries. Labor intensity in cattle production is 3 man-days per head-year or 1.71 man-days per hectare-year as compared with 50 to

65 man-days per hectare-year for most annual crops and with 120 to 300 for most vegetable and fruit crops [69].

Daines [70] gives another measure of the low labor intensity and high investment requirements to generate new employment in livestock. In livestock $\frac{1}{2}$ labor is 1.0 to 4.3 percent of total costs as compared with 20 to 68 percent for most crops. And it takes from 6,300 to 26,255 dollars of investment to generate a new direct man-year of employment in livestock as compared with 300 to 3,270 dollars for most crops.

The failure of the supply of livestock commodities to react to the high demand elasticity by which they are characterized is attributable not only to production difficulties but also to the problems created by current marketing systems. The deficiencies affecting the rounding-up and transport of livestock, as well as slaughter and beef distribution, are manifold.

The marketing system is extremely fragmented; many small buyers and commission agents serve the ranchers, and there are many slaughterhouses, many agents placing meat in slaughterhouses, and many small stores selling meat. Most animals are bought on the farm, and are usually purchased with little consideration of weight or quality. Cattle are shipped directly to "ferias" or stockyards, which are located throughout Colombia. Medellin is the most important market and often sets the price standard for the country.

 $\frac{1}{1}$ Includes wool, eggs, poultry, pork and beef.

The high cost of transporting livestock has been and will continue to be one of the most serious marketing problems confronting the industry. Serious losses of weight occur during on-the-hoof movements. Cattle trailed for long days--from the Llanos for instance--lose up to 15 percent of their weight, in addition to which mortality must be taken into account. Severe weight losses are also registered in animals taken by boat; in some cases the time between the departure from the farms, arrival at the port of loading and transport to the place of destination may be as long as 15 days or longer. Although truck and rail transport cause fewer losses, these means are deficient and costly and the animals are badly mishandled in transit.

Methods of slaughtering and slaughterhouse services are extremely old-fashioned in most municipalities. Conditions are unhygienic, and as a general rule there are no veterinary services for proper inspection of the cattle on the hoof and the meat. One of the chief drawbacks is too many small slaughterhouses where the volume of operations is not large enough to finance the equipment, construction and services which would be required for efficient organization. Only about 5 percent of the slaughterhouses are located in major cities and provide technical and hygicnic services.

Among other serious deficiencies in the slaughter of livestock and the handling of meat are the inefficient utilization of slaughter by-products and the lack of

refrigeration facilities, even in torrid climates where meat spoils in a very few hours.

Prices of cattle and beef have been rising with the general inflation that has prevailed in Colombia for several years. Prices of cattle at the ranch, on a liveweight basis, were approximately 5.17 pesos per kilogram in early 1970, equivalent to U. S. \$0.28 per kilogram liveweight. Deflated consumer beef prices have increased by 18 percent from 1964 through 1969, while prices at the ranch only increased by 13 percent [41].

In addition to the rising secular trend, beef prices show both seasonal and cyclical variations with cycles averaging about seven years in length. Seasonal price fluctuations are caused directly by the occurrence of dry seasons and the lack of irrigated pastures and forage storage.

Until 1964 the official exports of livestock products were negligible, but have since shown substantial increases. In 1970, these exports reached over U. S. \$21 million [63]. In 1974 livestock exports are expected to range between a low target of U. S. \$51 and a high target of U. S. \$107 million [35].

Exports of beef (frozen, refrigerated and chilled), viscera and processed meat have been increasing in importance. Estimated values for 1971 were U. S. \$9.4 million for beef and U. S. \$0.17 million for viscera and processed meat as compared with U. S. \$2.4 million and U. S. \$1.08 thousand in 1965, respectively.

Peru has been the most important market for live cattle followed by the Dutch Antilles, French Gulana and Venezuela. Illegal exports of live animals, predominantly to Venezuela, have been estimated between 100,000 and 300,000 head annually. Spain, Peru, the French Antilles and French Gulana have been the most important markets for beef.

Beef Consumption

Registered or controlled slaughter in 1970 was 2.366 million head, but total slaughter was estimated at 2.603 million head after increasing the former by 10 percent to account for unregistered or clandestine slaughter [63].^{1/}

Although the trend in cattle slaughter has been upward, there have been significant variations [23, 61]. From 1954 to 1957 slaughter increased 27.5 percent and then turned downward during 1958 and 1959. From 1960 to 1964 it increased again by 30 percent. From 1965 to 1968 slaughter decreased by 3.3 percent and then turned upward again during the next three years.

Slaughter of male cattle fluctuates less than that of females and averages about 60 percent of total slaughter. Female slaughter averages about 40 percent of total slaughter and has ranged from 34 percent in 1954 to 44 and 43 percent in 1951 and 1965 respectively [16, 61]. Consequently, the year-to-year variations in total cattle slaughter have been

 $[\]frac{1}{This}$ refers to domestic consumption. Official statistics usually include registered exports of beef and live cattle.

largely due to changing policies of farmers who withhold females for breeding purposes. Apparently, accumulation and liquidation phases in the cattle cycle are completed, on the average, every seven years.

The trend in per capita consumption of carcass beef has been slightly downward due to the more rapid rate of growth in the human population than in total beef production. Registered cattle slaughter decreased from 123 head per 1000 inhabitants in 1951 to 110 head in 1970. Per capita consumption decreased from 29.6 kg. in 1951 to 22.4 kg. in $1970, \frac{1}{}$ but when unregistered slaughter is considered, per capita consumption in 1970 increases to 24.6 kg. Yet, unequal distribution of income aggravates the nutritional problem, leaving peasants and low income urban groups with a consumption of 18.0 or less kilograms [31]. Undoubtedly beef consumption by the mass of the population is below the recommended nutritional requirements set at 28.0 kg. [9].

Increases in domestic demand will depend on population growth, per capita income and income and price demand elasticities. Assuming no price changes, domestic demand is expected to grow at approximately 4.8 percent annually [31]. This assumes an income elasticity of .6, and annual rates of growth in population and real income per capita of 3.2 and 3.0 percent respectively [31].

 $[\]frac{1}{\text{Estimate based on an overall dressed carcass}}$ average of 200 kg.

Riley [61] and ICA [31] have projected domestic consumption in 1975 using different estimates for the average consumption per capita. According to Riley, if per capita consumption remains at 23.75 kg. annually--the average for the 1958-60 period--domestic consumption in 1975 would be 571 thousand tons or a 64 percent increase over the 1958-60 average of 347 thousand tons. If per capita consumption rises to 29.06 kg., domestic consumption would double the base period average. The ICA estimate shows that if per capita consumption is 25.9 kg. annually--the average for 1964--domestic consumption would increase to 640 thousand tons by 1975 or about 85 percent over the base period.

The parameters determining the rate of growth in demand and the estimates of domestic consumption suggest the need for well coordinated government policies if the goals of improved nutrition, production incentives and increased foreign exchange are to be attained. If beef supplies are not increased substantially, the income of the lower income group is not raised, or beef substitutes are not available, large numbers of the population will continue to be undernourished.

National Policies Toward the Cattle Industry

<u>Taxation</u>. Incentives for beef cattle production in Colombia are crucially affected by government policy. Cattle raising is subject to the same income and complementary taxes (net worth and excess profit) as any other economic

activity. But certain special provisions by which costs and income are computed favor the cattle producing taxpayer.

The essence of this tax policy relates to the cost basis on which profits are calculated. For tax purposes, the cost of livestock sold is the purchase price only if acquired during the tax year. If cattle are sold in the year following that of purchase, then the approximate market value at the end of the previous year is taken as the purchase price. The difference between the purchase price and the assessed end-of-the-year market value is treated as an increase in capital and is not subject to income tax.

The tax policy is also designed to encourage ranchers to engage in breeding activities or to hold females rather than males in inventory to build up the national cattle herd. A net worth tax exemption and two taxes support this policy. The first is a slaughter and export tax which differentiates between the sexes: 50 pesos per head for males and 100 pesos per head for females. The second is a selective inventory tax equivalent to the value of 4 kg. liveweight per head which applies only to males over one year of age. The amount of this tax varies from year to year. In 1971 it was 18.40 pesos per head.

A final element in government taxation of the livestock industry is a general inventory tax. Any individual or corporation whose investment in livestock exceeds Ps. 15,000 at the close of any year from 1959 through 1980 is subject to a levy of 1 percent on the net investment.

Taxpayers who elect to subscribe for shares of Banco Ganadero and the Fondos Ganaderos at par, in an amount equal to the total tax due, are exempt from cash payment of the tax. This is in fact the customary form of payment, and it provides an important part of the capital of these credit institutions.

A property tax of 4.2 mills on the assessed value of land is also levied on the cattle subsector. Additional surtaxes of three and two mills are levied on assessed properties in the areas comprising the Corporacion Autonoma Regional de Valle del Cauca (CVC) and the Corporacion Regional de la Sabana (CAR), $\frac{1}{}$ respectively.

In 1971 the Colombian Government proposed the use of presumptive techniques for a more effective income taxation of agriculture, and finally in 1972 passed a law for approval by Congress [56]. Now, cadastral value of the land alone serves to assess farm income. Yet only a proportion of the cadastral value is used: (1) 50 percent for permanent crops and cattle raising, (2) 75 percent for temporary crops, and (3) 80 percent for annual crops. The presumed income is 10 percent for all crops and cattle fattening, 4 percent for cattle breeding, $\frac{2}{}$ and is also subject to the normal progression of the income tax. The reformed tax law also phases

 $\frac{1}{These}$ are regional development corporations with headquarters in Cali and Bogota respectively.

2/Javier Ayala, "Nueva Propuesta Sobre Ley Agraria," El Tiempo, (Bogota) January 20, 1972, p. 1. out the special inventory tax on males and provides tax incentives on reinvestments in the farm. Fifty percent of farm income in excess of the presumed income is exempted from taxation if reinvested during the year following the fiscal year.

Land Reform. Large holdings and extensive methods in cattle raising have made grazing lands an easy target for expropriation and land distribution schemes. Under the provisions of Law 135--the agrarian reform law--most pastoral estates are considered inadequately utilized and could be expropriated at the least favorable terms.

With the increasing need for farm products, and considering that threats of expropriation have discouraged long-term investments and hampered agricultural development, the government in 1971 undertook a major revision of Law 135. The revised law, $\frac{1}{}$ now pending approval by Congress, provides for more protection against expropriation of adequately utilized farms and for more favorable compensation terms if expropriation occurs.

The designation of farms as adequately utilized has been tightened; it now includes, among other things, the attainment of minimum levels of productivity, and the improvement of the level of living of the workers employed by the landowner.

Cash payments for adequately utilized farms have been increased to 40 percent of the land value if the value

<u>1/161d</u>.

is 500,000 pesos or less, with this proportion decreasing gradually as the total land value increases. The balance will now be paid in five years with interest bearing and negotiable government notes.

<u>Credit</u>. Government direction of agricultural credit is carried on through a complex of official rediscount facilities, reserve requirements and direct legislation. The Monetary Board, appointed in 1963, has legislative control of the banking system and is responsible for setting legal reserve requirements, interest rates and term of loans.

Lending to the agricultural sector has been growing faster than in the economy as a whole. But within the agricultural sector, livestock increased slower than the growth in overall credit in the economy [41]. The livestock portfolio's share of the total has been relatively constant, reaching a low of 18.3 percent in 1966 from a peak of 21.7 percent in 1963. Over the period 1958-1967 the livestock portfolio averaged 19.5 percent of the total portfolio.

Among legislative measures, Law 26 of 1959 has increased the supply of credit to agriculture and strengthened the activities of the Banco Ganadero and Fondos Ganaderos through allocation of the general inventory tax. These credit institutions which specialize in livestock development, must loan not less than 70 percent of their funds for breeding and growing. The law also requires that commercial banks loan not less than 15 percent of their deposits for agricultural purposes.

At present, the Caja Agraria and Banco Ganadero are the two most important sources of credit to livestock producers. In addition, commercial banks are required by law to lend 15 percent of their deposits to agriculture, including loans for livestock development. In 1967, the Caja Agraria held 47.9 percent of the livestock portfolio; the commercial banks 30.0 percent; Banco Ganadero 18.6 percent; and Banco Popular 3.5 percent.

Institutional credit is available to cattle producers at varied interest rates and terms. In general, interest rates charged to small and medium producers range from 8 to 12 percent annually, which are below the current commercial rate of 14 percent. Interest rates charged to large producers are more in line with the commercial rate.

Terms for repayment vary greatly according to the purpose of the loan. For fattening activities terms do not exceed one year, while for breeding and land improvements terms range from three to twelve years. Grace periods from one to four years have been introduced to accommodate better the repayment obligations to the slow return from investments characteristic of the cattle industry.

Special funds from foreign and domestic sources are administered by the Caja Agraria and Banco Ganadero as part of the overall cattle development plan. Small cattle producers within the INCORA-supervised credit programs receive loans, mostly in kind, from the Caja Agraria and Banco Ganadero. The Caja-INCORA scheme is financed by a loan from

AID to INCORA, and funds for the Banco program come from INCORA's Fondo Rotatorio. The Caja also administers a loan from the World Bank for livestock development programs. The Banco Ganadero has been using funds from the Inter-American Development Bank (IDB), the Dutch Government and AID for the same purpose.

Loans from these programs are being devoted mainly to beef production in the Atlantic Coast and the Eastern Plains. Ranchers borrowing from these funds have to participate with 20 percent of the estimated cost, receive technical assistance and invest up to 70 percent of the loan on farm improvements. Interest rates are 14 percent annually, the term of repayment is up to twelve years with a three- to four-year grace period.

Another form of credit quite common in Colombia, known as "cattle-in-partnership," is made available in the form of cattle for which the rancher provides pasture and supervision. Profits are shared when the cattle are sold. The cattle are financed by the private sector and the Fondos Ganaderos (livestock funds), for which financing is provided by departmental and national governments and by the Banco Ganadero and Caja Agraria. A usual profit sharing arrangement is 60 percent rancher, 40 percent financier. While such an arrangement has the advantage of not impairing the rancher's borrowing capacity, it is probably equivalent to a loan with interest between 15 and 20 percent (depending on the profit shared).

Despite the priority given by the government to agricultural credit and the increased channeling of resources to it, there is still an unsatisfied demand for long-term credit. Recent agricultural credit policies have been oriented toward increasing the availability of funds and raising the interest rates to ensure a better utilization of scarce capital resources.

Changes in agricultural credit policies have included:

- (1) Increased use of supervised credit. Credit is now considered an effective means of introducing technological change.
- (2) Increased terms and interest rate of loans under Law 26. Beginning in August 1969, the Monetary Board increased terms of repayment up to seven years with a grace period of two and one half to three years. Interest rates were changed from 8 and 9 percent annually to a variable rate that is 10 percent the first year and increases every year thereafter by one-half of 1 percent.
- (3) Increased and preferential rediscount quotas for loans made by Caja Agraria and Banco Ganadero.
- (4) Preferential rediscount rates for Caja Agraria, Banco Ganadero and INCORA.
- (5) Obligatory investment by commercial banks for 32 percent of its loan portfolio for development. The latter includes Law 26 loans and other loans of the agricultural sector.
- (6) Maintenance of subsidized interest rates for small producers.
- (7) New program for credit to land reform beneficiaries organized in cooperative or commercial operations.
- (8) New program for personal credit to small farmers based on expected returns on the investments.

Disease Control. With an international commitment

brucellosis. To meet its commitment, and with the financial assistance of the Inter-American Development Bank and the technical assistance of the Pan-American Center Against Foot-and-Mouth Disease, ICA prepared a two-stage plan beginning in 1971.

During the 1971-1975 stage the campaign will be concentrated in the Atlantic Coast region where 83 percent of the cattle population will be treated by the end of the period. In the same year the proportion of cattle treated in the rest of the country will be approximately 58 percent.

In the next period--1976-80--control measures will be intensified in all producing zones and the proportion of cattle treated will be very close to 100 percent.

While the control of FMD is restricted to priority areas, the control of brucellosis will be spread over the entire country. The campaign aims at having 100 percent of the female population free of brucellosis by 1974.

<u>Development Plan</u>. In 1972 the Ministry of Agriculture prepared a comprehensive livestock development plan for Colombia.^{1/} The objective was to establish livestock production goals for the next decade and then to outline in detail the necessary plan of action to help achieve the desired goals.

The most important policy instruments are: (1) tax incentives for breeding and farm improvements; (2) increased

<u>l</u>/"El Gobierno Modifica Su Politica Ganadera," <u>El Espectador</u> (Bogota), November 7, 1972, p. 1.

3 1.

availability of credit and easier credit terms; (3) increased association of credit with technical assistance and subsidized technical assistance for small producers; and (4) protection against land expropriation if certain levels of productivity and use of resources are attained.

Exports. General measurer to promote exports are amore flexible exchange rate policy and a 15 percent tax bonus (Certificado de Abono Tributario--CAT) incentive for all exports, except coffee, raw cattle hides, and petroleum. CATs may be traded at a discount or used after one year of issuance for tax payments. More specifically, the government has begun to promote beef exports through a semi-public lending institution, Corporacion Financiera Agrope cuaria (COFIAGRO). About 80 percent of COFIAGRO'r portfelio is in enterprises engaged in the export of beef, but it also lends to ranchers for fattening operations at one-year terms and at an effective interest rate of 16.28 percent.

The government has recently taken two new measures intended to regulate the domestic and export markets. Beginning July $1972^{1/}$ beef has been banned two days a week from restaurants, hotels and similar public outlets. Beginning January $1973^{2/}$ a quota system regulates exports to avoid

<u>l</u>/Jaime Sotomayor, "Veda de Carne Dos Dias a la Semana," <u>El Espectador</u>, (Bogota), June 30, 1972, p. 1.

^{2/&}quot;El Gobierno Fija Cupos de Exportacion de Carne," El Tiempo (Bogota), December 11, 1972.

domestic shortages. This measure requires the gradual phasing out of export of live animals and an increase in beef and processed meat exports.

Domestic marketing of beef is also being improved; the Banco Ganadero in cooperation with USAID has placed special emphasis on financing the modernization of slaughtering facilities.

CHAPTER 3

THE REGIONAL SETTING OF THIS STUDY

The Geopolitical Setting

The states or departments of Atlantico, Bolivar, Cesar, Cordoba, Magdalena, and Sucre considered in this study and known as the Costa, are part of the Atlantic or Caribbean plain which is one of the five geographic regions into which Colombia is divided. $\frac{1}{}$ The capitals are Barranquilla, Cartagena, Valledupar, Monteria, Santa Marta and Sincelejo, respectively. In 1964 these six states had a population of about three million within an area of 112,055 square kilometers; these figures were 18 and 10 percent of the total Colombian population and area, respectively [17].

The Atlantic plain is located between the Caribbean sea and the base of the Andean range in the northern part of Colombia. It is characterized by flat and swampy lands in the bottom of the alluvial valleys and the coastal plain, and by slightly undulating to rugged lands in the areas above the valleys floors and in the surrounding mountains. With the

 $[\]frac{1}{\text{The other four are:}}$ Andean region, Pacific Coast, Orinoco region and Amazon region. Geographically the department of Cuajira and the Antioquian region of Urabá are included in the Atlantic plain, but for all practical purposes this study will refer to the six departments listed.

exception of the Sierra Nevada de Santa Marta in the northeast, the altitude varies between 0 and 500 meters.

The most important rivers are the Sinú in the west, and the Magdelena with its three major tributaries, the San Jorge, the Cauca and the Cesar. The Magdalena, Cauca, San Jorge and Sinú rivers are navigable and serve as important means of transport.

The region has a relatively good network of roads which connect the main urban centers, but access roads to the agricultural areas are few and inadequate, especially during the rainy season.

The railroad connects the port of Santa Marta with Bogotá, Medellín and Cali. Air transportation is available both for passengers and cargo from the airports in the capitals and from air strips throughout the area. The sea ports of Barranquilla, Cartangena and Santa Marta have modern facilities and serve a substantial part of the Colombian export-import trade.

The Population

The Costa population has four major attributes, most of them characteristic of other regions in Colombia. First, the total population in the Costa has been increasing at an increasing rate. The annual average rate of population growth is estimated to be 3.23 percent.¹/ If it continues at this rate, the population will double in approximately 22 years.

 $\frac{1}{Rate}$ of growth estimated for the period 1938-1964.

Second, the population is unevenly distributed. The department of Atlantico has the greatest density (219 inhabitants per square kilometer) and Cesar the lowest (11 inhabitants per square kilometer). In 1964, about 61 percent of the population was urban, and approximately onehalf of this was concentrated in the cities of Monteria, Cartagena, Barranquilla, and Santa Marta.

Third, throughout the region the population has been shifting fairly constantly since 1938. These movements can be classified as: (1) permanent migration from rural areas to major towns of the region (population growth in the four major cities mentioned earlier is estimated to be near 5.0 percent annually); (2) migration from urban and rural areas to the neighboring labor-short Venezuela; (3) seasonal in- and out-migration of the rural labor force to accommodate the demand for harvest labor, especially for cotton, in the region and in the rest of the country; (4) migration from the rest of the country and from the region toward the new rural frontier areas along the Valle del Cesar, the Magdalena, and the low Cauca; and (5) out-migration toward other regions, especially the more developed urbanindustrial departments.

Fourth, education, occupational status, and income per capita are unevenly distributed, not only between the urban and rural populations but among the departments in the region. Literacy ranges from 40 percent in the more agricultural departments to 62 percent in the more urbanized

and industrialized department of Atlantico. The proportion of economically active population engaged in agriculture, forestry, hunting and fishing ranges from 60 percent in the departments of Cordoba, Bolivar, Sucre, Cesar and Magdalena to 16 percent in the department of Atlantico [17].

Although the 1964 census lists no figures on income per capita, income is probably higher in the urban than in the rural areas, and higher in the department of Atlantico than in the more agricultural departments. (These estimates are based on information collected by the Departamento Administrativo Nacional de Estadistica [DANE].) In 1970, $DANE^{\frac{1}{2}}$ examined the family income of urban and rural workers in the Atlantic region, in four other Colombian regions, and the city of Bogotá. For the Atlantic region, DANE estimates that 63 percent of the employed urban population and 84 percent of the employed rural population had a monthly income of 1,000 pesos^{2/} or less.

The 1970 DANE sample estimated over unemployment in the Atlantic region to be 10.96 and 7.73 percent of the economically active population for the urban and rural areas, respectively. But unemployment is more serious than these figures suggest. The number of people suffering from shortage of work is probably larger than the observed numbers actively

 $\frac{1}{DANE}$, Encuesta de Hogares 1970, Bogotá, June (1971), pp. 1-59.

 $\frac{2}{0}$ One U. S. dollar equals approximately Ps 20.

<u>h4</u>

seeking work or longer hours, because the unemployed or underemployed not openly seeking work might do so if unemployment decreased.

Ecological Zones

The Atlantic region can be divided into four distinct ecological zones characterized by the climate and natural vegetation: (1) the tropical dry-humid savannah in the littoral, east from the Sinu river outlet; (2) the tropical humid savannah in the center; (3) the tropical dry-humid forest south of the humid savannah; and (4) the tropical humid forest in the extreme south (Figure I.2).¹/ In turn, each zone can be divided into two special natural regions-the flood plains and the uplands--distinguished by their soils and the crops cultivated. These ecological zones are identifiable and reasonably distinct, although the boundaries between them are arbitrary. The three geographical features that determine the agricultural activities in these zones are climate, soil moisture and soil types.

Climate and Natural Vegetation

Rainfall and temperature are the two most important climatic features. The region alternates between two contrasting rainfall patterns: a low rain or dry period from December through March, and a high rainfall period

 $[\]frac{1}{1}$ In this and the following two sections, 1 have drawn heavily on the Magdalena Mission Report [13].



FIGURE I.2. Ecological zones of Northern Colombia.

from April through November. In general, rainfall increases and dry periods are shorter from north to south.

In the dry-humid savannah of the north, total annual rainfall averages less than 900 mm. The humid savannah receives between 1000 and 2000 mm. annually, the dry-humid forest about 2000 mm., and the humid forest in the south over 2000 mm. An equally important feature is the seasonal distribution of rain. The littoral receives hardly any precipitation in the dry season, whereas the other zones receive a fairly substantial amount throughout the year (an average of 23 mm. per month during the dry season).

The annual average temperature is about 27°C. Throughout the rainy season, the humidity is over 80 percent. During the dry period, winds flow from the sea causing the temperature and humidity to drop slightly, but this effect decreases with increasing distance from the littoral.

The natural vegetation of the Atlantic region can be divided into three basic categories: (1) the dry-humid savannah in the north characterized by xerophitic and subxerophitic vegetation and grasses; (2) the humid savannah in the center characterized by a mixture of natural grasses, scattered shrubs and thin to thick forest in the more wet areas near the rivers and in areas with higher rainfall within the zone; and (3) the rain forest in the south. The distinguishing feature from north to south is the vegetation change from the sparse savannah to the lush rain forest associated with increasing abundance of precipitation.

In general, the four ecological zones provide a good habitat for grazing animals and crops. The dry-humid savannah in the north is more suitable for grazing, although annual crops are grown during the rainy season. Irrigation is required because of the dry periods and to allow for double cropping. The humid savannah, the largest and most important agricultural zone, produces most of the region's cotton. During the rainy season, cropping is safe; with drought-resistant and short-cycle crops such as sorghum, double cropping may be possible. This region also provides most of the grazing land. The dry-humid forest provides lush green pasture all year, but is considered too wet for annual crops other than rice. In the northeast banana belt, which is in the same climatic zone, the land is used mainly for grazing. The humid forest has the same land uses as the dry-humid zone.

Soils

Semi-detailed soil studies of the Costa region have been made by the Instituto Geográfico Agustin Codazzi (IGAC) [38], the Instituto de Fomento Algodonero (Cotton Development Institute) [36, 37], and the Mission for the Study of the Magdalena Valley [13]. These studies also contain information which correlates soil types and phases with potential use.

Soils in the region can be divided into four general groups according to their origin: alluvial or flood plains, quaternary, tertiary and mountain (see Figures I.3 and I.4).



FIGURE I.3. Costa--general soil groups.



Topography, fertility and use are closely related to these soil groups. Alluvial soils are characterized by slopes of 0 to 3 percent; soil textures vary from light to medium and heavy, with deep topsoil and drainage varying from welldrained to imperfectly and poorly drained. Often these lands have a high nutrient content with the exception of soluble nitrogen, which is low. Soil pH ranges between 6.3 to 7.3; it is lower in the poorly drained soils and higher in soils with some degree of salinity.

Quarternary soils or terraces are old alluvial deposits characterized by slopes of 0 to 3 percent, and a hard or clay pan at varying depths. Soil textures are light and drainage is imperfect. Soil pH ranges between 5.0 and 6.0 and the nutrient content is low.

Tertiary soils have undulating slopes ranging widely from moderately steep to steep (7 percent to 50 percent). Soil textures vary from light to heavy, and most soils are susceptible to erosion. Soil pH ranges between 5.5 to less than 6.0, and the nutrient content is low.

The mountain soils in the region are characterized by steep to very steep slopes. Because of the excessive relief, most of them are erodible. They are formed chiefly from igneous and metamorphic rocks. Though the high nutrient content of some of these soils would ordinarily make them suitable for coffee and other permanent crops, because of their erodability their best recommended use is in forest.

Based on studies by the IGAC [38], the Magdalena Mission [12], and the Cotton Development Institute [36, 37], an inventory of the soil resource base by department was made (Table I.2). This inventory includes the acreage of total land, soil classification according to origin, and land use. Table I.2 also shows the region's natural endowment for raising cattle.

The Agricultural Economy

The Costa economy is basically agricultural, with cattle the predominant activity, whereas manufacturing is low and concentrated in the cities of Barranquilla and Cartagena. Agriculture employs 50 percent of the economically active population.

The Costa agriculture is characterized by the same problems that affect agriculture in all of Colombia: (1) slow rate of growth; (2) low productivity and high cost per unit of production; and (3) unequal distribution of wealth.

Private ownership is the predominant characteristic of land tenure in the region. A 1964 survey showed that 60 percent of farms were privately owned and included seven-eights of the agricultural land [12]. Other striking features of the Costa's land tenure are the high degree of concentration and absenteeism. As Table I.3 shows, approximately two-thirds of the farms are less than ten hectares, while about 1 percent of the farms are over

Region and Use Land Class	Atlantico	Bolivar and Sucre	Cordoba	Magdalena and Cesar	Total	Recommended Land Use
<pre>1-Well drained 2-Monflooded imper- fectly drained</pre>	21.60 59.CC	90.60 20.00	38.50 3.30	183.70 116.60	334.40 198.90	Crops and cattle (533.30) Cattle, crops occasionally Cattle (2,813.00)
3-Periodically flooded & poorly drained: a-Vater logging		103.14	155.63	66.20	324.97	
t-One-month floodings c-Four-month floodings d-More trac four-	5.64 7.63	141.82 317.84	37.80 82.48	226.09 154.32	411.35 562.27	
month floodings 4-Saline	20.00 17.20	1,452.90 52.30	133.30 15.20	86.83 61.60	1,693.08 146.30	
Total Land 5-Lakes	131.07 25.00	2,178.CO 47.10	466.21 25.20	895.39 129.82	3,671.27 227.12	
Tctal Area	156.07	2,225.70	491.41	1.025.21	3.898.39	
6-Quaternary terraces 7-Tertiary soils	0.50	65.90	218.10	381.40	665.90	Cattle, cash crops occasionally, food crcps (1,822.40) Cattle (2,042.66)
a-Slope < 25% b-Slope > 25% 8-Sandy scils and dures	22.00 52.00	315.20 573.00	422.40 310.00	396.90 761.00	1,156.50 1,696.00	
35 < slope ≤ 25%	33.36			313.30	346.66	
Total Land 9-Eroded and Mountain Soils	107.96	954.10	950.50	1,852.60	3,865.06	
	£2.10	511.70	1,075.60	1,791.70	7,441.10	Forest
Grand Total	326.03	3,691.50	2,517.51	4,669.51	11,204.55	

TABLE I.2. Costa-Land Classes and Recommended Use by Departments. ('000 has.)

Adapted from [13, 36, 37, 38], and versonal information from the Geographic Institute (IGAS).

			Area Occupied by Farms			
Size Categories (has.)	No.	Percent	Has. (000)	Percent	Average (Has.)	
Less than 10	120,793	67.5	241	3.4	2.0	
Fro m 10 to 100	43,741	24.4	1,491	21.3	34.1	
From 100 to 500	12,225	6.8	2,348	33.5	192.1	
Over 500	2,356	1.3	2,928	41.8	1,242.8	
TOTAL	179,115	100.0	7,008	100.0	39.1	

TABLE I.3. Costa--Distribution of Farms According to Size, 1960

Source: CIDA [12, p. 72]

500 hectares and occupy 42 percent of the land in farms. According to $DANE^{1/}$ 6,706 administrators operate one-third of the total agricultural land, or an average of 370 hectares each (mostly pastoral states). The majority of landlords visit these haciendas infrequently--rarely on a weekly basis and in some cases only once or twice a year.

Although crops have been increasing in importance, land use is dominated by pasture. While in 1959 crops occupied only 7.5 percent of all land and 12 percent of land in farms, pasture accounted for approximately two-fifths of all land or approximately three-fifths of land on farms [14, 15]. Artificial grasses are a low proportion of total

<u>1</u>/DANE, <u>Censo Agropecuario 1960</u>, <u>Resumen Nacional</u>, Bogotá, February (1964), p. 21.
grasslands; 20 percent according to DANE [18], and 45 percent according to FAO [66].

Cotton is the most important commercial crop; in 1969 the region had 137,000 hectares in cotton and produced 62 percent of the total Colombian production [7]. Sesame and rice are also important crops, accounting for 59 percent and 23 percent, respectively, of the total Colombian production in 1969 [7]. Sorghum has become increasingly important, particularly as a double crop with cotton; at the same time, the land devoted to raising bananas for export has declined from 20,000 hectares in the mid-sixties to approximately 5,000 hectares in the seventies. Sugarcane, tobacco, and coffee are also grown but to a lesser extent.

The most important staple crops are cassava and corn, both in terms of the number of producers and the number of hectares. In 1969 [7] 79,000 hectares were planted in cassava and the production accounted for 40 percent of the total Colombian production. About 203,000 hectares were planted in corn which accounted for 26 percent of the total Colombian production. Plantain followed in importance (30,000 hectares), and still less land was used for beans and fruits.

Although the introduction of commercial crops to the region during the past two decades has changed a number of traditional agricultural practices, average yields are still fairly low. Yet the potential for high yields clearly exists as has been demonstrated in properly managed commercial and

experimental farms. In the former, yields of cotton-seed, irrigated rice and sorghum have been doubled and that of corn tripled, while experimental yields for corn and irrigated rice have been 6 and 2.5 times as high, respectively [41]. $\frac{1}{}$

Low yields have been attributed to: (1) a large proportion of small holders producing under traditional methods; $\frac{2}{}$ (2) a lack of adaptive research and extension; (3) inadequate distribution and high cost of modern inputs; and (4) a lack of price incentives.

Soil conservation practices are ignored and the continual tillage of steep, erodible slopes with clean-cultivated crops is accelerating soil depletion.

As pointed out, cattle raising is the most important economic activity in the Costa where the same general characteristics and problems affecting the Colombian cattle industry also apply. The major production problems in the region can be summarized as follows:

- (1) Management and economics
 - (a) Lack of farm accounting and record keeping to establish cost relationships and operational efficiency

 $\frac{1}{\text{Average yields (M.Ts/Ha)}}$ in the Costa for the period 1965-1969 have been: cotton-seed 1.4, corn 1.11, irrigated rice 2.3, sesame .66, and sorghum 1.8 [7].

 $\frac{2}{1n}$ 1960 according to DANE [15], 98 percent of sesame, 83 percent of cotton, 95 percent of rice, and 96 percent of corn were produced in plots less than 10 hectares.

- (b) Lack of basic knowledge on returns to the different factors of production needed for an efficient allocation of resources and for considering organization alternatives
- (c) Inefficient markets for both products and inputs, including capital
- (2) Human
 - (a) Low level of education
 - (b) Lack of skills and training
 - (c) Poor health
- (3) Technological
 - (a) Inadequate soils and range management
 - (b) Inadequate breeding, pest and disease control practices
- (4) Environmental
 - (a) Poor use of natural resources
 - (b) Downgraded quality of rural life

Cattle Production

Although there are not reliable time series estimates of the cattle population in Colombia, it seems that the Costa supports between 40 and 50 percent of the total Colombian cattle population. Table I.4 shows the age and sex distribution of cattle in the Costa and the rest of Colombia according to the 1968 sample survey [18], but care should be taken when considering these figures. This survey, the sample surveys of 1964, and 1965, and the 1960 agricultural census made by DANE seemed to have underestimated the total cattle population by 2.5 million head [29].

•	Less tha	n 2 years	Two and	•	
Region	Males	Females	Males .	Females	TOTAL
Costa:					· · · · · ·
Atlantico	34.95	52.86	14.90	133.18	235.89
Bolivar and Sucre	481.81	487.12	172.40	1,167.15	2,308.49
Cordoba	381.76	349.50	418.82	1,059.27	2,209.35
Magdalena and Cesar	464.46	471.27	394.50	1,018.06	2,348.29
TOTAL COSTA	1,362.98	1,360.75	1,000.62	3,377.66	7,102.01
Non-Costa	1,965.77	2,438.40	1,634.84	4,559.16	10,598.17
TOTAL	3,328.75	3,799.15	2,635.46	7,936.82	17,700.15
		1	ļ	1	

TABLE I.4. Cattle Numbers in the Costa and Non-Costa Herd, 1968. ('000 head)

Source: DANE [18, p. 1]

The Costa is specialized in beef production from herds formed predominately by the cross-breeding of Cebu cattle with the native cattle. Within this type of cattle operation there are three productive phases which are performed separately as specialized activities or in conjunction with each other on individual farms. These are: (a) breeding--cows and calves; (b) growing--males and females, 1 to 4 years of age; and (c) fattening--males, 3 to 6 years of age and females discarded from the breeding herds.

. . .

The dual production of beef and milk has been a common practice among small and medium sized cow herds. The 1971 CIAT survey [9] showed that 62 percent of the cattle farms have breeding and milk production as their main economic activities. In 1968 DANE [18] reported one-third of the cows (females over 2 years) were milked, but this proportion seems to have been increasing with improved access to markets and increased demand. Yet yield of milk per cow is low, ranging between 3.06 liters daily during the rainy season and 2.54 liters during the dry season, with lactating periods varying from 90 to 250 days a year depending on the quality of management. Despite low yields, milk production is an important source of income for ranchers and a contribution to improved nutrition in the region.

As Table I.5 shows, the Costa is a major surplus area, and the major deficit area is Western Colombia. The Eastern Plains (excluding Meta) and the South, not shown in this table, are minor surplus areas. Yet, comparing

Region	Year	Production (Tons)#	Domestic Consumption (Tons)	Difference Between Production and Consumption
Atlantic Plain <u>a</u> /	1969 1970 1971 1972	198,082 213,669 240,325 260,080	62,231 64,720 67,309 70,001	+135,851 +148,949 +173,016 +190,079
Western Colombia <mark>b</mark> /	1969 1970 1971 1972	88,525 96,423 105,763 115,360	183,533 189,224 195,092 201,141	- 95,008 - 92,801 - 89,329 - 85,781
Central Colombia ^{_/}	1969 1970 1971 1972	74,574 79,666 86,036 92,380	125,955 130,490 135,187 140,054	- 51,381 - 59,824 - 49,151 - 47,674
North Eastern Colombia <u>d</u> /	1969 1970 1971 1972	66,669 71,178 76,829 82,443	60,680 62,136 63,628 65,155	+ 5,989 + 9,042 + 13,201 + 17,288
<u>a</u> / Includes A Sucre, Gua	tlanti jira.	co, Bolivar,	Cesar, Cordo	ba, Magdalena,
b∕ Includes V Antioquia,	alle, and C	Cauc <mark>a, Ris</mark> ar aldas.	alda, Quindio	, Nariño,
<u>c</u> / Includes C	undina	marca, Tolim	a, Huila, Bog	otá, Meta.
<u>d</u> / Includes B	oyacá,	Norte de Sa	ntander, Santa	ander.
*Figures are i	n metr	ic tons.		
Source: Adapt <u>Poten</u> en la (Bogo	ed from cial para Decada ta, Nor	m Max F. Bow ara la Expor a de 1970. vember, 1969	ser. <u>Prerecu</u> tacion de Cari Agricultura Ti), 704-708.	<u>isitos y</u> ne en Colombia ropical, XXV

TABLE 1.5. Surplus and Deficit Relationships in Beef Production Among Regions, Colombia, 1969-1972

regional estimates of beef production and consumption to identify surplus-deficit areas does not provide an adequate measure of total cattle produced by regions and interregional movements in terms of number of animals, since much of this movement involves feeders. It has been estimated that the Costa provides approximately two-thirds of the cattle that move in interregional trade [61].

The surplus cattle movements from the Costa include finished, stocker and feeder animals which move south mainly through the markets of Medellin and Bucaramanga. In some instances the cattle move southward to the central and upper Magdalena Valley region by means of the Magdalena River and the railroads. Cattle for export is handled through the Caribbean ports of Coveñas, Barranquilla, Cartagena and Santa Marta. Cattle are moved to Venezuela as well, but estimates of this flow are not available.

The results from the 1960 census provide a detailed breakdown of cattle operations by farm size and by number of head (summarized in Table I.6). As this table shows, the size of the cattle farms follows a skewed distribution.

Only one-fourth of the farms had cattle at the time of the census, but these farms included approximately threefourths of all land in farms. The average number of cattle per farm was 68 with a range that included 182 ranches with an average of 1900 head each. Cattle inventories were concentrated on a relatively small percentage of farms of more than 500 hectares each. Conversely, the smaller farms make

	Un:	Lts	Cattle in Farms		
Size Categories (has.)	No.	Percent	Head (000)	Percent	Average (head)
Less than 10	11,382	25.0	99.8	3.0	9
From 10 to 100	22,478	49.0	681.0	22.0	30
From 100 to 500	9,598	21.0	1,053.7	34.0	110
Over 500	2,195	5.0	1,256.2	41.0	572
TOTAL	45,653	100.0	3,090.7	100.0	68

TABLE I.6. Costa--Cattle Distribution According to Size

Source: Adapted from DANE [15].

up a high percentage of farms having cattle but control a small percentage of total cattle inventories; three-fourths of the farms were less than 100 hectares and controlled onefourth of the cattle with an average inventory of less than 50 head of cattle.

Agricultural Services

Although the agricultural service structure of the Costa appears to be rather complete from the standpoint of physical facilities, it seems in many instances they are not performing efficiently considering the needs of the region.

Research is undertaken by the Instituto Colombiano Agropecuario (ICA) in three experiment stations and eight centers for the study and diagnosis of cattle diseases. In addition to ICA, the Land Reform Institute (INCORA) maintains experimental farms in some of its irrigation projects. Beginning in 1969, the Centro Internacional de Agricultura Tropical (CIAT) started a training and research program in animal sanitation and farm management.

Extension activities are mainly undertaken by ICA, but technical assistance, usually in conjunction with supervised credit, is also provided through INCORA and the banking system. Extension-type services are also provided by the national federations of cotton, rice, cereals, and cattle growers, and by various firms selling plant protection products, herbicides and fertilizers.

Institutional credit is supplied by INCORA, the commercial and the development banks. Another common source of credit for the cattle subsector are the cattle-in-partnership agreements. These are found in both the private and public sectors. The latter are Departmental Organizations, called Fondos Ganaderos, which devote most of their efforts to breeding programs. Under the partnership arrangement, the rancher provides supervision of the cattle and pasture in exchange for a share of the profits when the cattle are sold.

Although agricultural credit programs are available within a wide range of interest rates and terms, the general consensus is that such credit is in short supply and is unevenly distributed. The CIAT survey showed an excess demand for credit (for buying stock) at the prevailing rates of interest.

Modern inputs are marketed through various channels: private supply stores, the INCORA cooperatives, grower federations, and the Caja Agraria. But despite this diversity of outlets, the region suffers from distribution problems: shortages and unavailability are common in the more isolated areas. Since production and distribution do not perform efficiently, prices are too high and incentives for increased use are lacking.

With the exception of cotton and bananas for export, for which there are very well organized marketing channels, the marketing of farm products performs poorly. This is characterized by an excess of middlemen, a shortage of medium- and long-term credit, lack of storage facilities, and inadequate means of transportation. The Institute for Crop and Livestock Marketing (IDEMA) is taking an active role in improving the marketing performance through minimum support prices and by providing storage facilities and credit. INCORA has also been contributing to the coordination of the market through establishment of farmers' cooperatives.

Cattle Marketing

Farmers sell most of their cattle to country dealers or ship directly to "ferias" or stockyards located within the region and as far as Medellin. The CIAT survey showed that five-sixths of the producers sell their cattle to the country dealers who come to the farms to bargain. The cattle are then transported to local slaughterhouses and ferias or

shipped to more distant markets. These dealers also act as intermediaries on feeder cattle transactions.

Cattle are usually purchased on the basis of estimated live weight or by the head. Delivery and transportation arrangements are agreed upon. In almost all cases buyers furnish transportation and assume all costs and risks involved in moving the cattle from the farm or ranch. Contract sales in conjunction with credit have been introduced in the region with the development of the beef export market. These contracts are arranged with the packing houses and for the finishing period.

Most of the cattle in the region are moved by truck, but cattle still move long distances on foot. Some of this movement involves the transfer of feeder cattle from breeding areas to fattening areas. Fat cattle and feeder animals also must be moved on foot to highway, rail, or river shipping points.

Severe weight losses occur during these hauling and on-foot movements. CIAT [9] reports losses up to 40 kilograms per head for finished males after a ten-hour haul to the market.

Live animal prices are affected by both seasonal and long-term cycles. The long-term cycles have averaged about seven years in length. Although the origin of the long-term cycles has not yet been clearly understood, prices, biological timetable of cattle reproduction and nutrition equilibrium seem to be some of the underlying causal forces. Seasonal price fluctuations are caused directly by the occurance of dry seasons and the lack of irrigated pastures or forage storage. During the dry season, cattle are sold as available forage is used, and prices are driven down. The opposite occurs during the rainy season when pastures are abundant.

Estimatates of the marketing margins for beef vary. A recent estimate made by Sarmiento [63] shows that the rancher receives between 80 and 85 percent of the final value. From the standpoint of the producer's share, the beef market seems to be performing well.

Slaughtering of livestock and the handling of meat have the same serious deficiencies described in Chapter 2. Most slaughter facilities are small rural community or village operations, but modern slaughterhouses are in operation in Barranquilla, Cartagena, Santa Marta and La Gloria--a Magdalena river port 300 kilometers south of Santa Marta. These plants are operated mainly to supply the needs of the export market.

CHAPTER 4

GENERAL SPECIFICATIONS AND PROCEDURE

Given the variety of land classes, climatic zones and types of cattle operations that exist within the region, and considering the multiplicity of changing variables that make up its social and economic environment, some simplifying assumptions and/or restrictions are needed to confine the study to workable proportions.

Area of Study

The study will be confined to the Departments of Cordoba, Bolivar, Atlantico, Sucre, Cesar and Magdalena which carry the largest cattle population in the Atlantic plain. The area has a vast potential for successful beef production, easy access to domestic and foreign markets, and expanding facilities for modern slaughter and meat processing.

Farming Sectors

Although there are four distinctly separated ecological zones in the Costa (see Chapter 3), the model is broadly disaggregated into two farming sectors, three agricultural regions and two subregions of competing farming activities based on land use capability (disregarding climatic conditions). Figure I.5 is a hypothetical cross section of these sectors.



FIGURE I.5. Agricultural regions of Northern Colombia.

Sector 1 or lowland is the flat land area in the valley floor formed by recent alluvial deposits. Of the two agricultural regions identified in this sector, region 1 includes lands which are permanently flooded or are subject to seasonal floodings and are used only for grazing. Region 2 is the flood-free area where cash and food crops compete with cattle for land and capital. Region 2 can expand into region 1 as the latter is drained and becomes available for cropping but cannot expand beyond the natural limit of lowlands.

Sector 2, or upland, is the nearly flat to rolling land above the valley floor. It is formed of quarternary terraces and tertiary soils with slopes of 1 percent and over. This sector comprises agricultural region 3 and subregions 1 and 2 where farming is mixed, and includes cash and food crops and cattle production. Subregions 1 and 2 are roughly determined by topographic conditions; subregion 1 is suitable for mechanized cropping and is the area where cash and food crops compete with cattle for land and capital. Subregion 2 is characterized by a more rough and complex topography and is suitable for food crops and grasses which also compete for land and capital. Region 3 cannot expand beyond the natural limit of uplands, but in subregion 1, land in cash crops can contract and expand within its natural limit.

These farming regions are not entirely internally homogeneous with respect to climate and cropping potentials. They occur within the four ecological zones but compromises were made to delineate these regions as homogeneous areas.

The primary reasons for this are twofold. First, despite climate variations, the pattern of farming is very similar in all ecological zones, and behavioral characteristics of farmers who control land use and modernization decisions are assumed to be identical throughout the four zones. Secondly, at the present state of aggregation of the model we are not interested in performing a separate accounting for each ecological zone.

Ranching Practices

While ranching in these sectors is of a mixed type, cattle breeding and growing is predominant in the uplands and fattening is done in the lowlands. But the cattle from the two sectors are aggregated into one herd when simulating the animal demography and computing the major outputs of the model. When the new alternatives of production are introduced, the cattle population of the Costa is disaggregated into two populations, one traditional and one using modern techniques (see Figure 1.6). The "traditional" cattle population is assumed to subsist on the flood-free (lowland and upland) areas during the rainy season. During the dry season, crop residues and additional grazing land which becomes available as the flood waters recede during dry months also add to the nutrient supply. It has been estimated that about 400,000 head from Sucre alone are moved from the uplands to the lowlands in search of water and forage as the dry season advances and food becomes scarce. \perp

 $\underline{1}$ /Personal information.



FIGURE 1.6. A functional flow diagram of cattle production.

Animals in the "modern" sector are assumed to be situated on the flood-free pasture lands where adequate nutrition is available from properly managed grassland and supplemental feed obtained from land devoted specifically to forage production. The level of husbandry is also assumed to be upgraded: diseases and parasites are controlled and improved breeding techniques are used.

Modern Alternatives

In considering the alternatives to traditional cattle production care has been taken to select those which embody a rather simple technology and are deemed to be both feasible and easily transferable given the resources at hand and the behavioral characteristics of ranchers in the Costa. Thus the alternatives considered are focused on investments in relatively simple improvements that will advance management and increase output. Outlays are spent on the most elementary of inputs: fences and stock water supply to permit the beginnings of managerial control; yards and corrals to offer the beginnings of health protection measures; seeds and fertilizers to begin to increase fodder production.

Since a major problem for cattle in the region is a lack of adequate dry season nutrition resulting in substantial weight losses, lower calving rates, higher death rates, and "delayed" maturation, the alternatives emphasize methods of increasing pasture production and growing and storing forage. These not only improve nutrition but also step up

the average carrying capacity, allowing either for a larger or a constant cattle population in the face of expanding crops and shrinking pasture area.

The modern alternatives evaluated in the model are:

- (1) Pasture lands are kept with the grass species already present. Fences, stock water supply and corrals are established to pursue the beginnings of managerial control and health protection measures. Proper grazing rate and pasture rotation are introduced to increase fodder production and improve nutrition.
- (2) The same ranching practices as in Alternative 1 with artificial pastures substituting for natural pastures.
- (3) The same as in Alternative 1 with forage crops being used to provide feed during the dry season.
- (4) The same as in Alternative 2 with forage crops being used to provide feed during the dry season.

At the present stage of development of the model, the modern alternatives are not competing among each other for land and capital; they are evaluated in separate computer runs, each one at a time.

Static Restrictions

Handling all the variables in unrestricted dynamics requires a team effort which is beyond the scope of this study. Thus, analytical restrictions are imposed that keep some of the variables fixed. Patterns of consumption and of ownership of resources, and hence an implied distribution of private real income is assumed fixed; the regional population is assumed constant, and the institutional set-up of the economy is assumed fixed. The implication of these assumptions for the outcome of the study will be discussed in the last "chapter.

Another restriction imposed on this study is that its primary focus is on the beef production process with only general considerations of the related crop subsector, and rudimentary considerations of the marketing element of the beef subsector.

Procedure

To accomplish the objectives of this study, a multicomponent but non-maximizing model of a micro-macroeconomic nature has been developed. The computer simulation model is composed of five tasic components or building blocks (see Figure I.7) which are closely interrelated as the outputs from one block serve as inputs of others. Informationfeedback mechanisms build into the system add to the dynamic interaction of variables within and among the various components.

The first, the land allocation and modernization decisions component allocates land between crops and cattle in the regions of competing farming activities. Land use in food crops in all regions and cash crops in region 2 are exogenously determined, but in subregion 1 land use decisions are based on perceived relative profitabilities of the cattle and cash crops enterprises. Cattle modernization decisions are based on perceived relative profitabilities and the availability of credit, investment capital, and information either from farmer-to-farmer in a diffusion process or from extension agents as part of modernization promotion efforts. Then,



FIGURE I.7. Building blocks of the Costa beef model.

expansion of total cultivated land and/or of modern entitle operations may occur as a result of these economic decisions.

The second principal component takes the allocation of land from the land allocation component and, given prices, costs and yields, computes agricultural production and farm income. The principal element of this component is the cattle demography which is computed endogenously for the region and exogeneously for the rest of Colombia.

A third unit of the model (the price generator) generates world, market and producer prices for cattle, and producer prices for crops. Although market prices for cattle are based on total demand and supply, prices for crops are exogeneously determined. Since we are concerned with farmer decision makers, the streams of future revenues and costs (Equations 5.35) used in the profitability calculations should reflect the farmers' expectations. Thus, the producer prices used here are five-year exponential averages of recent prices. In the production component, however, current prices are used to determine short-run supply responses of cattle.

The remaining two components are the primary entry and exit points of the system. As policy entry points, cattle production campaigns (land, herd and management improvement, and animal health) are specified and conducted and credit, taxes, subsidies and export policies are set. Finally, in the criteria and general accounting component, several alternative criteria functions which might influence a policy maker's choice of development programs are calculated at both interim and final stages of the simulation experiment. Farm incomes derived from meat, milk, and crops are computed, as are capital investment and operating costs incurred through implementing various modernization policies. Thus, several relative benefit/cost relationships for experimental modernization policies are summarized by computed performance functions which include discounted net income from cattle, cattle and crop incomes, foreign exchange balances, beef output, and government revenues. The ability of the industry to meet the increase in demand for beef is determined by computing the domestic consumption when projected exports are achieved.

Each of these five building blocks of the Costa cattle model will be described in some detail in Chapters 5 through 9.

PARTII

MODEL DESCRIPTION

Introduction

The computer simulation model of the cattle economy of northern Colombia is composed of the five basic components or building blocks summarized in Chapter 4: the land allocation and modernization decisions component, the agricultural production component, the price generation component, policy entry points, and the criteria and general accounting component. These building blocks, composed of interrelated functional relationships and linkages between them (an output from one either being an input to another or a performance variable), are an attempt to represent the physical, biological, economic, social, political, and cultural relationahips within and among the major segments of the Costa's cattle economy.

Most of the equations describing these components are applicable to all four modern alternatives considered. But, in cases where there are structural differences arising from the underlying assumptions, specific formulations are shown and discussed. Moreover, the efficiency and economy of the computer model is increased with the use of subroutines written in the FORTRAN programming language.

These subroutines permit simulation of the behavior of the various alternatives and their components over time with one general model.

Each of the five building blocks of the Costa model will be described in some detail in Chapters 5 through 9. A copy of the computer program displaying all equations, value of parameters and initial conditions used in these components and their related subroutines is shown in the Appendix.

CHAPTER 5

LAND ALLOCATION AND MODERNIZATION DECISIONS--CROPS/CATTLE (LAMDAC)

Component LAMDAC of the simulation model allocates land to the production of cash crops, food crops and grasses in each of the farming sectors and three agricultural regions described in detail in Chapter 4 (see Figure I.5). Briefly, region 1 is subject to seasonal or permanent floodings and is kept under traditional grazing practices; region 2 is the flood-free area in the alluvial valleys, and region 3 is the area of quarternary terraces and tertiary soils above the valley floor. In making these allocations, LAMDAC simulates farmers' choices among the alternative uses for their land based on economic and cultural factors. Modernization of current cattle practices is an alternative as is transferring land into the production of crops or cattle.

Land Uses

In general, the land uses in the agricultural regions include grasses, annuals, perennials, and wood lots. Since crops are of secondary concern in this study it was not considered necessary to design a detailed land allocation mechanism for the various crops. Instead they are handled as constant weighted averages of the major commodities grown in the region. Although this assumption has the advantage of simplifying model computations at this stage, it prevents increases in average values--caused by a rapid expansion of crops with high profitabilities--affecting the pasture/ crops allocation decisions. Cash crops in the lowlands are defined as a weighted composite of the major commercial crops grown in the Costa: sesame, cotton, corn, sorghum and rice. Bananas for export are grown in the banana belt north of Magdalena and are computed separately. Cash crops in the uplands include the same group of crops as in the lowlands with the exception of rice and commercial bananas. Further, it is assumed that 47.5 percent of sesame, 63 percent of cotton, 28 percent of corn, and 48 percent of sorghum are grown in the lowlands.

Food crops in all regions are defined as a weighted composite of the major staples grown and consumed in the Costa: plantain and cassava.¹/ Although corn is also an important staple and it is grown mainly in small or subsistence plots, it was included here as a cash crop because of government interest in promoting its production with the aim of generating an exportable surplus [57].

 $[\]frac{1}{1}$ The weights used are as follows: (a) cash crops in lowlands: sesame--.04; cotton--.265; corn--.244; sorghum--.017; and rice--.433; (b) cash crops in upland: sesame--.054; cotton--.18; corn--.74; and sorghum--.021; (c) food crops: plantain--.34; and cassava--.66. These weights are derived from hectares in production as reported in [7].

Pastures are defined as a weighted composite of artificial and native grasses, each having a different yield and nutritional value. Differences in pasture yield and nutritional value also arise from the four range and herd management improving alternatives considered in the model.

Availability of Agricultural Land

Farming area in each region has expanded into the land in forest as the latter has been cleared due to price incentives, population growth, etc. Without detailed information on how these factors have influenced such expansion, as an initial approximation the model exogenously computes increases in the agricultural land base through time using first order, delay equations. These equations simulate the gradual addition of new land to the agricultural land base until the latter reaches the maximum area potentially available in each case [21, 52, 54]. A more elaborate computation would make land expansion a function of endogenous decisions and would likely make use of higher-order distributed delays to simulate more realistically the time response of these decisions.

$$TLAVL(t) = TLAVL(t-DT) + \frac{DT}{DEL15}*(TLAVLO - TLAVL(t-DT)) \quad (5.1a)$$

$$TLAVU1(t) = TLAVU1(t-DT) + \frac{DT}{DEL16}*(TLAVUO1 - TLAVU1(t-DT)) \quad (5.1b)$$

 $TLAVU2(t) = TLAVU2(t-DT) + \frac{DT}{DEL17} (TLAVU02 - TLAVU2(t-DT))$ (5.1c) $T0LSF1(t) = TGLSF1(t-DT) + \frac{DT}{DEL18} (T0LSF0 - TGLSF1(t-DT))$ (5.1d)

where:

TLAVL	flood-free agricultural land actually available in region 2 (has)
TLAVLO	flood-free agricultural land potentially available at time zero in region 2 (has)
TLAVUI	agricultural land actually available in subregion 1 (has)
TLAVUOI	agricultural land potentially available at time zero in subregion l (has)
TLAVU2	agricultural land actually available in subregion 2 (has)
TLAVU02	agricultural land potentially available at time zero in subregion 2 (has)
TGLSF1	grazing land actually available in region 1 (has)
TGLSFO	<pre>= total potential grazing land in region l at time zero (has)</pre>
DEL15,16	5,17,18 = lag parameters (years).

Additions to the cropping area in region 2 due to the performance of flood control and drainage projects (RLDRN) are exogenously determined by Equation 5.2. Irrigation and drainage schemes are accompanied by land distribution and are undertaken and administered by the government through INCORA. Administrative and construction time lags occur in this process which cause the drained land to be immediately unavailable for cultivation. These time lags are introduced by means of the third order distributed delay of Fquation 5.9. Changes in the rate at which land is drained (RLDRN) and in the time required for the completion of a project (DEL11) will allow testing the effect of various policies toward land reclamation on the total grazing area.

TLDRN(t) = min(TLDRN(t-DT) + DT*AUX10(t-DT), TLDRNL0) (5.2) CALL DELAY(RLDRN(t-DT), AUX10, CROUT11, DEL11, DT, K11) (5.3)

where:

- TLDRN = total grassland drained and added to cropping area (has)
- TLDRNLO = total flooded land capable of drainage and flood control at the beginning of simulation (has)
- RLDRN = unlagged rate at which flooded land is drained (has/yr)
- AUX10 = actual (lagged) rate at which drained land becomes productive (has/yr)
- DELAY = is a FORDYN subroutine which introduces distributed delays with various properties
- CROUT11 = an array of intermediate rates necessary in simulation of the drainage delay
- DEL11 = average length of time in the drainage process (years)
- Kll = the order of the delay (=3 in this case)
- DT = time increment of the model (years).

Food Crops

Food land is land on which either subsistence or cash food is actually in production. It is assumed there is no competition among food crops and that they remain at a constant relative proportion. Justification for this assumption and its likely effects in the model's output have been discussed in the section on land uses. Exogenously promoted yield increases are allowed as part of the modernization of agriculture, but there is no disaggregation between traditional and modern production.

Food crop land is assumed to grow exponentially with time and population growth as determined by the following equations:

RLFC(t)	-	AL2#TLFCO#EXP(AL2#t)	(5.4)
TLFC(t)	=	TLFC(t-DT) + DT#RLFC(t-DT)	(5.5)

where:

RLFC	rate of change of land in food crops (has/yr)
TLFCO	land in food crops at time zero (the start of the model) (has)
TLFC	total land in food crops (has)
AL2	a model parameter (very nearly the annual population growth rate)
EXP(AL2*t)	this is an exponential function defined as equivalent to e ^{αt} where α = AL2
DT	time increment of the model (years)
	time in years.

81.

Food land in the uplands is given by:

where:

TLFCU = to se	etal land in food crops in upland ector (has)
TLFCU1 = to (h	tal land in food crops in subregion l as)
TLFCU2 = to (h	tal land in food crops in subregion 2 as)
TLAVU1 = ag	ricultural land available in subregion l Equation 5.1b (has)
TLAVU2 = ag 	ricultural land available in subregion 2 Equation 5.1c (has)
CFD1 = a to	model parameter allocating food crops subregion l
C249 = a to	model parameter allocating food crops the uplands (region 3)
min[a,b] = a wi	function equal to the minimum of terms thin the brackets.
Food land in	the lowlands is given by:

TLFCL(t) = min(TLFC(t) - TLFCU(t), TLAVLO + TLDRN(t)) (5.9)

where:

TLF	CL =	total land in food crops in region 2 (lowlands) (has)
TLA	VLO _. =	flood-free agricultural land available at time zero in region 2 (has)
TLDI	RN =	total land drained and added to cropping area in the lowlandsEquation 5.2 (has)

Cash Crops

Cash crop land is land on which commercial crops are actually in production. As for food crops, cash crops enter the model at a constant relative proportion without competing for land among themselves. Exogenously promoted yield increases are allowed also, but there is no disaggregation between traditional and modern production.

The expansion of cash crops in region 2 is exogenously determined by a constant rate near the historical trend (RLCRL). In subregion 1 of region 3 this expansion is determined endogenously and is discussed below in conjunction with the allocation of pasture. The general assumption is made that cash and food crops compete for a limited amount of land available for cropping (TLAVL, TLAVU1) in each region.

Cash crop land in region 2 (lowlands) is given by the following equations:

ALNDL(t) = TLAVL(t) + TLDRN(t) - TLFCL(t) - TLBAN(t)	(5.10)
TLCRL(t) = min(TLCRL(t-DT) + DT*RLCRL, ALNDL(t))	(5.11)
TLCRLR(t) = TLCRL(t) + TLBAN(t)	(5.12)

• .

where:

ALNDL = allocatable land (cash crops and pasture)
in region 2 (has)
TLBAN = total land in commercial banana (has)
TLCRL = total land in cash crops--excluding
commercial banana--in region 2 (has)

TLCRLR = total land in cash crops in region 2 (has) RLCRL = rate of change of land in cash crops (banana excluded) in region 2 (has/yr).

Total land in bananas for export is determined from time series data as reported by Caja Agraria [7, 8] and assumed to remain after 1970 at its present area of 5,000 hectares which is the hurricane-free, adequately irrigated region within the banana belt:

		20,000	,	1	0 <u>< t <</u> 6	
TLBAN(t)	=	18,520;	8,950; 7,860	. (6 < t <u><</u> 9	•
14		5,000		ł	: (t), <u>></u> , 10	(5.13)

Pasture and Cattle

Pasture land is land on which cattle are actually in production. The area in pasture is computed separately for each region and then aggregated to obtain the total for the Costa. In region 2 and in subregion 2 the assumption is made that cattle are a less profitable activity than growing crops and no decision mechanism is considered; therefore, pasture land is computed as a residual after subtracting land in crops from the total available land in each region. This simplification is based on the empirical evidence that in the more productive lowlands the profitability of cash crops outweighs that of cattle. Likewise, in small farms where food crops are grown in both region 2 and subregion 2 cattle production is not considered as a profitable alternative. In regions 1 and 2 it is also assumed that once land is drained it is placed into a more intensive use and subtracted from

grazing. The allocation of land in subregion 1 is based on the relative returns to land from the cash crops and cattle activities. It is thus responsive to exogenously

determined crop prices and cattle prices determined endogenously by the model.

Pasture or grazing land in the seasonally flooded and in the flood-free areas (regions 1 and 2) is computed as:

$$TGLSF(t) = TGLSF1(t) - TLDRN(t)$$
(5.14)

TGLL(t) = max(TLAVL(t) - TLFCL(t) - TLCRLR(t), 0) (5.15)

where:

TGLSF1	= as defined in Equation 5.1d
TGLSF	= total pasture land in region 1 (has)
TGLI.	<pre>= total grazing land in region 2, the flood-free area (has)</pre>
max[a,b]	= a function equal to the maximum of terms within the brackets.
The rate	of change in grazing land in region 2,

the flood-free area of the lowlands is computed as:

$$RTGLL(t-DT) = \frac{TGLL(t) - TGLL(t-DT)}{D'T}$$
(5.16)

where:

RTCLL = rate of change of total grazing land in region 2 (has/yr).

Pasture land in subregion 2 of the upland is simply computed as:
TGLU2(t) = TLAVU2(t) - TLFCU2(t)(5.17)

where:

TGLU2 = total grazing land in subregion 2 (has). The rate of change in grazing land in subregion 2 is given by:

$$RTGLU2(t-DT) = \frac{TGLU2(t) - TGLU2(t-DT)}{DT}$$
(5.18)

where:

RTGLU2 = rate of change of total grazing land in subregion 2 (has/yr).

The mechanism for allocating land to cash crops and pasture (traditional and modern) in subregion 1 is more complex than the one described for the other regions. This part of the model gradually shifts land to the activity with the greatest return per unit of land. First, the model allocates farming land to food crops as shown in Equations 5.4 through 5.7. The remaining area is then allocated to the land in the most profitable of the two other cash earners --cash crop and pasture.

The following equations apply:

ALNDU1(t) = TLAVU1(t) - TLFCU1(t)(5.19) DLFCU1(t) = TLFCU1(t) - TLFCU1(t-DT)(5.20)

where:

ALNDU1 = allocatable land in subregion 1 (has)

DLFCU] = change in food crops land in subregion t (has).

Land shifts to the more profitable activity at a

rate that is proportional to:

- 1) the percent difference in cash returns per unit of land that exists between the two activities,
- 2) the amount of land currently allocated to the less profitable activity, and
- 3) a model parameter, CLl, which can be varied to match prevailing farmer behavior.

The rate of land transfer is given by:

$$RLCRU(t) = \frac{CL1*XTLU(t)*(DCRU(t) - DRLAV(t))}{|DCRU(t)|}$$
(5.21)

where:

- RLCRU = rate of change of cash crops land in subregion 1 (has/yr)
- DCRU = discounted sum of returns over the planning horizon for cash crops in subregion 1--Equation 5.34 (Ps/ha)
- DRLAV = discounted sum of returns over the planning horizon for cattle production. This is an average of traditional and modern production --Equation 5.33 (Ps/ha)
- XTLU = total grazing land if DCRU > DRLAV
 - = total land in cash crops if DCRU < DRLAV (has)
- CL1 = a model parameter that controls the speed of land adjustment
- $|\ldots|$ = the absolute value.

Positive rates of transfer mean shifts from pasture and cattle grazing to cash crops; negative rates mean shifts from cash crops to pasture. Land transferred from grasses to cash crops is immediately available for crop production, but the rate at which land transfers from cash crops to grasses is delayed to account for the time needed for grasses to become well established and to begin production at full grazing capacity.

This time lag is simulated by the following distributed delay:

CALL DELAY(AUX1(t-DT), AUX2, CROUT4, DEL4, DT, K4) (5.22) where:

AUX1 = unlagged rate of transfer of cash crop land to grazing land: < 0 if DRLAV > DCRU = 0 if DRLAV < DCRU (has/yr) AUX2 = lagged rate of transfer of cash crop land CROUT4, DEL4, DT, K4 = elements of the DELAY subroutine defined earlier (p. 84).

Additions to food land in subregion 1 can come either from land in the least profitable of the remaining activities or from both cash crop and grazing land in specified proportions. Land also could be allocated to food from grazing land even if the latter is more profitable than cash crops.

Such allocation is performed in the model by means of the variable AUX4 and the parameter CL2. If cash crops are more profitable than cattle, the rate of change of land in cash crops is positive (RLCRU > 0) and all the increase in food land comes from land in grasses. If cattle are more profitable than cash crops, the rate of change of land in cash crops is negative (RLCRU < 0) and the increase in food land comes from land in cash crops and/or land in grasses depending on the value of the model parameter $CL2(0 \le CL2 \le 1)$:

If CL2 = 0 all new land in food crops comes from land in grasses, modern and traditional

If CL2 = 1 all new land in food crops comes from land in cash crops.

The variable AUX4 is then computed as:

$$AUX4(t) = \begin{cases} CL2*DLFCU1(t), DRLAV \ge DCRU & (5.23a) \\ 0, DRLAV < DCRU & (5.23b) \end{cases}$$

where:

- AUX4 = transfer of land from cash crops to food (has)
- CL2 = a model parameter that allocates the change in food land between cash crop land and grassland.
- Given the above allocation mechanism, total land in cash crops in subregion 1 is:

TLCRU(t) = max[min(TLCRU(t-DT) - AUX4(t) + DT*(AUX2(t-DT))]

+
$$AUX3(t-DT)$$
, $ALNDU1(t)$), 0] (5.24)

.

where:

TLCRU = total land in cash crops in subregion 1 (has)

AUX3 = rate of transfer of grazing land to cash crops land (= RLCRU) if DRLAV < DCRU

= 0 if DRLAV > DCRU (has/yr).

Equation 5.24 essentially computes the time integral of the total rate of change of cash crop land limited to preclude the possibility of negative land and expansion beyond the allowable land limits (ALNDU1--Equation 5.19).

Given total allocatable land from Equation 5.19 and total land in cash crops from Equation 5.24, the model computes total grazing land (in subregion 1) as:

$$TGLU1(t) = ALNDU1(t) - TLCRU(t)$$
(5.25)

where:

TGLU1 = total grazing land in subregion 1 (has).

The rate of change in grazing land in subregion l is given by:

$$RTGLU1(t-DT) = \frac{TGLU1(t) - TGLU1(t-DT)}{DT}$$
(5.26)

where:

RTGLU1 = rate of change of total grazing land in subregion 1 (has/yr).

The total grazing land in the Costa is simply the sum of grazing land in each of the farming sectors and is given by the following equations:

TGLU(t) = TGLU1(t) + TGLU2(t) (5.27)

 $TGL(t) \sim TGLL(t) + TGLU(t)$ (5.28)

TGLR(t) = TGL(t) + TGLSF(t)*C9(5.29)

TGLU		total grazing land in upland sector (region 3) (has)
TGLU1	=	total grazing land in subregion l (has)
TGL	E	total (flood free) grazing land in regions 2 and 3 (has)
TGLR	=	total grazing land in the Costa region (has)
C9	-	a model parameter that adjusts seasonally flooded grasslands to permanent grazing land.

Grazing land in region 1 is subject to periods of flooding that last from a few weeks to six months and longer. Since we are interested in the permanent stocking capacity of grasslands, seasonally flooded grazing land is adjusted in the model to a permanent grazing land equivalent. This is done by the model parameter C9 of Equation 5.29 which is a weighted composite of area and length of flooding.

Alternatives

where:

In principle, every current land use is a conceivable alternative to every other present use in the same farming sector. In practice, however, certain assumptions can be made which will reduce the multitude of alternatives and will simplify the model. Since the model is focused primarily on cattle production, only alternatives concerned with the introduction of technological changer in this activity will be considered in detail. The alternatives to traditional cattle production included in this study are those described in Chapter 4 and considered as feasible in every agricultural region with the exception of region 1. Pasture land in this region remains under traditional management throughout the simulation.

Implicit in the allocation of land in region 2 and subregion 2 is the assumption that both food and cash crops are more profitable than cattle. Here, the allocation of land to food production has priority in order to meet the nutritional requirements and consumption preferences of the population. The remaining land is then first allocated to cash crops and finally to pasture. In subregion 1 a more complex allocation mechanism has been described; land is first allocated to food production and then to cash crops and cattle based on their relative profitabilities.

Although the present structure of the model restricts consideration of cattle production alternatives to one each simulation, future expansion could include competition among modern grazing alternatives. The model also could be expanded to include competition among crops and a more realistic, though more complex, decision-making mechanism for the allocation of land between individual crops and pasture in all regions.

Cattle Modernization Decisions

Land use decisions between cattle and cash crops have been discussed above in Equations 5.19 to 5.23. In this section we are mainly concerned with the more complex decision mechanism of cattle modernization. The rate at which cattle modernization takes place depends on the relative profitability

96

of each alternative, on modernization promotion efforts, on diffusion effects, on the availability of capital, and on the behavioral characteristics of the farmers making decisions. These considerations will be discussed in detail below.

Profitabilities

Farmers' decisions among the alternative uses for their land are based upon their perceptions of the relative profitabilities of the available alternatives. These decisions have been .estricted to the allocation of land between pasture and crops in subregion 1, and between traditional and modern cattle in all regions but region 1. In the first case, the relative profitability is given in Equation 5.21 above. The relative profitability of the modern cattle production alternatives is given by:

$$PDR(t) = \frac{DRLAM(t) - DRLAT(t)}{|DRLAT(t)|}$$
(5.30)

where:

- PDR = the relative profitability differential (dimensionless)
- DRLAM = discounted sum of returns over the planning horizon for modern cattle production--Equation 5.31 (Ps/ha)
- DRLAT = discounted sum of returns over the planning horizon for traditional cattle production---Equation 5.32 (Ps/ha).

Land use profitabilities are defined as the present value of the stream of net income farmers expect to receive over some relevant planning horizon. The model computes the sum of the discounted present value of returns to a land use from the present to the planning horizon. This discounted sum is the "profitability" of that land use. But while expected revenues and costs from modern cattle vary over the planning horizon, those from traditional cattle and cash crops are assumed to remain constant. Equation 5.31 computes the profitability of modern cattle:

$$DRLAM(t) = \sum_{i=1}^{n} \frac{(TRLAM_{i}(t) - TCLAM_{i}(t))}{(1 + DIR)^{i}}$$
(5.31)

where:

- DRLAM = is defined in Equation 5.30
- TRLAM = total revenue from modern cattle--Equation 5.35c (Ps/ha-yr)
- TCLAM = total costs of modern cattle--Equation 5.35d (Ps/ha-yr)
- DIR = the relevant discount rate (proportion/ year)
- 'i = indexes the n years of the planning horizon.

The profitability of traditional cattle is given by;

$$DRLAT(t) = (TRLAT(t) - TCLATL(t)) * \sum_{i=1}^{n} \frac{1}{(1 + DIR)^{i}}$$
(5.32)

where:

DRLAT = as defined in Equation 5.30
TRLAT = total revenue from traditional cattle-Equation 5.35a (Ps/ha-yr)
TCLATL = exponential average of total costs of
traditional cattle--Equation 5.35b (Ps/
ha-yr)

DIR = as defined in Equation 5.31

i = indexes the n years of the planning horizon.

The average profitability of all cattle used in

Equation 5.21 is given by:

$$DRLAV(t) = \frac{(DRLAT(t) + TTGLR(t) + DRLAM(t) + (TLMOD(t) + TRSL(t)))}{TGLR(t)}$$

(5.33)

where:

- TTGLR = total traditional grazing land in the Costa region--Equation 5.51 (has)
- FLMOD = total grazing land in modern production--Equation 5.48 (has)
- FRSL = total land in transition from traditional to modern cattle production--Equation 5.46 (has)
- TGLR = total grazing land in the Costa region--Equation 5.29 (has).

The profitability of cash crops in the uplands is

computed as:

$$DCRU(t) = (TRCRU(t) - TCCRUL(t)) * \sum_{i=1}^{n} \frac{1}{(1 + DIR)^{i}}$$
(5.34)

where:

DCRU = discounted sum of returns over the planning horizon for cash crops production in the uplands (Ps/ha)

• .

- TRCRU = total revenue from cash crops--Equation 5.35e (Ps/ha-yr)
- TCCRUL = exponential average of total costs of cash crops--Equation 5.35f (Ps/ha-yr)

DIR = as defined in Equation 5.31.

With the purpose of making all discounted present values comparable, the profitabilities are computed using a planning horizon common to all. In this case, this period of time is 12 years, the planning horizon selected for modern cattle. The discount rates used to compute the present value of future returns are behavioral parameters in the model. They reflect farmers' rates of time preference and the varying risks of each alternative. In general, the more risky and unfamiliar the alternative, the higher the discount rate.

Since we are concerned with farmer decision makers, the streams of future revenues and costs (Equations 5.35) used in the profitability calculations should reflect farmers' expectations. These expectations are assumed to be reflected in five-year exponential averages of recent producer prices. Prices of cattle are determined endogenously (Equation 7.2) but prices of crops are determined exogenously and projected into the future with the same trend as costs. The form and computation of producer price averages and trends are discussed in detail later in the description of the price generating component (Chapter 7).

Similarly, the stream of crop yields farmers expect are the yields they currently experience rather than the potential production reported by experiment stations. Increased yields are considered later in Chapter 12 as part of the policy experiments. Additions to expected revenues are any cash and/or price subsidies which may be offered as part of a modernization program, and the payment of development credits.

The cost side includes taxes on land and cattle, biological, chemical, labor, and capital input requirements over the planning period. Associated input prices are exogenous in the model and are projected into the future according to rate of increase in farm costs. Production costs of crops are averaged and lumped in one figure while those of cattle are computed separately; but all costs are also exponentially averaged when they enter in the computation of profitabilities. Exponential averages of past costs are used here to reflect farmers' expectations of future cost streams. The computation of costs is discussed more fully in Chapter 6.

Total revenue and total cost of traditional cattle are computed as:

 $TRLAT(t) = \frac{(EPAP(t)*ESLSPT(t) + PRMT*EQMT(t))}{TTGLR(t)}$ $+ AGSUBT(t) \qquad (5.35a)$ $TCLATL(t) = TCLATL(t-DT) + \frac{DT}{DEL19}*(TCLAT(t-DT))$ $- TLCLATL(t-DT)) \qquad (5.35b)$ where:

EPAP = the expected producer price of finished males--Equation 7.5a (Ps/animal)

ESLSPT	m _	the expected animal sales(animals/year) $\frac{1}{}$
EQMT	E1	the expected production of milkEquation 6.21b (liters/year)
PRMT	=	the price of milk (Ps/liter)
AGSUBT	=	subsidies paid to traditional cattle (Ps/ha-year)
TTGLR	=	as defined in Equation 5.33
TCLAT	•	total traditional cattle costs (unlagged) (Ps/ha-year) <u>1</u> /
DT	=	time increment of the model (years)
DEL19	=	lag parameter (years).

The stream of revenues and costs over the planning horizon for modern cattle production reflect different expectations according to the alternative adopted. Changing expectations throughout the planning horizon are simulated by a set of coefficient arrays (SINCR, TMINCR) that increase output and costs over traditional cattle as perceived by farmers. Values of these coefficients for each of the four modern alternatives are shown in Table II.4 in Chapter 6. Total revenue and total cost over the planning horizon, i=1, ..., n, are computed as:

 $TRLAM_{1}(t) = (SINCR_{1} * EPAP(t) * ESLSPT(t) + EPAP(t) * BINCR_{1}$

#AUXL12(t) + TMINCR, #PRMT#EQMT(t))/TTGLR(t)

+ $ELOAN_{i}(t)$ + $AGSUBM_{i}(t)$ (5.35c)

 $\frac{1}{2}$ For detailed computation of these variables see subroutine AGACC in the Appendix.

$$TCLAM_{i}(t) = EOPCLM_{i}(t) + EOCLNM_{i}(t) + ETCEC_{i}(t) + EDBSER_{i}(t) + ETXC_{i}(t) + EVLDTX_{i}(t) + CADEM_{i} (5.35d)$$

۰.

where: A HARLES REAL AND A COMMENT SINCR = expected increase in sales over the planning r (manufactor from the modern herd (dimensionless) "TMINCR = expected increase in milk production over the planning horizon (dimensionless) N Envir BBINCR = expected increase in inventory over the planning horizon (animals/year) AUXL12 = finished male equivalent of cattle prices (proportion)1/ AGSUBM = subsidies paid to modern cattle producers (Ps/ha-year) Equation 6.40g (Ps/ha-year) 1.11 EOPCLM = expected operation costs of modern cattle "no 1 + 7 --Equation 6.40a (Ps/ha-year) EOCLNM = expected operation costs of modern grazing land--Equation 6.40b (Ps/ha-year) Section 1 •ETCEC = expected total cash establishment costs--1 g 1 1 1 -Equation 6.40f (Ps/ha-year) EDBSER = expected debt service of development credits--(Ps/ha-year)2/ = expected taxes on modern cattle---Equation ETXC 6.40d (Ps/ha-year) EVLDTX = expected taxes on modern grazing land--Equation 6.40c (Ps/ha-year) ECADEM = expected depreciation on equipment and improvements--Equation 6.40c (Ps/ha-year). . .

 $\frac{1}{For}$ a detailed computation of this variable see subroutine DEMOG in the Appendix.

 $\frac{2}{\text{For a detailed computation of this variable see}}$ subroutine AGACC in the Appendix.

Total revenue and total cost of cash crops in the uplands are computed simply as:

$$TRCRU(t) = EPCRPU(t) * EYLDCU(t)$$
(5.35e)
$$TCCRUL(t) = TCCRUL(t-DT) + \frac{DT}{DEL22} * (TCCRU(t-DT))$$

$$- TCCRUL(t-DT))$$
(5.35f)

where:

TCCRU = total costs (unlagged) (Ps/ha-year) $\frac{1}{}$ EYLDCU = expected yield--Equation 6.1 (Tons/ha-year) EPCRPU = expected producer price--Equation 7.5b (Ps/Ton) DEL22 = lag parameter (years).

Promotion and Diffusion

In the process of estimating the profitability differentials of the various alternatives, farmers need certain information. The information required by farmers include future producer prices, expected yields, government or private subsidy and loan programs, and expected costs. In the model this is provided as part of the promotion effort and in the form of "information units." These information units not only include extension agents, the main instruments of disseminating information, but also any other means of mass communication (radio broadcasts, films, and newspapers).

$\frac{1}{F}$ For a detailed computation of this variable see subroutine AGACC in the Appendix.

While promotional information units (extension agent equivalents) are exogenously generated as a policy (Chapter 8), the model also computes (Equation 5.39) the demonstration effect of farmers learning from one another about alternative land uses.

Transition Responses

Changes in land use patterns reflect farmers' responses to the perceived profitabilities of the available grazing alternatives. The assumption is made that all farmers modernizing their systems of production, either because of promotion or diffusion, will receive the same type of public and private incentives. Therefore, the perception of the profitability of the new methods will be the same in both cases.

The profitability response function (Equation 5.36) determines how many hectares of land an information unit can transfer per year from traditional to modern management. This calculation depends on the profitability of the alternative and the behavioral characteristics of the farm decision makers (see Figure II.1).

PR1(t) = max[E7*(1-EXP(-E8*(PDR(t)-E9))), 0] (5.36)

where:

PR1 = a variable which introduces the effects of the profitability criterion, PDR, upon the adoption rate (proportion)

E7 = maximum proportion attainable--a model parameter

- E8 = the rate of promotion response with respect to profitability (dimensionless)
- E9 = the promotion response threshold (dimensionless)
- EXP = exponential function
- max = takes the maximum of the term within the brackets
- PDR = the relative profitability differential--Equation 5.30 (dimensionless).

As shown in Figure II.1, the parameter E7 determines the maximum value of the function. The threshold (E9) and response rate (E8) parameters reflect the farmers' attitudes and behavioral characteristics which affect the rate of the: response to the relative profitabilities of their various alternatives. These two parameters represent a wide range of attitudes toward risk involved in the new methods, uncertainty related to social stability, government programs



FIGURE II.1. The profitability response function.

and promises in general, and the land tenure system. It is clear that a wide range of adaptor behaviors can be simulated by appropriately assigning values to these three parameters. Since farmers will have different attitudes toward extension agents (or other promotional efforts) than they will toward one another, the values of the parameters may be different for promotion responses than for diffusion responses.

The profitability and behavioral criteria are instrumental in determining the rates at which farmers will respond in a diffusion process or to overt campaigns introducing modern methods. The rate at which land enters a modernization process as a result of overt promotion is given by Equation 5.37.

$$RLMPI(t) = E3(t) * PR1(t) * EXT1(t)$$
 (5.37)

where:

- RLMPI = rate at which grazing land enters modernization due to promotion (has/year)
- EXT1 = units of pre-campaign promotion--a policy variable
- E3 = the maximum feasible adoption rate per unit of EXT1 (has/year per unit of EXT1)

PR1 = as defined in Equation 5.36.

As the program progresses, the promotion effort becomes more efficient and the adoption rate likely will be increased. This phenomenon is simulated by Equation 5.38. Here, E3 has its maximum value (E31-E32) at the beginning of a campaign (TCAM=0) and approaches its maximum value (E31) when TCAM is large. Again, a wide range of real-world situations can be simulated by appropriately assigning values to the model parameters.

$$E_3(t) = E_{31} - E_{32}*E_{23}*T_{CAM}(t))$$
 (5.38)

where:

TCAM = the length of time the production campaign has been in operation (years) E31, E32, E33 = model parameters.

The rate of adoption due to demonstration effects depends on the differential between modern and traditional productivities and on the behavioral characteristics of farmers. The diffusion rate (Equation 5.39) is also a function of the land which remains under traditional management (TTGL) and the land which has been modernized (TLMOD). If there is no land in either use, there is no demonstration effect and no diffusion, while the diffusion rate is greatest when there is as much land in the alternative use as in the present use. Thus, the rate at which diffusion takes place reflects the S-shaped curve of diffusion theory [1].

$$RLMDI(t) = \frac{PR1(t)*TTGL(t)*TLMOD(t)}{TGL0}$$
(5.39)

where:

- RLMDI = rate at which grazing land enters modernization due to diffusion (has/year)
- TGLO = total (flood free) grazing land at the beginning of the simulation (has)

The total land entering modernization, RLMI, due to the combined effect of promotion and diffusion is simply computed as:

$$RLMI(t) = RLMPI(t) + RLMDI(t)$$
(5.40)

The transition rate (Equation 5.41) is constrained by available capital and lagged to account for delays in establishing a modern alternative. Since the Costh is a surplus area (see Chapter 3), requirements of capital for buying cattle are not considered in the aggregate for the region, though this is an important factor in analysing individual farms. Thus, the demand for capital for investment is restricted to farm improvements. The carital available (NCFR) includes carital generated endogenously as income (after allowing for consumption), transfers from the crops subsector, and potential credit. The model does not consider competition for cupital between investments and consumption; it is assured that consumption to a first claimant to farm income and investment is treated as a residual. This oversimplification of the consumption/saving decisiers is justified here because of the lack of information on the consumption and saving patterns of farmers in the region. In addition, a realistic simulation of the consurption/saving and asset balance decisions of farmers requires the establishment of a preference function in order to maximize toth satisfaction as well as returns from different investments

under conditions of changing interest rates, risk, and uncertainty.

The availability of capital and credit will be discussed more fully in the discussion of the Accounting and Performance Criteria Component (Chapter 9). The requirements of credit for development (Equation 5.42a) are determined by subtracting from the total establishment costs the proportion of these costs met by the farmers' own resources. Establishment costs are reduced to an annual basis, so the credit required in the year the transition is made can be related to whatever credit funds are available that same year.

CRM(t) = max(min(RLMI(t), [min(ARM1(t), ARM2(t)) +

ARM3(t)]), 0)

(5.41)

where:

CRM	= unlagged rate of modernization constrained by development credit and farmers' invest- ment capital (has/year)
RLMI	= unlagged and unconstrained rate at which grazing land enters modernization== Equation 5.40 (has/year)
ARM1	= allowable rate of modernization depending on availability of development credit== Equation 5.42a (has/year)
ARM2	= allowable rate of modernization depend- ing on capability of farmer to meet his required share of the establishment costs Equation 5.42b (has/year)
ARM3	= allowable rate of modernisation depending on farmer's capability to meet total estab- lishment costs without development credit Equation 5.24c (has/year)

- min[a,b] = a function equal to the minimum of terms
 within the brackets
- max[a,b] = a function equal to the maximum of terms
 within the brackets.

The variables ARM1, ARM2, and ARM3 (Equations 5.42) $\frac{1}{2}$ that determine the constraints to the rate of modernization require further explanation. If credit for development is not available (ARM1 = 0; ARM3 > 0) the rate of modernization depends on the farmers' capability to meet the total establishment costs. If development credit is in ample supply but the investment capital of the participating farmers is not enough to meet their required share of the establishment costs (ARM1 > ARM2; ARM3 = 0), the rate of modernization depends on the farmers' available investment capital. If development credit is available and the investment capital of the participating farmers is more than enough to meet their share of the establishment costs (ARM1 < ARM2; ARM3 > 0), the model assumes that the remaining capital is reinvested in the farm and the rate of modernization is increased by an amount equal to the allowable rate without credit support (ARM3). This is to say that resources in modern cattle production are being used at a low opportunity cost.

$$ARM1 = \frac{CREDT(t)}{CRTREQ(t)}$$
(5.42a)

 $ARM2 = \frac{NCFR(t-DT)*LT1}{TCEC(t)*RPTN}$

(5.42b)

 $[\]frac{1}{Equations}$ 5.42b and 5.42c have implicit in the denominator a one-year multiplicative factor that provides the denired units.

ARM3(t) =

(NCFR(t-DT)-min(ARM1(t), ARM2(t))*RPTN*TCEC(t)/LT1)*LT1 TCEC(t)

(5.42c)

where:

- ARM1 = allowable rate of modernization depending on availability of credit (has/year)
- ARM3 = allowable rate of modernization depending on ranchers capability to meet total establishment costs without development credit support (has/year)
- CREDT = credit available for modernization--Equation 9.4 (Ps/year)
- CRTREQ = per hectare credit requirement (Ps/ha)
- LT1 = time over which development loans are paid-a policy variable (years)
- RPTN = farmers' participation of total establishment costs--a policy variable (0 < RPTN < 1) --proportion
- NCFR = net investment capital of farmers--Equation 9.13 (Ps/year)
- TCEC = total cash establishment costs--Equation 6.37 (Ps/ha).

The modernization process is simulated as a series of exponential delays which allow for the possibility of "dropouts" and represents the phenomena of random modernization times for individual farms in the aggregate. Equations 5.43 describe this process.

$$R1(t) = R1(t-DT) + \frac{DT}{XDEL}*(CRM(t-DT) - R1(t-DT))$$
 (5.43a)

$$XR1(t) = R1(t)*A5$$
 (5.43b)

$$R2(t) = R2(t-DT) + \frac{DT}{XDEL}*(XR1(t-DT) - R2(t-DT))$$
 (5.43c)

$$RLM(t) = RLM(t-DT) + \frac{DT}{XDEL}*(R2(t-DT) - RLM(t-DT))$$
 (5.43d)

where:

- RLM = average rate land leaves the modernization
 process and begins producing at modern levels
 (has/year)
- XDEL = one-third of the average time required for modernization (years)
- A5 = one minus the proportion of land that "drops out" due to shortage of technical assistance and credit--Equation 5.44.
- Rl, XRl, R2
 = intermediate rates (has/year).

The "dropouts" response function (Equation 5.44) determines the proportion of land that remains, A5, after land "drops out" due to the shortage of extension workers and/or development credit.

$$A5(t) = \min(E12*\frac{EXTA(t)}{EXTR(t)}, 1)*(AUX7(t) + AUX9(t)*min$$

$$(E13*\frac{ACRDT(t)}{DCRDT(t)}, 1))$$
(5.44)

where:

EXTR		extension workers (or the equivalent) re- quired to sustain the modernization program Equation 5.45 (man-years)
EXTA	=	extension workers available (man-years)
ACRDT	۵	credit allocated for modernizationa policy variable (Ps/year)

<u>_</u> D <u>C</u> RDT	8	total demand for development credit Equation 9.3 (Ps/year)
AUX7	H	actual proportion of total land being modernized with resources other than develop- ment credit
AUX9	=	actual proportion of total land being modernized with development credit resources
E12, 1	E13 =	} adjustable model parameters
minโล	.b	

= the minimum of a and b.

The calculation of A5 involves the combined effect of two functions which have the same response patterns. One depends on the ratio of available extension workers to the number of extension workers required to sustain the modernization programs; the other on the ratio of credit allocated for modernization to the total demand for development credit. As Figure II.2 shows, the parameter El2 (El3) which governs the shrpe of the response function determines the threshold at which dropouts from the modernization process start and the dependence of the dropout rate upon the ratios EXTA/EXTR and ACRDT/DCRDT.

The extension workers required (EXTR) are computed in Equation 5.45 as the number of man-years needed to provide technical assistance to the land in transition from traditional to modern management (TRSL).

EXTR(t) = C257*TRSL(t) (5.45)

where:

C257 = the extension workers to grazing land ratio-a model parameter (man-years/ha).

In order to compute inputs required for modernization it is important to know how much land is in the modernization process at any given time. The land in modernization is simply the sum of the time integrals of Equations 5.43. Equation 5.46 computes TRSL--the land in transition from traditional to modern practices due to overt promotion and diffusion.

$$TRSL(t+DT) = (XR1(t) + R2(t) + RLM(t)) * XDEL$$
 (5.46)



FIGURE II.2. The land "dropout" function.

A final economic decision to be made is whether some modern grazing land should be reverted to traditional practices if the profitability of modern methods drops significantly due to declining output prices, increasing input prices, etc. Figure II.3 shows how the model (Equation 5.47) handles this decision. It is clear that no reversion will occur unless the profitability cirteria relating modern returns to traditional returns (PDR) drop below some threshold value, in which case, the reversion to traditional practices will occur at an increasing rate, up to a maximum, as the profitability continues to fall. The adoption response function and the reversion response function, which are not symmetrical, attempt



FIGURE II.3. The reversion response function.

116

to capture by proxy the farmers' investment and disinvest-۰. ment decisions. PR2(t) = max(Ell*(1-EXP(-E81*(E91-PDR(t)))), 0) (5.47) . . * where: PR2 = proportion of total land in modern production which reverse to traditional practice PDR = the profitability criteria of Equation 5.30 Ell = maximum proportion that will be reverted (proportion/year) E81 = a parameter regulating the reversion rate (dimensionless) E91 = reversion threshold (Ps/ha). The rate by which modern land reverts to traditional uses is simply the product TLMOD(t) #PR2(t).

where:

TLMOD = the total land in modern production (has).

Given this reversion rate and the rates land is being modernized by production campaigns and diffusion, it is possible to compute the total modern land, TLMOD, assumed to produce at modern productivities. This is done in the model separately for agricultural region 2 (lowland) and for subregions 1 and 2 of agricultural region 3 (uplands). Equation 5.48 first allocates grazing land entering modernization among the three regions in the same proportion grazing land in each region is of total flood-free grazing land, then subtracts the modern land reverting to traditional practices, and finally allocates any change in total grazing land over time between traditional and modern practices.

$$TLMOD_{1}(t+DT) = max\{min[TLMOD_{1}(t) + DT*(RLM(t)*\frac{GLTOT_{1}(t)}{TGL(t)} - TLMOD_{1}(t)*PR2(t) + \frac{ENT*TLMOD_{1}(t)*RTGL_{1}(t)}{GLTOT_{1}(t)},$$

$$GLTOT_{1}(t)], 0\} (5.48)$$

where: '

- RLM = as defined in Equation 5.43d
- GLTOT = total grazing land in the given agricultural region (has)
- **RTGL** = rate of change of total grazing land in the given agricultural region $(has/year)^{\perp}$
- ENT = a model parameter (E6, E10, E14) that determines the percentage of land entering or leaving pasture production that enters or leaves modern production
- **TGL** = total (flood-free) grazing land (has)
- i = indexes the agricultural regions, i=1, ..., 3.

The inclusion in Equation 5.48 of the term involving RTGL requires further discussion. Since, over time, the land allocated by decision makers to cattle production will change, there is a question about how these changes should be allocated to traditional and modern production. The model formulation permits the user to make a number of assumptions about this through adjustment of the parameter ENT. For example if ENT = 0, the model allocates all increases and decreases in total land to traditional production. If ENT = 1, the model allocates changes in land area to traditional and modern

 $[\]frac{1}{7}$ For detailed computation of rate of change in lowlands, subregion 1 and subregion 2 (RTGLL, RTGLU1 and RTGLU2, respectively) see subroutine LANDAL in the Appendix.

production proportionately according to the percentage each is of total land. Further, if:

ENT = 1 when $RTGL \ge 0$ and

ENT = 0 when RTGL < 0,

the model allocates net increases in total land proportionately to modern and traditional production and subtracts all decreases from traditional production, etc.

Total flood-free land in modern (TLMODR) and traditional (TTGL) production in the Costa is computed as:

$$TLMODR(t) = \sum_{i=1}^{3} TLMOD_{i}(t)$$
 (5.49)

$$TTGL(t) = TGL(t) - TLMOD(t) - TRSL(t)$$
 (5.50)

where:

TTGL = total flood-free grazing land in traditional production in the Costa (has).

Finally, Equation 5.51 computes flood-free and seasonally flooded grazing land in traditional production (TTGLR) in the Costa.

$$TTGLR(t) = TTGL(t) + TGLSF(t) = C9$$
 (5.51)

where:

- TGLSF = total pasture land in the seasonally flooded region--Equation 5.14 (has)
- C9 = a model parameter adjusting seasonally flooded grasslands to permanent grazing land.

Cattle Transfers

As pasture land is modernized and forage production increased, cattle are moved to graze in these lands under improved husbandry practices. The rate that animals are added to the modern grazing lands is a function of the rate of increase of their nutrition levels and the relative difference between the achieved nutrition and the desired one. This is computed by Equation 5.52.

 $RAA(t) = \frac{RTDN(t) + C12*TOPOPM(t)*(TDNAM(t) - TDNREQ)}{TDNREQ} (5.52)$

where:

- RAA = rate animals are added to the modern sector (animals/year)
- RTDN = rate of increase of TDN in the modern sector (tons/year-year)
- TDNAM = TDN per animal in the modern sector (tons/ animal-year) $\frac{1}{2}$
- TDNREQ = desired TDN per animal in the modern sector (tons/animal-year)
- TOPOPM = total animal population in the modern sector (animals)¹/
- C12 = a model parameter that determines the influence which the difference between the achieved nutrition level and the desired nutrition level in the modern sector has on the rate animals are added to the modern sector (proportion/year).

The rate at which males and females, summing to RAA, are added to the modern population is given by Equations 5.53. It is assumed that the sex ratio of transferring animals is

120

 $[\]pm$ /For detailed computation of these variables see subroutine DEMOG in the Appendix.

the same as that of the traditional population (this could also be a policy variable). It is also assumed that only fertile females are transferred from the producing cohort.

$$RFGTT(t) = \frac{RAA(t) * PFGT(t)}{TOPOPT1(t)}$$
(5.53a)

$$RFPTT(t) = \frac{RAA(t)*FERT(t)}{TOPOPT1(t)}$$
(5.53b)

$$RMGTT(t) = \frac{RAA(t)*PMGT(t)}{TOPOPT1(t)}$$
(5.53c)

$$RMPTT(t) = \frac{RAA(t)*PMPT(t)}{TOPOPT1(t)}$$
(5.53d)

where:

RFGTT	п	rate growing females are transferred <u>out of</u> the traditional sector (animals/year)
RFPTT	3	rate producing females are transferred <u>out</u> <u>of</u> the traditional sector (animals/year)
RMGTT	=	rate growing males are transferred <u>out of</u> the traditional sector (animals/year)

- RMPTT = rate producing males are transferred <u>out</u> of the traditional sector (animals/year)
- FERT = total traditional fertile producing females --Equation 6.18 (animals)
- TOPOPT1 = total cattle population in the traditional sector net of females unsuitable for reproduction--Equation 5.54 (animals).

The rates animals leave the modern sector are the negatives of RFGTT, RFPTT, RMGTT and RMPTT (negative departures are arrivals):

RFGTM(t)	= -RFGTT(t)	(5.53e)
RFPTM(t)	= -RFPTT(t)	(5.53f)
RMGTM(t)	<pre>-RMGTT(t)</pre>	(5.53g)

 $\operatorname{RMPTM}(t) = -\operatorname{RMPTT}(t)$

The traditional cattle population base for these transfers (TOPOPT1), computed in Equation 5.54 below, excludes all females which are unfit for reproduction, i.e., old cows, infertile cows, and cows with severe cases of mastitis.

TOPOPTI(t) = TOPOPT(t) - OLDFT(t) - FINFT(t) - FMAST(t) (5.54)

where:

TOPOPT	=	total cattle population in the traditional sector (animals)
OLDFT	=	traditional population of old females Equation 6.12 (animals)
FINFT	a	traditional producing females which are infertileEquation 6.17a (animals)
FMAST	=	traditional producing females with mastitisEquation 6.17b (animals).

The preceding equations define the most relevant variables and structural relationships of the land allocation and modernization component. The interested reader can find the complete list of equations performing the land allocation and modernization decisions described here in subroutines LANDAL, MODCRD and MODRAT of the computer program shown in the Appendix. The output of component LAMDAC is used as an input to the component AGPRAC described in the next chapter.

Table II.1 at the end of this chapter shows the values of a selected number of variables used in component LAMDAC.

Definition (Equation 1	Value	
DT (5.1a)	Determines the time increment of the model (years)	. 25
AL2 (5.4)	Determines the rate of growth in food crops land (proportion/year)	.03
CFD1 (5.8)	Determines the allocation of food crop land to subregion 1 (proportion)	.5
CL1 (5.21)	Determines the speed of land adjustment in subregion l (proportion/year)	.03
CL2 (5.23a)	Determines the allocation of food crop land between cash crop and grasslands (proportion)	•5
C9 (5.29)	Determiner permanent grazing land equivalent (proportion/year)	• 55
C12 (5.52)	Determines the speed of animals transfer (proportion/year)	1
C249 (5.6)	Determines the allocation of food crop land to region 3 (proportion)	. 8
C257 (5.45)	Determines extension workers requirements (man-years/ha)	.0005
E7 (5.36)	Determines maximum adoption rate (proportion)	1
E11 (5.47)	Determines maximum proportion of reversion to traditional practices (proportion/year)	.5
E12 (5.44)	Determines the extension "drop- outs" response threshold (dimensionless)	1.2
E13 (5.44)	Determines the credit "dropouts" response threshold (dimension- less)	1
E31 (5.38)	Determines maximum value of E3 when accumulated time is large (has/year per unit of extension)	4,000

TABLE II.1. Selected Coefficients and Initial Values in the Land Allocation and Modernization Decisions Component (LAMDAC).

TABLE	II.1.	(continued)

Definition (Equation N	Value	
E32 (5.38)	Determines minimum value of E3 at time zero (has/year per unit of extension)	2,000
E33 (5,38)	Determines the rate at which E32 decreases over time (dimension- less)	.3
RLCRL (5.11)	Determines the rate of growth in cash crop land in region 2 (has/year)	4,500
RLDRN (5.3)	Determines the rate flooded land is drained (has/year)	13,500
RPTN (5.426)	Determines farmers' participation of total establishment costs (proportion)	.2
EXTA (5.44)	Determines extension workers available (man-years)	250
TDNREQ (5.52)	Determines TDN desired for modern animals (tons/animal-year)	1.85
PRM (5.35a)	Determinen the price of milk (Ps/liter)	l
XDEL (5.43a)	Determines one-third of the time to complete modernization (years)	1
TLAVLO (5.1a)	Initial flood-free agricultural land potentially available in region 2 (thous, har)	533.3
TLAVL(0) (5.1a)	Initial flood-free agricultural land actually available in region 2 (thous, has)	431.0
TLAVUO1 (5.1b)	Initial agricultural land potentially available in sub- region 1 (thous, has)	2,169. 06
FLAVU1(0) (5.1%)	Initial agricultural land actually available in subregion 1 (thous, has)	1,720.73
FLAVU02 (4.10)	Initial applicultural land potentially available in subregion ? (thous, has)	1,676.0

TABLE II.1. (continued)

Definition (Equation No	Value	
TLAVU2(0) (5.1c)	Initial agricultural land actually available in subregion 2 (thous. has)	1,293.02
TGLSF0 (5.1d)	Initial potential grazing land in region l (thous. has)	3,137.95
TLDRNLO (5.2)	Initial land capable of drainage (thous. has)	3,137.95
TLFCU1(0) (5.20)	Initial land in food crops in subregion 1 (has)	40,915

Source: [13, 15, 36, 37, 38, 42, 53, 58, 59] and initial guesstimates and model tuning.
CHAPTER 6

AGRICULTURAL PRODUCTION--CROPS/CATTLE (AGPRAC)

Component AGPRAC generates the production of crops, pastures and cattle, and determines the yields of farm crops and the sales of cattle.

Crop Yields

Crop yields are a composite of the major crops grown in the Costa and are assumed to remain constant throughout the simulation. Increases in crop yields over time are allowed in the model as part of the modernization policies that will be discussed in Chapters 8 and 12.

Cash crop yields are computed for each agricultural region according to the crop mix assumed in Chapter 5. Further, it is assumed that yields in the more fertile lowlands (YLDCL) are 10 percent higher than in the uplands (YLDCU). With these assumptions and using the average yields derived from the Caja Agraria crop reports [7], it is possible to compute the average yield of individual commodities for each agricultural region. Then, using the same crops allocating weights of component LAMDAC, it is possible to compute a composite yield for each agricultural region. Table II.2 shows the computed average yield of each commodity, and Table III.3 on page 232 shows the composite yields used in

computing crop production and income below (Equation 6.29a). The computation of food crop yields (YLDFC) is simply a composite of the average yields of plantain and cassava as reported by the Caja Agraria for the period 1965-1969 [7]. $\frac{1}{}$

The five-year average yield of cash crops in the uplands used in the land allocation decisions of component LAMDAC is computed as:

$$EYLDCU(t) = EYLDCU(t-DT) + \frac{DT}{DEL8} (YLDCU(t-DT) -$$

$$EYLDCU(t-DT))$$
(6.1)

where:

EYLDCU	=	exponential average of cash crop yields	3
		in the uplands, used in Equation 5.35e	
		in LAMDAC (tons/ha-year)	

YLDCU = average yield of cash crops grown in the uplands (tons/ha-year)

DEL8 = average lag (years).

The increase in crop yields as a result of crop modernization efforts is computed in Equations 6.2. Here, it is assumed that crops reach their maximum target yield gradually, over a period of several years (DEL9 and DEL10). This length of time can be interpreted as being responsive to the crop modernization campaigns and could be a policy

 $[\]frac{1}{A}$ verage regional yields (tons/ha-year) of each commodity are: sesame--.66; cotton--1.4; corn--1.11; sorghum --1.8; rice--1.9; plantain--8.38; and cassava--8.28.

	Y1 (Tons	eld /ha-yr)	Cost* (Ps/ha-yr)			
Crop	Uplands	Lowlands	Uplands	Lowlands		
Sesame	.63	.693	400	408		
Cotton	1.34	1.47	1,639	1,672		
Corn	1.08	1.19	313	319		
Sorghum	1.73	1.9	564	575		
Rice		1.9		1,270		
Plantain	8.	38	600			
Cassava	8.	28	787			

TABLE II	.2. Av	verage	Annual	Yield	and	Initial	Costs	by
	Fa	arming	Sectors	3.				-

*Costs are reported for 1970 and adjusted to 1960 prices. Source: [7, 57, 69] variable. Target or desired yields have been derived from targets set by the Ministry of Agriculture for Colombian major crops $[58]^{1/}$ and then weighted by the crop allocating weights of component LAMDAC to obtain a composite average for each agricultural region.

$$YLDCU(t+DT) = YLDCU(t) + \frac{DT}{DEL9} * (DYLDCU - YLDCU(t))$$
(6.2a)

$$YLDCL(t+DT) = YLDCL(t) + \frac{DT}{DEL9} (DYLDCL - YLDCL(t))$$
(6.2b)

$$YLDFC(t+DT) = YLDFC(t) + \frac{DT}{DEL10} (DYLDFC - YLDFC(t))$$
(6.2c)

where:

YLDCU	the projected yield of cash crops in the uplands (tons/ha-year)
YLDCL	the projected yield of cash crops in the lowlands (tons/ha-year)
YLDFC	the projected yield of food crops (tons/ ha-year)
DYLDCU	the target yield of cash crops in the uplands (tons/ha-year)
DYLDCL	the target yield of cash crops in the low- lands (tons/ha-year)
DYLDFC	= the target yield of food crops (tons/ha-year)
DEL9, DI	EL10 = average lag (years).

Pasture Production

Before generating the production of cattle, component AGPRAC determines the output of fodder as total digestible

<u>1</u>/Target yields (tons/ha-year) for the major crops grown in the Costa are: sesame--.75; cotton--1.8; corn--1.6; sorghum--3.8; rice--4.1; plantain--10; cassava--10.

nutrients (TDN) from traditional and modern grasslands. First, the component computes (Equations 6.3) the regional average TDN yields of artificial and native grasses under both traditional and modern practices. Production of artificial and native grasses is estimated as an average of dry and rainy season yields based on the permanent carrying capacity of these pastures. Traditional artificial pastures in the uplands under continuous grazing and without fertilizer, yield 5.5 tons of dry forage per hectare annually or approximately 3.48 tons of TDN (1 kilogram of dry forage produces .633 kilograms of TDN [33].) This is enough feed to support 1.9 head of cattle throughout a year on the basis of an average nutritional requirement of 1.82 tons TDN/head-year. Further, it is assumed that artificial grasses in the more productive lowlands (mostly para grass) yield about 10 percent higher than upland pastures (mostly guinea and puntero grasses). It is also assumed that native grasses yield two-thirds less than artificial grasses [61, 66]. Improved pastures are estimated to have a carrying capacity 40 percent higher than pastures under traditional management. The average yields used in these computations are shown in Table III.3 on page 232.

$$CGO(t) = \frac{(CGOU + (TTGLU1(t) + TTGLU2(t)) + CGOL + TTGLL(t))}{TTGL(t)}$$
(6.3a)

$$CGOl(t) = \frac{(CGOUl*(TTGLUl(t) + TTGLU2(t)) + CGOLl*TTGLL(t))}{TTGL(t)}$$
(6.3b)

 $CG2(t) = \frac{(CGU2*(TLMOD2(t) + TLMOD3(t)) + CGL2*TLMOD1(t))}{TLMOD(t)}$ (6.3d)

· .

,CGO =	average TDN yield from traditional artificial grasses in the Costa (tons/ha-year)
CG01 =	average TDN yield from traditional native grasses in the Costa (tons/ha-year)
CG1 =	average TDN yield from improved artificial grasses in the Costa (tons/ha-year)
CG2 =	average TDN yield from improved native grasses in the Costa (tons/ha-year)
CGOU =	average TDN yield from traditional artificial grasses in the uplands (tons/ha-year)
CGOUl =	average TDN yield from traditional native grasses in the uplands (tons/ha-year)
CGU1 =	average TDN yield from improved artificial grasses in the uplands (tons/ha-year)
CGU2 =	average TDN yield from improved native grasses in the upland (tons/ha-year)
CGOL =	average TDN yield from traditional artificial grasses in the lowlands (tons/ha-year)
CGOL1 =	average TDN yield from traditional native grasses in the lowlands (tons/ha-year)
CGL1 =	average TDN yield from improved artificial grasses in the lowlands (tons/ha-year)
CGL2 =	average TDN yield from improved native grasses in the low lands (tons/ha-year)
TTGLL, 1	TTGLU1, TTGLU2 total land in traditional grazing in region 2, subregion 1 and subregion 2, respectively (has)

- TTGL = total Costa flood-free land in traditional pasture production--Equation 5.50 (has)
- TLMOD1, TLMOD2, TLMOD3
 - = total land in modern grazing in region 2, subregion 1 and subregion 2, respectively--Equation 5.48 (has).

Total digestible nutrients available to the traditional sector come from pasture in the flood-free and in the seasonally flooded lands. Crop residues, particularly of cotton, are added to the nutrient supply. Since there are indications of overgrazing in the region, the deteriorating effect of this practice is introduced into the model by Equations 6.4.

$$GRT(t) = \frac{TTGLR(t)}{TOPOPT(t)}$$
(6.4a)

where:

.

GRT	<pre>= grazing rate in the traditional sector (has/animal)</pre>
TTGLR	<pre>total traditional grazing land in the CostaEquation 5.51 (has)</pre>
TOPOPT	<pre>total traditional cattle population (animals)1/</pre>

$$RCON(t+DT) = max(RCON(t) + DT*C5*(GRT(t) - GRE), .1) \quad (6.4b)$$

RCON	range condition (a dimensionless number)
GRE	equilibrium grazing rate (which results in constant range condition) (has/animal)
C5	a parameter that determines the extent of influence of grazing rate upon range condition

 $[\]frac{1}{For}$ a detailed computation of this variable see subroutine DEMOG in the Appendix.

 $\max(a,b) = the maximum of a and b.$

Range condition is prevented from diminishing below an unrealistic limit by establishing a lower bound for RCON. The above equations stipulate that range condition increases or decreases over time if GRT is respectively greater or less than GRE. Given range condition, it is now possible to compute the total TDN available from the flood-free traditional grazing land.

TDNTG(t) = RCON(t)*(CGO(t)*CPLPT*TTGL(t) + CGO1(t)

$$*(1 - CPLPT)*TTGL(t))$$
 (6.5)

Т	CDNTG =	total traditional TDN from flood-free grasslands (tons/year)	
_ C	PLPT =	proportion of artificial grasses in traditional grazing land	
С	:GO =	as defined in Equation 6.3a	
С	GO1 =	as defined in Equation 6.3b	
Т	TOL =	as defined in Equation 6.3a.	
Т	'otal TI	DN available from crop residues and sea	sonally
flooded l	ands i	s computed as:	
TDNRE(t)	= C7*C	220#(TLCRU(t) + TLCRL(t))	(6.6)
TDNSF(t)	= TGLS	F(t)#C9#C10	(6.7)
where:	. •		
Ť	DNRE =	TDN available to traditional animals f crop residues (tons/year)	rom
T)	LCRU =	land in cash crops in the uplandsEqu 5.24 (has)	ation

- TLCRL = land is cash crops in flood-free lowland--Equation 5.11 (has)
- TDNSF = TDN available to traditional animals from seasonally flooded land (tons/year)
- TGLSF = total pasture land from seasonally flooded areas--Equation 5.14 (has)
- C7 = TDN yield of crop residues (tons/ha-year)
- C220 = a model parameter determining the proportion of cash crops producing residues used to feed traditional animals
- C9 = proportion of time that flooded land in available for grazing
 - Cl0 = TDN yield of grasses from seasonally flooded lands (tons/ha-year).

Finally, the total TDN available annually to the traditional sector, TDNT, is simply computed as:

```
TDNT(t) = TDNTG(t) + TDNRE(t) + TDNSF(t) (6.8)
```

Total TDN available to the modern sector depends on the alternative adopted. Land in transition from traditional to modern practices (TRSL) is considered part of the modern sector, but as producing forage at a rate intermediate between traditional and improved pastures. Briefly, alternatives 1 and 3 consider the improvement of both native and artificial grasses, as well as the production of forage crops in the latter to supplement nutrition during the dry season. Alternatives 2 and 4 consider the improvement of artificial pastures and the substitution of improved artificial grasses for traditional native grasses. In addition, alternative 4 includes the production of forage crops to supplement nutrition from grasses. The component computes the average yield

from improved and transition grasses as well as the average yield from forage crops in the form of silage. Given the average yields and the land in pasture and forages, it is possible to compute the total TDN available to the modern sector from each alternative. Forages are planted to the extent needed to make up for the deficit in nutrition as shown by the difference between the TDN obtained from grasses and a target level of TDN (TDND in Equation 6.9f). Although TDND could be a policy parameter, the model assumes this target is set at the nutrition level required to support four head per hectare, the current average carrying capacity in the Costa during the rainy season. Equation 6.9g, which computes area in forages, implies that land is taken out of modern production and transition in the same proportion. Mathematically, these computations are carried out in Equation 6.9.

$$CGA(t) = CG1(t)*CPLPT + CG2(t)*(1 - CPLPT)$$
(6.9a)

CTR(t) = CG3 + CPLPT + CG4 + (1 - CPLPT) (6.9b)

$$TDNF = \frac{C253*C254*TDNSG}{C250}$$
 (6.9c)

Further, CGA(t) = CG1, and

CTR(t) = CG3 for alternatives 2 and 4.

where:

CGA	-	TDN	yield	from	improved	gr	asses	(tons	/ha-year)
CTR	=	TDN ha-v	yield vear)	from	grasses	in	transi	ltion	(tons/	

- CG4 = TDN yield from transition native grasses (see Table III.3) (tons/ha-year)
- TDNF = TDN yield from forages (tons/ha-year)
- TDNSG = TDN yield from silage (tons/ton of silage)
- C250 = a model parameter to account for weight
 losses when green forage is converted into
 silage
- C253 = yield of green forage (tons/ha-cutting)
- C254 = the number of times forages are harvested during the growing season (cuttings/year).

In computing Equation 6.9c it has been assumed that forages are grown without irrigation and the growing season is thus restricted to the rainy period.

$$TMPL_{i}(t) = TLMOD_{i}(t) + TRSL_{i}(t) - TLF_{i}(t)$$
 (6.9h)

where:

TMPL = total land in pasture (has)
i = indexes the alternatives--i=3,4.

Finally, TDN available to the modern sector from

alternatives 3 and 4 is computed as:

```
TDNM_{i}(t) = TDNF*TLF_{i}(t) + TDNG_{i}(t)*TMPL_{i}(t), i=3,4 (6.91)
```

where:

TDNM = TDN output from modern land (tons/year).

Cattle Production

Demography

Curtle inventories and output are modeled dynamically as populations distributed over time and stage of production. The demographic model of the cattle population is divided into three age cohorts for females and two age cohorts for males (Figure II.4). The respective cohort length of flect the three production stages the model identifies: a prowing stage, a producing stage, and a stage in which animals without reproductive capabilities remain. The aging of animals through the first two cohorts is modeled by alcoributed lags. When females finally enter the old age cohort, their aging rate is no longer modeled, and cows remain there affected by deaths and sales through herd management policies. Based on the census data available [15] the model assumes that all animals two years old and less are included in the growing cohort; the producing cohort includes those anirals over two years.

The preceding production process is simulated by four calls to PFLET subroutine [52, 54], one for each sex cohort. Since the structure of the subroutine is alike for each cohort and differences in output arise only from differences in inputs, only the production of growing females

 $[\]frac{1}{The}$ distributed lag model used here has been adapted following Abkin [1, Chapter 3].



FIGURE 11.4. Cattle production cohorts.

will be discussed in detail here. The growing males cohort uses the male birth rate as an input, and the producing cohorts use as an input the cutput of the growing cohorts. Equations 1.10 define traditional or modern populations depending upon whether DLLDT is supplied with traditional or modern data.

```
CALL DELDT(BP(t-DT), RFOUl(t), RINTF1(t), DOROF, IDTF1,
DT, KOROF) (6.10a)
```

- BP = rate females enter the first cohort, i.e., the female birth rate--Fquation 6.19b (animals/year)
- **RFOUL = rate** growing females leave cohort 1 and enter the producing stage (animals/year).

- DGROF = average length of time females remain in Cohort 1 (years)
- KGROF = a parameter that determines the probability distribution for the length of time individual females remain in Cohort 1
- RINTF1, IDTF1
 = other variables associated with the use of
 the DELDT subroutine

DT = time increment of the model (years).

The purpose of this call to subroutine DELDT is to compute RFOUl(t), the rate females leave Cohort 1. This rate minus any losses (due to deaths, sales and transfers) becomes the input to Cohort 2, the producing stage, RFOUP1(t):

RFOUPI(t) = RFOUI(t) - RFOUI(t)*(DRL1(t) + PSFG(t))

```
+ PPFGT(t))*DT
```

(6.10b)

where:

- DRL1 = death rate of the growing population--Equation 6.16 (proportion/year)
- **PSFG** = proportion of growing females sold (proportion/year)1/
- PPFGT = proportion of growing females transferred out of a given sector (modern or traditional) (proportion/year)1/

The output of the male producing cohort, RMOU2, is the number of finished males available annually for immediate consumption, and the output of the female producing cohort, RFOU2, is the rate at which females leave the producing stage

 $[\]frac{1}{For}$ detailed computation of these variables see subroutine DEMOG in the Appendix.

as a result of old age. Given the basic model of the cattle demographic process, it is possible to compute the population in each cohort, total population and other variables of importance in the model.

The number of animals in the female and male cohorts are computed as time integrals of population flow rates. As stated earlier, old females are not transferred but stored for sales decisions based on herd management policies. Since the structural equations are alike for Cohorts 1 and 2, the computation of the number of growing females will be shown in Equation 6.11 below. Yet it must be remembered that each equation uses the variables relevant to each cohort, and that the producing cohorts use as an input the output of the growing cohorts. Equation 6.12 computes the population of old females. Total cattle populations in the traditional and modern sectors and in the region are computed simply by adding the populations from each cohort.

PFG(t+DT) = PFG(t) + DT*(BF(t) - DRL1(t)*PFG(t) - SLSFG(t) -

$$RFGT(t) - RFOUP1(t)$$
 (6.11)

.

where:

	PFG	=	population of growing females (animals)
ž,	SLSFG	=	<pre>sales of growing femalesEquations 6.22 (animals/year)</pre>
	RFGT	=	rate growing females are transferred Equations 5.53 (animals/year)
	RFOUP1	=	rate females leave the growing stage and enter the producing stageEquation 6.10b (animals/year).

$$OLDF(t+DT) = OLDF(t) + DT*(RFOUP2(t) - SOLDF(t) - DRL2(t)$$

 $*OLDF(t))$ (6.12)

- OLDF = population of old females (females which have concluded the reproductive life) (animals)
- SOLDF = sales of old females--Equation 6.22e (animals/year)
- RFOUP2 = rate females leave the producing stage and enter old age--Equation 6.10b where transfers = 0 (animals/year)
- DRL2 = death rate of the producing population--Equation 6.16 (proportion/year).

Once cattle demography has been simulated, the model computes the variables affecting the transition rates of this population. Live birth rates and death rates are computed as a function of the level of nutrition (TDNA). Death rates are computed separately for the growing and producing populations as shown in Equations 6.13. The table look-up functions [52, 54] used in these equations compute traditional or modern birth and death rates depending upon whether the VAL arrays are supplied with traditional or modern data. Figures II.5 graphically depict these functions.

BR(t) = TABLIE(VALB, SMALLB, DIFFB, KB, TDNA) (6.13a)
DRG(t) = TABLIE(VALDG, SMALLDG, DIFFDG, KDG, TDNA) (6.13b)
DRP(t) = TABLIE(VALDP, SMALLDP, DIFFDP, KDP, TDNA) (6.13c)



FIGURE II.5.b. Modern birth rate versus total digestible nutrients.



FIGURE II.5.d. Modern death rate of growing cohort versus total digestible nutrients.



FIGURE II.5.e. Traditional death rate of producing cohort versus total digestible nutrients.



FIGURE II.5.f. Modern death rate of producing cohort versus total digestible nutrients.

BR

E	live	bir	th	rat	e-	-pro	portic	on of	pro	oducin	Ľ.
	femal	les	cal	vin	g	per	year.	ln	the	model	is
	also	tak	en	ลร	а	preg	snancy	rate			

- DRG = death rate--proportion of growing population dying per year
- DRP = death rate--proportion of producing population dying per year
- VAL = an array of numbers which defines the dependent argument of the function
- DIFF = the fixed differences between values of TDNA
- K = the number of line segments used to approximate the birth or death rate functions
- TDNA = total digestible nutrients (tons/animal-year) --the independent argument of the function. $\frac{1}{2}$

In reality, births and deaths do not change instantaneously with changes in nutritional levels and/or population sizes, but rather lag behind changes in these variables. The variables BR, DRG and DRP must therefore be operated on to introduce these lag effects. Equation 6.14 shows this computation for birth rates:

$$BR2(t) = BR2(t-DT) + \frac{DT}{DEL1} (BR(t-DT) - BR2(t-DT))$$
 (6.14)

 $[\]frac{1}{F}$ For a detailed computation of this variable see subroutine DEMCG in the Appendix.

BR2 = actual live birth rate (proportion/year)
DEL1 = lag parameter (years).

The actual death rate of the growing population, DR1, and the actual death rate of the producing population, DR2, are computed using similar equations to the one above.

Nutrition and management are not the only factors affecting birth and death rates. Diseases have a major role in determining the value of these variables, and we are particularly concerned here with brucellosis and footand-mouth disease (FMD) which are epidemic in the region. The effect of brucellosis on birth rates is introduced into the model by the variable CBANG which depends on the proportion of cows treated. This is shown in Equation 6.15 where the variable BR2 is taken as a pregnancy rate.

$$BRL2(t) = BR2(t) + (1 - CBANG(t))$$
 (6.15)

where:

BRL2 = the effective live birth rate (proportion/year)

CBANG = proportion of pregnant cows aborting due to brucellosis, where $0 \le CBANG \le .04.1/$

The effect of FMD on death rates is introduced into the model by the variable DRA which depends on the proportion of animals treated annually. This effect is shown in Equation 6.16 for the growing cohort.

 $[\]frac{1}{For}$ a detailed computation of this variable see subroutine DEMOG in the Appendix.

DRL1(t) = DR1(t) = DRA(t)

where:

- DR1 = the actual death rate of growing animals--Equation 6.14 (proportion/year)
- DRA = proportional increase in death rates due to FMD, where $1 \le DRA \le 1.33$.

Before computing total births it is necessary to determine the number of cows capable of calving found in the population of producing females. This is done in the model with Equations 6.17 by computing the number of infertile cows and those affected by severe cases of mastitis, caused by FMD, that have to be discarded from the breeding herd. Equations 6.17 are the time integrals of female population flow rates where affected animals come out from each of the transition rates in the same proportion. Equation 6.17a incorporates both the effects of malnutrition and infectious abortion on fertility (see Chapter 2). The effect of malnutrition is introduced by the variable PIFNU which depends on the level of nutrition available and a rate of response to changes in nutrition. The function has an upper bound that is reached when nutrition is below a predetermined level; the lower bound indicates that a minimum of infertility is always present (due to heredity and/or other causes) despite high levels of nutrition [24]. The effect of brucellosis is introduced by the variable, CB(t) = BR2(t) * CBANG(t), which determines the proportion of cows aborting (see Equation

6.15); it is also assumed that 10 percent of the cows aborting become infertile [31].

$$FINF(t+DT) = PFINF(t)*[PFP(t) - DT*(DFP(t) + RFOU2(t) + SLMAS(t))] - SLINF(t)*DT + DT*(CB(t)*0.10 + PIFNU(t))*[(PFP(t) - DT*(DFP(t) + RFOU2(t) + SLMAS(t))*(1 - PFINF(t))) + DT*(RFOU1(t) - RFPT(t) - SLFER(t))] (6.17a)$$

•	FINF	=	producing females which are infertile (animals)
	PFINF	=	proportion of producing females which are infertile
	PFP	=	population of producing femalesEquation 6.11 (animals)
	DFP	×	actual deaths of producing females Equation 6.20b (animals/year)
	RFOU2	=	actual rate at which producing females leave the producing stageEquations 6.10 (animals/ year)
	SLMAS	=	sales of producing females with mastitis due to FMDEquations 6.22 (animals/year)
	SLINF	=	sales of infertile femalesEquation 6.22 (animals/year)
	СВ	=	producing females aborting per year (proportion/year)
	PIFNU	=	producing females becoming infertile due to malnutrition where $.01 \le PIFNU \le .05$ (proportion/year)
	RFOUl	-	actual rate at which females leave the grow- ing stageEquations 6.10 (animals/year)
	RFPT	=	rate producing females are transferred out of a sectorEquations 5.53 (animals/year)
	SLFER	Ħ	<pre>sales of fertile producing femalesEquations 6.22 (animals/year).</pre>

Equation 6.17b below assumes that 18 percent of the cows without treatment against FMD get the disease; further, it is assumed that 5 percent of the cows getting FMD are affected by severe mastitis [31]. This equation also implies, as a simplification, that treatment against FMD is applied to animals in each cohort in the same proportion.

```
PMAS(t+DT) = PFMAS(t)*[PFP(t) - DT*(DFP(t) + RFOU2(t))
```

```
+ SLINF(t))] - SLMAS(t)*DT + (1 - PATAF(t))
*DT*0.18*0.05*[(PFP(t) - DT*(DFP(t) + RFOU2(t)
+ SLINF(t)*(1 - PFMAS(t))) + DT*(RFOU1(t) -
RFPT(t) - SLFER(t))] (6.17b)
```

where:

	FMAS = producing females with mastitis due to FMD (animals)
	PFMAS = proportion of producing females with mastitis
	PATAF = animals treated against FMDEquation 6.28c (proportion/year).
	Finally, the number of fertile cows is simply computed
as:	

$$FER(t) = PFP(t) - FMAS(t) - FINF(t)$$
(6.18)

Given the population of fertile females and the effective birth rate, it is possible to compute total births. It is assumed in the model that births are evenly distributed between males and females. BA(t) = BRL2(t)*FER(t) (6.19a) BF(t) = 0.5*BA(t) (6.19b)

BM(t) = BA(t) - BF(t) (6.19c)

where:

BA = total animal births (animals/year)
BF = total female births (animals/year)
BM = total male births (animals/year)
BRL2 = as defined in Equation 6.15.

Animal deaths are computed for growing and producing animals and for each sex category. Equations 6.20 show this computation for females only.

DFG(t) = PFG(t) * DRL1(t) (6.20a)

DFP(t) = PFP(t) * DRL2(t) (6.20b)

where:

DFG		=	actual deaths of growing females (animals/year)
DFP		=	actual deaths of producing females (animals/year)
PFG		=	population of growing females Equation 6.11 (animals)
PFP		=	population of producing females Equation 6.11 (animals)
DRL1,	DRL2	=	as defined in Equation 6.16.

Milk and Animals Output

Next, the model computes the output from cattle in the form of milk and total sales, and the marketing equations used in the generation of prices. The quantity of milk produced depends on the number of fertile cows, cows getting foot-and-mouth disease (FMD) and the level of nutrition. It is assumed that cows affected by FMD produce 15 percent less milk than healthy cows [31].

```
QM(t) = [FER(t)*(PATAF(t) + (] - PATAF(t))*0.82 + (] -
```

PATAF(t))*0.18*0.85)]*PFLAC(t)*C202*YMA*TABEXE

(6.21a)(VAL6, .31, .31, 4, TDNA)

where:

QM	= quantity of milk produced (lite	ers/year)
PFLAC	; = proportion of females lactating	5
YMA	= average milk output per cow (15	lters/cow-year)
C202	= a model parameter determining t lactating cows which are milked	the numbe r of 1 (proport ion)
TABEX	<pre>KE(VAL6,, TDNA) = a subprogram which introduces a duction factor determined by th nutritionTDNA.</pre>	a milk pro- ne level of
The e	expected production of milk used in	n Equation
5.35a of comp	oonent LAMDAC is computed as:	
EQMT(t) = FEF	<pre>RT(t)*PFLACT(t)*CT202*YMAT</pre>	(6.21b)
where:		
EQMT	= the expected milk production in -ional sector (liters/year).	the tradi-

Cattle sales are computed for each cohort and are part of the herd management policies introduced into the

model. Although the sales policies have been designed with enough flexibility to permit simulation of farmers' behavior the current formulation allows little supply response. Cash flow imbalances that might induce sales are not considered as a factor influencing sale decisions, and the model assumes that these decisions depend on prices and the level of nutrition. Price changes may have a short-term response effect but, in the long run, sales seem to be dominated by nutritional considerations. As a general sales policy the model assumes the following priorities (for other than finished males): (1) old cows; (2) cows with reproductive problems; . (3) growing males; (4) growing females; (5) fertile cows; and (6) producing males.

According to the preceding assumptions, sales are used to control the cattle population to maintain a prescribed level of nutrition, and animals exceeding carrying capacity are marketed following the order discussed earlier. If pastures are being undergrazed, the decision mechanism operates to reduce sales and increase the retention of animals until the appropriate grazing rate is achieved. This nutrition effect is introduced in the sales equations by the variable PAN. This variable is recalculated each time a sale is performed.

The sales equations describe a family of supply curves which first are completely inelastic, then are positively sloped, and finally become completely inelastic (Figure II.6). The inelastic portions of the curve place an upper bound to sales preventing the herd from being liquidated, and a lower



FIGURE II.6. The cattle sales function.

bound indicating that a minimum of animals are marketed from the herd despite low price incentives and/or excess carrying capacity. These bounds are set by the model parameters BMN, BMX, AMX and AMN which permit simulating farmers' behavior and herd management policies (see Table II.3). Since there are similar relationships between management of the traditional and modern herds and for each of the six sale groups, the sale of old cows will be shown in detail here. Response to price is given by the relationship $\left(\frac{PAP}{PAPO}\right)^{ELAS}$ in Equation 6.22c, and nutrition relationships are given by $\frac{PAN}{COHS}$ in Equation 6.22d. $\frac{1}{2}$

 $[\]frac{1}{COHS}$ is used here as a general form to indicate the number of animals in a given sales group.

	Management F	ractice
Sales Group and Parameter $\frac{1}{}$	Traditional	Modern
Old Cows:		
AMX1 AMN1 BMX1 BMN1	0.75 0.50 0.75 0.50	1.00 0.90 1.00 0.90
Intertile cows:	0.25	1 00
AMX2 AMN2 BMX2 BMN2	0.15 0.18 0.15	0.80 0.95 0.80
Cows with mastitis:	1 00	1 00
AMX 3 AMN 3 BMX 3 BMN 3	0.70 1.00 0.70	0.80 1.00 0.80
Growing males:	0.00	0 20
AMX4 AMN4 BMX4 BMN4	0.23 0.11 0.14 0.11	0.11 0.20 0.11
Growing females:	0.01	0.25
AMX5 AMN5 BMX5 BMN5	0.13 0.17 0.13	0.13 0.19 0.13
Fertile cows:	0.06	0.06
AMX6 BMX6 BMN6	0.02 0.05 0.02	0.02 0.05 0.02
Producing males:	0.20	0.20
AMX7 AMN7 BMX7 BMN7	0.20 0.10 0.15 0.10	0.10

TABLE II.3. Maximum and Minimum Proportions of Cattle Sales.

 $1/_{AMX_1}$, BMX₁ and AMN₁, BMN₁ are model parameters determining upper and lower bounds to cattle sales (Equations 6.22c and 6.22d)--proportion/year.

Source: Guesstimates and model tuning.

$$TOPEQ(t) = TDN(t)*NREQ$$
 (6.22a)

$$PAN(t) = TOPOP(t) - TOPEQ(t)$$
(6.22b)

- NREQ = the reciprocal of the TDN required per animal (animal-year/tons TDN)

.

PAN = defines the difference between the current animal population TOPOP and the equilibrium population TOPEQ (animals).

$$PRES1(t) = min{BMX1, max[BMN1, PRSI1* $\left(\frac{PAP(t)}{PAPO}\right)^{ELAS1}$]} (6.22c)

$$PSFO(t) = min{AMX1, max[AMN1, PRES1(t) + C206* \frac{PAN(t)}{OLDF(t)}$$
]} (6.22d)

$$SOLDF(t) = OLDF(t)*PSFO(t)$$
(6.22e)

$$PAN(t) = PAN(t) - SOLDF(t)$$
(6.22f)$$

Market Model

Component AGPRAC computes the demand and supply of beef which are part of the simple market model used in determining the price of cattle. First, the model determines the number of animals marketed for consumption and/or export from the Costa herd (Equation 6.24). Although some of these animals are finished in other regions, it has been assumed they are slaughtered and enter the retail market as they leave the Costa farms. Nevertheless, this simplifying 156

assumption does not greatly alter the total supply of beef as computed in the model.

(SLSMGT(t) + C212*SLSMGM(t)) (6.24)

SUPCTA	=	supply from the Costa herd (animals/year)
SLSCCT, SLSCCM	=	sale of cows with reproductive problems traditional and modern, respectively (animals/year)
SLSMLT, SLSMLM	12	sale of mature males traditional and modern, respectively (animals/ year)
SOLDFT, SOLDFM	=	sale of old cows traditional and modern, respectively (animals/ year)
TDTHST, TDTHSM	=	total animal deaths traditional and modern, respectively (animals/year)
SLFERT, SLFERM	-	<pre>sale of fertile cows traditional and modern, respectively (animals/ year)</pre>
SLSFGT, SLSFGM	*	sale of growing females traditional and modern, respectively (animals/ year)
SLSMGT, SLSMGM	æ	sale of growing males traditional and modern, respectively (animals/ year)

Non-Costa cattle population, TOPOPK, and supply are exogenously determined in Equations 6.25. Cattle population is assumed to grow in a non-cyclical, exponential fashion and its rate of growth could take different values in order to test the effect of government policies on the development of the non-Costa cattle economy. Beef supply is computed as the off-take for exports and slaughter from this population, where the extraction ratio reflects the oscillations of the long-term cycle (see Chapter 3). This cycling effect is approximated in Equation 6.25b by a TABLIE function which completes a cycle every seven years with the extraction ratio reaching a simulated peak at .17 and a bottom at .118.

TOPOPK(t+DT) = TOPOPKO*EXP(C217*t)(6.25a)

```
SUPB(t) = SUPCTA(t) + TOPOPK(t)*TABLIE(VAL9, 0, 1, 7, AMOD(t,7)) (6.25b)
```

торорк	=	total non-Costa cattle population (animals)
тороко	8	non-Costa cattle population at the beginning of simulation (animals)
EXP	=	the exponential function
C217	8	rate of growth of non-Costa cattle popula- tion (proportion/year)

= total Colombian supply of beef (animals/ SUPB year) TABLIE(VAL9, ..., AMOD) = a subprogram which introduces a cycling factor on extraction ratios determined by the length of the long-term cycle--seven years = simulated time in the model (years). t Domestic demand for beef is computed in Equation 6.26a and its growth is due to population, income and price effects. . 15 25-1

$$DEM(t+DT) = DEM(t) + DT*(DEM(t)*RDEM(t))$$
(6.26a)

RDEM(t) = ELASI*C237 + ELASP*C238 - ELASD*

$$\frac{PA(t) - PA(t-DT)}{PA(t-DT)*DT}$$
(6.26b)

where:

DEM	=	domestic demand for beef (animals/year)
RDEM		the rate of growth of demand (proportion/ year)
PA	E	market price of finished malesEquation 7.2 (Ps/animal)
ELASI	=	income elasticity of demand for beef
ELASP	E	population elasticity of demand for beef
ELASD	=	price elasticity of demand for beef
C237	=	rate of increase in income (proportion/year)
C238	=	rate of increase in population (proportion/ year).

Total demand for Colombian beef, TDEM, is simply computed as the sum of domestic demand, official exports, EXPL, and illegal exports, UNEXPL. Due to the lack of statistics on illegal exports, these are handled as a constant throughout the simulation. Yet it would be realistic to treat UNEXPL as a variable since it can be expected this border trade will be responsive to market conditions in Colombia as well as in the neighboring countries (mainly Venezuela and Ecuador). Official or registered exports are computed from recorded statistics between 1964 and 1971, and from projected targets from 1972 forward (see Table I.1, p. 19 for figures and sources).

$$TDEM(t) = DEM(t) + EXPL(t) + UNEXPL \qquad (6.26c)$$

Disease Control

Since foot-and-mouth disease and brucellosis seriously impair cattle production, and the Colombian government is committed to their control and eradication (see Chapter 2), it is relevant to the model to include some equations to test the effect of control measures. Component AGPRAC includes a simple exogenous model which permits evaluation of disease control policies. Here it is assumed that all the effort is directed toward the traditional sector and that all animals in the modern sector are treated according to recommended practices. Further, it is assumed that before the campaign starts, animals are treated at a constant proportion, but afterwards (i.e., after 1971) this proportion gradually increases until it reaches the value one. At this point the model indicates that all the cattle population is being treated. This effect is introduced in the model by the exponential function involving the variable TCAD(TCAD = t -TDO) where TDO is the year at which the disease control program starts. The treatment against brucellosis is shown in Equations 6.27 below. This treatment is applied only once when heifers are three to six months old, but for simplicity the model assumes that heifers are treated at birth.

ATABTT(t) = BFT(t) * max[1 - C198 * EXP(-C199 * TACAD), C242] (6.27a)

where:

- ATABTT = heifers treated against brucellosis in the traditional sector (animals/year)
- BFT = total traditional female births--Equation 6.19b (animals/year)
- C242 = proportion of heifers treated without the campaign
- **C198** = proportion of heifers left untreated (C198 = 1 - C242)
- C199 = model parameter regulating the shape of the exponential curve.

The movement of the treated heifers is tracked through the growing stage until they reach the producing stage, and this allows the model to compute the proportion of cows treated, FCTAB, which is needed in determining the variable CBANG used in Equation 6.15. The equations computing PCTAB have the same structure for both the traditional and modern sectors, with the exception that the latter keeps track of the treated females transferred from the traditional sector. Equations 6.27b through 6.27d show this computation for the traditional sector. POFTB(t) =

$$\frac{\text{ATABTT}(t) \text{*}\text{DT} + \text{PGFTBT}(t-\text{DT}) \text{*}(\text{PFGT}(t) - \text{BFT}(t-\text{DT}) \text{*}\text{DT})}{\text{PFGT}(t)} \quad (6.27b)$$

CTABT(t) = RFOUT1(t-DT) * PGFTBT(t-DT) * DT + PCTABT(t-DT) *

$$(PFPT(t) - RFOUT1(t-DT) = DT)$$
(6.27c)

$$PCTABT(t) = \frac{CTABT(t)}{PFPT(t)}$$
(6.27d)

where:

PGFTBT	=	proportion of growing females treated against brucellosis in the traditional sector
PFGT	=	population of growing females in the traditional sectorEquation 6.11 (animals)
CTABT	×	cows treated against brucellosis in the traditional sector (animals)
RFOUT1	=	rate females leave the growing stage Equation 6.10b (animals/year)

PCTABT = proportion of cows treated against brucellosis in the traditional sector.

Although the campaign against FMD and brucellosis is being carried out simultaneously, the data available [31] do not allow a breakdown of expenditures between the two programs. This problem was simplified in the model assuming that treatment against FMD was the only one depending on government expenditures. But since farmers have been and continue to treat a part of the herd on their own, the model computes both animals treated privately and by campaign personnel. Equation 6.28b implies that eventually all animals could be treated by the campaign, in which case farmers who have been
treating their herds privately will be charged for the service. The current program involves charging services to medium- and large-size farmers, and providing subsidized services to small farmers [31].

Since treatment against FMD is applied two or three times every year to all animals, the computation of the proportion of animals treated, PATAFT, used in determining the variable DRA in Equation 6.16 is more straightforward than the preceding for brucellosis.

$$ATAFT(t) = min\left(\frac{EXPAFT(t)}{COSTFT}, TOPOPT(t)\right)$$
(6.28a)

where:

ATAFT	 government treated animals against FMI the traditional sector (animals)) in
EXPAFT	government expenditures against FMDa policy variable (Ps/year)	3
COSTFT	<pre>government cost of treatment against l (Ps/animal-year)</pre>	PMD
TOPOPT	<pre>total traditional cattle population in Costa (animals).</pre>	n the
ATAFPT(t) = (TO	POPT(t) - ATAFT(t))#max[1 - C200#EXP	
(-C	201#TACAD), C244)]	(6.28b)
$PATAFT(t) = \frac{ATA}{TOP}$	FTT(t) OPT(t)	(6.28c)
where:		
ATAFPT	privately treated animals against FMD traditional sector (animals)	in the

ATAFTT = total privately and government treated animals against FMD in the traditional sector (animals)

PATAFT	<pre>= proportion of animals treated against FMD in the traditional sector</pre>
C244	proportion of animals treated privately without the campaign
[.] C200	<pre>proportion of animals left untreated (C200 = 1 - C244)</pre>
C201	model parameter regulating the shape of the exponential curve.

Equation 6.28b above also implies a promotion and/or diffusion effect among farmers due to the campaign. As time of campaign passes, the proportion of animals treated privately increases approaching one.

Agricultural Accounting

Finally, component AGPRAC performs the macroeconomic accounting for the agricultural production. This section also simulates farmers' varying expectations about the accounting variables during the planning horizon. These expectations are introduced in component LAMDAC for the computation of discounted profitabilities. First, revenues and costs of crops production are generated. Costs are a composite of the major crops grown in each of the agricultural regions and are computed using the same crop allocating weights of component LAMDAC. It is also assumed that costs increase over time at the inflation rate for farm inputs. Table II.2 on page 128, shows the computed average cost of each commodity, where costs in the lowlands are slightly higher to account for increased harvesting costs because of higher yields. Equations 6.29 show these computations for cash crops in the uplands.

ARCRU(t	;) = TLC	RU(t)#YLDCU(t)#PCROPU(t)	(6.29a)
where:			
	ARCRU	 accounting revenue from cash crops in uplands (Ps/year) 	the
	TLCRU	<pre>= total land in cash crops in the upland Equation 5.24 (hectares)</pre>	s
	YLDCU	the projected yield of cash crops in t uplandsEquation 6.2a (tons/ha-year)	he
	PCROPU	= the projected producer price of cash c from the uplandsEquation 7.4a (Ps/to	rops n).
ACCRU(t) = TLCI	RU(t)*CSTHCU(t-DT)*(1 + DT*RCST)	(6.29b)
where:			
	ACCRU	accounting cost of cash crops in the uplands (Ps/year)	
	CSTHCU	the average cost of cash crops in the uplands (Ps/ha-year)	
	RCST	<pre>= the rate of increase in farm costs (proportion/year).</pre>	

Next, the model generates costs and revenues from cattle. Operating costs have been computed separately for the pasture lands and for the herd. The operating cost of land in the traditional and in the modern sectors are computed by Equations 6.30 below. Equation 6.30a is flexible to account for the seasonally flooded lands in TTGLR where it is likely that expenditures on range management are kept to a minimum.

OPCLNT(t) = TTGLR(t) * CLNDT(t) * C267(6.30a)

}

where:

OPCLNT	=	total operating costs of traditional grass- lands (Ps/year)
TTGLR	Ħ	total traditional grazing land in the Costa Equation 5.51 (hectares)
CLNDT		average operating cost of traditional grass- lands (Ps/ha-year)
C267	æ	a model parameter controlling total land where costs are incurred (proportion).

Equation 6.30b has been designed with enough flexibility to be used in any alternative. If the model is supplied with data for alternatives 1 and 2, this equation reduces to TLMOD(t)*CLNDM(t). Maintenance costs in lands in transition (TRSL) are included in establishment costs.

OPCLNM(t) = TLMOD(t) * CLNDM(t) * (1 - CPLF(t)) + TLF(t-DT) *

CSRFGH(t) + TLF(t) * CSHARV(t) * C253 * C254 (6.30b)

where:

OPCLNM = operating cost of modern grasslands (Ps/year)

- TLMOD = total land in modern grazing--Equation 5.49 (hectares)
- CLNDM = the average operating cost of modern grasslands (Ps/ha-year)
- CPLF = proportion of modern land in forage crops--Equation 6.9f
- TLF = total land in forage crops--Equation 6.9g (hectares)
- CSRFGH = cost of replanting and growing forages (Ps/ma-year)
- CSHARV = cosi of harvesting and storing forage (Ps/ton)
- C253 = a parameter determining the yield of forages
 per cutting (ts/ha-cutting)

C254 = a parameter determining the number of cuttings per year.

Operating costs per animal include labor, drugs, supplemental feed if any, and other miscellaneous costs. The computation of total animal costs for the traditional and modern sector use the same structure shown in Equation 6.31 for the traditional herd.

$$OPCLAT(t) = TOPOPT(t)*CSTANT(t)$$
 (6.31)

where:

OPCLAT =	operating costs of traditional cattle (Ps/year)
торорт =	total cattle population in the tradi- tional sector (animals)
ĊSTANT =	average operating cost of traditional animals (Ps/animal-year).

Another component of operating costs is depreciation of grazing land capital and equipment, as well as taxes on land. Because of lack of data on initial capital stock, the model simplifies the computation of replacement investments in cattle production by a lump annual sum per unit of land in production. Total value of depreciation is determined exogenously in Equation 6.32 for the traditional and modern sectors. The value of EQLM varies with the alternative chosen^{$\frac{1}{2}$} and the corresponding value (EQLT) for the traditional sector is adjusted by a model parameter to account for the flooded grasslands (see discussion for Equation 6.30a

 $\frac{1}{EQLM} = 110$; 120; 150; 170 Ps/ha-year for alternatives 1, 2, 3 and 4, respectively. above). The value of the property tax is based on the assessed capitalized value of land (see Chapter 2) and is computed in Equation 6.33 for the traditional sector.

$$CAPDEM(t) = TLMOD(t) * EQLM$$
(6.32)

where:

CAPDEM	modern sector replacement investment in grasslands and equipment (Ps/year)	
EQLM	<pre>= capital costs for modern cattle produc- tion (Ps/ha-year)</pre>	i
TLMOD	= as defined in Equation 6.30b (hectares)	•
VLDTXT(t) = VLA	ANDT(t-DT)*TAXLND*C248 (6.33)

where:

VLDTXT	=	value of taxes on land in traditional cattle production (Ps/year)
VLANDT	=	capitalized asset value of land in traditional cattle productionEquation 9.15 (Pesos)
TAXLND		the land tax rate (proportion/year)
C248	2	a model parameter adjusting the capitalized value of land to the cadastral (assessed) value (proportion).

Finally, the special taxes on cattle discussed in Chapter 2 are computed in Equations 6.34. The general inventory tax (TAXC3) based on the net investment on cattle has been the most difficult to estimate because of the complexity involved in the accounting of assets and liabilities of farmers. This problem was circumvented in the model by assuming a constant tax rate per animal estimated from Ministry of Agriculture sources. $\frac{1}{}$ The assessed liveweight tax rate (PKGR) used in computing the selective inventory tax (TAXC1) was recorded from values set by the government between 1967 and 1970 $\frac{2}{}$ and then extrapolated by means of a TABEXE function [52]. These computations for the traditional sector are shown below:

TAXCT1(t) = (PMGT(t) * C222 + PMPT(t)) * C223 * PKGR . (6.34a)

where:

TAXCT1 = the traditional cattle selective inventory tax (Ps/year)

PMGT, PMPT

- = traditional growing and producing male population, respectively--Equation 6.11 (animals)
- PKGR = the assessed liveweight tax rate (Ps/ kilogram-year)
- C222 = a model parameter determining the proportion of growing males over one year of age (proportion)
- C223 = the animal-liveweight conversion factor (kilograms/animal).

TAXCT2(t) = FEMSCT(t) * C277 * SFTAX + SLSMLT(t) * C278 * SMTAX (6.34b)

where:

TAXCT2 = the traditional cattle export and consumption tax (Ps/year)

<u>l</u>/Ministerio de Agricultura, "Estudio Sobre la Renta Presuntiva." Bogota, 1971, pp. 1-111. (Mimeographed.)

 $\frac{2}{\text{The Ministry of Agriculture sets the liveweight}}$ price at the end of each fiscal year. The values used in the model were obtained by personal information.

- FEMSCT = total traditional females sold for consumption (animals/year)
- SLSMLT = total traditional adult males sold for consumption (animals/year)
- SFTAX = the female consumption tax rate (Ps/animal)
- SMTAX = the male consumption tax rate (Ps/animal)
- C277 = a model parameter determining the proportion
 of females sold for immediate consumption
 (proportion)
- C278 = a model parameter determining the proportion of males sold for immediate consumption (proportion).

Equation 6.34b above needs a further discussion. First, it implies that the tax on animals sold for export is paid by the producer. The coefficients C277 and C278 introduce flexibility into the model to determine those animals which are sold to be finished in other regions and whose tax is not paid by the Costa producers. The variable FEMSCT includes cows with reproductive problems, old cows, fertile cows and heifers; and SLSMLT includes finished steers and males sold out of the producing cohort.

 $TAXCT3(t) = TOPOPT(t) * C279 \qquad (6.34c)$

where:

- TAXCT3 = the traditional general inventory tax (Ps/year)
- TOPOPT = total traditional cattle population (animals)
- C279 = the estimated general inventory tax rate (Ps/animal-year).

After computing operating costs, the model generates the cost of establishing any modern alternative at market factor prices, and then determines the actual cash outlays made by farmers. Equation 6.35 is a composite of costs of improving native and artificial pastures, planting artificial pastures and forages, and building storage for forages. It is clear that not all these costs apply to every alternative; a subprogram in the model assigns the relevant costs to each alternative, and in addition, computes costs per unit of land weighted by the proportion of land in artificial pastures and forages.

TEC(t) = CSIMNP(t) + CSIMAP(t) + CSPLAP(t) + CSPLFG(t)

where:

TEC	Ħ	total	establishment	costs	at	market
		prices	(Ps/ha)			

- CSIMNP = average cost of improving native pastures (Ps/ha) <u>1</u>/
- $CSIMAP = average cost of improving artificial pastures (Ps/ha) \frac{1}{2}$
- CSPLAP = average cost of substituting artificial for native pastures (Ps/ha)
- CSPLFG = average cost of establishing forage crops $(Ps/ha) \frac{1}{2}$
- CSTGH = average cost of building forage storage (Ps/ha)l/

 $\frac{1}{For}$ a detailed computation of these variables see subroutine MODCRD in the Appendix.

Equation 6.35 above is the general approach of accounting for establishment expenditures at their opportunity cost. Yet some of the inputs required can be supplied on the farm at no extra cash expense, decreasing the need for the use of credit and/or savings. Examples of these inputs are family labor, materials for fencing and building. existing tools and equipment, etc. The function ALPH1 computed in Equation 6.36 is an attempt to simulate the response of farmers' behavior to changing profitabilities. This behavior includes changing attitudes toward work and leisure, a more efficient use of the inputs at hand (including management), and incentive to utilize more fully the farm natural resources. As shown in Figure II.7 ALPH1 depends on a profitability threshold (C235) below which there is no incentive for farmers to use their resources intensively. As profitability increases, farmers exploit their resources more fully until they reach a limit (C234) where it is assumed that the ability to use on-farm resources has been exhausted. A parameter (C236) determines how rapidly the attitudes change with changes in the profitability criterion. It is clear that a wide range of farmers' behavior can be simulated by appropriately assigning values to these three parameters.

ALPH1(t) = C234 + min[1 - C234, (1 - C234)*EXP(-C236*)]

(PDR(t-DT) - C235))] (6.36)



FIGURE II.7. The on-farm resource use response function.

where:

ALPH1	= a variable which introduces the effect of the profitability criterion PDR upon total
	outlays for establishing an alternative (proportion)

- PDR = the relative profitability differential of Equation 5.30 (dimensionless)
- C23[#] = a model parameter determining the minimum proportion of establishment costs met with outside resources (proportion)
- C235 = the on-farm resources intensity of use response threshold (dimensionless)
- C236 = the rate of on-farm resource use response with respect to profitability (dimensionless)
- min[a,b] = the minimum value between a and b

EXP = the exponential function.

Finally, the total cash requirements for establishing an alternative, TCEC, are simply computed as:

$$TCEC(t) = TEC(t) * ALPH1(t)$$
(6.37)

Given the components that enter in the formation of cattle production costs, it is possible to generate the accounting costs, ACLA. Equation 6.38 makes this computation for the modern sector. Accounting costs in the traditional sector are computed using only the first five terms of Equation 6.38 below.

ACLAM(t) = OPCLAM(t) + OPCLNM(t) + CAPDEM(t) + VLDTXM(t) +

TAXCM(t) + ALINT(t) + ALREP(t) + (TCEC(t) -

$$ALPH1(t) * CSTGH(t)) * \frac{TRSL(t)}{3* XDEL} + RPCAPT(t)$$
 (6.38)

where:

ACLAM	=	total accounting costs of modern sector (Ps/year)
OPCLAM		operating costs of modern cattleEquation 6.31 (Ps/year)
OPCLNM	•	as defined in Equation 6.30b (Ps/year)
CAPDEM	a	as defined in Equation 6.32 (Ps/year)
VLDTXM	×	value of taxes on land in modern cattle productionEquation 6.33 (Ps/year)
TAXCM		total special taxes on modern cattle
		$(TAXCM(t) = \sum_{i=1}^{3} TAXCM_{i}(t)) - Equations 6.34$
		(Ps/year)
ALINT		interest payments on development credit

Equation 9.8b (Ps/year)

ALREP	×	repayment of development creditsEquation 9.5 (Ps/year)
CSTGH	=	cost of building forage storage (Ps/ha)
TCEC	=	total cash establishment costsEquation 6.37 (Ps/ha)
ALPH1	8	as defined in Equation 6.36 (proportion)
TRSL	=	total land in transition from traditional to modern practicesEquation 5.46 (hectares)
3*XDEL	=	time required to complete a land improvement programEquations 5.43 (years)
RPCAPT	=	the rate farmers' cost is increased by execution of additional storage capacity

Accounting revenues from cattle are computed from sales of milk and animals and increased by any direct subsidy paid to farmers. In the modern sector, revenues are also increased by the payment of development loans, but due to the difficulty in allocating commercial loans between the two sectors, these are computed in a more aggregated accounting in Chapter 9. Since the market model only generates the price of finished males, the pricing of other animals sold is computed as a proportion of the price of finished males. This computation is done with a set of coefficients estimated from time series recorded by the Central Bank and published by Garcia Samper [23]. Equations 6.39 show the computation of revenues from the modern sector.

(Ps/year).

SLSPM(t) = (SLSCCM(t) + SOLDFM(t))*C224 + SLSMPM(t)*C225

+ SLSMFM(t) + SLFERM(t)*C226 + SLSFGM(t)*C227

+ SLSMGM(t)#C228

(6.39a)

YAM(t) =	PAP(t)#SLSPM(t)	
----------	-----------------	--

where:

SLSPM = sales from the modern sector weighted by
price relationships (i.e., finished males
equivalents) (animals/year)

(6.39b)

- SLSCCM, SOLDFM, SLFERM = as defined in Equation 6.24 (animals/year)
- SLSMPM, SLSMFM = as defined in Equation 6.24 (animals/year)
- SLSFGM, SLSMGM = as defined in Equation 6.24 (animals/year)
- YAM = income from sales of modern animals (Ps/year)
- PAP = producer price of finished males--Equation 7.3 (Ps/animal)
- C224, ..., C228 = model parameters determining price relationships between finished males and other sale groups (proportion)

ARLAM(t) = YAM(t) + YMM(t) + ALOAN(t) + AGSUM(t) (6.39c)

where:

- ARLAM = accounting revenues from modern cattle (Ps/year)
- YMM = income from milk in the modern sector (Ps/year)
- ALOAN = credits paid for farm development--Equation 9.4a (Ps/year)

AGSUM = subsidies paid to the modern sector (Ps/year). Accounting costs and revenues provide estimates at a given point in time. Yet when farmers are considering the adoption of a new method they require an estimate of the future stream of revenues and costs throughout a relevant planning horizon. These projections into the future and within a planning horizon are simulated in the model by assigning a weight to each year and for each accounting variable involved based on farmers' past experience and on their judgment about changes brought about by the new methods. As can be expected, each alternative produces a set of different expectations and therefore the above-mentioned weights vary accordingly (see Table II.4). Expected revenues and costs are computed in Equations 6.40 below:

$$EOPCLM(t) = \frac{CSTANTL(t) * CAINCR_{1}}{GRE}$$
(6.40a)

$$EOCLNM_{i}(t) = CLNDTL(t) * CLINCR_{i}$$
 (6.40b)

$$EVLDTX_{1}(t) = VLTXTL(t) * VLTXTP_{1}$$
(6.40c)

$$ETXC_{1}(1) = TAXCTL(1) * TXCP_{1}$$
 (6.40d)

$$ECADEM_{1}(t) = EQLT * C268 * CAPDTP_{1}$$
 (6.40e)

where:

EOPCLM	<pre>= expected operating costs of modern cattle (Ps/ha-year)</pre>
EOCLNM	expected operating costs of modern grasslands (P3/ha-year)
EVLDTX	<pre>= expected taxes on land in modern cattle production (Ps/ha-year)</pre>
ETXC	<pre>= expected special taxes on modern cattle (Ps/ha-year)</pre>
ECADEM	<pre>= expected capital depreciation in the modern sector (Ps/ha-year)</pre>
CSTANTL	the exponential average of traditional animals cost (Ps/animal-year)

					P	lanni	ng Hoi	rizon	(yea:	rs)	·····		
Parameter Definition (Eqn. No	o.)	1	2	3	4	5	6	7	8	9	10	11	12
SINCR(5.35c) determ change in sales Alternative: TMINCR(5.35c) deter change in milk outp Alternative:	ines 1 2 3 4 mines ut 1 2 3	1 1 1 1.1 1.2 1.3	1 1 1 1.2 1.22 1.37	1.15 1.15 1.5 1.5 1.25 1.29 1.57	1.45 1.7 2.7 2.7 1.25 1.36 1.8	1.65 2 3.2 3.2 1.35 1.43 2	1.9 2.25 3.9 3.9 1.35 1.46 2.2	1.9 2.45 4 1.35 1.53 2.22	2 2.6 4 1.4 1.57 2.22	2 2.6 4 1.4 1.57 2.22	2 2.6 4 1.4 1.57 2.22	2 2.6 4 1.4 1.57 2.22	2 2.6 4 1.4 1.57 2.22
BINCR(5.35c) determ change in inventory Alternative:	4 ines 2 3 4	1.32 0 0 0 0	1.39 0.05 0.08 0.35 0.35	1.6 0.15 0.22 0.45 0.45	1.82 0 0.15 0.5 0.5	2.2 0.15 0.15 0.4 0.4	2.4 0 0.1 0.2 0.2	2.4 0 0.05 0.2 0.2	2.4 0.1 0.05 0 0	2.4 0 0 0 0	2.4 0 0 0 0	2.4 0 0 0 0	2.4
CAINCR(6.40a) detern change in animal cos Alternative:	mines sts 2 3 4	1.17 1.17 1.2 1.2	1.27 1.32 1.67 1.67	1.42 1.47 1.97 1.97	1.42 1.47 2.47 2.47	1.57 1.62 2.87 2.87	1.57 1.62 3.07 3.07	1.57 1.62 3.27 3.27	1.67 1.77 3.27 3.27	1.67 1.77 3.27 3.27	1.67 1.77 3.27 3.27	1.67 1.77 3.27 3.27 3.27	1.67 1.77 3.27 3.27

TABLE II.4. Perceived Changes in Cattle Output and Costs During the Planning Horizon.

TABLE II.4. (continued)

~

					Plar	nning	Hori	zon (;	years)			
Parameter Definition (Eqn. No.)			2.	3	4	5	6	7	8	9	10	11	12
CLINCR(6.40b) dete change in land cos Alternative:	ermines ts l 2 3	1	1	1.25 1.25 1.6	1.25 1.25 1.8	1.4 1.4 2	1.4	1.4 1.4 2.5	1.5	1.5	1.5	1.5	1.5
VLTXTP(6.40c) dete	4 rrmines	ī	1.2	1.4	1.6	1.8	2.2	2.2	2.2	2.2	2.2	2.2	2.2
Alternative:	1 2 3 4	1 1 1 1	1 1 1	1 1 1 1	1.2 1.25 1.3 1.32	1.2 1.25 1.3 1.32	1.2 1.25 1.3	1.3 1.35 1.4	1.3 1.35 1.4 1.42	1.3 1.35 1.4 1.42	1.3 1.35 1.4	1.3 1.35 1.4 1.4	1.3 1.35 1.4
TXCP(6.40d) determ in value of cattle	ines change taxes	_		_							±• 'C	4.72	1.72
Alternative:	1 2 3 4	1.1 1.1 1.1	1.16 1.25 1.35	1.19 1.3 1.6	1.21 1.33 2	1.26 1.35 2.4 2.4	1.27 1.36 2.55	1.28 1.37 2.6	1.3 1.4 2.6	1.3 1.4 2.6	1.3 1.4 2.6	1.3 1.4 2.6	1.3 1.4 2.6
CAPDTP(6.40e) dete change in capital tion	ermines deprecia-	1.1	1.35	1.0	2	2.4	2.55	2.0	2.0	2.0	2.0	2.0	2.0
Alternative:	1 2 3 4	1 1 1.1 1.1	1.5 1.6 2 1.9	2 2.5 2.8 2.7	2.5 2.8 3 2.9	2.5 3 3.4 3.3	2.5 3.4 3.3	2.5 3 3.4 3.3	2.5 3 3.4 3.3	2.5 3.4 3.3	2.5 3 3.4 3.3	2.5 3.4 3.3	2.5 3 3.4 3.3

Source: [42] and initial guesstimates and model tuning.

GRE	=	equilibrium grazing rate (has/animal)
CLNDŢL	=	the exponential average of traditional grasslands cost (Ps/ha-year)
VLTXTL	u	the exponential average of traditional land tax (Ps/ha-year)
TAXCTL	=	the exponential average of special taxes on traditional cattle (Ps/ha-year)
EQLT*C268	=	capital costs for traditional cattle productionEquation 6.32 (Ps/ha-year)
CAINCR	=	the expected increase in animal costs (dimensionless)
CLINCR	=	the expected increase in range management costs (dimensionless)
VLTXTP	=	the expected increase in land tax (dimensionless)
CAPDTP	8	the expected increase in capital costs (dimensionless)
i	=	indexes the planning horizoni=l,, n

Expected establishment costs are computed as equal allotments during the years required to complete a modern alternative. A model subprogram allocates these values among the relevant years in the planning horizon.

$$ETCEC(t) = \frac{TCEC(t)}{3*XDEL}$$
(6.40f)

where:

ETCEC	=	expected cash establishment costs of a modern alternative (Ps/ha-year)
TCEC	=	as defined in Equation 6.37 (Ps/ha)
3*XDEL	H	time required to complete a modern alter- nativeEquations 5.43 (years).

The expected debt service, EDBSER, is generated by a model subprogram that first computes interests paid on the entire credit received for development during the period of establishment and the grace period. Next the subprogram computes repayments in equal allotments during the repayment period, and charges interest on the unpaid balances. Finally, the expected credits paid are computed in a way similar to ETCEC above. The time during which credits are paid (LT1) may not necessarily be equal to the time required to establish the alternative (3*XDEL). A model subprogram allocates these values among the relevant ;ears in the planning horizon.

$$ELOAN(t) = \frac{TCEC(t)*(1 - RPTN)}{LT1}$$
(6.40g)

where:

- ELOAN = expected credits paid for farm development (Ps/ha-year)
- RPTN = farmers' participation of total establishment costs--a policy variable (0 < RPTN < 1) (proportion)
- LT1 = time development loans are paid--a policy variable (years).

Table II.5 at the end of this chapter shows the values of a selected number of variables used in component AGPRAC.

Definition (Equation N	Io.)	Value
DYLDCU (6.2a)	Determines target yield of cash crops in the uplands (tons/ha- year)	1.65
DYLDCL (6.2b)	Determines target yield of cash crops in the lowlands (tons/ ha-year)	2.9
DYLDFC (6.2c)	Determines target yield of food crops (tons/ha-year)	10
GRE (6.4b)	Determines the equilibrium graz- ing rate (has/animal)	.74
CPLPT (6.5)	Determines proportion of artificial grasses	.45
TDNSG (6.9c)	Determines the average TDN from sorghum silage (tons TDN/ton silage)	.2343
TDND (6.9f)	Determines target yield of TDN of grazing lands (tons/ha-year)	7.4
NREQT (6.22a)	Determines appropriate tradi- tional animal nutritional re- quirements (animal-year/ton TDN)	•55
ELASI (6.26b)	Determines income elasticity of demand for beef	.6
UNEXPL (6.26c)	Determines illegal cattle exports (animals/year)	300,000
COSTFT (6.28a)	Determines cost of treatment against foot-and-mouth disease (Ps/animal-year)	4.5
TAXLAND (6.33)	Determines the land tax rate (proportion/year)	.0042
SFTAX (6.34b)	Determines the female consumption tax rate (Ps/animal)	100
SMTAX (6.34b)	Determines the male consumption tax rate (Ps/animal)	50
CT202 (6.21b)	Determines number of traditional lactating cows milked (proportion)	•4

TABLE II.5. Selected Coefficients and Initial Values in the Agricultural Production Component (AGPRAC).

TABLE II.5. (continued)

Definition (Equation)	No.)	Value
C217 (6.25a)	Determines rate of growth of non- Costa cattle population (proportion/year)	.0285
C223 (6.34a)	Determines the liveweight tax rate (kilograms/animal-year)	4.0
C224 (6.39a)	Determines price relationship between finished males and cull cows (proportion)	.68
C225 (6.39a)	Determines price relationship between finished and producing males (proportion)	.92
C226 (6.39a)	Determines price relationship between finished males and breeding cows (proportion)	.88
C227 (6.39a)	Determines price relationship between finished males and growing females (proportion)	.42
C228 (6.39a)	Determines price relationship between finished and growing males (proportion)	.535
C237 (6.26b)	Determines the rate of increase in income (proportion/year)	.1
C238 (6.26b)	Determines the rate of increase in population (proportion/year)	.032
C242 (6.27a)	Determines proportion of heifers treated against brucellosis without campaign (proportion)	.05
С244 (б.28ъ)	Determines proportion of animals treated against foot-and-mouth without campaign (proportion)	•3
C248 (6.33)	Determines the assessed value of land (proportion)	•5
C250 (6.9c)	Determines the relationship between green forage and silage (tons forage/ton silage)	1.2
C253 (6.9c)	Determines yield of green forages (tons/ha-cutting)	40
		· · · · · · · · · · · · · · · · · · ·

TABLE II.5. (continued)

Definition (Equation No.))	Value
C254 (6.9c)	Determines the number of forage cuttings (cuttings/year)	3
C279 (6.34c)	Determines the cattle general inventory tax rate (Ps/animal- year)	3.1
CSTHCU(0) (6.29b)	Initial cost of cash crops in the uplands (Ps/ha-year)	560
CSTHCL(0) (6.29b)	Initial cost of cash crops in the lowlands (Ps/ha-year)	1,097
CSTHFC(0) (6.29b)	Initial cost of food crops (Ps/ha-year)	723
CLNDT(0) (6.30a)	Initial operating cost of traditional grazing lands (Ps/ha-year)	35.8
CLNDM(0) (6.30a)	Initial operating cost of modern grazing lands (Ps/ha-year)	100
CSRFGH(0) (6.30b)	Initial cost of replanting and growing forages (Ps/ha-year)	570
CSHARV(0) (6.30b)	Initial cost of harvesting and storing forages (Ps/ton)	6.25
CSTANT(0) (6.31)	Initial operating cost of traditional animals (Ps/animal- year)	24.14
CSTANM(0) (6.31)	Initial operating cost of modern animals (Ps/animal-year)	56
CSINPH(0) (6.35)	Initial cost of improving native grasses (Ps/ha)	576
CSIAPH(0) (6.35)	Initial cost of improving artificial grasses (Ps/ha)	576
CSPAPH(0) (6.35)	Initial cost of planting artificial grasses (Ps/ha)	660
CSPFGH(0) (6.35)	Initial cost of establishing forage crops (Ps/ha)	590
CBSTG(0) (6.35)	Initial cost of building forage storage (Ps/m ³)	8.5

Source: [19, 20, 23, 31, 32, 33, 42, 57, 58, 61, 66, 69, 71] and initial guesstimates and model tuning.

CHAPTER 7

PRICE GENERATION (PG)

Component PG generates world prices for beef and market and producer prices of cattle, cash crops and food crops. In addition, five-year exponential averages of the producer prices are computed for use by component LAMDAC in the profitability calculations for the land allocation decisions.

Export and Market Price of Cattle

Colombian beef exports have been mainly live cattle, but the government has announced plans to export only dressed animals after 1974. The model assumes this change in export policy will be effectively implemented, and that the relevant world price is for frozen carcass beef. For the period 1964 to 1974, prices are for a composite of live animals and frozen carcass beef. Although it has been assumed that exports will continue in the form of frozen carcass beef, it is clear that the export price will be modified if chilled, refrigerated or processed beef is exported. All export prices are given as live animal prices; carcass beef is convorted to live animals by use of the factor 4.3 which is the number of animals to produce a metric ton of carcass. World prices for cattle (U. S. \$/animal) are exogenously generated by Equation 7.1 below. Since cattle exports were negligible until 1964, no world prices are generated for the years prior to 1964. For the period 1964 to 1970, prices are a composite of live animals and carcass beef as reported by Sarmiento [63]. From 1971 to 1974 these composite prices are projections based on Instituto de Comercio Exterior (INCOMEX) estimates [35]. Finally, after 1974 world prices are projected assuming various trends that will be discussed in Chapter 12.

	ſo	0 < t < 4
	117, 115.5, 127, 147, 164, 163.5, 174	$4 \leq t \leq 10$
WPB(t) =	WPB1970*(1+WPBR*(t-10)	10 < t <u><</u> 14
	$\left(\frac{WPBC(t)}{4.3}\right)$	t > 14
		(7.la)

$$WPBC(t) = WPBC1970*(1+WPBCR*(t-10))$$
 (7.1b)

where:

WPB = world (FOB) price of beef (US\$/animal)
WPBC = world (FOB) price of frozen carcass beef (US\$/metric ton)
WPB1970, WPBC1970

recorded world price in 1970

WPBR = rate of change of world price after 1970

as a proportion of 1970 price. This is a composite of live animals and frozen carcass beef (proportion/year)

WPBCR = rate of change of world price of frozen carcass beef after 1970 as a proportion of 1970 price.
t = simulated time (t=0 is 1960)--years.

The market price of cattle is computed in Equation 7.2. This equation, which generates the market price of finished males as a function of excess demand, is derived directly from the definition of demand price elasticity ε :

$$\varepsilon = \frac{\Delta q/q}{\Delta p/p}$$

where $\Delta q = q_t - q_{t-DT}$ and $\Delta p = p_t - p_{t-DT}$ and the ratios are taken relative to the initial price and quantity demanded, p_{t-DT} and q_{t-DT} .^{1/}

PA(t) = PA(t-DT) + DT*C219*PA(t-DT)*

$$\frac{(\text{TDEM}(t-DT) - \text{SUPB}(t-DT))}{\text{ELASD*TDEM}(t-DT)}$$
(7.2)

where:

PA	a	market price of finished males (Ps/animal)
SUPB	Ħ	total Colombian supply of beefEquation 6.25b (animals/year)
TDEM	8	total demand for Colombian beefEquation 6.26c (animals/year)
ELASD	=	price elasticity of demand for beef
C219	=	a model parameter regulating the beef price response to excess demand (proportion/year).
Equat	Lor	7.2 assumes that the target change in

quantity, Δq , will be the excess demand in the previous period and that the equilibrium price will not necessarily be reached in one period, i.e., if DT#C219 < 1. The

 $\frac{1}{For}$ a detailed derivation see Abkin [1, Chapter 5].

domestic price of other types of cattle are computed as a constant proportion of the price of finished males. This computation will be more fully discussed in Chapter 9.

Producer Prices and Price Averages

Next, producer prices are computed for cattle and crops. Producer prices of crops are exogenously determined and are a composite of the major crops grown in each of the agricultural regions. The model assumes a constant profit margin in the production of crops and projects increases in their prices over time at the same rate that costs of production are increased. Producer prices in the base year (1960) are as reported by the World Bank report on Colombia [41] and weighted by total yields in each agricultural region. Weights are derived from hectares in production reported by DANE [15] and average yields reported by the Instituto de Mercadeo Agropecuario (IDEMA)^{1/2} as shown in Table II.6. Further, it is assumed that 50 percent of the harvested yield of sesame and sorghum, 35 percent of that of cotton, and 70 percent of that of corn come from the uplands.^{2/}

<u>l</u>/IDEMA. <u>Informe Sobre la Superficie Sembrada</u>, <u>Produccion Total y Rendimientos de los Productos Basicos</u>. Bogota. Mayo 1971. Unpublished Report.

^{2/}Weights used for 1960 are, for cash crops in the lowlands: sesame--.012; cotton--.13; corn--.24; sorghum--.015; and rice--.6. For cash crops in the uplands: sesame--.018; cotton--.11; corn--.85; and sorghum--.023. For food crop: plantain--.48; and cassava--.52.

Crop	Area Planted (has)	Average Yield (tons/ha_yr)	Price (Ps/ton)
Sesame	9,787	• 59	1,519
Cotton	38,815	1.34	1,726
Corn	162,806	1.19	474
Sorghum	2,952	2.46	369
Rice	74,588	1.98	883
Planțain	48,578	8.38	224
Cassava	53,243	8,28	303

TABLE II.6. Land in Crops, Average Yields and Producer Prices in the Costa, 1960.

Source: As indicated above. Price of sorghum was obtained by personal information.

The producer price of cattle is computed in Equation 7.3 below:

$$PAP(t) = PA(t)*(1-MKM)$$
(7.3)

where:

PAP = producer price of finished males (Ps/animal) MKM = the marketing margin for cattle--proportion $(0 \le MKM \le 1)$ (see Table III.2).

Because of a lack of information on producer price changes, crop prices are increased over time at the same rate of increase in farm costs (RCST). The assumption of maintaining a constant profit margin throughout the simulation is an accounting simplification that finds justification in the secondary concern of the model on the crop subsector. Equations 7.4 compute these prices as:

$$PCROPU(t+DT) = PCROPU(t)*(1+DT*RCST)$$
(7.4a)

$$PCROPL(t+DT) = PCROPL(t)*(1+DT*RCST)$$
 (7.4b)

$$PFCROP(t+DT) = PFCROP(t)*(1+DT*RCST)$$
(7.4c)

where:

, ,,

	PCROPIJ	Ξ	the projected producer price of cash crops from uplands (Ps/ton)
	PCROPL	8	the projected producer price of cash crops from lowlands (Ps/ton)
, ,	PFCROP	H	the projected producer price of food crops (Ps/ton)
	RCST	=	the rate of increase in farm costs (proportion/year).

Exponential price averages are computed in Equations 7.5 for use in determining land allocation decisions (Equations 5.35a and 5.35e).

$$EPAP(t) = EPAP(t-DT) + \frac{DT}{DEL5}*(PAP(t-DT) - EPAP(t-DT)) \quad (7.5a)$$
$$EPCRPU(t) = EPCRPU(t-DT) + \frac{DT}{DEL7}*(PCROPU(t-DT))$$

$$-$$
 EPCRPU(t-DT)) (7.5b)

where:

- 1	EPAP =	exponential average of finished males producer price (Ps/animal)
	EPCRPU =	exponential average of upland cash crops producer price (Ps/ton)

190

DEL5, DEL7

= averaging lags (years).

The values of a selected number of variables used in generating prices are shown in Table II.7.

Definition (Equation No	Value	
WPBR (7.la)	Determines change in live-carcass world price of beef (proportion of 1970 price/year)	.11ª/
WPBCR (7.lb)	Determines change in frozen carcass world price of beef (proportion of 1970 price/year)	.384 <u>b</u> /
ELASD (7.2)	Price elasticity of demand for beef (dimensionless)	.7
RCST (7.4)	Determines change in cost of farm inputs (proportion/year)	.12
SUPB(0) (7.2)	Initial total Colombian beef supply (animals/year)	1,889,100
TDEM(0) (7.2)	Initial total demand for Colombian beef (animals/year)	1,889,100
PCROPU(0) (7.4a)	Initial composite price of cash crops in uplands (Ps/m.ton)	630
PCROPL(0) (7.4b)	Initial composite price of cash crops in lowlands (Ps/m.ton)	892
PFCROP(0) (7.4c)	Initial composite price of food crops (Ps/m.ton)	265
PA(0) (7.2)	Initial price of finished males (Ps/animal)	1,067
WPBC1970 (7.lb)	World price for Colombian carcass beef in 1970 (dollars/ m.ton)	588

TABLE II.7. Selected Coeffic. nts and Initial Values in the Price Generation Component (PG).

- <u>a</u>/ Based on price increase between 1964 and 1968 and assuming a gradual increasing weight of carcass beef on exports from 1970 to 1974.
- b/ Based on 1970 price reported by Sarmiento [63], and on 1972 price (US\$1040/m.ton) reported by <u>El Espectador</u>, Bogota, April 8, 1973.

Sources: [15, 16, 23, 41, 63, 67]

CHAPTER 8

POLICIES FOR THE CATTLE INDUSTRY

In a policy-oriented model, there are a number of places in which the policy maker can interact with researchers to perform experiments in a simulated system. These experiments may involve changing system parameters and technological coefficients to see the effect on the model's performance or direct policy experimentation. In the latter course, policies and programs are specified explicitly and the consequences are simulated as a result of the system structure of the model. When experimenting with different values of system parameters and/or technological coefficients, the policy maker acquires a better judgment about those parameters to which the model is insensitive and about those which significantly affect the system performance and therefore would play a role in future policy and planning decisions. In addition, technological research may be suggested by policy runs speculating on the likely consequences of the introduction of an innovation which may not actually be developed at the moment.

Simulation runs testing parameter sensitivity and conducting direct policy experiments are discussed in Chapters 11 and 12.

Policies

Three basic policy strategy alternatives are structured in the simulation model. Others could be added, but the three included seemed to be quite relevant for Colombian policy makers at the time the model was defined. Folicies may be set and experimented with in any one or a combination of the following: production campaigns, tax policies and export policies.

Production Campaigns

Production campaigns make up the first class of policies which may be investigated. Promotion efforts aimed at modernizing cattle production can generate substantial returns to both the public and private sectors. Such modernization may entail the introduction of better animal husbandry and/or the encouragement of improved pasture management practices and increased fodder production. The increase in output can then result in higher incomes for the farmers, increased availability of basic food for the mass of the population, and increased tax revenues and foreign exchange earnings for the public sector. The nonagricultural sector, though not modeled, could be expected to grow also as a result of increased demands from the agricultural sector.

Associated with this modernization effort is a policy of providing credit for farm improvements at special lending terms. These terms include the interest rate, the grace period, the repayment schedule, the farmers' participation

on the total cost of the project and the technical assistance. Both the promotional information units discussed in component LAMDAC, Equation 5.37, and the funds for development credit discussed in component CRTACC, Equation 9.4, are generated exogenously using a promotion and credit allocation routine. The maximum yearly size (VMAX) and time spans (TO, T1, T2, TF) of these services may be specified by the experimenter, and the model generates the time profile shown in Figure II.8. 1

Other policies related to production campaigns are the control of foot-and-mouth disease and brucellosis, and the improvement of crop production. Currently, the model takes the disease control campaign as aimed toward the traditional sector since it is assumed that animals in the modern sector are appropriately treated as part of the improved hustandry adopted. As indicated in Chapter 6 (Equations 6.27 and 6.28), animals are treated regardless of their profitability. All that is required is an exogenous rate of vaccination (C199 and C200) which can be set as a policy. Government expenditures in the Costa on control of foot-andmouth disease, EXPAFT in Equation 6.28a, are generated using a TAPLIE function which steps up these expenditures from the years preceding the campaign until they reach a maximum, and

 $[\]pm$ This routine has been adapted from the modernization budget executive routine used in the simulation of the Nigerian Agricultural Economy [53].



FIGURE II.8. Promotion and development credit profile.

then the function projects them at a constant annual rate. The values used in this function have been approximated from estimates supplied by ICA [31] and show the more limited efforts that have been achieved in this program prior to 1971. The profile generated by the table function could be changed to one similar to Figure II.8 as part of the policy experiments. Finally, the improvement of crops is determined exogenously using the simple model described by Equations 6.2, and its effect is measured in the land allocation decisions and in farm income.

Taxes

The second major policy which can be investigated with the model is the area of taxing policies. Taxes are levied on net income, net worth, cattle, and land, but we are concerned here only with the last two categories. Taxes on cattle as described in Chapter 2 affect the cost of production and therefore cause adjustments in the use of factors of production that affect farm income and output. Taxes on land are amortized to its value and therefore decrease the asset position of farmers. Yet both cattle and land taxes are a main source of government revenue. With simulation runs incorporating different levels of tax rates for both cattle and land, questions can be answered regarding the likely consequences these policies will have on production levels, agricultural income, and other relevant economic performance criteria.

Export Policies

Finally, the model allows experimentation with several kinds of export policies aimed at generating foreign earnings and regulating domestic supply. Specifically, targets can be set on cattle exports, exchange rates can be varied and subsidies can be paid to exporters. In addition to these, the effects of different levels of exports on domestic consumption and price can be investigated and their consequences projected. Further, the value of transfers from public revenues to the private sector in the form of an export subsidy can also be examined.
CHAPTER 9

ACCOUNTING AND PERFORMANCE CRITERIA (CRTACC)

Component CRTACC completes the farm accounting and computes the performance indices used to evaluate the outcome of the cattle and crops policy experiments.

Budget Accounting

Given incomes, costs of crops and cattle and the rate of land improvment, it is possible to determine credit requirements, debt service, investment constraints and farm income on a regional basis. The farm development budget is modeled dynamically as cash flows distributed over time. The development budget model is divided into three stages of varying lengths. The respective stage lengths reflect the three investment periods which the model identifies: a period of expenditures on farm improvements and credit payments, a grace period, and a period of credit repayments. These lengths (LT) are policy variables which allow testing the effect of different credit schemes.

The preceding budget flow is simulated dynamically by three calls to BOXC, a "boxcar train" subroutine, one for each stage. This subroutine is used to delay a flow for a considerable period of time, with no outflow until the delay time is over [21, 52]. The credit-investments period

uses as an input the outlays required for farm improvement, net of farmers participation and use of private funds. These requirements are determined by the rate at which land enters modernization (Equation 5.41) and the establishment costs (Equation 6.37). The remaining stages use the output of the preceding stage as an input. Equations 9.1 describe this process.

CALL BOXC(AUX5(t-DT), BOUT1(t), TRAIN1, NCOUN1, NOCY1,

```
(9.1a)
LT1, SUMIN)
```

CALL BOXC (BOUT1 (t-DT), BOUT2(t), TRAIN2, NCOUN2, NOCY2,

(9.1b)LT2, SUMIN)

CALL BOXC(BOUT2(t-DT), BOUT3(t), TRAIN3, NCOUN3, NOCY3,

```
(9.1c)
LT3, SUMIN)
```

where:

AUX5 = credit needs for land entering modernization $(Ps/year) \frac{1}{2}$ BOUT1 = rate credit investments leave the first stage (this is the output variable of the first call to BOXC) (Ps/year) BOUT2 = rate credit investments leave the second stage (this is the output variable of the second call to BOXC) (Ps/year) BOUT3 = rate credit investments leave the third stage (Ps/year) LT1 = time development loans are paid (years)

 $\frac{1}{For}$ detailed computation of this variable see subroutine MODCRD in the Appendix.

LT2 = time after completing farm development adding up to the grace period (LT1 + LT2) (years) LT3 m time development loans are repaid (years) TRAIN, NCOUN, NOCY, SUMIN other variables associated with the use of the BOXC subroutine. The purpose of these calls to subroutines BOXC is to compute TRNSL(t), the sum of credit invested in each stage. These levels are computed in Equations 9.2 as time integrals of credit flow rates. TRNSL1(t+DT) = TRNSL1(t) + DT*(AUX5(t) - BOUT1(t)) $- \text{TRAIN1}(3) \neq (1 - A5(t)))$ (9.2a) $TRNSL2(t+DT) = TRNSL2(t) + DT_{\#}(BOUT1(t) -$ BOUT2(t)) (9.2b)TRNSL3(t+DT) = TRNSL3(t) + DT*(BOUT2(t) -BOUT3(t)) : (9.2c)

where:

TRNSL1 = total credits paid during the first stage-the period when improvements are implemented
(Ps)
TRNSL2 = total credits completing the grace period (Ps)
TRNSL3 = total credits that have to be repaid (Ps)
A5 = the proportion of land remaining in the
program after dropouts--Equation 5.44.

The term involving TRAIN1(3) in Equation 9.2a needs further explanation. The credit rates that flow through the first BOXC delay are adjusted to allow for the possibility of "dropouts" from the program. This dropout effect is introduced by the variable A5 discussed in Chapter 5. Equation 9.2a also implies that dropouts, if any, occur during the first year after entering the modernization program.

Development Credit

Once the total outlays of investment credit are generated, it is easy to determine the credit constraints to development and the debt service. First the model computes demand for credit, DCRDT, and availability of credit, CREDT, which are used to determine the credit-based rate of modernization in component LAMDAC (ARM1 in Equation 5.41).

$$DCRDT(t) = \frac{TRNSL1(t)}{LT1}$$
(9.3)

$$CREDT(t) = max(ACRDT(t) - DCRDT(t), 0) \qquad (9.4)$$

where:

- DCRDT = demand for development credit from ranchers already in the program (Ps/year) CREDT = credit available for additional modernization
- ACRDT = total credit allocated for modernization--a policy variable (Ps/year).

(Ps/year)

Next, the model computes the debt service on development loans. As a simplification, the model assumes that all farmers entering a modernization program receive credit if it is available. This is to say that:

$$ALOAN(t) = min(CREDT(t), DCRDT(t))$$
 (9.4a)

$$ALREP(t) = \frac{TRNSL3(t)}{LT3}$$
(9.5)

$$ALOANA(t+DT) = ALOANA(t) + DT = ALOAN(t)$$
 (9.6)

$$ALREPA(t+DT) = ALREPA(t) + DT = ALREP(t)$$
(9.7)

ALREP	= development loans repaid (Ps/year)
ALOAN	<pre>= development loans paid (Ps/year)</pre>
ALREP	A = accumulated development loans repaid (Ps)
ALOAN	A = accumulated development credits paid (Ps).
Inter	ests on development loans are charged on the
outstanding d	ebt balance. These are computed in Equations
9.8 below:	

$$DBTOUS(t) = ALOANA(t) - ALREPA(t)$$
 (9.8a)

$$ALINT(t) = DBTOUS(t) * RINTL$$
 (9.8b)

where:

DBTOUS	=	the outstanding development debt (Ps)
ALINT	-	interest payments on development loans (Ps/year)
RINTL		interest rate on development loansa policy variable (proportion of debt/year).

Commercial Credit

Commercial credit as used in the model is shortterm credit, usually for one year provided through the development and private banks under a variety of governmentregulated schemes (see Chapter 1). Since interest charges on these loans vary widely, the model uses an estimated average rate (RINTC) when computing the short-term debt service. As a general policy, short-term credit is supplied to cover operating costs and buying feeder cattle, although the latter use has been ruled out in the model (see Chapter 5, p. 109). Begional allocation of commercial credit by the banking system determines its availability to farmers and this could be an important constraint in the model. Yet lack of information on this matter led us to the simplifying assumption that the only constraint to the use of commercial credit was the farmers' capacity to provide an acceptable security. It is clear that with more information the allocation of commercial credit by the banking system, ACRDTC, could be a policy variable. Equations 9.9 below compute the availability of commercial credit and its debt service.

CRDAV(t) = min(PEQCR = EQPOS(t), ACRDTC(t)) (9.9a)

EQPOS(t) = VLAND(t) + VACAPL(t) - CDEB(t) - DBTOUS(t) (9.9b)

where:

CRDAV	<pre>= commercial credit available to the cattle sector (Ps/year)</pre>
EQPOS	<pre>= equity position of cattle producers (Ps)</pre>
ACRDTC	<pre>= commercial credit allocated (Ps/year)</pre>
PEQCR	<pre>proportion of equity which can be used as a credit base (proportion/year)</pre>
VLAND	<pre>= capitalized asset value of land in cattle productionEquation 9.15c (Ps)</pre>
VACAPL	<pre>value of cattle inventoriesEquation 9.14 (Ps)</pre>

CDEB = commercial debt of cattle sector--Equation 9.9f (Ps)

DBTOUS = as defined in Equation 9.8a (Ps).

Total demand for commercial credit, TDCRDC, is determined by the difference between net income from cattle, increased by internal transfers of capital from the crops sector, and the expenditures for consumption.

TDCRDC(t) = max(-FARIL(t) - C239*FARMIC(t))

where:

.

TDCRDC	=	total	demand	for	commercial	credit	by	the
		cattle	e sector	r (Ps	s/year)		•	

- FARIL = aggregated net farm income from cattle--Equation 9.10c (Ps/year)
- FARMIC = aggregated net farm income from crops--Equation 9.11 (Ps/year)
- EXPLIV = aggregated consumption expenditures of cattle producers--Equation 9.12a (Ps/year)
- C239 = a model parameter determining the proportion
 of income from crops internally transferred
 to cattle production.

Commercial loans paid to cattle producers, CLOAN, are computed as:

CLOAN(t) = min(CRDAV(t), TDCRDC(t)) (9.9d)

The repayment of the commercial debt is computed as:

CREP(t) = max(CREPR*CDEB(t-DT), 0)(9.9e)

CREP = commercial loans repaid (Ps/year)

4

CREPR = repayment rate (proportion of debt/year).

Given credit payments and repayments, it is possible to compute the outstanding commercial debt and interest payments by the cattle sector.

$$CDEB(t+DT) = CDEB(t) + DT*(CLOAN(t) - CREP(t))$$
(9.9f)

$$CINT(t) = RINTC*CDEB(t)$$
 (9.9g)

where:

CDEB	=	commercial debt of cattle sector (Ps)
CINT	=	interest payments on commercial debt (Ps/year)
RINTC	=	interest rate on commercial loans (proportion of debt/year)

Aggregated Income and Consumption

Component AGPRAC generates revenues and costs of traditional and modern cattle (Equations 6.38 and 6.39) disregarding the accounting effect of commercial credit. Now it is possible to incorporate this effect into the general accounting and determine farm income on a regional basis. First, Equations 9.10 compute aggregated income from cattle.

$$ARLSK(t) = ARLAT(t) + ARLAM(t) + CLOAN(t)$$
(9.10a)

where:

ARLSK = aggregated revenues from cattle (Ps/year)

ARLAT = accounting revenues from traditional cattle Equation 6.39c (Ps/year)
ARLAM = accounting revenues from modern cattle Equation 6.39c (Ps/year)
CLOAN = as defined in Equation 9.9d (Ps/year).
ACLSK(t) = ACLAT(t) + ACLAM(t) + CREP(t) + CINT(t) (9.10b)
where:
ACLSK = aggregated costs of producing cattle (Ps/year)
ACLAT = accounting costs of traditional cattle Equation 6.38 (Ps/year)
ACLAM = accounting costs of modern cattleEquation 6.38 (Ps/year)
CREP = as defined in Equation 9.9e (Ps/year)
CINT = as defined in Equation 9.9g (Ps/year).
Finally, aggregated net farm income from cattle,
FARIL, is simply computed as:

```
FARIL(t) = ARLSK(t) - ACLSK(t) (9.10c)
```

Next, aggregated farm income from crops is computed from net income of cash and food crops (as discussed in Chapter 6). The value of property tax used in this computation is based on the assessed capitalized value of all land in crops as will be discussed later in this section.

$$FARMIC(t) = FARICC(t) + FARIFC(t) - VLDTXC(t)$$
 (9.11)

FARMIC = aggregated net farm income from crops
 (Ps/year)1/
FARICC = net farm income from cash crops (Ps/year)1/
FARIFC = net farm income from food crops (Ps/year)1/
VLDTXC = value of taxes on land in crop production
 (Ps/year)1/

Due to a lack of data on family expenditure patterns and statistics on the number of family heads operating cattle farms, the computation of living expenditures posed a difficult problem. This was circumvented in the model by developing a simply income-consumption submodel which uses estimates based on experienced judgments. The number of cattle farm operators was roughly estimated from the number of farms supplied by DANE [14] and an annual minimum consumption expenditure was set for the region based on reasonable living requirements per family.2/ The incomeconsumption equations attempt to incorporate into the model consumers' behavior related to income elasticity of demand, wealth effects and Engel's law. Savings in the cattle economy are implied when the combined income from cattle and crops exceed consumption expenditures. Equations 9.12 determine the income-consumption relationships discussed above.

1/For detailed computation of these variables, see subroutine AGACC in the Appendix.

^{2/}The assumed number of farm families is 40,000 and the minimum expenditures for consumption Ps 5,000/family-year.

EXPLIV(t) = max(ALPH2(t)*GINC(t), EXLMIN)

where:

EXPLIV	 living or consumption expenditures of cattle farmers (Ps/year)
GINC	aggregated gross income from sales of animals and milk (Ps/year)
ALPH2	= as defined in Equation 9.12b (proportion)
max(a,b)	= the minimum value between a and b.

ALPH2(t) = C261 + (C262 - C261) * EXP(-C263*C264*max)

(UINC(E) = EADPIN, U)) (J.LE	(GINC(t)	- EXLMIN,	. 0))		(9.12)
------------------------------	----------	-----------	-------	--	--------

where:

ALPH2	 a varia	ible	which	intro	luces	the	effect of
	income	upon	const	umers'	behav	1or	(proportion)

- C261 = a model parameter determining the minimum proportion of income which is consumed
- C262 = a model parameter determining the maximum proportion of income which is consumed
- C263 = the rate of consumption expenditures response with respect to changes in income (dimensionless)

C264 = a scale factor

max(a,b)
 = the maximum value between a and b

EXP = the exponential function.

Equations 9.12 imply that a minimum level of consumption always takes place despite low incomes. Further, when income is low a higher proportion of it is consumed,

1/Por detailed computation see subroutine AGACC in the Appendix.

but as this increases the proportion consumed decreases until it is stabilized when high levels are attained. This expenditure pattern reflects consumers' behavior with respect to changes in income. It is clear that by assigning different values to the model parameters in the function ALPH2, a wide range of consumers' behavior can be simulated. The preceding computation can be improved and probably reformulated as more information on this subject is available.

Once aggregated income and consumption are determined, the model generates the farmers' investment capital that is used in component LAMDAC as a constraint to the rate of land modernization (Equations 5.41 and 5.42c). This is computed in Equation 9.13 below, which assumes an internal transfer of capital from the crops sector to the cattle sector.

NCFR(t) = max(FARIL(t) + C239*FARMIC(t) - EXPLIV(t), 0) (9.13)

where:

NCFR = net investment capital of farmers (Ps/year)
FARIL = as defined in Equation 9.10c (Ps/year)
FARMIC = as defined in Equation 9.11 (Ps/year)
EXPLIV = as defined in Equation 9.12a (Ps/year)
C239 = a model parameter determining the proportion
of income from crops internally transferred
to cattle production.

Capital Formation and Export Incentives

Two measurements of internally generated capital by the cattle subsector are considered in the model: the value of cattle inventories and the value of land. Changes in asset value of cattle over time reflect changes in both prices and cattle population. The asset value of land considered here is based on its capability to generate an income stream independent of location, population pressure and other external factors. Any increase in the asset value of cattle and land not only increases the "wealth level" of farmers but also the collateral value of their assets, enabling them to borrow more capital for further agricultural expansion.

The value of cattle in the model is related to the price of finished males since this is the only price generated by the market model (see Chapter 6, Equation 6.39a for a detailed discussion). It is clear that an expanded market model pricing of each animal category will provide a better estimate of the value of cattle inventories. The model accounts for a likely higher value of animals in the modern sector, and assumes that finished males are not kept in the herd but are marketed as soon as they complete the fattening period.

VACAPL(t) = PAP(t)*[C227*(PFGT(t) + C212*PFGM(t)) + C226*

(PFPT(t) + C212*PFPM(t)) + C224*(OLDFT(t) +

C212*OLDFM(t)) + C228*(PMGT(t) + C212*PMGM(t))

+ C225*(PMPT(t) + C212*PMPM(t))] (9.14)

```
where:
```

ł	VACAPL	Ľ	value of cattle inventories (Ps)
	PAP	щ.	producer price of finished males Equation 7.3 (Ps/animal)
	PFGT, PFGM	=	population of growing females, tradi- tional and modern, respectively Equation 6.11 (animals)
	PFPT, PFPM	=	population of producing females, traditional and modern, respectively Equation 6.11 (animals)
	PMGT, PMGM	=	population of growing males, tradi- tional and modern, respectively Equation 6.11 (animals)
	РМРТ, РМРМ	н	population of producing males, tradi- tional and modern, respectively Equation 6.11 (animals)
	OLDFT, OLDFM	U	population of old females, traditional and modern, respectivelyEquation 6.12 (animals)
	C212	H	a parameter accounting for heavier animals from the modern sector
	C224,, C	22	B model parameters determining price relationships between finished males and other sex and age groups (pro- portion).

The asset value of pasture land used in the model is its capitalized value which is obtained by dividing the annual average returns in a hectare of land by the prevailing interest rate. The total capitalized value in the Costa is the sum of the values of the total land in the traditional and modern sectors. It should be mentioned that the capitalized value of a hectare of agricultural land can be increased by the increase in output, output price and decrease in the cost of production. Furthermore, the change in the interest rate in the economy affects the capitalized land value. In the model, when the average returns from traditional cattle become negative, the land is valued at a salvage price assumed to be one peso per hectare. But when average returns from modern cattle or crops are negative, the assumed salvage value of land is that of land in traditional cattle production. The capitalized value of land in cattle production is computed in Equations 9.15 below, where RINT attempts to represent the opportunity cost of capital rather than the interest rate of bank loans.

$$VLANDT(t) = max(SVALT*TTGLR(t), \frac{FARILT(t)}{RINT})$$
 (9.15a)

where:

- VLANDT = capitalized asset value of land in traditional cattle (Ps)
- TTGLR = total Costa land in traditional cattle--Equation 5.51 (hectares)
- FARILT = net farm income from traditional cattle (Ps/ year)

RINT = the current rate of interest (proportion/year)

SVALT = the salvage value of traditional land (Ps/ha.). The computation of FARILT does not account for income and/or liabilities arising from credit. This accounting procedure implies that borrowing does not affect land values and the procedure is also applied in the modern sector. The capitalized value of crop land, VLANDC, is computed in a fashion similar to traditional pasture and is not shown here. Yet there are two exceptions: the salvage value of land discussed earlier, and the farm income that is an aggregate of all cash and food crops. Capitalized values per unit of land are simply computed dividing total value of land by total land in each use.

VLANDM(t) = max(VLNDHT(t)*(TRSL(t) + TLMOD(t)),

$$\frac{FARILMI(t)}{RINT}$$
 (9.15b)

VLAND(t) = VLANDT(t) + VLANDM(t) (9.15c)

where:

VLANDM	= capitalized asset value of land in modern cattle (Ps)
VLAND	<pre>= capitalized asset value of land in cattle in the Costa (Ps)</pre>
VLNDHT	the per hectare capitalized value of the traditional land (Ps/ha.)
FARILMI	<pre>= farm income from modern cattle net of credit accounts (Ps/year)</pre>
TLMOD	<pre>= total grazing land in modern production Equation 5.48 (hectares)</pre>
TRSL	= land in transition from traditional to modern practicesEquation 5.46 (hectares)

Finally, the model generates the variables associated with the export sector which are needed to evaluate policy alternatives toward cattle exports in Chapter 12. Since the instruments of export promotion have been mainly the payment of subsidies and adjustments in the exchange rate, we will be concerned with these two policy elements here. Export subsidies are paid as a proportion of the peso value of exports and are computed as follows:

SUBSE(t) = FOREX(t) * EXCHR(t) * EXSUB (9.16)

- SUBSE = export subsidies paid to cattle sector (Ps/year)
- FOREX = foreign exchange earnings from cattle exports--Equation 9.17 (US \$/year)
- EXCHR = the official exchange rate--a policy variable--Equation 9.18a (Ps/dollar)
- EXSUB = the export subsidy--a policy variable (proportion of value of exports).

Foreign exchange earnings are simply computed as:

```
FOREX(t) = WPB(t) * EXPL(t) (9.17)
```

where:

- WPB = world (FOB) price of beef--Equation 7.1a
 (US \$/animal)

Exchange rates are computed for the relevant period of cattle exports starting in 1964. Between 1964 and 1966, exchange rate values are those recorded by the International Monetary Fund [44], and average values for 1967 and at the beginning of 1968 are those recorded by the Central Bank [3]. From 1968 forward the exchange rate is projected at the rate of increase observed during the period 1967 to 1972 [45]. These computations performed in Equations 9.18 below incorporate into the model the effect of a fluctuating exchange rate introduced by the Colombian government in March 1967.

EXCHR(t) = $\begin{cases} 0 \text{ (i.e., undefined)} & 0 < t < 4 \\ 12.77, 13.5, 13.5, 14.5, 15.76 & 4 \leq t \leq 8 \text{ (9.18a)} \\ \text{EXCHR(t-DT)} + \text{DT*RCHEX(t-DT)} & t > 8 \end{cases}$

E	EXCHR = aa	defined in Equation 9.16 (Ps/dollar)
· · · · · · · · · · · · · · · · · · ·	RCHEX = th ra	e rate of change in the official exchange teEquation 9.18b (Ps/year).
RCHEX(t)	= AL1#EXC	HR1968#EXP(AL1#(t-8)) (9.18b)
where:		
J	AL1	<pre>= the annual exchange rate growth rate (proportion/year)</pre>
· I	EXCHR1968	the official exchange rate at the begin- ning of 1968 (Ps/dollar)
I	EXP	= the exponential function
•	t	= time (years).

Given exchange rates, export subsidies, and domestic and world price of beef from component PG, the model computes the export margin, EXMAR, as a proxy for the competitive position of Colombian beef in international markets. Another way of looking at EXMAR is as the profit for beef exporters. When exports are made in the form of carcass beef it is assumed that revenues from viscera, hides and other byproducts cover slaughtering and handling costs. In Equation 9.19 below, the export subsidy, EXSUB, is reduced by a factor of .8 to account for the discounted price at which the tax certificates used to pay the subsidy are sold in the market. A negative export margin indicates that Colombian beef is priced out of international markets at the going effective rate of exchange.

EXMAR(t) = WPB(t) = EXCHR(t) = (1 + 0.8 = EXSUB) - PA(t)(9.19)

EXMAR	=	the profit margin of beef exports (Ps/ animal)
EXCHR	=	as defined in Equation 9.16 (Ps/dollar)
WPB	=	world price of beefEquation 7.1a (US \$/ animal)
EXSUB	8	the export subsidya policy variable (proportion of value of exports)
PA	=	market price of finished malesEquation 7.2 (Ps/animal).

Performance Criteria

Equations 9.20 through 9.23 compute a number of performance variables of the Costa model. These include: (1) farm assets and income; (2) foreign exchange and government revenues; (3) government expenditures on modernization campaigns; and (4) beef consumption. Other performance measures which may be useful in evaluating alternative modernization policies include output variables from other components. Examples of these are total cattle population, extraction ratios, and animals treated against contagious diseases.

Equations 9.20 compute value of capital in cattle production and farm income.

VALCAP(t) = VACAPL(t) + VLAND(t)	(9.20a)
FARMI(t) = FARIL(t) + FARMIC(t)	(9.20b)
FARMIA(t+DT) = FARMIA(t) + DT#FARMI(t)	(9.20c)
FARILA(t+DT) = FARILA(t) + DT#FARIL(t)	(9.20d)

DSREVL(t+DT) = DSREVL(t) + DT + PARIL(t) + EXP(-DIR+t) (9.20e) where: VALCAP = total value of animal population and grazing land (Ps) FARMI = total agricultural income (Ps/year) FARMIA = accumulated agricultural income (Ps) FARILA = accumulated farm income from cattle (Ps) DSREVL = discounted future returns from cattle (Ps) = the discount rate (proportion per year). DIR Foreign exchange earnings and government revenues are computed by Equations 9.21. FOREXA(t+DT) = FOREXA(t) + DT + FOREX(t)(9.21a)

GOVREV(t) = TAXCT(t) + TAXCM(t) + VLDTAX(t)(9.21b)

GOVREVA(t+DT) = GOVREVA(t) + DT = GOVREV(t)(9.21c)

where:

- FOREXA = accumulated foreign exchange earnings from cattle (Ps)
- GOVREV = government revenues from the cattle sector (Ps/year)
- TAXCT, TAXCM
 - = value of taxes on traditional and modern cattle, respectively--Equations 6.34 (Ps/year)
- VLDTAX = value of taxes on land based on the aggregated capitalized land value--Equations 6.33 and 9.15c (Ps/year)
- GOVREVA = accumulated government revenues from the cattle sector (Ps).

Government expenditures on modernization campaigns, Equation 9.22, include allocation of funds for development credit and control of foot-and-mouth disease (FMD).

ACRDTA(t+DT) = ACRDTA(t) + DT*ACRDT(t) (9.22a)

 $EXPDS(t+DT) = EXPDS(t) + DT \neq EXPAFT(t)$ (9.22b)

where:

ACRDTA = accumulated funds allocated for development credit (Ps)
EXPDS = accumulated public expenditures on control of FMD (Ps).

Finally, beef consumption is computed in Equations

9.23 for the total Colombian population.

POP(t) = POPO*EXP(C282*t)(9.23a)

$$PERCAP(t) = \frac{C281 * (SUPB(t) - EXPL(t) - UNEXPL)}{POP(t)}$$
(9.23b)

where:

POP	=	total Colombian population (habitants)
POPO	=	total Colombian population at the begin- ning of simulation (habitants)
C585	=	the rate of growth in population (por- portion habitants/year)
PERCAP	=	the Colombian per capita beef consumption (kgs/habitant-year)
SUPB	H	total Colombian beef supplyEquation 6.25b (animals/year)
EXPL	æ	registered beef exportsEquation 6.26c (animals/year)
UNEXPL		non-registered beef exportsEquation 6.26c (animals/year)

C281 = the average dressed careass weight (kgs/animal).

Table II.8 shows the values of a selected number of variables used in component CRTACC.

Definition (Equation N	o.)	Value
LT1 (9.1a)	Determines the time development loans are paid (years)	3
LT2 (9.1b)	Completes the grace period (years)	1
LT3 (9.1c)	Determines the time develop- ment loans are repaid (years)	8
RINTL (9.8b)	Determines the interest rate on development loans (proportion of debt/year)	.14
PEQCR (9.9a)	Determines the proportion of equity which can be used as a credit base (proportion/year)	•5
CREPR (9.9e)	Determines the repayment rate on short term loans (proportion of debt/year)	1
RINTC (9.9g)	Determines the interest rate on short term loans (proportion of debt/year)	.1
EXLMIN (9.12a)	Determines the farmers aggre- gated minimum living expendi- tures (thous. Ps/year)	200,000
SVALT (9.15a)	Determines the salvage value of traditional grasslands (Ps/ha)	1
RINT (9.15a)	Determines the opportunity cost of capital (porportion/year)	.18
EXSUB (9.16)	Determines the export subsidy (proportion of value of exports)	.15
AL1 (9.18b)	Determines the rate of growth in the exchange rate (pro- portion/year)	.0728
C212 (9.14)	Determines increased weight of "modern" animals (dimensionless)	1.4
C239 (9.9c)	Determines internal transfer of capital from crops to cattle production (proportion)	.1

TABLE II.8. Selected Coefficients and Initial Values in the Accounting and Performance Criteria Component (CRTACC).

.

TABLE	тт.8.	(continued)
TADUL	TT * 0 *	(construction)

Definition (Equation No	».)	Value
C261 (9.12b)	Determines the minimum consump- tion from gross income (pro- portion)	.25
C262 (9.12b)	Determines the maximum consump- tion from gross income (pro- portion)	•75
C263 (9.12b)	Governs response rate of consump- tion to changed in income (dimensionless)	.3
C264 (9.12b)	A scale factor (dimensionless)	10-8
C281 (9.23a)	Determines the average dressed carcass weight (kgs/animal)	200
C282 (9.23a)	Determines the rate of growth in population (proportion habitants/year)	.032
EXCHR1968	Official exchange rate at the beginning of 1968 (Ps/dollar)	15
POPO (9.23a)	Initial Colombian population (thous. habitants)	15,415.7

Sources: [3, 5, 17, 41, 42, 45] and initial guesstimates and model tuning.

PART III

VALIDATION AND TESTING

Introduction

Model testing is an ongoing process which should continue even after a model is implemented and in routine use. Testing, refining and validating a model are closely connected processes. A simulation model is tested both to check its internal consistency and to assure that it is an adequate representation of the complex processes of the real world. The validity of a model has to be established with some degree of confidence before a decision maker can base policy decisions on the experimental results of that model.

There are primarily three ways in which a model may be validated. The first method compares the structure of the model and its simulated output, using alternative assumptions about its behavior established by experts and from other published sources. This test uses the intuitive knowledge and expertise of people who have experience in Colombia and other developing countries.

The second approach attempts to compare the behavior predicted by the model under various conditions with what actually occurs as real time passes under the same conditions. Or alternatively, the model can be used to reproduce

historical data from the real world which are not used in the construction of the model. Once the model has been implemented, it is tuned and updated as an ongoing process with such comparisons.

Finally, sensitivity tests, which identify which of the model's parameters outcomes are most sensitive to their value changes, can also be conducted to validate the logic and internal consistency of the model.

Chapters 10 and 11 briefly discuss the model's data requirements and problems and examine two of the approaches commonly used to deal with these problems. These are tuning the model to track recorded time series and analyzing the model's sensitivity to variations in parameter value.

CHAPTER 10

DATA USAGE AND MODEL TUNING

Data requirements for the model are extensive. Data were obtained from a diversity of sources $\frac{1}{1}$ that included Ministry of Agriculture reports, FAO reports, World Bank reports, FEDEGAN reports, INCORA reports, Caja Agraria reports and statistics, DANE and Central Bank statistics, other published reports and informal guesstimates. Other data used were "synthesized" or "simulated" from various combinations of data. Often costs were one point estimates obtained from published sources that were later converted to base year values by means of indexes reflecting the rate of inflation in prices of farm inputs. Aside from the secondary sources, some informal primary information was obtained from a two-week survey of the Costa made in the summer of 1971. Yet, the data problems encountered were many and ranged from nonexistent information to unreliable and contradictory estimates. In Colombia, as in other developing countries, existing statistics on agricultural production are so deficient and deserve so low a degree of confidence that they create a problem in planning for

 $[\]frac{1}{Detailed}$ references are found in Chapters 1 through 3.

agricultural development. A few exceptions include estimates on area harvested and of the production of cotton, coffee, tobacco, bananas for export and sugarcane for sugar production. In the case of other crops and cattle the estimates are no more than conjectures [40].

However, researchers, planners and policy makers cannot wait for accurate and reliable data to recommend, plan, and make decisions on policies and programs for development. Models have to be designed on the basis of the best information that is readily available, and techniques may be used not only to improve the quality of data but also to make best use of the data available at the time.

The system simulation approach offers three ways of coping with the information problem. Sensitivity tests (discussed in the next two chapters) can reveal the implications of parameter variability both for the validity of the model and for policy formulation. These tests can also provide guidance for determining priorities for data gathering activities. Secondly, given coarse probability distributions for a set of key parameters, running the model in a Monte Carlo mode can generate directly output statistics reflecting data uncertainties [1, Chapter 4]. Finally, the model may be tuned to track a number of recorded reliable time series by adjusting uncertain parameter values. This procedure will be discussed later in this chapter.

The entire data input to run the model can be found in various chapters where this information is relevant and

and in the Appendix. Tables present the numerical values of the parameters and coefficients used to run the model in its deterministic mode, since time constraints precluded use of a stochastic determination of their values.

Model Data Requirements

Data for the Costa cattle/crops model fall into four general categories: system parameters, technological coefficients, initial conditions, and historical time series. The data requirements of the first three categories are extensive and obtained from a diversity of sources including descriptive information and guesstimates from knowledgeable persons. In this section, we will briefly discuss the first three categories of data and their sources. Historical time series, used in tuning the model, will be discussed in the next section.

System Parameters

Fundamentally, system parameters reflect the behavioral characteristics of the system being modeled. These parameters and the interconnected basic equations, in fact, define the system. A few examples of the many system parameters of the Costa model are:

- 1. the land modernization and disinvestment profitability response parameters (E9, E8 in Equation 5.36 and E91, E81 in Equation 5.47);
- 2. the land use response rate parameter (CL1 in Equation 5.21);
- 3. the profitability discount rate (DIR in Equation 5.31);

- 4. the many delays and averaging and smoothing lags of the model (e.g., XDEL in Equations 5.43, DEL19, 16 and 17 in Equations 5.1);
 - 5. the sales restriction parameters (BMN, BMX, AMX and AMN in Equations 6.22);
 - 6. the farmers' resource use profitability response parameters (C235, C236 in Equation 6.36);
- 7. the income elasticity of expenditure parameters (C261, C263 in Equation 9.12b).

There is little or no information on most of the behavioral system parameters and acquisition of this type of data would entail survey research that has never been The values used in the early stages of building conducted. and testing the model were educated and intuitive guesstimates acquired from various secondary sources [e.g., 29, 42, 53], from experiences in other developing countries (mainly experience acquired by the Michigan State University simulation team for Nigeria) and from such primary sources as interviews with Colombian officials and farmers in the Costa. Although values of selected system parameters are shown in Tables III.1 and III.2, we will take a close look here at the pasture land modernization and reversion transition response thresholds as an example (E9 and E91 in Table The value of E9 shown (.5) means that the alter-III.1). native to traditional grazing must be at least 50 percent more profitable before farmers will transfer the land to modern management. And the value of E91 (.3) means that the profitability of the modern operation must be at most 30 percent higher than that of the traditional one before farmers will reverse the modernization process. The relative

		•		
•		Alternative Uses		
Variables (Eqn. No.) (definition)	Present Uses	Modern Grazing	Traditional Grazing	
E9 (5.36)	Traditional Grazing	•5		
(response threshold) E91 (5.47) (response threshold)	Modern Grazing		.3	
E8 (5.36)	Traditional Grazing	1.0		
(governs response rate) E81 (5.47) (governs response rate)	Modern Grazing		2.0	
DIR (5.31) (discount rate)		.15	.15	

TABLE III.1. Profitability Response Parameters for Traditional and Modern Grazing (Dimensionless).

Source: Initial guesstimates and model tuning.

РАВЬЕ	111.2.	Cattle	Production	Parameters.
		(

	ر مر هم هم محمد معمد معمد معمد معمد معرف معرف معرف معرف معرف معرف معرف معرف	
Definition (Equation N	Value	
MKM (7.3)	Marketing margin	.15
C213 (6.24)	Proportion of dead animals consumed	0
C214 (6.24)	Proportion of fertile cows sold for slaughter	.6
C215 (6.24)	Proportion of growing females sold for slaughter	.2
C216 (6.24)	Proportion of growing males sold for slaughter	0
C242 (6.27a)	Proportion of heifers treated against brucellosis	.05
C244 (6.28b)	Annual proportion of animals privately treated against foot-and-mouth disease	•3

Source: [29, 31, 63, 64] and initial guesstimates and model tuning.

values hypothesize different farmer attitudes (e.g., risk aversion and uncertainty as discussed in Chapter 5) toward cattle modernization.

Despite the lack of accuracy in parameters such as those highlighted above, they play an important role in the validation of the model. Some of them provide a range of values which may be tested in tuning the model to track historical time series and to improve the model's behavior in comparison to "reality." Some others, as shown by sensitivity tests, are not crucial to the model's performance and therefore the results are more sensitive to other elements in the system.

Technological Coefficients

Technological coefficients are probably the easiest to obtain and handle in the model. The various sources of data include Ministry of Agriculture reports, FAO reports, World Bank reports, INCORA reports, FEDEGAN reports and many other published reports [e.g., 5, 9, 20, 29, 31, 32, 33, 42, 43, 57, 58, 60, 61, 66]. The existence of data for these parameters does not mean they are completely reliable. Instead, more research and field work will be necessary to increase their level of confidence.

Some examples of technological coefficients used in the model are:

- 1. crop yields (YLDCL, YLDCU and YLDFC in Equations
 6.2);
- 2. pasture yields (e.g., CGOU, CGOUL, CGOL and CGOLL in Equations 6.3);

- 3. costs of production (e.g., CSTHCU, CSTHCL and CSTHFC in Equation 6.29b);
- 4. average carcass weight per slaughtered animal (C281 in Equation 9.23b);
- 5. mean times spent in the cattle production stages (DGROF, DGROM, DPRODF and DPRODM in Equation 6.10a).

Almost all the technological coefficients remain constant throughout a simulation run. Some exceptions are costs of production and price of crops that change with domestic inflation, and crop yields. Learning curves for yields are discussed in detail above in component AGPRAC, Equation 6.1. Values of selected technolgoical coefficients are presented in Tables III.3 and III.4.

Initial Conditions

Initial conditions (1960) define initial values of all levels (and some rates) that must be given before the first cycle of model computations can begin. Since their values change during the course of a run they must be reset at the start of each run. Some of these include:

- 1. land usage (e.g., TLFCO, TLCRL and TLCRU in Equations 5.4, 5.11 and 5.24);
- 2. cattle population in each cohort (PMG, PMP, PFG, PFP and OLDF in Equations 6.11 and 6.12);
- 3. crop prices and price averages (component Price Generation, Chapter 7)

4. total demand for beef (TDEM in Equation 6.26c).
Some of these variables present no data problems.
For instance, assuming all cattle population at time zero
(1960) is traditional, we determine that modern population.
in each cohort is zero. But the model is quite sensitive

to the initial cattle inventories, as we shall discuss later (Chapter 11), so more complete and accurate cattle population estimates would increase this model's accuracy. Values of selected initial conditions are shown in Table III.5.

It must be stressed that the model can be useful for planning economic development, in spite of imprecise parameter estimates, for it is not necessarily the aim of a development model to forecast in absolute terms the values that will be attained by certain variables at a specified time. The aim is to design a strategy of development by experimenting with the model under various assumptions and then by comparing alternatives.

Tuning

The major components of the Costa model were programmed, simulated and tested individually as part of the overall modelbuilding process. During this process, conceptual and programming errors were detected and corrected, and then the components were integrated into the Costa model. Extensive model tests were performed on the larger model to eliminate programming errors and inconsistencies between related model components, and to examine its correspondence with the real system. Checking the model against time series of past behavior and adjusting the values of certain system parameters, adding new mechanisms, or modifying structural relationships is what is known as "tuning" the model. These checks are made before the model is implemented and they suggest which

					5
			5 % S		· · · ·
ULA DE 12	111 2	Posturo	and Cron	-Yioldg	(Long/ha-vear).
TADD	111	1 (14) 1444 17	CLERKA VEL VEP	1 1 1 1 1 1 1 1 1 1	(we will be a set of the set of

Pastures (Average TDN Yjelds)* (Variable) (Eqn. No.)	
<u>Uplands</u> Traditional artificial (CGOU) (6.3a) Traditional native (CGOUI) (6.3b) Modern artificial (CGUI) (6.3c) Modern native (CGU2) (6.3d)	3.48 1.16 5.0 1.7
Lowlands Traditional artificial (CGOL) (6.3a) Traditional native (CGOL1) (6.3b) Modern artificial (CGL1) (6.3c) Modern native (CGL2) (6.3d)	3.8 1.26 5.1 1.7
Land in Transition Transition artificial (CG3) (6.9b) Transition native (CG4) (6.9b)	3.3 1.16
Crops Cash crops in lowland (YLDCL) (6.29a) Cash crops in upland (YLDCU) (6.29a) Food crops (YLDFC) (6.29a)	1.56 1.11 8.3

*TDN = Total digestible nutrients

Sources: [7, 33, 61, 66] and

.

- Instituto Colombiano Agropecuario (ICA), <u>Dia de</u> <u>Campo-Pastos y Forrajes--Ganado de Carne</u>. (Monteria, September 12, 1970), 28-58.
- . "Algunos Aspectos de la Fertilizacion de Pastos." <u>Pastos y Ganados para la Costa</u> <u>Atlantica</u>. Boletin Tecnico No. 15 (Bogota, 1967), 33-42.
- No. 029. (Bogota, April, 1970), 1-4.
- . <u>El Pasto Guinea</u>. Hoja Divulgativa No. 029. (Bogota, March, 1971), 1-4.

.
TABLE III.4. Mean Length of Cattle Production Stages (years).

Production Cohorts* (Variable)(Eqns. No.)		
Growing females (DGROF) (6.10a) Growing males (DGROM) (6.10a) Producing females (DPRODF) (6.10a) Producing males (DPRODM) (6.10a)	u.	2.5 2.5 10.0 3.0

*Traditional and modern

Sources: [5, 29, 61, 66]

TABLE III.5. Selected Initial Conditions (1960).

Cattle	Population	(thousand	head)	(Variable))(Eqn.	No.)

Growing males (PMGT) (6.11)	1,174
Growing females (PFGT) (6.11)	1,183
Producing males (PMPT) (6.11)	943
Producing females (PFPT) (6.11)	2,424
Old cows (OLDFT) (6.12)	427

Land Use (thousand hectares) (Variable) (Eqn. No.)

Cash crops in lowlands (TLCRL) (5.11)	140.5
Cash crops in uplands (TLCRU) (5.24)	281.0
Food crops (TLFCO) (5.4)	101.73
Export banana (TLBAN) (5.13)	20.0
Grazing land in region 1 (TGLSF1) (5.1d)	2,510.36
Grazing land in region 2 (TGLL) (5.16)	319.8
Grazing land in subregion 1 (TGLU1) (5.26)	1,471.7
Grazing land in subregion 2 (TGLU2) (5.18)	1,252.3

Sources: [Table I.2, 8, 15, 16, 29] and initial guesstimates and model tuning. parameters need adjustment, or where a structural relation must be added to the model to improve its behavior in comparison to "reality."

Despite the deficiency of Colombian statistics on agricultural production, four time series (1961-1970) were used initially in tuning the Costa model: Colombian supply of beef, market price of finished males, land in crops, and cattle population in the Costa. Since the modernization campaign promotion started in 1965, the tuning process included many of the parameters and structural relationships used in the modernization decisions, and the simulated series reflect the effects of the first five years of campaign implementation. The combination of traditional management before 1966 and modern management with improvement of artificial grasses and substitution of artificial for native grasses (alternative 2) from that year on was used as a standard run for tuning the model. Alternative 2 was selected because it patterns more closely the modernization program currently carried out in the Costa. Plots of the above simulated series along with the actual ones are depicted in Figures 111.1 and 111.2. Fable III.6 displays the four time series resulting after the initial coarse tuning. Data values generating this fit were used in the policy runs discussed in Chapter 12.

Although mathematical measures of the goodness-offit could have been used as a criterion to measure past behavior characteristics [1, 21, 53], at this stage in the



Figure III.1. Results of "coarse" model tuning--cattle population in the Costa and Colombian beef supply time series against simulated series. Cattle inventories are taken at the end of the year.



Figure III.2. Results of "coarse" model tuning--market price of finished males and land in crops time series against simulated series.

TABLE III.6. Time Series Tracking.

Year	Colombia (thous. DATA	n Beef Supply animals/yr.) SIMULATED	Cattle Pop Costa (tho DATA	oulation in the ous. animals) SIMULATED	Land in C Costa (th DATA	rops in the nous. has.) SIMULATED	Price of F (Pesc DATA	Finished Males os/animal) SIMULATED
1961	1,965	1,970	6,200	6,235	533-3	555.1	1,058	1,150
1962	2,145	2,123	6,400	6,356	548.2	565.9	1,054	1,236
1963	2,305	2,292	6,560	6,492	552.5	576.7	1,117	1,272
1964	2,418	2,350	6,680	6,656	n.a.	587.7	1,346	1,322
1965	2,469	2,324	6,800	6,856	619.4	599.0	1,629	1,459
1966	2,228	2,156	6,920	7,054	614.7	610.4	2,089	1,718
1967	2,205	2,006	7,160	7,181	616.7	620.6	2,345	2,097
1968	2,360	2,235	7,480	7,284	589.7	622.7	2,606	2,425
1969	2.624	2,485	7,800	7,403	589.6	633.4	2,599	2,523
1970	2,820	2,710	8,080	7,555	n.a.	642.4	2,729	2,567

n.a. = not available

Sources: As indicated on pp. 238-39.

development of the model more emphasis was placed on the general qualitative nature of particular interrelationships in the various model components that, in concert, produce the behavior that interests us in the actual system. Nevertheless, a close look at Figures III.1 and III.2 reveal a closeness between simulated and observed real-world data that increases confidence in use of the model as an experimental tool to study the effect of structural and policy changes in the actual system.

During the period 1961-1967, the observed prices plotted in Figure III.2 are the national average of current prices as reported by Garcia Samper [23]. But between 1968 and 1970, prices are a weighed average of finished males at the Medellin stockyard as reported by the Central Bank.¹/ Since Medellin prices for live animals are usually higher than in other markets, the average prices of finished males computed in Table III.6 for the period 1968-1970 are likely overestimated in relation to nationwide averages. This means that if national averages were used instead of Medellin averages for the last three-year period of observations, the match of the recorded and simulated price series in Figure III.2 would have been even closer. Estimates of cattle population and beef supply were obtained from $DANE^{2/}$ and

2/Departamento Administrativo Nacional de Estadistica, Pronosticos y Estimaciones Agricolas, (Bogota, 1971), p. 24.

<u>l</u>/Banco de la Republica, "Resumen de las Principales Ferias de Ganado en el Pais," <u>Revista del Banco de la</u> <u>Republica</u>, XL-XLIII (1967-1970), Bogota.

those of land in crops from Caja Agraria. $\frac{1}{}$ Cattle population in the Costa was assumed to remain at a constant 40 percent of Colombian cattle population.

Consistency Checks

Consistency checks are useful in evaluating the basic equations used to simulate real world behavior and to test the value of their associated parameters. Changes in parameter values and functional relationships may result after comparing the model's data with reliable data given by others. The demographic model with input data from alternative 2 was used to make comparisons with some of the productivity relationships given by the CIAT survey [9]. These relationships include average grazing rate, average daily per hectare milk production and average cows per hectare. As Table III.7 shows, the results of these checks indicate that, on the basis of data from actual records, the model's program is performing reasonably well in reproducing real world situations. In the following development of internal consistency checks, these definitions are used.

TOPOPR	<pre>= total cattle population in the Costa (animals)</pre>
TGLR	<pre>= total grazing land in the Costa (hectares)</pre>

<u>l</u>/Caja de Credito Agrario, "Calculos de Produccion Agricola de 1958 a 1963," Carta Agraria, Anexo al No. 144 (Sept. 1964), Bogota, I-IV.

Idem, "Estimativos Sobre Areas Cultivadas y Produccion Obtenida Para Los Principales Cultivos, 1965-1969," <u>Carta Agraria</u>, No. 244 (Nov. 1970), 5-14.

Productivity Index	Model (1970)	CIAT (1971)
Cattle head/ha.	1.46	1.45
Cows/ha.	• 59	.56
Lts. milk/haday	.6	.93

TABLE III.7. Results of Consistency Checks.

PFCT, PFCM	=	total traditional and modern cow population (producing and old females), respectively (animals)
QMCT, QMCM	=	total traditional and modern quantity of milk produced, respectively (liters/cow-day)
FERT, FERM	ш	total number of traditional and modern fertile cows, respectively (animals)
PFLACT, PFLACM	=	proportion of traditional and modern lactating cows, respectively
CT202, CM202	E	proportion of traditional and modern lactating cows which are milked, respectively.

The following equations define some of the prodictivity indices tested:

Cattle head/ha. = $\frac{\text{TOPOPR}(t)}{\text{TGLR}(t)}$

۰.

.

$$Cows/ha. = \frac{PFCT(t) + PFCM(t)}{TGLR(t)}$$

Lts. milk/ha.-day = $\begin{cases} \frac{PFCT(t)}{PFCT(t)} * (QMCT(t) - QMCM(t)) + \\ \hline \\ QMCM(t) \end{cases} * \frac{(FERT(t)*PFLACT(t)*CT202 + FERM(t)*PFLACM(t)*CM202)}{TGLR(t)} \end{cases}$

General Validation

Although the Costa model has been designed to experiment with four alternatives to traditional cattle management (see Chapter 4), manpower and time constraints required this policy experimentation be limited to one alternative. Briefly, alternative 1 considers the improvement of native and artificial grasses; alternative 2 considers the improvement of artificial grasses and the substitution of artificial for native grasses; alternatives 3 and 4 add the production of forages and silage to the improvement of range lands in alternatives 1 and 2, respectively. While alternatives 3 and 4 presented the problem of predicting the consequences of practices never done in the past, and alternative 1 implied little improvement over traditional practices, alternative 2 showed the ability to reproduce a pattern similar to the actual system. This ability was demonstrated during the process of tuning the model discussed earlier. It was fortunate that a modernization program was in effect in the region during the last five years of the period used to tune the model. This circumstance allowed evaluation of the reasonableness of the data and functional relationships employed in building the model, and increased confidence in alternative 2 as a predictor of the value of the real system over the future and as the "standard" or "base" run for policy experimentation.

Nevertheless, as part of the general validation of the model, the four alternatives were tested and compared under the same underlying general assumptions described in

Chapter 12 for the base run. As a first approximation, most of the system parameters and technological coefficients used in testing alternatives 1, 3, and 4 were the same adopted for alternative 2. However, new auxiliary equations and technological coefficients were introduced, and some system parameters changed as required by the nature of a particular alternative. While it is true that no strict statistical or econometric methods of verification and validation were used, less rigorous objectivity tests were applied repeatedly in assembling data, modifying and developing model components, combining smaller into larger model components and in evaluating model output. The objectivity of concepts or empirical theories requires that they be:

1. consistent with observed and recorded experience,

- 2. internally and logically consistent,
- 3. interpersonally transmissable, and
- 4. workable when used to solve problems.

Table III.8 summarizes the output of selected variables at the end of 25 years of simulation time. The results of this test suggested that more experimentation was necessary in order to better fit the behavior of the real system. This was especially true of all behavioral parameters and more specifically of those that control the modernization decisions in Equation 5.36.

As Table III.8 shows, no modernization takes place (TMPL = 0) when alternative 1 is considered as a means of improving cattle production. This result indicates that,

Variables Type of	Cattle Population in the Costa	Land in Forages	Land In Cash Crops	Modern Pasture Land	Traditional Grazing Land	Accumulated Farm Income from Cattle	Accumulated Government Revenue from Cattle	Living Expenditures	Market Price	Beef Consumption Per Capita
Modern Management	TOPOPR (thous nead)	TLF (thous nas)	TLCC (thous nas)	TMPL (thous nas)	TTGLR (tnous has)	FARILA (mill Ps)	(mill Ps)	(mill Ps)	(Ps)	(Kgs)
Alternative 1	7,077	0	628.0	0	5,125	30,980	2,561	2,334	10,570	18.8
Alternative 2	9,211	0	625.7	725.0	4,375	32,280	2,721	2,750	9,916	19.96
Alternative 3	16,720	553.1	625 .2	2,019	2,556	1,860	3,334	3,860	7,176	25.42
Alternative 4	15,010	225.6	625.7	1,512	3,390	18,910	3,364	3,404	7,274	24.34

TABLE III.8. Selected Results of Alternatives to Traditional Cattle Production--1985.

given the model assumptions, the expected profitability of this alternative over the planning horizon does not produce the necessary incentive to induce farmers to change their traditional practices. We must remember that in the decision mechanism set in the model (Equation 5.36) modernization does not begin unless the perceived relative profitability (PDR) of the modern operation is larger than the value of the threshold parameter (E9). This and the response rate parameter (E8) are used to simulate farmers' attitudes toward the new methods of production, and as stated earlier, they have been assigned the same values as in alternative 2. But the minimum risk involved in alternative 1 suggests that farmers might react differently than they do when facing the more risky alternative 2. This consideration does point out the need of experimenting with other parameter values in the transition response function of alternative 1.

In alternatives 3 and 4 the transition response is the greatest as shown by the total modern pasture land (TMPL) and the cattle population (TOPOPR). However, these alternatives involve the greatest risk since they include the use of silage during the dry season which is a practice unknown in the region except for some experimental trials. In addition, they require the planting of large areas in forages that, in the case of alternative 3, almost equal that of cash crops. This implies a great effort on the part of both farmers and agencies supplying services--particularly in a region where cropping has not been the leading agricultural activity.

The effect of a too rapid rate of modernization in alternatives 3 and 4 is shown by the declining and lower value of the accumulated income from cattle (FARILA). As a matter of fact, income from the modern operation becomes negative in some years as a result of lower animal prices arising from larger supplies, increased operating costs from producing silage and larger expenditures on land improvement. The preceding discussion suggests that the simulated decisions of farmers in alternatives 3 and 4 have to be adjusted to a rational behavior more attuned with their profit and cash flow positions, and market situations.

Tuning the standard run to track four time series was a necessary but insufficient condition to validate the model. Confidence is confirmed by the correspondence of total model behavior to that of the actual system. And this is accomplished in a process of intuitive, theoretical, and empirical consistency analyses. This general process of model validation is very judgmental and should be viewed as an iterative and ongoing process of the model's development and application. Although different validation techniques have been applied to complex general systems simulation and econometric models used for decision making, the quest for better model evaluation methods continues among researchers. Detailed discussions of the methodological and philosophic issues involved in the process of validating and verifying complex models are found in Rossmiller, et al. [62], Johnson and Zerby [49], Fromm [22] and Shapiro [65].

CHAPTER 11

RESULTS OF SENSITIVITY ANALYSIS

The primary purpose of sensitivity tests is to indicate those areas of the model in which changes in parameter values or formulations have a significant impact on model results. Such information is useful, not only for model tuning and validation, but also for policy making and as a guide to data collection priorities. Sensitivity tests on an individual or combination of parameters are essential in model development since they enable us to check the internal consistency of the model against the theoretical and empirical knowledge we have on the real system. Additionally, they indicate those portions of the model which deserve the greatest additional research. Time and money can be saved by not studying parameters that have little or no effect on the results. Thus, the efficiency of both research and the decision-making process are improved. The preceding uses of sensitivity analysis have been explained in detail and exemplified by Abkin [1, Chapter 9]. An analysis of the results of a series of sensitivity runs of the Costa model is presented in the following section.

Analysis of Results

Methodology

The series of 17 sensitivity runs conducted with the Costa model investigate the effects of varying 45 selected coefficients. These 45 parameters are broken down into: (a) 12 production coefficients; (b) 12 land allocation coefficients; (c) 11 price and cost coefficients; and (d) 10 time lag or smoothing parameters. The production coefficients include proportion of cows milked, initial proportion of land in artificial pasture, initial cattle population, rate of transfer of animals from traditional to modern management, illegal cattle exports, and crop yields. Land allocation coefficients include initial land in the three agricultural regions, profitability response parameters of traditional cattle, internal transfer of capital from crop to cattle production, and consumption expenditures. Price and cost coefficients include price of crops and milk, cattle marketing margin, rate of inflation of farm inputs, and operating costs of crops and cattle. And finally, time lag or smoothing parameters include average lengths of time for averaging output, price and cost of crops, and cattle. Some of these parameters were tested individually while others were tested in combination in the same run; each run simulates agricultural production in the Costa over a 25-year period, 1960-1985.

A strict test of parameter sensitivity would require that all parameters be varied in the same way, e.g., plus

20 percent. The results could then be compared on the basis of a given deviation. This procedure would contribute nothing, however, to understanding the nature or overcoming the problem of the uncertainty arising from varying degrees of confidence in parameter values. If we increase a parameter value by 20 percent, we obtain a score value which may be either less than one standard deviation from the mean and have a small variance, or greater than one standard deviation and have a large variance. The resulting deviations in the output variables would be quite different in the two cases. Therefore, in statistical terms it would be meaningless to compare the relative consequences of 20 percent variations in each of two parameters whose means may be varied so differently.

The general rule followed, therefore, in defining the sensitivity runs described here was to vary each parameter by an amount which I felt covered most of its variability --two standard deviations. In the absence of informed judgments or other estimates of relevant distributions, the variations used were guesstimates on my part. They certainly do not imply the degree of precision suggested by the term "two standard deviations," but do reflect the relative width of subjective confidence intervals. But when estimates were available from several sources (for example CPLPT and MKM) their reported values were entered in the sensitivity runs. In this case sensitivity tests are helpful in establishing the confidence that can be placed on different sources of

information by checking the consistency of the output generated by a given estimate against the knowledge we have on the real system. Where possible, similar parameters were varied by similar amounts. For example, the cattle population cohorts were each decreased 25 percent, the lag parameters were each decreased 40 percent, and the initial land parameters were decreased by 10 percent each.

Output criteria, parameter deviations and results are tabulated in Table III.9. The output criteria include farm income from cattle (FARILA), farm income from crops (FARMICA) and government revenues from cattle (GOVREVA) accumulated over the 25-year simulation. In addition, effects of parameter variations are shown for cattle population in the Costa (TOPOPR), capitalized value of pasture land (VLAND), value of capital on cattle (VALCAP), and Colombian beef per capita consumption (PERCAP).

The discussion of results has been limited to the major explanations of the more appreciable output deviations. The test of some production parameters from Runs 2 through 5 indicates that the initial proportion of grazing land in artificial pasture is quite important in determining the performance of the model. Results of Run 3 are the same as those of traditional management under "worst" nutrition conditions. $\frac{1}{}$ This means that farmers are not motivated to adopt the new production methods. The higher

 $[\]frac{1}{"}$ "Worst" and "best" nutrition conditions as used in this analysis are related to the initial proportion of land in artificial pastures.

1						Performance Criteria ^{2/}							
						TOPOPR	FARILA	FARMICA	GOVREVA	VLAND	VALCAP	PERCAP	
1						Base Run Value							
			Percent		Base Run Value	9,211- thous. animals	32,280 m111. Ps	56,070 mill. Ps	2,721 mill. Ps	9,361 mill. Ps	65,300 mill. Ps	19.96 Kgs/person- year	
lun	Farameter Tested	Parameter Definition	From Base Run	Run Value			Pe	rcent De	parture	From B	ase Run		
2	CT202	Proportion of "tradi-	15.000										
	CM202	tional" cows milked Proportion of "modern"	+25.0	-5	-4	1-4.46	8.05	.12	.48	-16.5	-6.02	-2.3	
3	CPLPT	cows milked Proportion of initial	+25.0	-5	• 4	p							
1		(1960) grazing land in	-55.55	2	.25	-28.5	-16.3	. 09	-20.51	-68.52	- 31 . 1	-9.7	
4	C12	Effect of nutrition on						,				-9.1	
5	UNEXPL	Illegal exports (thou-	-50.0	.>	1.0	30	22	0.	07	-21.0	-30.0	-1.05	
6	E8	rand head/year)	-50.0	150	300	-11.8	-24.8	•2	-7.72	-74,4	-25.9	4	
		response of traditional	22.22			1. 12	6				10.7		
7	29	Profitability threshold	-33.33			-0.95	02	0.	-2.2	-31.11	-10.7	-3.01	
		of response of tradi- tional to modern cattle	+100.0	1.0	.5	1							
	E91	Profitability threshold				1.18 0		nh		1.2 .			
		vestment in modern	1.11.21			114.5	2.2	.04	-5.9		-20.12	-5.81	
	DIR	Discount rate over the	+66.67	.5	• 3	1			1.1.1				
		planning horizon	1.11.1		1	1.00							
9	C239	Proportion of Internal	+20.0	.15	.15	-5.7	-1.12	0.	-2.13	-25.9	-8.6	-2.61	
		transfer of capital from crops to cattle											
10	P 261	production	-50.0	.05	.1	-1.6	.74	0.	15	27	-1.6	15	
10		ALPH2 tends as income				1.1				1			
11	C263	gets large Rate of response of	-20.0	.2	.25	.2	46	0.	0.	-27.4	-3-3	-1.05	
		consumption expendi-											
12	-	gross income from cattle	+16.67	- 35	.3	o.	0.	0.	0.	0.	0.	0.	
12	PHM	(Ps/liter)	+50.0	1.5	1.0	-7.5	15.8		- 48	-18 0	-0.21		
13	мкм	Marketing margin of				1			46	-10.9	-9.21	-4.0	
		(proportion of price)	+60.0	.24	/ .15 ^t	-13.7	-34.7	-25	-9.41	-83.9	-34.01	-5, 41	

TABLE III.9. Results of Sensitivity Tests on Selected Parameters of the Costa Model.

目的思想

10000000

١.

250

CPC

Χ.

TABLE III.9. (continued)

								Per	formance	Criter	1aª/			
	Parameter Tested				Base	TOPOPR	FARILA	FARMICA	GOVREVA	VLAND	VALCAP	PERCAP		
						Base Run Value								
			Percent	Tast		9,211 thous. animals	32,280 mill. Ps	56,070 mill. Ps	2,721 mill. Ps	9,361 mill. Ps	68,300 mill. Ps	19.96 Kgs/person- year		
Hun		Parameter Definition	From Base Run	Run Value	Run Value		P	ercent D	eparture	From B	ase Run			
14	RCST	Rate of increase in				1								
	CSTHCU	year) Initial average cost of	-16.67	.1	.12									
	CSTHCL	ha-year) (1960) Initial average cost of	+18.93	666	560		20.8			167 4		6		
	CSTHEF	ha-year) (1960) Initial average cost of	+18.51	1300	1097									
	CTLBLT	(1960) Initial cost of labor	+18.95	860	723	14.1	39.8	-50.4	10.5	147.4	31.3	3.51		
	VETRAC	in traditional sector (Ps/animal-year) (1960) Initial veterinary cost	+19.02	21.9	18.4									
	CLNDT	in traditional sector (Ps/animal-year) (1960) Initial average operat-	+19.67	7.3	6.1									
		ing cost of traditional grazing land (Ps/ha- year) (1960)	+18.72	42.5	35.8									
15	CSTHCU	Initial average cost of upland cash crops (Ps/			55.0	ĥ					.)			
	CSTHCL	Initial average cost of lowland cash crops (Ps/	+13.5/	030	500									
	PCROPU	Initial average price of upland cash crops	40	1092	1097									
	PCROPL	(Ps/ton) (1960) Initial average price of lowland cash crops	+6.03	668	630	03	٥.	3.25	0.	.1	0.	0.		
	YLDCU1	(Ps/ton) (1960) Initial average yield	+.67	898	892									
	YLDCL1	(ton/ha) (1960) Initial average yield	+8.18	1.19	1.1									
		(ton/ha) (1960)	+.64	1.57	1.56									

TAPLE 111.9. (continued)

								fer	formance	Criter	taª/		
						TOPOPR	FARILA	FARMICA	GOVREVA	VLAND	VALCAP	FERCAF	
1						Base Run Value							
Parameter In Tested	Parameter		Percent Change	Test	Base	9,211 thous. animals	32,280 mill. Fa	56,070 m111. Ps	2,721 mi11. Ps	9.361 mi11. Ps	08.330 m111. Fa	istar Ega. person- year	
	Definition	Base Run	Value	Value		Pe	rcent De	parture l	From Ba	se Run			
6	DEL5	Average length of time of exponentially averaged				1							
	DEL6	(years) Average length of time of exponentially averaged traditional cattle sales	-40.0	3	5								
	DEL7	(years) Average length of time of exponentially averaged upland cash crops price	-40.0	3	5								
	DELS	(years) Average length of time of exponentially averaged upland cash crop yields	-40.0	3	5								
	DEL12	(year) Average length of time of exponentially averaged per hoctare taxes on traditional cattle	-40.0	3	5								
	DEL13	(years) Average length of time of of exponentially averaged per hectare traditional land tax (years)	-40.0	3	5	÷ (1.3	4.3	16	9.33	18.1	17.7	8.5	
	DELIG	Average length of time of exponentially averaged per hectare traditional cattle operating costs											
	DEL20	Average length of time of exponentially averaged operating costs per traditional animal	-40.0	3	5								
	DEL21	(years) Average length of time of exponentially averaged per hectare operating	-40.9	3	5								
	DEL22	costs of traditional grazing land (years) Average length of time of exponentially averaged per hectare upland cash	-40.0	3	3								
		crop costs (years)	-40.0	3	5								

and the second se

TABLE 111.9. (continued)

Run	Parameter Tested	Parameter Definition	Percent Change Prom Base Run	Test Run Value	Base Run Value	Performance Criteria ^{2/}						
						TOPOPR	FARILA	FARMICA	GOVREVA	VLAND	VALCAP	PERCAP
						Base Run Value :						
						9.211 thous. animals	32,250 mill. Pc	56,070 mill. Ps	2,721 mill. Ps	9,361 m111. Ps	68.300 m111. Fs	19.96 Kgs/person- year
						Percent Departure From Bane Run						
17	PMGI	Initial size of growing male herd (thousand animals) (1960)	-25.0	850	1174	1						
18	PMPT	female herd (thousand animals) (1960) Unitial size of produc-	-25.0	887	1183							
	PFPT	ing male herd (thou- sand animals) (1960) Initial size of pro-	-25.0	707	943	-12.8	-8.33	16	-8.1	-49.8	-19.33	-5.31
	OLDFT	(thousand animals) (1960) Initial size of old	-25.0	1818	2424							
	TGLSFO	female herd (thousand animals) (1960) Initial seasonally	-25.0	320	427							
	TLDRNLO	(thousand has) (1960) Initial drainable	-10.0	2824	3138							
	TRAVLO	(1960) Initial flood-free agricultural land in	-10.0	2824	3138	-5.95	9.73	39	-4.81	1.8	-1.33	-2.05
	TLAVU01	lowland (thousand has) (1960) Initial agricultural	-10.0	480	533-3							
	TLAVU02	land in subregion 1 (thousand has) (1960) Initial agricultural	-10.0	1950	2169							
		(thousand has) (1960)	-10.0	1508	1676							

a/ Values are at the end of a 25-year simulation run (1985). All monetary values are in Colombian pesos and weight units are metric tons. Variables are: TOPORE = cattle population in the Costa FARILA = time integral of farm income from cattle FARICA = time integral of farm income from crops GOVREVA = time integral of government revenue from cattle VLAND = capitalized value of pasture land VALCAP = value of capital on cattle FERCAP = Colombian beef consumption per capita

b/ [Hector A. Sarmiento], <u>Colombia as a Cattle Exporter</u>, in <u>Pactors Affecting the Beef Cattle Industry's Return on Capital and Development Possibilities in Selected Areas in South America</u>, by Roderlen Von Oven (unput linkes, 1966), 21.

g/ James E. Grunig, Information, Entrepreneurship, and Economic Development, (unpublished Ph.D. dissertation, University of Wisconsin, 1968), p. 335, guoting Supply Problems of Basic Agricultural Products in Colombia, Latin American Agricultural Marketing Institute (ILMA), (Bogota, 1964).

cost of planting a larger area in artificial grasses lessens the relative profitability differential of improved over traditional cattle production to a value that is below the minimum set in the model before farmers begin modernization (the threshold parameter of Equation 5.36). Although by 1975 this relative profitability differential is greater than .5, the absence of promotional activity from that year on precludes the adoption of improved management practices (see Equation 5.37). This seems to be an unrealistic behavior of the model that may require further consideration since this implies that ten years of campaign promotion do not produce any traceable effect in the region, or that farmers are unable to take advantage of profit opportunities by themselves. Increasing the proportion of cows milked (Run 2) licreases cattle income but decreases the relative proficuoility differential of improved over traditional cattle production. Although this slows the adoption of improved management practices, the impact on the output variables is not significant. While controlling the rate of animal transfer from traditional to modern has no major effect on most of the output variables (Run 4), the volume of illegal exports has an important effect on the performance of the model (Run 5). In general, given the moderately inelastic price elasticity of demand for beef (-0.7), changes in domestic supply have an impact on prices which, in turn, significantly affect the overall performance of the model. This effect will be discussed in more detail in the next chapter as part of the policy experiments.

Runs 6-11 test the sensitivity of a number of land allocation coefficients. Of the land allocation coefficients the profitability response parameters are the most sensitive (Runs 6-8), particularly the adoption response threshold parameter (E9 in Run 7). Although in Run 7 the relative profitability differential of improved over traditional cattle production is greater than one after 1975, its output is the same as that of traditional management under "best" nutritional conditions. This performance experiences the same unrealistic behavior of Run 3 that was discussed earlier. The parameter controlling the response rate (E8) and the discount rate appear to be somewhat sensitive (Runs 6 and 8). A look at Figure II.1 in Chapter 5 will explain why increases in E9 have greater consequences on model output than decreases in E8. Since traditional and modern cattle profitabilities are not too far apart, the modern alternative is not too far out on the PDR axis, so the response is nearly at its asymptotic minimum. Increases in the threshold, i.e., shifting the curve to the right, has greater consequences than flattening the whole curve by decreasing the response rate. This is a fortunate circumstance since it is easier to determine the threshold than the response rate.

The parameters tested in the last three runs of this group (Runs 9-11) affect the transition response of grasslands by changing the cash flow position of farmers and their balances available for investment. Decreasing total cash balances in the cattle subsector by decreasing the

internal transfer of capital is of little consequence on the output of the criteria variables (Run 9). Decreasing consumption or living expenditures as in Run 11 leaves more capital for investment in the early years of the simulation, but its effect on the output of the model is nil. $\frac{1}{2}$ Since the modernization decisions are dominated by the profitability response and adoption rate parameters, an increased availability of capital for investment at the beginning of modernization does not have any effect on the model's performance. Later in the simulation period the consumption expenditures of the Base Run and Run 11 are equal as the coefficient determining the proportion of gross income that is consumed (C261 in Equation 9.12b) reaches its lower bound of .25 in both runs. Although the consequences of decreasing the lower limit to the proportion of income from cattle that is consumed (Run 10) is more conspicuous than changing the rate of response, its effect on the output of the criteria variables is not significant.

Despite the apparently low sensitivity of model output to changes in the consumption or living expenditures variable (EXLIV), this is an area that requires more careful consideration and refinement because of its welfare effects and its importance in making decisions at the farm level.

 $[\]frac{1}{Since}$ the exponential curve of Equation 9.12b is negatively sloped, any increase in C263 is reflected in a decrease in the variable ALPH2.

20 percent of the rice is grown in the uplands and average prices and yields have to be adjusted accordingly. Although income from crops is increased, this effect is of little consequence in the model.

The time lag constants tested in Run 16 appear to be quite sensitive. These parameters are used in the model in the exponential smoothing or averaging of prices, costs, and yields on which farmers base their expectations of future income streams. Reducing these time delays makes farmers respond more quickly to changing economic conditions and their decisions are then reflected in the model's output. Determining the length of smoothing time constants appears to be of utmost importance for plausible model behavior. In some instances during the validation and policy experiments we have been confronted with the paradox of farmers perceiving profitable conditions for modernization even when they are actually experiencing losses in the modern operation. This apparent noneconomic behavior points out the need for refining some formulations in the model in order to determine a more consistent response to situations of rising and declining profits. Changes in profits may lead to reorganization of the cattle enterprise by moving to different production functions.

The initial cattle and land parameters tested in Runs 17 and 18 appear to be quite sensitive. As might be expected, reducing the land base has a drastic effect on the animal population that can be supported in the region.

These results suggest the usefulness, for the purposes of this model, of obtaining accurate cattle inventories and land surveys.

Summary

In closing this chL_{P} er we can summarize some general observations and conclusions. Income from crops remains fairly stable throughout all runs with the notable exceptions of Runs 14 and 15. The major source of variation in Run 14 arises from changes in the relative profitability differential between cash crop and cattle production in the uplands (subregion 1). Through the land allocation mechanism of Equation 5.21 these changes have consequences on the total land in crops and consequently on their aggregated income. The policy implications of this effect will be discussed in the next chapter.

Cattle population in the region (TOPOPR) is, as expected, greatly influenced by the nutritional base. Nutritional conditions are affected, among other things, by the initial allocation of artificial pasture to grasslands and by the rate at which land modernization takes place, if any. The sensitivity tests suggest that the estimate of 20 percent of grasslands in artificial pastures provided by DANE in $1968^{1/2}$ is biased downward, and that the FAO^{2/2} estimate

 $[\]frac{1}{DANE}$, Encuesta Agropecuaria Nacional 1968, (Bogota, April 1970), p. 6.

^{2/}United Nations, Food and Agriculture Organization, <u>Livestock in Latin America: Status, Problems and Prospects-</u> <u>Colombia</u>, 1962, p. 14.

is more consistent with cattle censuses. A comparison of DANE estimates on cattle numbers from Table III.6 on page 237 with the output of Run 3 indicates that the latter is 12 and 22 percent lower than DANE's for 1965 and 1970, respectively, while deviations of the base run which uses the FAO estimate (45 percent) from DANE's data are +.08 and -6 percent, respectively. Herd size greatly influences off-take and market prices which, through interactive effects in the model, determine farm income from cattle (FARILA), government revenues (GOVREVA), value of land (VLAND), value of capital in cattle (VALCAP), and Colombian beef consumption per capita (PERCAP). Feedback effects of price changes caused by variations in supply and demand, and/or by changes in marketing margins determine farmers' decisions on farm management and investments which, in turn, affect cattle numbers. When domestic beef supply is increased in Run 5 by reducing illegal exports 50 percent, all long run performance variables depending on cattle decrease; after short run price declines, consumers become worse off as the lower price curtails cattle population and output and raises price.

Changes in costs of production as shown in Run 14 have a major impact on farmers' income, their capability to carry on land improvements, and their attitudes toward modernization that must be taken into account when deciding on policies to improve cattle production. Subjective expectations of farmers toward future prices, costs, and yields also have an important role in determining model output as shown in Run 16. In general, the capitalized value of land (VLAND) seems to be more sensitive than the other performance variables to changes in parameters value. This is, in part, explained by the response of VLAND to changes in annual income during the base year and not to its accumulated value. Accumulated income smoothes the annual oscillations of revenues from cattle and does not help in showing the cash flow position of farmers that affect their decisions on a yearly basis.

The preceding results show that the accuracy of the model will greatly be increased by improving the estimates on number of cows milked, proportion of land in artificial pastures, the marketing margin of animal sales, initial cattle inventories, land surveys, cattle traded at the border, and behavioral responses of farmers. In this chapter we have seen how sensitivity tests provide essential information for model building and validation by pointing out possible programming and modeling errors and by contributing to a better understanding of the model itself as well as of the system it is simulating. Sensitivity analyses may also be used to examine potential policies and to suggest priorities for data collection. The detailed analyses of the runs testing 45 selected parameters presented here illustrate these capabilities.

PART IV

MODEL APPLICATION IN DECISION MAKING

Introduction

The problems of planning for economic development are characterized by the uncertainty necessarily inherent in any process of planning for the future. This uncertainty arises both from the quantity and quality of available data and from the difficulties of forecasting how a large-scale system of complex interactive and feedback relationships will respond to policy inputs. With this in mind, the model developed in this dissertation, though restricted to a region and a subsector of the Colombian agricultural economy, sets up a workable and reliable alternative for dealing with these problems. The system approach used here, by modeling specific causal and structural relationships and by projecting time paths of behavior, provides at least some of the flexibility necessary to deal with the complexity and uncertainty of planning.

Chapter 12, in reporting and analyzing the results of a series of policy runs, illustrates how the model could be used in actual planning situations. Chapter 13 concludes the dissertation with a summary of results and a discussion of needed improvements and extensions of the current model.

CHAPTER 12

POLICY EXPERIMENTS ON THE NORTHERN COLOMBIA BEEF INDUSTRY

A system simulation model can be useful to policy makers in two principal ways: improving their understanding of the socioeconomic system they are concerned with, and formulating development policies. The model-building process and sensitivity tests, discussed in Chapter 11, can contribute substantially to an improved understanding of and sharpened intuitions regarding the development process, in general, as well as the particular socioeconomic system of concern.

The goals of development are part of a country's political process and, once established, policies must be defined for their attainment. Thus, the definition of policies is an important input to the policy-making process represented by the system simulation model. The computerized simulation model allows experimentation with alternative strategies of development under various assumptions, and then comparison of their likely outcomes. A set of plans or strategies are considered acceptable only if they are relatively effective in reaching the multiple goals under a wide variety of circumstances. During this planning exercise and through the interaction between

policy makers and researchers, goals may be set to more attainable levels or redefined toward the satisfaction of more urgent needs indicated by the simulation results. This chapter discusses a set of cattle development policies and presents an analysis of the results of a series of deterministic policy runs.

Policy Experimentation

The policies selected for this analysis are either currently in use or their implementation is being considered by the Colombian government as a means of attaining a diversity of objectives which are, more often than not, competitive. $\frac{1}{2}$ Examples of these objectives are: creation of incentives to increase production, measures to capture a larger surplus from cattle producers, cattle export incentives to increase foreign exchange earnings, and improved human nutrition through increased supplies. The policies tested include: disease control, taxes on land and cattle, credit for development, price and export targets, crops modernization, and cattle export incentives. With few exceptions, these issues have been unresolved controversial issues between the government and interested groups.

 $[\]frac{1}{The}$ analysis was made following very limited interaction with Colombian policy makers, though official policies were known through several published sources.

Although this dissertation is concerned with the Costa region, some of the policies tested have national implications since their implementation would not be restricted to this region. Furthermore, cattle production in the Costa weighs heavily in Colombian cattle output; therefore, any policy affecting production in this region will have an effect on the national cattle economy.

Run Definitions and Organization

Policy experiments were conducted with 21 simulation runs which cover the time period 1960-1985 (Table IV.1). The results analyzed here are for the period 1965-1985, with campaign promotion initiated in 1965 and policy implementation beginning in 1966 for cattle modernization, and in 1971 for disease control. There are two main reasons for including the period 1965-1973 in the policy analysis. First, this period, which is a part of the model's historical validation, includes the execution of both a cattle development plan as well as a disease control Second, we wanted to try a retrospective preprogram. diction of the likely consequences of starting the development plan in 1966 under a set of policies different from those prevailing at the time. Prospective predictions (forecasting) are considered after 1973 and projections are carried as far as 1985 to give the long-run diffusion response to the production campaign time to exert its major impact.

. 264

TABLE IV.1. Policy Simulation Runs.

Run No.	Run Sets	Run Definition
l	all	Standard Rundisease control; modernization of cattle production; normal land and cattle taxes; current credit terms; current export subsidy and fluctuating exchange rate; low export target and high world price; no modernization of crops or rest of national herd.
2	1	No disease control campaign for the traditional herd.
3	Ŀ	Increased government expenditures in foot-and-mouth disease control by 25 percent from year 1971.
4	\$ [,]	Taxes on cattle cutoff at years 1970 and 1972.
5	2	Property tax increased at year 1970.
6	3	Reduced interest rate on development credits from year 1966.
7	3	Increased terms of repayment on development credits from year 1966.
8	3.	Increased funding for development credit from 1966-1976.
9	3	Increased units of modernization promotion from 1965-1973.
10	4	Cattle exports set at intermediate target from year 1974.
11	· 4	Cattle exports set at high target from year 1974.
12	4,7	Run 11 with fixed exchange rate at Ps 25 and low world price, plus modernization of rest of national herd from year 1975.
1 <u></u> 3	5	Moderiis ion of cash and food crops from 1972- 1976.
·14	6	Combines Run 4, Run 6 and Run 7.
15	6	Combines Run 8 and Run 9.
16	6	Combines Run 14 and Run 15 plus growth of rest of national herd at 5 percent from year 1975 and high export target.
17	7	Run ll with low world price after year 1974.
ച8	7	Run ll with moderate world price after year 1974.

TABLE IV.1. (continued)

.

Run No.	Run Sets	Run Definition
19	7	Run 17 with export subsidy cutoff at year 1975.
20	7	Run 17 with fixed exchange rate at Ps 25.
21	7	Run 17 with export subsidy cutoff at year 1975 plus modernization of rest of national herd from year 1975.

With simulation, it is possible to increase the complexity of the combination of policies tested and at the same time keep the output format flexible enough to permit analysis of aggregated macroeconomic variables or the investigation of responses at a more micro level. Experiments can start with runs to evaluate single policies or programs (e.g., cutting off taxes on cattle) and then other policies and programs (e.g., reduced interest rate and increased terms of repayment) can be added succesively to investigate interactive effects. The 21 policy runs are organized to take advantage of these capabilities and they are grouped into seven sets (Table IV.1). All seven sets include Run 1, the base run, as a standard point of reference. The base run projects likely performance of alternative 2 under current policies and assumes that the non-Costa cattle population grows at 2.85 percent annually.

The output measures selected for these experiments provide information about relevant parameters at both the macro-and micro-economic levels. Annual variations in private and government revenues, capital formation and beef consumption resulting from the implementation of different policies can be examined in each set of runs. But the more simple evaluations of run sets 1 and 7 each use only the criteria variables (described below) considered most relevant for this analysis.
The performance variables of Runs 4 through 16 (sets 2, 3 and 4) include cattle population in the Costa (TOPOPR), Colombian beef consumption per capita (PERCAP), annual farm income from cattle (FARIL), average per hectare capitalized value of land (VLANDH), and annual government revenues from cattle (GOVREV). In this analysis, annual value of revenues and per unit value of land were considered more meaningful than the accumulated and more aggregated values used in the sensitivity tests. Runs 2 and 3 (set 1) use as performance variables cattle population in the Costa (TOPOPR) and extraction ratio from the Costa herd (ERR). Finally, runs 17 through 21 (set 7) use as performance variables domestic market price of finished males (PA) and export margin (EXMAR). The performance variables used in the sensitivity tests and the policy experiments are those considered most useful in the analysis of results in this dissertation, but they are only a small sample of the many output variables actually produced or that can be produced in the model. The following sections analyze and graphically depict the results of the 21 policy experiments.

Policies Related to Disease Control

Disease control, particularly for foot-and-mouth disease, have a major impact on the good health of cattle and the corresponding size and productivity of the Costa beef industry. In the first set of runs, Run 1 shows the results of a foot-and-mouth control program budgeted for

Ps 224 million over the period 1971-1985 (an annual treatment cost of Ps 4.50/animal is assumed). Run 2 explores the effect of maintaining precampaign disease control practices in the traditional herd. Run 3 investigates the results of increasing government expenditures on foot-and-mouth control by 25 percent up to Ps 280 million over the period 1971-1985. Although control of foot-and-mouth disease is highlighted here, improved practices include control of brucellosis as well as other diseases (black leg, anthrax and septicemia) and parasites.

The effect of upgrading disease control in the traditional herd (Run 1) is to increase cattle population and sales faster than under traditional control practices (Run 2). This effect is reflected in the different time path projections of Runs 1 and 2 as illustrated in Figures IV.1 and IV.2. But this is a short-term effect; if modernization proceeds steadily in the long run, output from Run 2 will approach that of Run 1 since the modern herd, treated with improved control practices, increases both absolutely and relatively. But this is not to say there should not be a disease control program. The advantages of starting this program early are many. On one account, eradicating brucellosis will increase not only the output of calves and milk in the traditional herd but in the modern herd as well since more females free from the disease are transferred to the latter.



and without a disease control program, 1965-1985.

- 1. Disease control, modernization of cattle production and continuation of present trends and policies (base run).
- 2. No disease control in the traditional herd.
- 3. Increased government expenditures in foot-andmouth control by 25 percent from year 1971.



Figure IV.2. Extraction ratio from the Costa cattle herd with and without a disease control program, 1965-1985.

In addition, the control of foot-and-mouth disease in the whole herd will have the double advantage of increasing cattle numbers and output, and easing sanitary barriers against Colombian beef exports in world markets.

Yet increasing government expenditures on footand-mouth control by 25 percent have a negligible impact on cattle population and sales as shown by Runs 1 and 3. Run 3 (with increased expenditures) shows a faster increase in population early in the campaign, but after 1975 this run assumes the same trend as Run 1 although population levels are slightly higher.

Increased government expenditures have the initial effect of accelerating the annual rate at which animals in the traditional herd are treated, but this slows down and almost equals that of Run 1 after the first five years of the campaign. Thus, its long-run effect on the traditional herd is small, and increases in population depend more on the rate at which grasslands are modernized and animals are transferred accordingly.

The projected cattle population is slightly higher in Run 3 because of differences in the long-term modernization responses of farmers which result from the varying expectations about future income streams. Such expectations are higher in Run 3 than in Run 1 throughout the planning horizon. This results in a higher rate of land modernization in Run 3 than in Run 1 which, in turn, affects the size of the modern herd. Policies Related to Cattle and Land Taxes

Taxes on cattle and land are investigated in the second set of runs, Runs 4 and 5. Since the cattle sector is considered hard to tax, with its contributions to governmental revenues greatly out of proportion to the sector's importance, a special scheme has been devised for the purpose of taxing the industry more effectively. These taxes (described in detail in Chapter 2) include a consumption tax, a selective inventory tax, and a general inventory tax. Although Law 26 of 1959 enacted the general inventory tax until 1970, Law 42 of 1971 made it effective through 1980. Run 1 assumes that further extensions will take place and this will continue in effect through the period of the simulation. Run 4 examines the effect of cutting off the general inventory tax in 1970 and the remaining special taxes after $1972.\frac{1}{}$

Run 5 compares the consequences of increasing the land or property tax from the current level of 4.2 mills to 14.2 mills. The property tax is the main source of municipal revenues, and it has remained at 4.2 mills since 1948. Although in recent years there has been a large number of proposals and laws to utilize the property tax, both as a measure to increase government

 $[\]frac{1}{\text{The } 1972}$ proposed legislation on presumptive taxation of agriculture intends to alleviate the tax burden of ranchers by eliminating the selective inventory tax.

revenues as well as to encourage improved land utilization, they were either never implemented or never put into effect [30]. In 1968, when the Musgrave Mission [11] presented its proposal for tax reform, the Colombian government again considered a change in the structure of the property tax, and in 1969 a surcharge of 10 mills (1 percent of the assessed value of land) was proposed for the purpose of financing public education. $\frac{1}{}$ Although this tax surcharge has not yet been levied on landowners, Run 5 attempts to evaluate the likely consequences of an early implementation by assuming that its collection started in 1970, one year after legislation was proposed.

Run 4 (cutting off special cattle taxes) results in values of all five performance variables below those of Runs 1 and 5. Cattle population, government revenue and per capita consumption in Run 4 remain below the other two runs for the period 1975-1985.

Maintaining the special cattle taxes in Run 5 while increasing by 3.4 times the tax rate on the assessed value of land results in a 29 percent increase in government revenue over the base run by 1985. But when the land tax remains at its current rate of 4.2 mills in Run 4 while the special cattle taxes are cut off after 1970, government revenue falls 80 percent below the base run in 1985 (Figure IV.7).

<u>l</u>/Rodrigo Llorente, <u>El Desafio de un Pueblo en</u> <u>Desarrollo</u>, (Bogota: Editorial Andes, 1972), p. 273

Although the tax revenues examined here need to be considered with caution due to probable estimation inaccuracies, a comparison of annual government revenues attained with Runs 1, 4 and 5 suggests that taxing cattle inventories and sales is a major regional source of government income from the cattle subsector (not considering income and net worth taxes). We can conclude that eliminating the special taxes on cattle would likely produce a big loss in government revenue unless this loss is compensated by other fiscal policies designed to tax more effectively the income, net worth and property of ranchers.

There are two unrealistic assumptions in the model affecting the output of all runs. First, because of inadequate information and accounting difficulties in computing the general inventory tax as a proportion of the net investment on cattle (Law 26, 1959), this is computed at a constant Ps 3.1 per head throughout the simulation. Thus, revenues from this tax change with the cattle population and not with the net worth of farmers. Second, the property tax is computed from the assessed value of land which is set at 50 percent of the capitalized value of land. Yet the use of the capitalized value of land as a tax base introduces a destabilizing effect on government revenues since this value changes with net returns and the rate of interest resulting in an unlikely automatic adjusting mechanism in the value of land and the related value of the property tax.

Total cattle population and per capita consumption follow the same rank of government revenues, i.e., Run 5 at the top and Run 4 at the bottom (Figures IV.3 and IV.4). Per capita consumption is used here as a measure of the effect of increasing cattle output from the Costa herd on the availability of beef to the Colombian population. We estimate that domestic consumption equals the residual of the total supply of beef after accounting for official and illegal exports; therefore, the program producing the largest cattle population and off-take will result in the highest supply to the domestic market. The cyclical pattern of consumption per capita is produced by the long-term (7 years) cycle of beef supply from the rest of the country which is forced into the regional model. Lower prices arising from increased supplies have the effect of improving the consumer's situation and worsening the producer's. Consumers are better off in the short run, but they may be worse off in the long run, since low prices reduce incentives to increase output through modernization and may eventually cause a reversion of land to traditional management which would curtail output more drastically. Such welfare effects must be faced when policy makers consider alternative strategies for development.

Annual cattle populations for the various runs depend on their projected composition as traditional and modern. As stated, differences in herd size are produced by varying rates of land modernization caused by differences in the farmers' long-term modernization responses. While it is

true that cutting off the special taxes increases income and leaves the farmer with more capital available for investment in farm improvements, (and possibly a higher rate of modernization) the actual rate is smaller. This paradox can be explained by examining the dynamic nature of the farmers' decision-making process. We have conceptualized that the rate of land modernization cannot be higher than the rate at which land enters modernization due to promotion and diffusion, and that this rate depends upon the farmers' perception of the relative profitability of the modern operation and the response rate parameters.

Since eliminating the special taxes makes profits proportionally higher in the traditional than in the modern herd, the perceived relative profitability of the modern operation throughout the planning horizon declines, and as a result the rate of land modernization in Run 4 is lower. This effect has its greatest impact after 1975. On the other hand, increasing the property tax reverses the situation described above, causing the perceived relative profitability of the modern operation and the resulting rate of land modernization to be higher in Run 5. Although these results apparently give support to those advocating increased land taxes in order to encourage an improved land utilization, no definite conclusions can be drawn at this point.

Aggregate farm income from cattle $\frac{1}{2}$ sharply increases in all cases from 1965 to 1975 showing the effect of modernization and rising prices. It then decreases rapidly (Figure IV.5). From 1966 to 1980 the time paths of the three runs are parallel with Run 4 at the top and Run 5 at the bottom as expected. But after 1980, Runs 1 and 5 reverse the trend while that of Run 4 continues downwards. The behavior of this performance variable is responsive to the interaction of several other variables in the model. As stated, the sharp increase in income from 1965 to 1975 coincides with a period of rising beef prices and increasing sales from a larger herd. But the long-run effect of increased supply slows the rate of increase in price from 1975 to 1980, and this, coupled with rising costs and the debt burden after 1976 (when payments of development credit cease) causes a decline in income from 1975 to 1980. After 1980, increases in total demand more than offset the initial impact of increased supplies and prices regain the upward trend of the period 1970-1975. Rapid rising prices and decreasing indebtness from development credits cause income to become an upward trend after 1980 in Runs 1 and 5. Furthermore, given the structure of the model, land modernization stops altogether

 $[\]frac{1}{Farm}$ income includes both sale of traditional and modern animals and milk as well as credits paid, but is net of operating costs, debt service and interests.

between 1975 and 1980 in these runs, eliminating all costs on land improvement. This happens in the model when costs and living expenditures exceed the value of sales and transfers from the crop sub-sector, cutting off the availability of capital for investment. In this case, commercial credit makes up the difference to cover operating costs and/or living expenditures.

Although Run 4 experiences the same trends in price and indebtness, it takes longer to slow down and reverse its downtrend. Since higher incomes leave the farmers with more capital for investment, more private resources are committed to land improvement, and modernization proceeds for a longer period in this run, though at a lower rate, before it ceases. This results in lower acreage in modern land, but extended establishment costs throughout the period 1980-1985. This behavior may suggest that, after 1985 when modernization stops, the downtrend of Run 4 could be tapered off or even reversed. Only by making projections after 1985 will this behavior be known with certainty.

Finally, the per hectare capitalized value of land follows very closely the pattern of farm income (Figure IV.6). Although the value of a hectare in modern grazing (based on its amortized annual average returns) is much higher (about double) than a hectare of traditional grazing, the relative proportion of the former in the total grazing land is so small during the simulation that changes in the composition of acreage in grazing have less effect on the capitalized land value of

- 1. Disease control, modernization of cattle production, current land and cattle taxes and continua-4. Taxes on cattle cut off at years 1970 and 1972.
 5. Property tax increased to 14.2 mills at year 1970.



- 1. Disease control, modernization of cattle production, current land and cattle taxes and continuation of present trends and policies (base run).
- 4. Taxes on cattle cut off at years 1970 and 1972.
- 5. Property tax increased to 14.2 mills at year 1970.



Figure IV.4. Beef consumption per capita of the Colombian population under various taxing policies, 1965-1985.

- 1. Disease control, modernization of cattle production, current land and cattle taxes and continuation of present trends and policies (base run).
- 4. Taxes on cattle cut off at years 1970 and 1972. 5. Property tax increased to 14.2 mills at year 1970.



Figure IV.5. Aggregated farm income from cattle production in the Costa under various taxing policies, 1965-1985.

- Disease control modernization of cattle production, current land and cattle taxes and continuation of present trends and policies (base run).
- 4. Taxes on cattle cut off at years 1970 and 1972.
- 5. Property tax increased to 14.2 mills at year 1970.



Figure IV.6. Average per hectare capitalized value of grazing land in the Costa under various taxing policies, 1965-1985.

- Disease control, modernization of cattle production, current land and cattle taxes and continuation of present trends and policies (base run).
- 4. Taxes on cattle cut off at years 1970 and 1972.
- 5. Property tax increased to 14.2 mills at year 1970.



Figure IV.7. Annual government revenues from cattle production in the Costa under various taxing policies, 1965-1985.

the region. Yet as modernization proceeds and the area in modern land increases (both absolutely and relatively), its impact on the average value of land will be greater. While it is true that the special annual land tax burden will be amortized in the form of lower land prices, thus reducing the net worth of farmers, its long-run allocative effect could result in increased land values and government revenues as more traditional farmers are encouraged to improve their lands and herds.

In general, increasing the land tax greatly increases government revenues without impairing farmers' income and wealth to a great extent. Probably the best strategy would be one of combining increased land tax and eliminating some of the special taxes on cattle. These results suggest that, when proper policies are implemented, more surplus could be taken from cattle producers in the form of higher taxes without greatly curtailing farm capital formation and output.

Policies Related to Promotion and Development Credit

Policies and programs related to promotion and credit for cattle modernization are examined in simulation Runs 6, 7, 8 and 9. Current arrangements for development credit assumed in Run 1 are those set by the Caja Agraria project: interest rate of 14 percent, a four-year grace period, eight years for repayment, and borrowing ranchers participating with 20 percent of the estimated cost. The total budget for the credit program is assumed to be Ps 750 million spread over a ten-year period (see Figure II.8). And precampaign promotion is assumed to be 3,000 units of promotion spread over an eight-year period (see Figure II.8). $\frac{1}{}$ Run 6 compares the effect of reducing the interest rate to 10 percent during the life of the program. Run 7 explores the effect of increasing the grace period to five years and the repayment period to ten years.

The remaining runs of this set examine policies of a more macroeconomic nature. Run 8 explores the effect of increasing the allocation of credit funds up to Ps 1,125 million over the same ten-year period. And finally, Run 9 investigates the effect of increasing the units of campaign promotion to 4,500 man-year equivalents over the same eightyear period.

The most striking observation that can be made about this set of runs is that Run 8 (increased funding for development credit) produces results practically identical with those for the base run, and that results from Runs 7 and 9 (improved terms for repayment and increased units of promotion, respectively) are very close. Results from Run 6 (reduced interest rate) are consistently higher than for the other runs with the exception of the period 1965-1971 for farm income and value of land and the last three years of the run for farm income only (Figures IV. 8, 9, 10, 11 and 12).

 $\frac{1}{0}$ One unit is equivalent to one man-year with annual cost of Ps 20,000.

The almost identical results (differences are so small they can't be detected in the graph) for Runs 1 and 8 indicate that allocating 50 percent more funds for development credit over a period of ten years has practically no effect on the output of the model. While it is true that farmers use more credit when it is available and that more of their cash balances are used to complement credit payments, the total area in modern land is not affected. It is the same in both runs. Runs 1 and 8 generate the same rate of land modernization which means that, given the present model's structure, more available credit does not affect farmers' decisions. This is to say that more available credit does not change farmers' perception of relative profitability nor their threshold and response rate parameters. Recalling the modernization mechanism in the model, we see that these parameters are used to simulate a wide variety of responses to environmental situations (social, political and Therefore, we can see that, unless the environmental economic). conditions are such as to make new investments attractive, increasing government funds for modernization do not have major impacts on development.

As stated above, the projected time paths of Runs 7 and 9 are very close and produce virtually the same results. Initially, the rate of modernization is dominated by the relatively higher net returns expected throughout the planning horizon when the debt burden is spread over a longer period, but, after a few years, the promotion effort builds and its

effect outweighs that of the relative profitabilities to produce a higher rate of modernization in Run 9 than in Run 7. Yet after 1973 when promotion ceases, the modernization decisions are based exclusively on the relative differences of the net present values and the other behavioral parameters, and as a result more land is modernized in Run 7 than in Run 9.

As discussed in the previous section, the projected time paths of the five performance variables respond to the interactive effects of changes in use of modern land. This interactive effect is more easily seen in the changing order of cattle numbers in Runs 7 and 9 which follow the varying rates of land modernization discussed above (Figure IV.8). Farm income behavior is more complicated since there are more factors involved. From 1965 to 1973 it is higher in Run 7 than in Run 9 as a result of the extended grace period on the development credit and less expenditures on land modernization by the end of the period. Then follows a period when increased modernization demands more expenditures in Run 7 than in Run 9 and aggregate income in the former drops below the latter (Figure IV.10). The increasing and decreasing trends in this variable have been discussed. The capitalized value of land is a reflection of farm income although, in the final period of the run, larger values of traditional land increase average assessed land value in Run 7. This is the effect of the capitalized higher price of cattle in Run 7 resulting from lower supplies as shown in Figure IV.9. Finally, government

revenues follow from cattle inventories, sales and assessed land values (Figures IV.11 and 12).

When outputs of Runs 1, 6, 7, 8 and 9 are compared, reducing the interest rate on development credits in Run 6 shows the greatest impact on all variables. Cattle population, consumption per capita and government revenues are consistently higher in Run 6 than in the other runs. Farm income, however, is the lowest in Run 6 from 1966 to 1971, and then is the highest until approximately 1983 when it is below Runs 7 and 9 but above the base run (Figure IV.10).

Reducing the interest rate greatly increases the net present value of the expected income stream from modern relative to traditional cattle, and the resulting higher profitability accelerates modernization in Run 6 above the other runs. The initial effect of this enhanced rate of modernization is to decrease farm income relative to the other runs as more resources are committed to farm improvements and more cattle are retained in order to build up the herd. Although total cattle off-take is reduced, the supply of beef is increased as a larger proportion of heavier animals from the modern herd are slaughtered. Yet this enlarged beef supply, in turn, lowers the price of cattle and reinforces the decline in income. At this juncture, it should be remembered that all cattle prices in the model are related to the market price of finished males, disregarding conditions of demand for different animal categories. For

a more precise accounting of cattle prices and income this part of the model needs more refinement.

After this initial impact, increased sales in Run 6 outweigh the decline in price and as a result income becomes larger until 1983 when expenditures in modernization continue at a faster rate than in the other runs, and income becomes lower. The capitalized value of land reflects the varying pattern of farm income while government revenues follow in relation to cattle inventories, sales and assessed land values. The average value of land at the end of the run is greatly influenced by a higher proportion of modern land in the total grazing land which is 34 percent in Run 6 compared with 15 percent in Runs 1 and 7, and 22 percent in Run 9 (Figure IV.11).

Policies Related to Domestic Supply

Pricing policies are investigated in Runs 10, 11 and 12. Retail prices have been under control at various times but have never been efficiently enforced. Therefore, coercive controls are ruled out here and the only price adjusting mechanism allowed is through control of domestic supply via the export market and/or increased off-take from the national herd. Cattle exports will come mostly from the Costa herd but excess supplies from other producing regions, if ever attained, will also be exported. Exports are exogenously determined according to targets set by the government. Low target projections are included in all runs between 1972 and 1974 (see Table I.1, p.19) but thereafter three different



Figure IV.8. Cattle population in the Costa under various promotion and credit policies, 1965-1985.

- 1. Disease control, modernization of cattle production, current credit terms and continuation of present trends and policies (base run).
- 6. Reduced interest rate on development credits from year 1966. 7. Increased terms of repayment on development
- credits from year 1966.
- 8. Increased funding for development credit from 1966 to 1976.
- 9. Increased units of modernization promotion from 1965 to 1973.



Figure 1V.9. Beef consumption per capita of the Colombian population under various pro-motion and credit policies, 1965-1985.

- 1. Disease control, modernization of cattle production, current credit terms and continuation of present trends and policies (base run).
- 6. Reduced interest rate on development credits from year 1966. 7. Increased terms of repayment on development
- credits from year 1966.
- 8. Increased funding for development credit from 1966 to 1976.
- 9. Increased units of modernization promotion from 1965 to 1973.



Figure IV.10. Aggregated farm income from cattle production in the Costa under various promotion and credit policies, 1965-1985.



Figure IV.11. Average per hectare capitalized value of grazing land in the Costa under various promotion and credit policies, 1965-1985.

- 1. Disease control, modernization of cattle production, current credit terms and continuation of present trends and policies (base run).
- 6. Reduced interest rate on development credits from year 1966.
- 7. Increased terms of repayment on development credits from year 1966.
- 8. Increased funding for development credit from 1966 to 1976.





Figure IV.12. Annual government revenues from cattle production in the Costa under various promotion and credit policies, 1965-1985.

targets are tested in the model. The base run continues projecting exports at a low level (329 thousand head annually), Run 10 makes projections at an intermediate level (496 thousand head annually), and Run 11 at a high level (692 thousand head annually) [35]. Run 12 compares the effect of attaining a high export target at the same time that the cattle herd in the other four producing regions is growing at 5 percent annually. At the time the simulation runs were completed, details of the new cattle development plan were released setting a revised export target at one million head annually by 1990, but due to time constraints this was not included in the study. $\frac{1}{}$ Illegal exports are kept constant in all runs at 300 thousand head annually.

The effect of setting up low, intermediate and high export targets accompanied or not by programs to develop the cattle herd in the rest of the country results in striking differences in outputs. In Figure IV.14 we see that the major contributor to high consumption per capita is the development of the cattle herd in the rest of the country beginning in 1975, even after sustaining the export of 692 thousand head annually. The lowest consumption per capita is attained in Run 11 with high export target and "normal" growth of the non-Costa herd.

 $\frac{1}{E1}$ Espectador, (Bogota), July 8, 1973.

Total (national) cattle population is the highest in Run 12 as expected, despite a substantial decline in the Costa inventories after 1975 (Figure IV.13). Yet, the non-Costa cattle projections for this run reveal unrealistic internal inconsistencies. In one account, the same conditions that discourage cattle production in the Costa might exist in the other producing regions, resulting in a lower or even zero rate of mcdernization. Likewise, the 5 percent annual rate of growth assumed in the government plan seems to be unrealistically high since the growth of the Costa herd in the base run has been about 1.6 percent for the period 1960-1985, and about 2.2 percent annually during 1960-1975, the period of the fastest growth.

The rising trend in the Costa cattle population of Run 12 (Figure IV.13) tapers off after 1975 and turns down after 1980 ending the projection with 8 percent fewer cattle than the other runs. The substantial increase in beef supply after 1975 causes a sharp decline in cattle prices (price of finished males are 45 percent lower in Run 12 than in Run 1 by 1985) that first reduces the comparative advantage of the modern operation and later turns it into a loss. At this point land and cattle are transferred back to traditional practices, and the nutritional imbalance caused by these shifts fosters sales that further depress prices. We must recall that the modernization mechanism in the model will return land to traditional management when the perceived relative profitability (net present values) becomes negative

indicating that the discounted returns at the base year for traditional cattle are higher than those of modern throughout the planning horizon. Therefore, the declining profitabilities discussed above, through their interactive effects in the model, result in reduced and early cessation of modernization, reversion of land and animals from modern to traditional, and firally in lower cattle inventories.

Despite larger incomes from higher cattle prices in Runs 10 and 11 than in Run 1, the projected rates of land modernization are very close and, as a result, the Costa cattle population is virtually the same in these runs. The reason for this is twofold: (1) discounted net returns in each base year from traditional and modern cattle experience about equal proportional increases for higher prices apply to sales from both operations. Consequently, the perceived relative profitabilities that are part of the modernization's decision mechanism remain very close in their time paths, and (2) the final rate of modernization is dominated by the behavioral responses of farmers despite the greater availability of investment capital arising from higher incomes, especially in Run 11. This effect is the same as that of having more credit available for modernization. Yet we can speculate that projections after 1985 might show a higher population in Run 11 as modernization expands because of the increased liquidity of farmers at a time when that of the other runs has been reduced. It is also worth noting that the higher average discounted returns from cattle in Runs 10 and 11 also have an effect on land use, reducing the area in crops in

the region. This estimated effect, through the land allocative mechanism in the model, and its possible inaccuracies will be discussed in the next section on crop modernization.

The most dramatic effects of regulating prices through domestic supplies are seen in the other three performance variables. Aggregated farm income from cattle (Figure IV.15) is the highest when the high export target is attained, followed in order by Runs 10, 1 and 12. Farm income in Run 12 not only falls sharply after 1975 when increased cattle outputs from non-Costa regions are marketed, but also becomes negative during the run's last five-year period. As income drastically drops, farmers have to resort to short-term credit to cover their operating costs and living expenditures. As short-term indebtedness increases, debt service expenditures further reduce farmers' liquidity. Although under these circumstances farmers reorganize their business, returning land and cattle from modern to traditional management, the model does not provide adequately for reorganizing production by reducing expenditures or, more drastically, going out of business. Contrarily, the upper bound on sales and the dominance of nutritional factors built into the sales mechanism preclude a liquidation of the herd and force farmers to stay in business even at a loss. This behavior points out the

need to refine the model so as to better handle this investment and disinvestment. $\frac{1}{2}$

The capitalized value of land (Figure IV.16) follows the projected time paths of farm income. However, Run 11 shows the effect of a higher value of traditional land on the weighted average of land in the region. The value of land in Run 12 decreases sharply after 1975 until it reaches the salvage value of one peso assumed in the model when income becomes negative.

Government revenues (Figure IV.17) are again the highest in Run 11 and the lowest in Run 12. Higher government revenues in Runs 10 and 11 than in the base run arise from higher assessed land values since cattle inventories and sales remain about the same. The loss in revenues in Run 12 is produced by reduced cattle inventories and mostly by lower assessed land values.

Policies Related to Crop Modernization

Run 13 attempts to highlight the interactive effects of increasing the production of cash and food crops via extension efforts to introduce new seed varieties and improved cultural practices, improving average cash crop yields in lowlands and uplands to 2,900 and 1,650 kg./ha., respectively, and average food crop yields to 10,000 kg./ha. This run

 $[\]frac{1}{1}$ In turn, this need indicates a disciplinary need for economists to develop a user cost theory as a basis for improved modeling of investment and disinvestment.



.Figure IV.13. Cattle population in the Costa and the rest of Colombia under various domestic supply policies, 1965-1985.



- 10. Cattle exports set at intermediate target from year 1974.
- Cattle exports set at high target from year 1974.
 Cattle exports set at high target from year 1974 plus modernization of rest of national herd from year 1975.



Figure IV.14. Beef consumption per capita of the Colombian population under various domestic supply policies, 1965-1985.

- 1. Disease control, modernization of cattle production, low export target and continuation of present trends and policies (base run).
- Cattle exports set at intermediate target from year 1974.
- 11. Cattle exports set at high target from year 1974.
- 12. Cattle exports set at high target from year 1974 plus modernization of rest of national herd from year 1975.



igure IV.15. Aggregated farm income from cattle production in the Costa under various domestic supply policies, 1965-1985.
- Disease control, modernization of cattle production, low export target and continuation of present trends and policies (base run).
- Cattle exports set at intermediate target from year 1974.
- 11. Cattle exports set at high target from year 1974. 12. Cattle exports set at high target from year 1974
 - plus modernization of rest of national herd from year 1975.



Figure IV.16. Average per hectare capitalized value of grazing land in the Costa under various domestic supply policies, 1965-1985.

- Disease control, modernization of cattle production, low export target and continuation of present trends and policies (base run).
- Cattle exports set at intermediate target from year 1974.
- Cattle exports set at high target from year 1974.
- Cattle exports set at high target from year 1974 plus modernization of rest of national herd from year 1975.



Figure IV.17. Annual government revenues from cattle production in the Costa under various domestic supply policies, 1965-1985.

assumes that yield targets are attained at the end of five years with the crop modernization effort starting in 1972 while cattle improvement programs are still in effect.

Run 13 examines the effect of a program to modernize cash and food crop production on the performance of the **cattle** industry. In general, the modernization of crops has an adverse effect on all performance variables but cattle population which shows a slightly higher trend in Run 13 than in the base run between 1980 and 1985, although for practical purposes the results can be considered the same (Figure IV.18). When cash crops are modernized, their profitability increases relative to that of cattle and more land is transferred from cattle production to crop production. As a result, total grazing land declines over time. But the land allocation and modernization mechanisms in the model interact to step up the proportion of modern land in total grazing land. The effect of the land allocative mechanism is to take out land from the least profitable cattle activity--the traditional in this case--and transfer it to the most profitable cash crop activity. Meanwhile, the modernization decision mechanism continues to shift land from traditional to modern cattle management. If the latter effect outweighs that of the land allocation, as in Run 13, this results in a higher proportion of modern pasture land relative to total pasture land that compensates any loss of grazing area. Yet it is likely that, in the long run, differences in cattle population will widen as more grazing land is

modernized in Run 13 after 1985 at a time when modernization has ceased in the base run. The reason for this is the greater availability of capital for investment brought about by larger revenue transfers from crops to cattle production as total income in the crops subsector increases. We should remember that the model assumes a 10 percent $\frac{1}{}$ internal transfer of net revenues from crops to cattle; and that, for the period 1972-1985, net crop income is greatly increased, being 190 percent higher in Run 13 than in the base run in 1985. While this effect has its major impact in the absence of development credit and when the liquidity of farmers is low the projected rates of land modernization are very close in the two runs, despite larger internal capital transfers in Run 13. As discussed, this result indicates that the final rate of modernization is dominated by similarities between the way farmers' behavioral responses are modeled in both runs.

Although in Run 13 there are conditions for a larger and faster shift of land from cattle production to crop production, the assumptions underlying the land allocation mechanism prevent profit incentives from exerting a greater impact on land use. It is assumed that cash crops in the lowlands expand at a constant 4,500 hectares annually throughout the simulation; this seems to be an unrealistic assumption since it is likely this rate would increase as the profitability

 $[\]frac{1}{The}$ accuracy of this estimate is highly uncertain. The rational for using a low proportion of capital transfer at this point is the current low level of reinvestment in cattle raising.

of crops relative to cattle becomes greater. In the uplands (subregion 1), however, we have reasoned that farmers' decisions to expand (or contract) their cash crop acreages are based on their perceived differences in net present value of future income streams from land in different uses, and on a response rate parameter. In the land allocative mechanism, we see that the effects of revenue differences between the allowed land uses are reduced by the response rate parameter which is assumed constant in the projected time paths for the two runs. This also seems to be an unrealistic assumption since it is likely that farmers will accelerate the rate of land transfer from pasture to crops as their perceived relative profitability becomes larger. As a result, the impact of relatively greater returns from crops on land allocation is minimized and the final estimate of land in crops is biased downwards. However, this transfer could be tapered off or even reversed if the average returns from cattle approximate or outweigh those from crops.

Although cattle population is slightly higher in Run 13 than in Run 1, the resulting per capita consumption is lower during the last five years of the run (Figure IV.19). This apparent paradox is explained by a higher pasture-toanimal ratio toward the end of Run 13 that results in a lower extraction ratio. In the sales mechanism of the model, the short-term response in off-take is dominated by nutritional conditions. Yet it is likely that the off-take in Run 13 will increase in the long run when modernization is completed

and the cattle herd is in equilibrium with the forage available.

As expected, aggregate farm income from cattle and the resultant capitalized value of land follow the same pattern. Income from cattle (Figure IV.20) follows almost the same time path in the two runs until 1980 indicating, again, the behavior and interactions discussed in previous sections. But after 1980, the two runs diverge widely and, while Run 1 reverses its downtrend, Run 13 continues to fall sharply. The reason for this is found in the greater availability of capital for investment brought about by increased revenue transfers from crop to cattle production--which enables farmers to continue investing in farm improvements for longer periods after the credit program has been cut off. This downtrend could be tapered off or even reversed if modernization slows down or stops altogether either for lack of incentives, lack of outlay balances for modernization, or lack of traditional land.

The capitalized value of land per hectare is a reflection of what happens to farm income (Figure IV.21). However, it is worth noting that in Run 13, after 1980 the decrease in land values is less than that of farm income. This is explained by the increased weight of modern land in the average capitalized value of land in the region resulting from the expansion of modern pasture and the contraction of traditional one. Finally, government revenues are lower in Run 13 after 1980 as a result of lower tax collections from cattle sales and assessed land values (Figure IV.22).

- 1. Disease control, modernization of cattle production, no modernization of crops and continuation of present trends and policies (base run).
- 13. Modernization of cash and food crops from 1972-1976.



Figure IV.18. Cattle population in the Costa with and without a crop modernization program, 1965-1985.

- 1. Disease control, modernization of cattle production, no modernization of crops and continuation of present trends and policies (base run).
- 13. Modernization of cash and food crops from 1972-1976.



Figure IV.19. Beef consumption per capita of the Colombian population with and without a crop modernization program 1965-1985.

- Disease control, modernization of cattle production, no modernization of crops and continuation of present trends and policies (base run).
- 13. Modernization of cash and food crops from 1972-1976.



Figure IV.20. Aggregated farm income from cattle production in the Costa with and without a crop modernization program, 1965-1985.

- 1. Disease control, modernization of cattle production, no modernization of crops and continuation of present trends and policies (base run).
- 13. Modernization of cash and food crops from 1972-1976.



without a crop modernization program, 1965-1985.



Figure IV.22. Annual government revenues from cattle production in the Costa with and without a crop modernization program, 1965-1985.

Effect of Various Policy Combinations

The set of runs which investigates the consequences of various combinations of policies and programs includes Runs 1, 14, 15 and 16 as defined in Table IV.1. Briefly, Run 1 projects present trends and policies (the base run); Run 14 implements the taxing policies of Run 4 with the credit policies of Runs 6 and 7; Run 15 investigates the effects of increasing the funds for development credit coupled with a 50 percent increase in the promotion effort; and Run 16 implements the programs of Runs 14 and 15 concomitantly with the development of the non-Costa cattle herd and the increase of beef exports from a low to a high target.

In general, the more micro-economic oriented policies of Run 14 have the greatest long-run impact on the performance variables in the model, followed by those of a more macroeconomic nature (Run 15). The combination of these policies in Run 16, coupled with a program of expanding exports and growth of the cattle herd in the rest of the country, produce varying results that go from a steady increase in consumption per capita to a gradual decline in cattle population in the Costa and a sharp decline in farm income, land value and government revenues.

By 1985, total (national) cattle population is 42 million head in Run 16, about 50 and 40 percent higher than in the base run and Run 14, respectively. The cattle population in the Costa in Run 16 at first experiences a faster rate of growth than in the other runs, but this tapers off

after 1975 when the development of the cattle industry in the rest of Colombia begins, and in the long run the growth rate will likely turn down (Figure IV.23). Although the negative effects of increased beef supplies from the rest of Colombia on land modernization and cattle population in the Costa have been discussed earlier, further discussion is necessary at this point to better understand the changes in cattle inventories in Run 16. Between 1975 and 1985, cattle population continues to grow at a decreasing rate although no more traditional land enters the process of modernization and conditions exist for a reversion of land from modern to traditional management. The reason for this is found in farmers' delayed response to falling prices and income which is built into the model. Since the perceived relative profitabilities are based on exponentially averaged costs and prices of the preceding five years, it takes several years before the decline in price has its full effect on expected returns. Increasing prices before the turning point in 1975 (see Figure IV.28) have the effect of increasing or maintaining averages in the following base years. Since the perceived relative profitability in 1985 continues to be positive and higher than the threshold parameter, it prevents the operation of the land reversion mechanism although modernization of new land is stopped by the other constraints imposed on the modernization decision mechanism. Nevertheless, the land in process of modernization from previous years (the land "stored" in the modernization delay) completes its development and enters full modern production a few years after modernization of new traditional land has been stopped; as a result, the area in modern land increases beyond 1975 and with it the more productive modern herd. However, it is to be expected that in the long run this process would reverse as discussed earlier in the section on domestic supply policies in connection with Run 12. It is highly unlikely that farmers would wait ten years before changing their expectations from profit to loss and start reorganizing production. This weakness of the model should be considered more carefully and corrected.

Run 14, which combines the incentives provided to modernization only, shows a steady uptrend in cattle population as a result of a higher and sustained rate of land modernization (Figure IV.23). Run 15, which combines the credit funding and promotion policies of huns 8 and 9, results in virtually the same cattle population of Run 9 alone (about 9.8 million head in 1985) indicating that adding increased availability of development credit to the promotion effort does not affect the output of the model. This result was explained in the section on promotion and development credit policies.

Consumption per capita is the highest when the cattle herd in the rest of Colombia is developed, even after sustaining a high level of exports. When the cattle development program is limited to only the Costa region, Run 14 yields the highest domestic supplies (Figure IV.24).

As depicted in Figure IV.25, aggregated farm income from cattle is greatly influenced by export policies and development programs which affect cattle prices and costs as well as expenditure commitments for farm improvement. Initially, farm income is inversely related to the rate of land modernization with Run 1 at the top followed by Runs 15. 14 and 16 respectively. But between 1970 and 1975, this order is completely reversed by the effect of the various policies followed. Run 16 shows the effect of increased prices because of high exports coupled with increased sales and cost reductions arising from the development programs. When the export target is low and domestic price drops as in Runs 1, 14 and 15, the production incentives provided in Run 14 result in the highest farm income. The changing trend in farm income in Runs 1, 14 and 15 after 1975 is explained by different levels of expenditures on land modernization at the end of the credit for development program. Farm income in Run 16 sharply decreases after 1975 and becomes negative after 1980 when increased supplies from the rest of the country greatly depress prices -- more than offsetting the positive effect of the other policies implemented in this run.

In general, the capitalized value of land (Figure IV.26) follows the projected time paths of farm income. In the initial period 1965-1970, the order differs from that of farm income because of varying proportions of traditional and modern land in the total grazing land that affect the average value of land in the region. Although farm income

in 1985 is lower in Run 14 than in Runs 1 and 15, the capitalized value of land is higher because of the larger value of traditional land which has the greatest weight in the average value of land. The value of land in Run 16 decreases sharply after 1975 until in 1985 it reaches the salvage value of one peso per hectare assumed in the model when income becomes negative.

Government revenues (Figure IV.27) are the highest when development is promoted while maintaining the special taxes on cattle (Run 15). When the special taxes are cut off, government revenues are sharply reduced even though the other policies implemented in Run 14 substantially increase revenues from the property tax. But government income is the lowest in Run 16 because of the lost revenues from cattle taxes and very low assessed land values.

Policies Related to Export Promotion

Policies aimed at the promotion of beef exports are examined in the last set of runs, Runs 12, 17, 18, 19, 20 and 21. Although the present structure of the model does not provide a feedback linkage between the export sector and the domestic market except for the simple effect of increasing the total number of animals demanded, the output of the study is appropriate for exploring the likely outcome of some world market conditions and/or domestic policies on the competitiveness of the Colombian cattle industry.

The export promotion policies now in operation were introduced in 1967 under Decree-Law 444, and those examined

- 1. Disease control, modernization of cattle production and continuation of present trends and policies (base run).
- Combines Run 4, Run 6 and Run 7.
 Combines Run 8 and Run 9.
- 16. Combines Run 14 and Run 15 plus modernization of rest of national herd from year 1975 and high export target.



Figure IV.23. Cattle population in the Costa under various policy conditions, 1965-1985.

- 1. Disease control, modernization of cattle production and continuation of present trends and policies (base run). 14. Combines Run 4, Run 6 and Run 7. 15. Combines Run 8 and Run 9.

- 16. Combines Run 14 and Run 15 plus modernization of rest of national herd from year 1975 and high export target.



Figure IV.24. Beef consumption per capita of the Colombian population under various policy conditions, 1965-1985.

<u>Run Definitions</u>



Figure IV.25. Aggregated farm income from cattle production in the Costa under various policy conditions, 1965-1985.



Figure IV.26. Average per hectare capitalized value of grazing land in the Costa under various policy conditions, 1965-1985.



Figure IV.27. Annual government revenues from cattle production in the Costa under various policy conditions, 1965-1985.

here at the tax credit certificate (CAT) discussed in Chapter 2, and a fluctuating exchange rate. The present 15 percent rate of return accruing to exporters through the CAT scheme is subject to annual adjustments depending on the competitive position of Colombian exports in foreign markets.

Although sales of live cattle have dominated Colombian exports, it is assumed that after 1974 only dressed animals will be exported in the form of frozen carcasses. Therefore, the three levels of world price considered here are for frozen beef. The high price assumes that after 1974 the world price will continue the trend for the period 1970-1972, one of rapid rising prices (during this two-year period, the price of Colombian frozen beef increased 38.4 percent on a yearly basis). The low price assumes that after 1974 the rising trend in world price will be approximately oneeleventh of the 1970-1972 period. And finally, the moderate price assumes a rising trend in the world price of about three-eights of the 1970-1972 period. In all cases the per ton price of carcass beef has been converted to a live animal equivalent by a factor of 4.3. Runs 17 and 18 investigate the effect of low and moderate world prices, respectively. In the remaining runs, Runs 12, 19, 20 and 21, policies related to exchange rate, export subsidy and promotion of cattle production in the rest of the country are examined under conditions of low world price. Run 19 investigates the effect of cutting off the export subsidy

325

. .

after 1975 while maintaining a flexible exchange rate. Run 20 speculates on maintaining export subsidies while returning to a fixed parity against the dollar when this reaches an approximate value of Ps 25. Run 12 combines the effect of maintaining export subsidies and fixing the exchange rate with an increased growth of the non-Costa herd at 5 percent annually after 1975. And finally, Run 21 combines the effect of cutting off export subsidies after 1975 and maintaining a flexible exchange rate with an increasing cattle population in the rest of the country.

In general, the international competitive position of Colombian beef is investigated assuming the least advantageous conditions in both the domestic and world markets. In the domestic market we always assumed high export targets which are reflected in higher domestic prices, and in the world market we assumed low prices in all cases but in Run 18. In the study, export margins, defined as the difference between the pesos equivalent of the world price per animal increased by the export subsidy and the domestic price per animal, are used as a proxy for competitiveness in world markets. The subsidy to exports considered in the study is that enacted by Decree-Law 444 of March 22, 1967. Although Decree 444 also introduced a fluctuating exchange rate, the study assumes a return to fixed parity against the dollar when this reaches an approximate value of 25 pesos (Runs 12 and 20). Exchange rate values are trend-like projections computed exogenously in the model by Equation 9.18a on page 213.

As Figure IV.28 shows, the domestic market price per finished male is greatly affected by policies related to exports and production expansion. When exports are at a low level and modernization is limited to the Costa, policies to increase production (Run 14) result in lower prices than the base run. If modernization is carried out only in the Costa with the same development policies as in Runs 1, 10 and 11, the larger export target in Run 11 results in the highest price. The major contributor to lower prices is the expansion of cattle population in the rest of the country as depicted by Run 12.

Although Runs 1, 10, 11 and 14, show a steadily rising trend through the period 1965 to 1985, the varying time path at each five-year interval shows the effect of the seven-year cycle of beef supply from the rest of the country. Run 12 experiences a sharp drop in price after 1975 when the increased production from the rest of the country herds is marketed. But an upward trend follows after 1980 which coincides with a declining path in the long-term supply cycle. Since cattle production is discouraged in the Costa by low prices and domestic demand is always increasing, it is likely that in the long run the price cycle in Run 12 will have a raising trend.

From Figure IV.29 we can see that export margins are consistent with world prices and with policies affecting domestic

prices and the pesos equivalent value of exports. $\frac{1}{2}$ Since Colombia's current and projected share of the international beef market is relatively small (the high export target sustained after 1975 is equivalent to approximately 161 thousand metric tons or about 10 percent of the total beef traded internationally), $\frac{2}{}$ we assume an infinitely elastic demand for Colombian exports. This means that Colombian sales to the world market do not have an impact on world prices and that beef exports will be traded at the going world price. However, if Colombia greatly increases its share of the world market in the future, the above assumption would not hold and conditions of world supply and demand would have to be more carefully considered. When domestic price is high (because of high exports) and world price is low as shown in Run 17, export margins are negative for a period of about ten years, even if export subsidies are paid and the pseo gradually depreciates against the dollar. But further depreciation of the peso, to the point when the exchange rate reaches a 1985 value of Ps 53.98 for each dollar, results in a rising export margin that becomes positive during the last three years of the run. If world prices are moderate, export subsidies are paid, and the exchange rate is flexible, as shown in Run 18, export

 $[\]frac{1}{0}$ One way of looking at export subsidies is by changing the effective exchange rate at which certain commodities are traded.

 $[\]frac{2}{Australian}$ Bureau of Agricultural Economics, "World Production and Trade," <u>The Beef Sutiation</u>, No. 15 (May, 1971), 12-14.

margins are always positive and by 1985 could be as high as Ps 15,160/head.

Runs 19 and 20 project the effect of two alternative policies for export promotion under conditions of high domestic price and low world price. Run 19 shows that if export subsidies are cut off in 1975, a flexible exchange rate is not enough to produce positive export margins, although the upward trend of the last ten years suggests that in the long run the margin will be positive as the peso further depreciates. Yet, as shown in Run 20, exporters are worse off when a subsidy is paid but the exchange rate is fixed at about Ps 25 to the dollar. The downward trend in this case indicates that the negative margin will continue to widen as increases in domestic demand bid up the price. Run 21 shows that under conditions of low domestic prices resulting from increased supplies from the rest of the country and low world prices after 1975, a flexible exchange rate alone is sufficient to cause export margins to change the downward trend and become positive. And finally, Run 12 shows that if a fixed exchange rate is used in combination with an export subsidy the export margin takes an upward trend from 1975 to 1980, but after 1980 it again shows a downtrend when domestic price rises after increases in demand more than offset the impact of an enlarged supply. It is clear in this run that export margins follow a reciprocal pattern to that of domestic prices, and therefore the margins are affected by the long-run supply cycle as well (see Figure IV.28).

- 1. Disease control, modernization of cattle production, low export target and continuation of present trends and policies (base run).
- Cattle exports set at intermediate target from year 1974.
- Cattle exports set at high export target from year 1974.
- Cattle exports set at high target from year 1974 plus modernization of rest of national herd from year 1975.
- 14. Combines Run 4, Run 6 and Run 7.



- 12. Same as Run 1 with high export target and low world price after 1974 plus fixed exchange rate at Ps 25 and modernization of rest of national herd from 1975.
- Same as Run 1 with high export target and low world price after 1974.
 Same as Run 1 with high export target and moderate
- Same as Run 1 with high export target and moderate world price after 1974.
- 19. Same as Run 1 with high export target and low world price after 1974 plus export subsidy cut off at year 1975.
- Same as Run 1 with high export target and low world price after 1974 plus fixed exchange rate at Fs 25.
- Same as Run 1 with high export target and low world price after 1974 plus export subsidy cut off and modernization of rest of national herd after 1975.



Figure IV.29. Competitive position of Colombian cattle in export markets under various policy alternatives assuming high export target and moderate and low world beef prices, 1965-1985.

Two comments on the assumptions about the exchange rate are relevant to this analysis. First, when the exchange rate is flexible or fluctuating, it is assumed to depreciate at 7.28 percent annually precluding any acceleration or tapering off of this rate over time. Second, when the exchange rate reassumes its fixed parity against the dollar it remains at about Ps 25 from approximately 1973 until the end of the run in 19d5. This extended period of fixed parity is likely to result in an overvaluation of the peso that would eventually affect Colombian foreign trade leading to balance of payment problems and eventual further devaluation. The above considerations make the assumptions related to exchange rates in the model likely unrealistic.

The annual value of the subsidy paid to exporters will depend on the volume of exports, the world price and the exchange rate. Given the same world price, the higher the exchange rate the higher the subsidy per animal exported. Although annual payments vary from year to year, at the end of the simulation in 1985 they are the highest when the world price is high, the exchange rate flexible and the export target high (Run 11) and amount to Ps 5,179 million. But subsidy payments drop to Ps 1,168 million when world price is low (Run 17) while exports and exchange rate are the same as above. If world price is high, the exchange rate flexible, but the export target is low as in the base run, subsidy payments are Ps 2,462 million. And finally, subsidy payments are the lowest amounting to Ps 546 million when the

exchange rate is fixed and the world price is low even though exports are high (Run 20).

The value of foreign exchange earnings will depend on the volume of exports and the world price and, therefore, its annual projection will experience a changing pattern that will also differ among the various runs according to their underlying assumptions. If exports are sustained at a high level, the projected value of foreign exchange earnings in 1985 will be \$144.3 million when the assumed world price is low; \$307.5 million when world price is assumed moderate; and \$639.7 million when world price is assumed high. $\frac{1}{}$ If world price is assumed moderate, the projected value of foreign exchange earnings in 1985 will be \$146.2 million when exports are maintained at a low level; and \$220.4 million when exports are intermediate.

The foregoing examples indicate that changes in world price provide the government with more latitude in the choice of policies for the attainment of foreign earning targets without greatly impairing other desirable goals. Recalling earlier discussions we see that increasing the level of exports reduces consumption per capita unless other measures are taken to increase domestic beef supply (see Figure IV.14). Yet it is also true that higher prices brought about by

 $[\]frac{1}{At}$ the end of the run, the projected world prices per live animal based on the price of frozen beef are: low \$208.5; moderate \$444.4; and high \$924.4.

enlarged exports could create the incentives for farmers to expand cattle production with the long-run effect of increasing the number of animals marketed for slaughter. Likewise, government revenues from the domestic sector are enhanced as profits, cattle inventories and land values are increased.

Conclusions

We have made eight major inferences from our cattle policy experiments.

First, given the assumptions behind the model, the availability of credit for development does not seem to be as crucial as originally thought. Although farmers use credit whenever it is available, there are substantial unused credit balances at the end of each year between 1966 and 1976. Furthermore, the model indicates that, in the aggregate, farmers could cover establishment costs of the land being modernized with resources generated internally in the agricultural sector. This is shown in the model by the allowable rate of modernization depending on the farmers' capability to meet the total establishment costs without credit support (ARM3 in Equation 5.42c) being higher than the combined rate of modernization due to promotion and diffusion (RLMI in Equation 5.40). As seen throughout this analysis, farmers' response to economic opportunities determine their use of private and public savings for farm improvement. However, cash flow problems developing late in the simulation suggest the need for a comprehensive credit

policy. These outcomes are seen more explicitly here when increasing the competitive edge of the modern over the traditional operation; in such cases, easing the credit terms, as in Runs 6 and 7, encourages modernization more than increasing development credit funding as in Run 8. And the sharp decline in annual income in all runs after 1975 is an indication of farmers' unbalanced cash flow. The preceding results imply that providing credit assistance for a longer period might be more effective than increasing the volume of funds within the ten-year period (1966-1976) set in the model.

Second, the model's output indicates that the performance indices depend heavily on farmers' attitudes toward adoption and continued use of the new production methods. This behavior is simulated in Equations 5.36 and 5.47 using principles of diffusion theory. But modeling farmers' decisions to expand or contract production on the basis of diffusion theory is a poor proxy to sound economic principles explaining such processes. Until more adequate user cost, investment, and disinvestment theory is developed for use in modeling expansion and contraction of agricultural production, the model would benefit from more research and experimentation on farmers' response patterns under the diffusion theory used in this study.

Third, the benefits of long-term output responses in modern pasture acreage expansion are reinforced by policies directed toward the modern sector only. Otherwise, the

relative profitability differential of improved over traditional cattle production is decreased and the adoption rate of improved management practices is reduced accordingly. The effect of production incentives and the production campaign, particularly promotion, are highly complementary in encouraging the farmers to modernize their system of cattle production.

Fourth, cattle inventories and output in the Costa depend decisively on controlling diseases in the traditional sector. But at the national level, the major effect is obtained when cattle production is also modernized in the other producing regions.

Fifth, the cumulative effects of policy combinations (Runs 14, 15 and 16) on the model's output are, in general, greater than those of single policies. Positive effects of single policies are reinforced by complementary measures when simultaneously applied. Reinforcing and offsetting policy effects are important characteristics to be considered by policy makers when designing strategies for economic development.

Sixth, the competitive position of Colombian cattle in world markets is decisively dependent on world price and policies concerning domestic supply and export incentives. In this analysis, a negative export margin indicates that domestic price is higher than world price as viewed by Colombian exporters and implies that they cannot compete in world markets at the price beef is being internationally traded. The results shown in Figure IV.29 suggest that the need for and the amount of transfers from the rest of the economy to cattle exporters in the form of export subsidies can be examined and determined in the face of both domestic and world prices and exchange rate policies.

Seventh, an alternative course to policy makers' active participation in the model was to assume certain policy changes during the period 1966-1973. In this period various programs were implemented (cattle development and animal disease control), legislation was proposed (increased property tax, presumptive taxation of agriculture and elimination of cattle selective inventory tax), and important decisions were made (extension of general cattle inventory tax after its expiration in 1970). Or, using another interpretation, retrospective policy experimentation is a demonstration of how the model could have been used as a planning and decision-making tool had it been designed at the time decisions were made and/or policies proposed. Quick legislative action, assumed in some cases in the model, could possibly have been obtained by the use of the model. Models designed to predict the outcome of alternative policies have the additional application of prompting government officials and legislators to action. The Costa model attempts to evaluate the effects of an early implementation of various proposed policies and the likely effects of having started cattle development programs applying different measures to those used at the time of their introduction. Initial time of alternative policy implementation can easily be changed to 1973 or any other future year. The model's

structure is flexible enough to allow these changes. Given the structural relationships and assumptions behind the model, it can be expected that its output variables will experience the same trends as shown for the period 1966-1973. Changes in initial time of policy implementation were planned in additional experiments with the model but were not accomplished because of time constraints. Meanwhile, it is believed that extrapolating results from policy implementation during the period 1966-1973 will produce legitimate predictions for another future period, say 1973-1992.

Finally, throughout this study we have demonstrated that a simulation model provides a useful experimental setting in which policy makers and researchers can interact at different stages in model creation and developmental planning. This interaction is instrumental in redefining goals and policies, and reformulating parameter values and interrelationships. With different and improved sets of goals, policies, parameter values and interrelationships, new experiments can be carried on and new outcomes analyzed. This interactive, iterative process can and should continue until results are judged to represent the real system reasonably well and until decision makers are satisfied with their goals and the display of projections indicating the effects of following their various development strategies.

In our case, each of the policy runs have projected different outcomes at different time phases of the planning horizon. These projected differences provide policy makers

with some basis for selecting a prefered policy option on the basis of tradeoffs among perceived values for the economy. For example, if the perceived dominant values for the cattle subsector during the next 20 years are farmers' income and the foreign exchange generated from cattle exports, Run 11 (which has the high export target) may be preferred. On the other hand, if a perceived paramount value is government revenue, Run 5, which gives the highest annual revenue and accumulated funds for government, is the logical option. However, if the dominant value is to ensure consumers a higher availability of beef, without impairing farmers' and government revenues, Run 6 (with the reduced interest rate feature) would be recommended. If the objective were to increase per capita consumption of beef up to or above the recommended nutritional requirement of 28 kilograms, the selected option would be Run 16 (with the non-Costa cattle modernization feature). But this run, even though it maximizes foreign exchange earnings from cattle exports, seriously impairs farmers' and government revenues. Hence, in this particular situation, the tradeoffs are between a loss in government revenue and an increase in output and personal income of farmers in the cattle economy; and between an increase in beef consumption and a loss in foreign exchange generated from cattle exports, government revenue and farmers' income.
CHAPTER 13

SUMMARY AND CONCLUSIONS

Introduction and Summary

Colombia, as many other developing nations, is facing a new pattern of food scarcity in the decade ahead. The increased demands of her rapidly expanding population are added to the impact of rising affluence on demand for food. To meet domestic needs and take advantage of world market opportunities to increase foreign earnings required to support development, Colombia must make a great effort to encourage agricultural production, particularly the output of the protein-rich food which is in greatest demand.

With vast natural resources suitable for cattle production, this industry has the potential for becoming a leading sector in the Colombian economy. To this purpose the Colombian government has committed resources for preparing and implementing a nationwide plan aimed at the development of beef cattle production. With about half the cattle population and with a regional comparative advantage for grazing, the Atlantic or Caribbean plain of Northern Colombia is receiving most of the development effort.

With the preceding considerations in mind and realizing the experienced usefulness of the systems simulation approach in overcoming many of the complexities of development

planning, this dissertation developed a simulation model which: (1) focused primarily on the production of cattle in the six departments of Northern Colombia (the Costa) that include most of the Atlantic plain, and (2) included only rudimentary considerations on the related production of crops. The model thus developed is capable of exploring the ramifications of the proposed government strategy and their resultant interactions and feedback effects.

This study is divided into four parts incorporating basic background information and material relevant to the model building and analysis. Part I describes the general problems of producing cattle in Colombia and the physical and economic setting of the Costa region, discusses the justification for using the systems simulation approach, and finally sets the model's specifications and procedure. Part II details the five components used to simulate the production of cattle which: (1) allocate land use according to the farmer's perceived profitabilities of cattle and crops subject to land and capital constraints; (2) calculate the yield and output of cattle and crops and their respective producer and market prices; (3) provide the instrumental linkages for the government revenue, export trade policies, and production campaign policies; and (4) generate the performance criteria necessary to evaluate the impacts of alternative programs on the cattle economy through time. Three major policy entry points are considered; production campaigns can be specified, cattle and property taxes can

be levied, and export targets and incentives can be regulated. Part III discusses data needs and methods used in dynamic system models to determine the correspondence between the model and the system being represented. These methods include sensitivity analyses, tuning the model to track recorded time series and general validation procedures. Sensitivity analyses not only reveal logical or theoretical inconsistencies but also can provide an indirect way to test policy options and suggest data collection priorities.

Part IV demonstrates the model's applicability to policy formulation. Chapter 12 analyzes the results of 21 runs that examine combinations of policy options which have recently been considered in Colombia.

Salient Features of the Costa Model

In this summary some of the salient features of the Costa model are discussed. Then some policy implications and areas for additional research will be discussed.

First, the model is mathematical. With a mathematically formulated model, assumptions about behavior, technology and institutions are translated into the universal and precise language of mathematics which makes them relatively explicit and open for examination.

Second, the model is operational. That is, it is a computerized model and can be operated without much difficulty and at very low cost. Using an operational model is a major step forward in the task of modeling sectorial and/or regional economics. Experiments can be performed and new improvements in the model introduced if needed. Repeated runs using refined data and structural relationships improve the model's representation of reality and its overall performance. Without an operational model, it would be difficult, if not impossible, to accomplish these ends.

Third, the model to a large extent is data based. That is, many of its relationships have been formulated and tested using actual data. The model's outputs can be checked against actual data. These points are important because the degree of confidence we have in a model depends to a great degree on the extent to which it is able to explain past variation of variables and to predict future variation. As discussed in Chapter 10, when regional data are poor in quality and quantity, vast data requirements make the task of formulating and implementing a model with measurable quantities particularly difficult. Although models can be built even when information is poor, the system simulation approach provides means for dealing with the data problem by indicating where improved regional information would yield high returns in terms of a better understanding of regional phenomena and superior models.

Fourth, significant steps forward are made in the modeling of cattle demography. Three age cohorts for females and two for males with age-specific birth rates and age-sexspecific death rates, sale rates, transfer rates, and disease treatment rates have been employed which make the model's outputs change in extremely significant ways in

response to changes in the age and sex distribution of the cattle population. Such changes feed back in the cattle demographic component through time delays and induced behavioral effects to influence the pattern of cattle population development that produces further changes, and so on. Those endogenous interactions between cattle demography and performance appear to be vital elements in the process of regional economic growth.

Fifth and last, although the model as presented here needs further work and is not ready for implementation, it affords a good example of how the system simulation approach provides an analytical framework within which researchers and policy makers can interact while formulating alternative cattle policies. We were specifically interested in evaluating the long-term economic impact of modifying cattle prices through exports, revising tax policies, and the proposed government production campaigns to expand cattle production in Northern Colombia. Crop improvement and export policies were included as a secondary objective.

To this end, the computerized cattle simulation model provided a very useful and a convenient means of predicting and comparing the outcomes of various combinations of cattle programs and policies. Based on the predicted time paths of the various performance indices of the cattle subsector, the merits of various policy alternatives were discussed. This capability of the model to project the time paths of various performance indices can be used to give the policy

maker a clear picture of the range of possible outcomes of each proposed policy.

In addition, policy makers not only benefit from the analysis of potential policy results but also from the process of formulating the simulation model itself. Planning officials taking part in the model development process are forced to specify, examine and study their assumptions, data sources, underlying interrelationships and impact of each policy upon the model structure and parameters. Thus the planners may refine and improve their decision process and the information used in it. As the process of simulation model development and experimentation proceeds, both researchers and planners gain greater insight into the mechanisms and likely patterns of change within the system being modeled. Further, the decision maker can play a more active role in the experimental system by making exogenous policy decisions at the end of any time period and allowing immediate feedback on the results of alternative decision patterns and policy choices. This iterative process involving close interaction among decision makers and system analysts engages decision makers in investigation activities that lead them to perform as researchers as well. Hence, the simulation model becomes not only a valuable analytical tool in helping decision makers in their planning, policy formulation, and program development activities, but also an educational tool that enhances their planning capacities [28].

Policy Implications from Simulation Experiments on the Costa Cattle Economy

There are six major inferences from the simulation experiments which may throw light on questions of public policy concerning a cattle development program. First, investments in government disease control programs are justifiable since the projected cattle output, given the same model assumptions, was higher when improvements in range and herd management were accompanied by disease control in the traditional sector. Nevertheless, long-run assessments of the pay-offs to "traditional" farmers require a more accurate accounting of costs with and without the extended treatments.

Second, measures aimed at improving the profitability of modern cattle alone are the most effective in encouraging cattle output and increasing farmers' incomes and government revenues. Easing the debt burden, particularly interest payments, had the greatest long-run effect on all performance variables. Although credit has been considered crucial for development, the results point out that farmers' attitudes constrain the use of capital resources to a point that increased credit funds went largely or totally unused. The pay-offs of "educating" the farmer, demonstrating the profit opportunities of the new practices and creating a socioeconomic environment amenable for investments seem to be very high. Further, the profitability of the cattle subsector relative to other sectors in the economy must be carefully considered if capitalization of the former is a

preferred goal. Otherwise capital will likely be diverted from the cattle subsector to more profitable sectors of the economy.

Third, increasing the domestic price level of cattle by increasing exports from low to high targets is not by itself an effective tool for expanding cattle modernization and output. While it is true that higher prices greatly increase farmers' income and wealth, they also have the effect of curtailing the incentive to modernize by reducing the profitability differential between traditional and modern Consequently, if modernization is to be enhanced practices. by taking advantage of the farmers' expanded revenue, it is necessary to implement other policies that would increase the profitability of the modern operation relative to traditional. Yet, when total Colombian cattle output is greatly expanded because of developmental efforts in the non-Costa regions, the simulation analysis indicates that a pricing policy through the export market is crucial for sustaining modernization and preventing farmers from incurring heavy income and capital losses. Likewise, government revenues would not likely be improved because of farmers' higher income unless there are more effective ways of taxing farmers.

Fourth, special taxes on cattle are the main sources of government revenue (not including income tax). Cutting off cattle taxes, in addition to removing this source of revenue, has the same counteracting effect on modernization as price increases (discussed above). However, any loss

In government revenue that might occur from cutting off special cattle taxes could be compensated by more effectively taxing the farmers' increased income and the asset value of their land; or by taxing their increased purchases of producer and consumer goods. Although increasing taxation on land has, from the point of view of government revenue, an equivalent effect to levying taxes on an expanded cattle population, the likely <u>allocative effect</u> of land taxation makes it a more preferred policy option. These results suggest the need for a careful reassessment of the taxing policies toward the cattle subsector and the consideration of alternative policies that will not impair government revenues and will not interfere with the allocation of resources on the farm.

Fifth, if improved cattle production is to be used to bring some sort of redistribution of income in the region, it is a requisite that the medium and small farmers are not left behind in the modernization effort. Alternatively, modernization might be accompanied by a change in the pattern of land ownership that prevailed in 1960. According to the agricultural census of that year, three-fourths of the cattle farms were less than 100 hectares; they controlled one-fourth of cattle and had an average inventory of less than 50 head of cattle [14].

Finally, the world price of beef greatly affects foreign earnings. It is, therefore, very important that the government secure the highest world price for exports.

This can be attained by exporting high quality dressed and/or processed beef to bring higher prices in international markets. In addition, this measure might increase the Colombian value added of exports with spill-over effects on the economy. The policy experiments show that the profit margin of cattle exporters varies according to domestic and world price conditions and measures affecting the effective rate of exchange. The combination of a fluctuating exchange rate and an export subsidy seems to give an ample profit margin to exporters. This suggests that the subsidy could be adjusted periodically based on cattle and foreign exchange market conditions. An appropriate exchange rate could maintain a competitive edge in international markets, thereby eliminating or reducing the need for large transfers from public revenues to exporters. Since cattle exporters compete with suppliers to the domestic market, the foreign earning targets set by the government may affect the nutritional levels of the Colombian population. By the same token, exports have a regulatory effect on domestic prices that in turn affect incentives to produce.

Improvements and Extensions of the Model

The simulation of dynamic human systems is a process of trial and invention that can never be completed. Each simulation result teaches and prompts additional questions leading to an iterative procedure that helps sharpen data and verify structural and causal relationships. It also may disclose problems in the original formulation of goals, feasibility and methodology that might need refinement and reformulation. Revisions may also be required by the changing needs of planners and policy makers. The extent of these experiments and changes demand costs that must be weighed against the expected returns of increased model accuracy, flexibility and relevance before a decision is made to proceed with the modifications.

A number of areas in the current Costa model need further attention in order to improve its performance. These are discussed below. Experiences with other regional models [1, 27, 53, 62] have suggested possible extensions to enable it to better address some of the major problems of economic development. These will also be discussed.

Needed Improvements in the Model

There are several aspects of the Costa model which need further development and verification. First, it is not certain that the model of the domestic cattle price mechanism (Equations 6.39a and 7.2) adequately or even realistically represents the actual operation of that market. In particular, the pricing of cattle other than finished males is an oversimplification of supply and demand for the various sale groups. Furthermore, the link between the Colombian market price and the regional market price is not clear since regional prices are sensitive to short-term fluctuations caused by seasonal changes in pasture yields. In addition, the supply response to price changes is not yet very well understood. Since the model is fairly sensitive to cattle prices, further research and eventual modification of this aspect of the model may be indicated.

A second area that could call for further work is the simulation of cattle demography. Although the use of different demographic groups is an improvement over previous regional cattle models [51, 53, 55] a further step could be to calculate the productivity of each age group relative to the feed resources consumed. This will permit more precise computation of feed requirements and total herd productivity (kilograms of gain per animal-year). The latter will enable the model to estimate the nutritional contribution to the Colombian population more accurately.^{1/} If policy makers and planners feel this is an area they would like to investigate more fully, revisions of the model will be necessary.

A major feature of the model which needs theoretical and empirical verification is the modernization decision mechanism (Chapter 5), particularly the value of parameters that determine the adoptors' behavior. A development program taking place currently in the region and resembling that described by alternative 2 (improvement and substitution of artificial pasture for native pasture) provided both an empirical base for assigning values to these and other model parameters as well as justification for focusing the simulation

 $[\]frac{1}{A}$ detailed demographic model of this sort is discussed by Johnson <u>et al.</u>, in [48].

experiments on this alternative. Yet other alternatives including forage crops, though with greater impact on the model's output and performance, were disregarded because of inconsistencies probably arising from the simulation of farmers' behavior in the face of the higher risk and uncertainty involved in these alternatives. Additional research is necessary to better understand the nature of this problem before further modeling work can be done on it.

Another unrealistic aspect of the decision mechanism is the assumption that capital is employed at a low opportunity cost in cattle production. This does not effectively represent the capital market in Colombia where transfers of capital out of cattle farming are certainly occurring with the subsequent effect of impairing the farm improvement effort. It also seems unrealistic that the private capital constraint on land modernization decisions (Equation 5.41) is practically inactive during most of the simulated time and only becomes effective toward the end of the simulation period when consumption expenditures are greatly increased. It is more likely that the capital outflow effect discussed above coupled to the demand for capital to buy cattle on an individual farm basis, which is not in the model, would put an earlier constraint on land modernization decisions.

Other constraints, which are not in the model at all, are the allocation of commercial credit through the banking system, the availability of labor (including management) and the availability of other inputs (fertilizers,

herbicides, seeds, and drugs, primarily). Thus, it may be desirable, if the model is to be implemented, to give high priority to modifying the model to realistically reflect actual input constraints on land allocation and production decisions.

Related to these possible shortcomings of the model are the problems of investing and disinvesting and the generation and use of farm-produced capital. The former is approximated in the model by the land modernization and reversion mechanisms of Equations 5.36 and 5.47, and the retention or sale of cattle accompanying the increased or reduced carrying capacity brought about by these land transitions. Yet, as pointed out by Johnson [47], a solution to these problems needs more development in economic theory concerning user costs which partially determine how many units of productive services are generated from fixed durable inputs in the intricate process of agricultural growth, change, and/or deterioration.

Two examples of other structures of the Costa model which may require further verification are the on-farm resource use response and the living expenditures adjustment mechanism. First, it may be questioned whether Equation 6.36, in assuming the functional relationship shown in Figure II.7 between the perceived relative profitability differential of traditional and modern cattle on the one hand and, on the other, the extent to which farmers make use of their on-farm resources, realistically or even adequately determines their preference function.

Secondly, further information may indicate weaknesses in the formulation of living expenditures, particularly if disaggregation by income groups is accomplished. Likewise, this formulation does not model income effects on consumption brought about by changes in the real income of farmers.

Finally, this model may always be improved by refining the data that go into it and by a closer interplay among specialists from related disciplines and policy makers.

Extensions

Seven additional ways in which the Costa cattle simulation model can be extended for policy analysis will be listed here. First, it should provide a more comprehensive basis than it currently does for analyzing government revenue and budget expenditures for agricultural modernization programs. The present model was not able, nor built, to generate the effect of a graduated income and wealth tax for cattle producers whose efficiency could be compared with alternative consumption, land and special taxes. Likewise, the present model did not provide comprehensive grounds for balancing and allocating the government budget for different modernization programs whether on a regional or national basis.

Secondly, to discuss meaningfully the distributional impact of government policies on investment in production campaigns, exports and public revenues, and the response of farmers to these policies, the cattle subsector may have to be disaggregated by farm size. Since farmers in each size category have a different command over resources and their own beliefs and values, such disaggregation may be helpful in analyzing problems of income distribution, and in simulating more realistically their decision making. It may be useful to further disaggregate the region by ecological zones and to subcategorize the cattle industry by type of activity, i.e., breeding, growing, fattening, and their combinations.

A third possible problem area which could call for an extension of the model is the question of growth and age distribution of the rural population. This is closely related to the problems of employment, labor supply, and pressure over natural resources, which in turn involve aspects of income distribution, rural-urban migration and land ewnership distribution. The dimension of income distribution, a pervasive one in all development plans, would be an important output criterion for evaluating alternative policies.

Fourth, as part of the process of improving the land use decision mechanism of Chapter 5, the crop subsector may have to be disaggregated by competing commodities in order to determine a more realistic crop mix. This will

allow the more accurate simulation of the competition between crops and cattle and, in addition, may suggest policies toward improvement of individual crops and regional specialization. Likewise, the model may be extended to relax the constraint of testing each modern alternative one at a time and include mechanisms for the adoption of various alternatives simultaneously. This may reflect more realistically the various responses of farmers with respect to profitability perceptions. The two extensions suggested here may have substantial impact on the performance variables that could be of interest to policy makers.

Fifth, the sensitivity of the model to cattle price changes suggests the need for including a semi-automatic decision-making mechanism whereby the government export policies in any one year depend on the interaction between the prevailing world and domestic price of beef. Export targets, subsidies and exchange rates could be set with the aim at maintaining both a price incentive to producers as well as a competitive position in international markets. The major benefit of such a model extension would be to help determine a more flexible governmental export policy which would stabilize farmers' income and net government revenues from exports, given the fluctuations of beef prices.

Sixth, one extension of a technical nature mentioned in Chapter 10 is running the model in a Monte Carlo stochastic mode rather than deterministically. Although there are various theoretical problems involved in the

use of Monte Carlo techniques [1, Chapter 11] such stochastic runs could be useful in dealing with data problems and in evaluating the relative stability and sensitivity of policy alternatives in the face of uncertainty. Giving a probability distribution to some of the data instead of a mean value, it is possible to incorporate methods of statistical sampling and inference into the outcome of the model. Such statistics would permit the evaluation of <u>ranges</u> and distributions of possible outcomes for different policy options rather than point predictions of absolute output levels.

Finally, there are extensions on the scope of the Costa model that would be relevant to policy makers in their task of solving the problems of development. First, the present version of the model can be easily adapted to the other four Colombian cattle producing regions and further integrated into a national subsectorial model. Secondly, based on other experiences, it could be expanded into an agricultural sector regional model or even a complete regional model including both the agricultural and nonagricultural sectors. The experience gained from these extensions and the information they contribute are of unquestionable relevancy for development planning in Colombia.

Concluding Remarks

We have discussed some of the shortcomings of this study and have suggested means by which they can be dealt with in order to improve the model's predictive and prescriptive capabilities. As we have seen most of these shortcomings

arise from data problems and from deficiencies in the economic theory required as a basis for modeling many of the activities found in developing economies [47]. The shortcomings we have encountered, however, also affect the accuracy of models built using simple, paper-andpencil techniques, or the more complex and specialized techniques discussed in Chapter 1. Nonetheless, we have reasoned that the systems simulation analysis as used here, with its flexible approach to many of the methodological problems found in studying economic development, provided an improved framework for policy, program and project problem analysis.

It must be stressed that the Costa model yields usable estimates of the consequences of following several policy strategy alternatives over a period of several years. Furthermore, the present work constitutes a major improvement over other cattle production studies made in Colombia as well as a useful contribution to the study of likely consequences upon regional growth of developing a leading agricultural industry. In the future, when more and better regional statistics and research and more advances in our knowledge of regional development are available, still more information could be introduced into the model to correct current inadequacies. The experience and lessons learned in the present work and others like it will be valuable in future modeling efforts. APPENDIX

COMPUTER PROGRAM

	PROGRAM SINCOL (INPUT, JUTPUT, TAPE1+INPUT, TAPE2+OUTPUT)	SINCOL	2
	COMMON /CONTRE/ 1, BIJICA, IRON, BEARAIJPRICHA, PRIVEIJPRIVEZJPRIVEZJPRIVEZ		2
2	HALSI 4003100	SINCO.	
	READADD ANNA	SINCOL	i
	DO 200 IRONAL, HAUM	81MC01	í
	CALL RUNDAT	SINCOL	,
	CALL PRISET	SINCOL	É
	CALL LANSET	SINCOL	
		S1HC0;	10
	CALL CRIPSET	\$1#C0	11
	CALL PROSET	SINCOL	12
		STHCO	13
	CALL PODSET	STHCOL	14
	CALL CRISET	SINCO	15
	PRT & BEGPRY	Ch5	3
	PRTVL . PRTVLI	SINCOL	16
- •		SINCOL	17
20	IPRINT = 0	SINCOL	18
		SINCOL	19
	IF (T.LT. PRT)GO TO 40	SINCOL	20
		642	3
	IF (T.EO. PRTCHG)PRTVLUPRIVLZ	SINCOL	21
	PRT & TOPRTVL	SINCOL	22
40	CALL PRIGEN	SINCOL	23
	CALL LANDAL	SINCO	24
	CALL AGPROD	SINCOL	25
	CALL DEMOG	91HC01	24
	IF(T.GE.THOD)GALL HODCRD	SINCOL	27
	CALL AGACC	SINCO	24
	CALL MODRAT	STHCOL	29
	CALL CHTACC	SINCOL	30
	1F(T,LT,DUH)GO TO 20	SINCOL	31
200	CONTINUE	STHCOL	32
	STOP	SINCO	33
900	FORMAT(I1)	STHCOL	34
	END		

•	SUBBOUTINE BUNDAT	RUNDAT	2
	COMMON /CONTRL / T.DT. DUR. IRUN. REGPRT. PRTCHG. PRTVLL PRTVL2, IPRINT.	CNS	1
	MAL - THODA TRD	CHS	2
	MAMEL LAT ANAMPUNA DT. DUN. BEOPRY. PHICHO. PRTVLS. PRTVLS. MAL. THOD. TDO	CH5	
		RUNDAT	5
		RUNDAT	•
	DER SA	CNZ	5
	FRICHU V AUU. Betul 7 a C	ČW2	6
		ČNŠ	5
		CuS	6
		RUNDAT	
	DEWFRI - 1, Thon - 140	RUNDAT	
		RUNDAT	10
		RUNDAT	11
		RUNDAT	ĨŽ
20	REAULA/MANRUMA	RUNDAT	13
	HK[16(2, MANKUN]	RUNDAT	14
		RUNDAT	- 15

SUBROUTINE PRIGEN PRIGEN 2 CONHON JACCOUN/ DRLAT, AMLAT, ACLAT, VLAUDT, VLDTYT, TAXCT, AGSUT, ACCOUN DELAN, ANLAN, ACLAN, VLAI, DY, VLDTYN, TAKCH, AGSUN, ACCOUN 1 1 DELAY, DERU, FARMIC, DITOUS, DERDT, RESTOT, EVPLIV, HEFR, ACCOUN 2 ALPHS, TEC. TCEC, TRUSLS, TLF, TLFP, CPLF, RINT, RINTL. ACCOUN 6 3 CREDT. DI4. C250, C253. C254. LT1. LT2. LT3. CBSTG. CSTGH. 4 C # 7 1 1 EPCRPJ, "CROPU. PCR JPL. PF CROP. ACCOUN 7 . EYLNCJ. YLOCU, YLOCU, YLOFC, DYLOCU, DYLOCL, DYLDFC ACCOUN 8 ACCOUN • REAL HOFR CONNON /CATTLE/ PEGT, FENT, PEPT, PHOT, PHOT, TOPOPT, TOPOPR, CATTLE 2 CATTLE PEPH, PH 34, PHPA, TOPOPH, VAGAPL, AUXL12, 1 PFGH. SLFERT, SLSCCT, SOLDFT, SLSHLT, SLSPT, SLSPTP, FEHSCT, CATTLE 2 4 SLFERM, SUBCCH, SOLDEM, SUSHLH, SUSPH, SUPR, FEMSCH, CATTLE 5 3 PPFGTT, PPFPTT, PPHOTT, PPHPTT, PPFGTH, Ā CATTLE PPFPT4, PPHGTH, FPHPT4, REGTT, REPTT, 3 9 CATTLE CATTLE R#GTT, RHPTT, RFGTH, RFPTH, RHGTH, RHPTH, 8 TONT, TOWAT, TOWN, TOWIP, TOWAN, PATADT, CHM, ONT, PFLACT, CATTLE . 1 YWAT, PA, PAP, PAPP, PRAT, ESPL, UNEXPL, CT202, C214, C244 CATTLE 10 COMMON /CONTRL/ T.DT.DUW, IRUN, BEGPRT, PRTCHG, PRTVL1, PRTVL2, 1PR14T, C#5 1 C×5 HAL,THOU,TDO 1 PRICEN 5 REAL MKM PRIGEN DIMENSION VALSO(11) 4 DATA C219, C237, C238 / 1., .1, .032 / DATA ELASD, FLASI, FLASP / .7, .6, 1. / DATA MKM, UNEXPL / .15, 300000. / PRIGEN 7 PRIGEN 8 PRIGEN . DATA PRF.UHEALC / DATA DEL7 / 5. / DATA SHALLSÖ, JIFFSÖ, KSÖ / 4., 1., 13 / DATA SHALLSÖ, JIFFSÖ, KSÖ / 4., 1., 13 / DATA VAL50 / 3100... 76023., 90375., 10040., 10140., 50270., C=5 7 PRÍGEN 11 PRIGEN 12 125700., 191700., 2450.0., 202000., 329000. / PRIGEN 13 1 PRICEY 14 DATA PAPO / 907./ PRIGEN 15 EPCRPU + EPCRPU+ (PCROPU+ERCRPU)+DT/DEL7 C # 4 \$ PCROPU + PCROPU+RCSTDT PCROPL + PCROPL+RCSTDT PFCROP + PFCROP+RCSTDT C¥4 2 Cu4 3 PRIGEN 19 PPA = PA PRICEN 21 PA . PA+(1,+C21++(TDE+=#JPD)+DT/(TDEH+ELASD)) PRIGEN 21 PAPP . PAP PRIGEN 22 PAP + PA+(1, +HKH) PRIGEN 23 PRAT & PAP/PAPS RDEN + ELASI+CZ37+ELASP+C238+8LA8D+(PÅ+PPA)/(PPA+DT) PRIGEN 24 PRIGEN 25 DEN # DEN+DT+(DEN+RCE*) PRIGEN 24 EXPL . 0. 15(T. 08.4.) FXPLPTABLIE(VAL50, SHALL50, 617790, K\$8.7) PRIGEN 27 PRIGEN 28 TOEN & DEN-EXPLOUNERPL PRICEV 27 IF (IPRINT.LT.1)RETURN PRIGEN 38 PRINT890,T PRINTO10, PASPAPSEXPL, DEM, TDEM, EPCAPU, PCADPU, PCROPL, PFCROP PRIGEN 31 PRIGEN 32 RETURN STO FORMAT(SOMIOUTPUT OF SUBROUTINE PRICES AT TIME, P8.2) PRIGEN 33 918 FORMAT (1H0, 9X. 2HPA, 18X, 3HRAP, 9X. 4HEXAL, 8X. 3HDEN. 9X, 4HTDEN. 8X. PRIGEN 34 ANEPCRPU, 6x, SHPCROPU, 6X, SHPCROPL, 6X, SHPPCROP/1H0, 8X, VE12, 4) PRICEY 3e 1 PRIGEN 36 ENTRY PRISET PRIGEN 37 PCROPU = 630. PRIGEN 38 PCROPL . 892. PRIGEN 39 PFCR0P = 245 PRIGEN 40 EPCRPU + PCROPU PRIGEN 41 PA = 1067. PRIGEN 42 TDEM . 1889100. PRIGEN 43 DEM = 1739180. PRIGEN 44 SUPE + 1889108. PRIDEN 45 PAP . 907.

RETURN End PRIGEN 46 PRIGEN 47

SUBROUTINE LANDAL LANDAL CONMON /CONTRL/ T.DT. DUN, IRUN, BEGPRT, PRTCHG, PRTVL1, PRTVL2, SPRINT, MAL, T400, TDO CHS 1 CHS TOL, TTOL, TTOLL, TTOLUS, TTOLUS, TOLSFO, TPSL, TLHOD, TLHODI, TLHODZ, TLHODS, TTOLR, TOLR, COMMON /LAND/ LAND LAND TLCAL, TLCBU, TLCRUP, TOLSP, TOLUSP, TLPC, TLC. LAND 3 XA1, R1, H2, PR2, PDR, CRH, RLH, RLH[, RLCRU, AUX1, AUX3, LAND 5 4 ATH1. ARM2. ARMS. AS. CV. GRT. CPLPT. XDEL. RPTH. GRE LAND DIMENSION CROUTA(3), CROUTEL(6), VAL49(5) LANDAL 9 DATA AL2,C249,CFD1,CL2 / .03, .8, .5, .5 / DATA VAL49 / 2000c., 18920., 8990., 7000., 5000. / LANDAŬ 6 LANDAŬ 7 DATA DEL4.DEL15 /1.7, 10./ DATA DEL19, DEL16, FEL17, DEL18 / 4+5, / DATA K4,K11,K40,DIFF40,SM4LL40, 3, 6, 4, Č14 6 LANDAL . DATA K4,K11,K49,D1FF49,SM4LL49/ 3, 6, 4, 1, 6, / DATA TLFC0,TGLSF0,TLDRNLG / 161733,, 3137950/, 3137950, / DATA TLAVLO,TLAVU01,TLAVJ02 / 933303, 2169060, 1676088; / DATA XDEL,RUDRN,RLCRL / 1,, 13908., 4900, / LANDAL 11 LANDAL 11 LANDAL 12 CH4 9 IF(T.LT.TH05)00 TO 28 LANDAL 14 TRSL = (XR1+R2+RLH)+XDEL LANDAL 15 E4 = 1 LANDAU 14 IF (RTOLL.LT.D. AND. TTOLL. ST. DT. (-RTOLL) E. ... LANDAL E10 • 1 LANDAL 18]F (RTGLU1.LT.0..AND.TTOLJ1.GT.DT+(+RTGLU1))E10+8+ LANDA. īī E14 = 1 LANDAL 28 IF (ATGLUZ.LT.8..AND.TTGLUZ.GT.DT.(-ATGLUZ))#14484 LANDA 25 LHODI . AMAXICAMINICTLMODI+DT+CRLM+TQLL/TQL+TLMDDI+PR2+E4+RTQLL+ LANDAL 22 TLHODI/TGLL), TGLL), Ö,) TLHODI/TGLL), TGLL), Ö,) TLHODZ = AMAX1(AHIN1(TLMOD2+DT+(RLH+TGLJ1/TGL+TLHOD2+PR2+RTGLU1+ 1 LANDAL 23 LANDAL 26 E12+TLHOD2/TGLU1), TGLU1), 0.) TLHOD3 = AHAX1(AMIA1(TLHOD3+DT+(RLH+TGLU2/TGL=TLHOD3+PR#+RTGLU2+ LANDAL 1 29 LANDAL 26 E14+TLMOT3/T3LU2), TGLU2), 0.) 1 Ž7 LANDAL TLHOD . TLHOD1+TLHOP2+TLHOD3 LANDAL 28 20 CONTINUE **LANDA** 29 TLAVE = TLAVE+(TLAVEC-TLAVE)+DT/DEL15 LANDAL 30 TLAVUI + TLAVUI+(TLAVJ01-TLAVU1)+07/DEL10 TLAVUZ + TLAVU2+(TLAVJ02-TLAVU2)+07/DEL17 LANDAL ĴĹ LANDAL JŽ RLFC + AL2+TLFCG+FXP(AL2+T) TLFC + TLFC+DT+RLFC LANDAL 33 LANDAL 34 LANDAL TLFCU . C249+TLFC 39 CALL DELAY(ALDRA, AUTIO, CAOUTII, DELII, DT, KII) TLDRN & AMINI(TLDPH)+DT+AUKID, TLDRNL0) LANDAL Ĵó LANDAL 37 TEFCE . AMINISTEFC. TEFCU, TEAVED. TEORY) LANDAŬ 38 TLBAN = 2000C. LANDAL 39 IF(T.GE.O.)TLHAN + TABLIE(VAL40,SHALL69,DIFF40,K69,T) Alndl = TLAVL+TLUPN-TLFCL-TLBAN LANDAL 41 LANDAL 4 TLCRL & AMINISTICAL OT ALCRL, ALNOL) TLCRL + TLCRL+TLPAN LANDAL 42 LANDAL 43 TGLLP . TGLL LANDAL 44 TGLL + AHAXI(TLAVL-TEFCL-TLCRLR, J,) HTGLL = (TGLL-TGLLP)/DT TGLSF1 = TGLSF1+(TGLSF0-TGLSF1)+DT/DEL18 LANDAL 45 LANDAL 46 LANDAL 47 TGLSF = TGLSF1-TLPRH LANDAL 48 TEFCUIP . TEFCUI LANDAL 49 TLFCUL + AMINI(CFN1+TLFCU, TLAYUL) LANDAL 50 DUFCUL + TUFCUL+TUFCULP LANDAĽ 51 ALNDUL . TLAVUL+TLFCUL LANDAL 92 AUX4 + 0. **LANDAL** 93 IF (ALCRU.LE.D.)AUX4+CL2+DLFCU1 TLCRUP + TLCRU LANDAL 54 **LANDAL** 55 TECRU + AMAYI (AMINI (TECHU+AUX4+DT+(AUX2+AUX3), ALNDU1), 4.) LANDAL 56 TGLU1P + TOLU1 LANDAL 57

2

Ż

2

ŝ

TGLU1 • AL	NDU1-7LCRV	LANUAL	20
RTGLUL . f	(TAL (1)+TGL (1)P)/OT	LANDAL	24
CALL DELAT	FAURT ALVZ. CRUITA. DEL 4. DT. BAL	LANDAL	66
		LANDAL	41
1LFCU2 = 4	MINICILLIVEIVED, CAVEN	LANDA	
TGLU2P + T			
16LU2 • TL	AVU2-1LFCU2	CTMDY!	
NTG1112 # 1	71112+151129)/11	LANDAL	
		LANDA	63
		LANDAL	
16LU + 16L	.U1+TGLU7	LANDAL	•/
*GL • TGLL		LANDAL	••
		LANDAL	67
		LANDAL	78
TGLU1 - T	GLU1-1L#012	LANDA.	24
TTGLU2 = T	(GLU2-TL#UP3	LANDAL	
TTGL = TGL		LANDA.	14
17616	CLINTCL SEAE9	LANDAL	73
		LANDA	74
IGER VIGE		LANDAL	75
IF(100141,	LT.1) / ETURN		
PR1NT893,T		LANUAL	.
PRINTY22.4	NE VICE TEAVESAE MOUSS TEAVUSS TEAVUSS TEORNS TEORN	LANDAL	,,
HOTLTONA T	TERLATERUS ALFOUSATIFCS TIREATICAL ALCALATERUSTECS	LANDA	78
		LANDA	7.
1 T		LANDA	
PR167914.T	TALL, TTALU1, TTALU2, TALU, TAL, TTAL, TALS / TALA	LANDAL	
PRINT924,P)*GLU_#*ALU1#K1GLU2#RLFC#AUX2#AUX4#AUX10	LANCAL	01
PRINTY30.C	ROUTA, CROUTIL	LANDAL	82
16/7 17 78		LANDAL	83
10 1 1 2 0 5 1 1		LANDA	84
PRINTVOUL			44
RETURY			
890 FORMAT(36H	(4^UTPUT OF SUBROUTINE LANDAL AT TIME,F8.2)	LANUA	
900 FORMAT(1H)),9X,5HALNC,7X,5HTLAY,,7X,6HALNDUI+6X,6HTLAVU1+6X,	LANDA	
1 41.1	1 44. 2 AY ANTI DAN. 78. 54TL HAN/140.98.7(E11.4+18)	LANDA	
	A CALL AND	LANDA	
YUB FURMALLIN.		LANDA	
1 401	LEC. BI, "HILLEL, 71, OHILLEGH BAS DHILLEU, 74344 (LCG) WAS		
2 Знт	LC/1HC,9x,1C(t11,4,1x))	LANDAL	
916 FORMAT(1HG	, 9%, 5HT TOLL, 7%, 6HT TOLJ1, 6%, 6HT TOLU2, 6%, 4HTOLU, 8%, 3HTOL,	LANDAL	72
1 97.	ANTTOL AN SHTULSF. 7X SHTTGLA/1HSAAN AF12.4)	LANDAL	93
	AV BULLETT ANDTEL 11. AT. ANDTEL 12 AV. ANDLEC. AV. ANDLEC. BY. ANALY2.	LANDAL	
V24 FURPATEING		43404	
1 51,	4+A. (4, 5%, 5%A. X10/102, 7%, /(E11, 4, 1%))	LANDAL	
930 FORMAT(1H)	;,9x,6HC90UT4,3NX,7HCR0UT11/1H0,9X,9(E11,6,1X})	LANDAL	70
950 FORMATCING	. 91. SHTLHCD. 71. 6HTLHOD1. 61. AHTLHOD2. 61. 6HTLMOD3. 61.	LANDAL	•7
1 447	WELLAY ANTOLISAT SHTOLUS 7X. SHTOLU2. 7X. ANTOLA/SHA.AX.	LANDAL	98
		LANDA	
< 9E1	6. •)	Peace	77
ENTRY LANS	SET	LAMPAL	100
c		LANDAĽ	101
ē	VALJES FOR LAND ALLOCATION	LANDAĽ	102
	ALAAPI A Buun Waxaaultaa	LANDAT	103
•	• · ·		
TLAVL # 43	1262.	LANUAL	144
TLAVU1 = 1	.72¢729.	LANDAL	107
TLAVU2 = 1	293017.	LANDAĽ	106
161561 .	A.TGI SFA	LANDAT	117
7,884 - 8	1 A = - 2 A = 1 A = - 2 A = -	LANDA	
		LANDA.	
LANN & SC	.030.	LANUAL	1.1
TLCRL = 14	IC7CS.	CN4	•
*LCRU = 28		C¥4	7
R. CRU - 15	b00.	CHA	
		I ANDAL	
		LATUAL	779
TLFC = 101			
TLFC = 101 TLFCU1 = 4	5915.	LANDAL	114
₹LFC = 101 *LFCU1 = 4 *LC = TLBA	15915. No TLCRL • TLCRU• TLFC	LANDAL LANDAL	114
TLFC = 101 TLFCU1 = 4 TLC = TLDA TLMOD = 0	10915. 10915. 1N+TLCRL+TLCRU+TLFC	LANDAL LANDAL LANDAL	114 115
TLFC = 101 TLFCU1 = 4 TLC = TLBA TLMOD = 0, TLMOD = 0,	IS915. IN•TLCRL•TLCRU+TLFC	LANDAL LANDAL LANDAL	114 115 116
TLFC = 101 TLFCU1 = 4 TLC = TLBA TLMOD = 0, TLMOD1 = C	1915. No TLCRL • TLCQU• TLFC	LANDAL LANDAL LANDAL LANDAL	114 115 116 117

	TINOD3 . A
	TCI 111 - 14717**
	TCLU2 - 1353135
	TCLL = 3104.4
	- GLL - J17864.
	AIGLUZ1,37.
	TOLL # -16376.
	PUX1 # 3.
	AUX2 = U.
	AUX3 # /000.
	PB2 = C,
	H2 = C.
	RLM = C.
	XM1 + C.
	CROUT4(1) . (.
	CPOUT4(2) = C
	CROUT4(3) = 0.
	30 991 JelyA
	CPOUT11(1) + 27:3.
996	CONTINUE
	RETURN
	END

LANDAL	119
LANUA	120
LANDA	121
LANDAL	122
LANDAL	123
LANDAL	124
LANDAL	125
LANDAL	176
LANDA	127
LANDAL	124
LANDAL	129
LANDAL	130
LANDAL	131
LANDAL	132
LANDAL	133
LANDAL	134
LANDAL	135
LANDAL	136
LANDAL	137
LANDAL	130
LANDAL	139
LANDAL	140
LANDAL	141
LANDA	142
LANDA	143
LANDAL	144
LANDA	145
LANDA	146
LANDAL	147

.

	SUBROUTINE AGPHOD	AGPROD	
	COMMON /ACCOUN/ DYLAT, APLAT, ACLAT, YLANDI, YLDINT, TAXCI, AGSUT,	ACCOUN	
		ACCOUN	
	S DEFA, DEAU, ANALIS, DEFOT, PESISISERPLIA, ASIA,	ACCOUN	
		ACCOUN	
	• C4EL1, D14, C2>0, C2>3, C2>4, L11, L12, L13, CUSTG, CSTGH,	C # 7	
		ACCOUN	
		ACCOUN	(
	REAL NOFH	ACCOUN	
	COMMON /CATTLE/ PEGT, FENT, PEPY, PMGT, PMPT, TOPOPT, TOPOPR,	CATTLE	
	1 PFGH, PFPH, PH 14, PHPH, TOPCPH, VACAPL, JUXL12,	CATTLE	
j	SLFEHT, SUSCET, SDUDET, SUSPER, SUSPER, FERSET,	CATTLE	
	SLEFRA, SUSCER, SOLOFA, SUSHLA, SUSPA, SUPA, FEASCA,	CATTLE	
1	4 PP, GTT, PPFPTT, PPMJTT, PP4PTT, PPFGTN,	CATTLE	
1	9 PPFPT4, PP4GTM, PP4PT4, RFSTT, RFPTT,	CATTLE	
1	b HMGTT, KMPTT, RFGTM, RFPTM, RMGTM, KMPTM,	CATTLE	
	7 TOUT, TOMAT, TOMA, 10449, TOMAM, PATADT, OMM, ONT, PELACT,	CATTLE	
	"YMAT, PA, PAP, PAPP, PRAT, EXPL, UYEXPL, GT202, G214, C244	CATTLE	1
	COMMON /CONTRL/ T.DT.DUM.IRUM.BEGPRT.PRICHG.PPTVL1.PRTVL2.IPRINT,	C45	
•	1 HAL,T"CU,TDD	C+5	
	COMMON /LAND/ TRL, TTGL, TTGLU1, TTGLU2, TGLSF0,	LAND	
1	1 TASL,TLMOD,TLMCD1,TLMOD2,TLMOD3,TTGLR,TGLR,	LAND	
i	2 TLCALITLCAUITLCAUPITGLSFITGLUIPITLFCITLCI	LAND	
	3 X91,P1,M2,PR2,PDP,CMM,ALM,PLM1,RLCRU,AUX1,AUX3,	LAND	
4	4 APM3, AKM2, ARM3, AS, C7, GRT, CPLPT, XDEL, RPTN, GRE	LAND	
	DATA CGOU,COCU1.CGU1.CGU2 / 3.48. 1.16. 5., 1.7 /	AGPROD	
	DATA CGOL,CACLI,CGL1,CGL2 / 3.8, 1.26, 5.1, 1.7 /	AGPROD	
	DATA C5, C7, C9, C10, C220 / . C03, 1.16, .55, 2.6, .21 /	AGPROD	
	UATA CPLPT, GRE, TDND, * CN35 / .45, .74, 7.4, .2343 /	AGPROD	1
	DATA CG3,CG4 / 3.3. 1.10 /	AGPROS	1
	DATA THODC, DEL 8. DFL 9. DEL 13 / 100., 305. /	AGPROD	1
	CG0 • (CGCU+(TTCLU)+TTCLU2)+CGCL+TTCL)/TTCL	AGPROD	1
	CG01 + (CC501+(TTSL01+TTSL02)+C30L1+TTGLL)/TTGL	AGPROD	1
	RCON . AMAX1 (+CCN+DT+C5+(3RT+GLE) . 1)	AGPROD	ĩ
	TDWTG + RCCV+(CC++CPLPT+TTGL+CCG1+(1.+CPLPT)+TTGL)	AGPRON	ĩ
	TDNRE & C7+C223+(TLCRJ +TLCRL)	AGPRON	1
		AGPROD	ī
	TENT . TENTS-TOURE-TONSE	ACPRON	
	EVIDEN . EV DELI-(VIDEL-EVIDEL)-DT/DELA	AGPROR	5
		AGREGO	5
		AGRADU	5
		AGPROU	5
		AGPRUS	
		AGPRUU	Ś
		AGPRUS	
		AGPROD	
		AGPRUD	<
	CO TO	AGPRUD	
		AGPROD	3
		AGPROD	3
	IF (FAL.CU.Z.UF.FAL,CU.4)UU IU 80	AGPROD	3
		AGPROD	2
	UTH • CUJ+CPLPT+C34*(1,*CPLPT)	AGPROD	3
	GO TO 160	AGPROD	3
	CGA • CG1	AGPROD	- 3
	CGA • CG1 CTR • CG3	AGPROD	2
	CGA = CG1 CTR = CG3 IF(MAL.GT.2)50 T0 140	AGPROD AGPROD AGPROD	2
	CGA = CG1 CTR = CG3 IF(MAL.GT.2)50 TO 140 TDNM = CGA+TLHOD+CTR+TP3L	AGPROD AGPROD AGPROD AGPROD	3
	CGA = CG1 CTR = CG3 IF(MAL.GT.2)50 T0 140 TDNM = CGA+TLM0D+CTR+TP3L G0 T0 200	AGPROD AGPROD AGPROD AGPROD AGPROD	3 3 4 4

		TONG A (COANTI MODECTRETHEL)/(TI 400+TREL)	AGPROD	43
			AGPROD	- 44
	145		AGPHOD	45
		- DNG	AGPROD	46
	724		AGPROD	47
			AGPROD	40
			AGPROD	4.
			AGPROD	50
			AGPROD	51
			AGPROD	52
	200	IF(IPHINT.LT.I)"E'U""	AGPROD	53
		PRINTED; I	AGPRON	54
		PRINT905, RCn1, CR5, CU01, UNION DURES UNDER SUMM	AGPROD	
		PRINTOIC, EVIDED, TEDICITEDIC	AGPROD	56
		IF (T. LT. TPOD) HETCHA	ACPROC	
		PRINT953, CGA, CTH, FPC, FINEC, FD IN	AGPRON	58
		RETURN.	AGPROD	59
	890	FORMAT (36H4 GTPHT C) SURAULINE AUTOD AL TIMETON C	AGPROD	60
	900	700MAY(1H0,9X,4HRCU*, X,3HCU0,9X,4HCUL1,8A,5HTU410,745,5HTU410,7	AGPROD	61
		1 SHTD:SF, 7X, 4HTT: //HC.9X, /(E11.4/1X/)	AGPROD	62
	910	FORMAT(1H0,9X,6HEVLICJ,6X,5HTLICU,7X,5HTLICC,7X,7HTLDC)	AGPROD	6.5
			AGPRON	64
	950	FORMAT (1HD, 9X, 3HCSA, 97, 3HCTR, 9X, 4 (CPEF, 8X, 3HIL) (4X) AN HE FORM	AGPROD	6
		1 4HTDUN/1HC,87,5E12.47	AGPRON	64
		ENTRY PPDSET	AGPROD	
С			AGPROD	61
Ĉ		VALUES FOR PRODUCTION	ACPROD	Ă
Č			AGP 800	7
-		GRT . (TGL+TGLSFC+C9)/10P3P1	AGPROD	7
		HCON = 1.	AGPHOD	7
		TDNM # 6.	ACPROD	2
		TLF = C.	ACPROD	
		TLEP = 3,	ACPROD	
		HETUHIA		2
		é nu		

SUBROUTINE DEMOG	DEMOG	2
CONNON /CATTLE/ PEGT.FERT.PEGT.PHGT.PHPT.TOPOPT,TOPOPR,	CATTLE	2
1 PESM, PEPH, PNGH, PHPH, TOPOPH, VACAPL, AUXL12,	CATTLE	3
2 SLFERT, SLSROT, SOLDFT, SLSHLT, SLSPT, SLSPTP, FEHSCT,	CATTLE	4
3 SLFERH, SLSCCH, SOLDFH, SLSHLH, SLSPH, SUPB, FENSCH,	CATTLE	5
4 PPFGTT,PPFPTT,PPHGTT,PPHTT,PPFGTM,	GATTLE	6
\$• PPFPTX*PPMTT**PFMTT**FFTT*	CATTLE	
A PHUTT, RHPTT, PFGTH, KFPTN, PHGTH, FHPTH,	GATTLE	
TONT, TONAT, TONA, TONAP, TONAH, PATAOT, 344, QMT, PFLAGT,	GATTLE	
8 YMAT, PA, PAP, PAPP, PPAT, EXPL, UNEXPL, GT202, G214, G244	GATTLE	10
COMMON /CONTRL/ T,DT,DUR, IRUN, BEGPPT, PRTCHG, PRTVL1, PRTVL2, IPRINT,	GHS	1
1 YAL, THOD, TOO	1410	2
COMMON /LAND/ TGL, TTGL, TTGLU, TTGLU, TTGLUZ, TGLUZ, TGLSFU,		1
		ũ
	LAND	Ś
3 XK1;K1;K2;FX2;FD4;GK1;KC1;FC1;FC0;GV4;GA2;F00;F	LAND	6
ARTISARIES ARTISARIES ARTISARIES (2) + 0 + 1 + 0 + 0 + 0 + 0 + 0 + 0 + 0 + 0	DEMOG	5
$\frac{1}{1}$	DEMOG	7
DIMENSION VALUE (Systemetro)	DEMOG	5
DTHENSION VAL9(A)	DEMOG	9
DIMENSION RINTFI1(15), RINTFM1(15), RINTFT2(15), PINTFM2(15)	GEMOG	10
DTHENSION RINTHT1 (15), RINTHM1 (15), RINTHT2 (15), RINTHM2 (15)	NEMOG	11
REAL NREQT.NZEOM	DEMOG	12
DATA 012 / 1. /	DEMOG	13
DATA C198,C199,C200,C2u1,CT202,CM202/.95, 2., .7, .4, .4, .4/	DEMOG	14
DATA 0206,0207,0208,0209,0210,0211/ 1., 1., 1., 1., 1., 1.	DEMOG	15
DATA C212, 0213, 0214, C215, C216, C217 / 1.4, C., .6, .2, D., .0205/	CH4	
NATA 6224,6225,6226,6227,6228 / . 18, 92, 88, 42, 535 /	05406	11
DATA (230, 2231, GT232, GH232, G233/+01, 65, 1+,1+5, 1+05/	05409	10
DATA 37241,07241,0242,0243,0244,02497 +9,495, +05,049,000	DENOG	2.1
NATA 3246 (286, 628) / 1., 1., 3. /	DENOG	21
DATA PPSIII, LASII, ARXII, ARNII, SORVIZ, AND $2 + 3 + 2 + 2 + 2 + 2 + 2 + 2 + 2 + 2 + $	DEMOG	22
UALA DASTIZ, ELANIZ, AMATZ, AM	DEMOG	23
DATA DESTISIELASTS ANYLA ANTLA BENTLA BEXTA 1 a 8 a 23 22 a 11 a 14 /	DEMOG	24
DATA DOSTES. EL ASTS. AMXTS. AMXTS. BMXT5. BMXT5/. 13 85 21.24.13 17 /	DEMOG	25
DATA PRSTTS	DEMOG	26
DATA PRSIT7. LAST7. AHXT7. AHNT7. 9HNT7. BHXT7/.13,.6,.2,2*.1,.15/	nF MOG	27
DATA PPSIM1.ELASM1,AMNH1,AMXH1,BMNM1,BMXH1/.5,1.,.9,1.,.9,1. /	DEMOG	28
DATA PPSIH2, CLASH2, AMNM2, AMXH2, BMNM2, BMXM2/.5, .7, .8, 18, .95/	DEMOS	29
DATA PPSIM3,ELASH3,AMNM3,AMXM3,BMNM3,BMXM3/.8,1.,.8,18,1/	DEMOG	30
DATA PPSIH4, ELASH4, AHNH4, AHXH4, BHNH4, BHXH4/.1, B, 11, 3, 11, 2/	DEMOG	31
DATA PRS145,ELASH5,AHNH5,AHXH5,3HNH5,BHXH5/.13,.85,.13,.25,.13,	DEMOG	32
1,19/	NEMOG	33
NATA PRSIM6,ELASM6,AMN(16,AMXH6,BMNM6,BMXH6/.902, ./02, .00,	DEMOG	34
		76
DATA PRS14/, LASH/, AMNH/, AMXM/, SMNH/, SMNH/, SMAH// 10, 10, 10, 10, 10, 10, 10, 10, 10, 10,	DEHOG	35
DATA $IDTF1, IUTF2, JUFF1, IUTF2, Z, Z, Z, Z, Z, Z$	DEMOG	37
0 1 1 1 1 1 1 1 1 1 1	DEHOG	30
1010 UNT 11 TOTES AND 1 TOTES	DEMOG	60
0444 SHALLIY AFFIYAA TONY 1009 00	DE HOG	41
π_{ATA} SHALL π_{1} (17.11) (17.11) (10.1	DEMOG	42
DATA SHALL M2. DIFFH2. KM2/ .3636. 5 /	DEMOG	43
DATA SHALL3.91FF3.K3/ .3635. 5/	DEHOG	44
NATA SMALLM3.NIFF43.K43/ .36, .36, 5 /	DEMOS	45
JATA SHALL 4, JIFF4, K4/ 0., 1., 1/	DEMOG	46
DATA SMALL5, DIFF0, <5/ 0., 1., 1/	DEMOG	- 47
74TA SMALL7, DIFF7, K7/ J., 1,, 7/	NEHOG	48
NATA SHALLS, NIFFO, KA/ 5., 1., 3/	DEMOG	49

	NATA SMALL9+0IFF9+<3/ (++1++7/	DENOG	E A
	NATA VAL1/ *** +25. +55. +53775/	DENOG	20
	DATA VALV1/ 000 30 56 7	DENOC	21
	TATA VAL2/ .761711	DENOG	76
	DATA VAL"2/ .712 9* 57032122 /	DENOC	73
	DATA VAL3/	DEHUG	24
	DATA VAL 437 .45	DEHOG	22
			70
			27
		UE HUG	20
		112406	24
		UEWUG	60
		DEHUG	51
	TATA COPTET COPTEM / A E J /	DEHOG	52
		DEMOG	63
		DEHOG	54
		DEHOS	55
	$\begin{array}{c} 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 $	DEMOG	-56
•	DATA THA', THAM / 575, 125 , 4	DEMOG	67
ä	IN-EXITE AND MASTILLS - NUMBERS	DEMOG	69
U		DEHOG	70
•	FINET = (PFINET*("FPT-UT*(DFDT+FFOUT2*SLMAST))-SLINET*DT+	DEMG	71
	1 UI * ("BT * 3, 13+PIFNUT) * ((PFPT-DT * (DFPT+RFOUT 2+SLMAST) *	DEMOG	72
	2 (1, J-PFINET))+OT*(PFOUT1-PFPTT-SLFERT)))	DEMOG	73
	TFMAST =PEMAST*(PEPT-DT*(DEPT*PECUT2+SLINET))=SLMAST*DT+(1.0-	DEMOG	74
	1 PATAFT)* TT*C.18*U.SC*((PFPT-DT*(DFPT+FFOUT2+SLINFT)*	DENOG	75
	2 (1.G-PFHAST))+OT+(RFOUT1-RFPTT-SLFERT))	DEMOG	76
	TE(T.LT.T401) 60 TO 20	DEMOG	77
	FINFM = (PFINFM*(PFPM-DT*(NFPM+RFOUM2+SLMASM))-SLINFM*DT+	DEMOG	78
	1 DT#(C3M+J.1_+FIFNUM)+((PFPM-DT*(DFPM+RFOUM2+SLMASM)+	DEMOG	79
	<pre>2 (1.J=PFINFM))+DT+(PF0UM1-PFPTM-SLFERM)))</pre>	DEMOG	80
	TFNASM = PFMASM+(PFPM-DT+(DFPM+RFOUM2+SLINFM))-SLMASM+DT+(1.]-	DEMOG	81
	1 PATAFM)*^**.18*0.05*((PFFM-DT*(OFPM+RFOUY2+SLINFM)*	DEMOG	52
	(1.7+FF(454))+0T+(RFOU 11-PFPTH-SLFERM))	DEMOG	43
•	23 CONTINUE	DEMOG	84
C		DEHOG	85
C	PEMOGRAPHY - TRADITIONAL	DEMOG	96
C		DENDG	87
	RFOUPT1 = PFSUT1-RFOUT1+(DRLT1+PSFGT+PPFGTT)*DT	DEHOG	A.A.
	PFGT = PFGT+DT\(PFT+)RLT1*PFGT+SLSFGT+RFGTT+PFOUPT1)	DENOG	89
	PFOUPT2 = REOUT2+REOUT2+(DRLT2+PSEPT+PPEPTT)+DT	DEHOG	an
	PFPT = PFPT+)T* (REQUET1+PRLT2*PFPT-SLSEPT-REPTT-PEQUPT2)	DEMOG	91
	QLOFT = (3FOUPT2-SOLDFI-ORLT2*QLDFT)*DI+QLDFT	DENOG	02
	9N0((PT1 = 2H3UT1+340)(T1+(02) T1+PSMGT+PPMGTT)+01	DENOG	31
	PMGT = PMGT+OT+(34T-ORI T1+PMUT-SI SMGT-R4GTT-D101PT1)	DENOC	55
	ROUPT2 = RHOUT2-PHOUT2+(0PL12+PSMPT+PPMPTT)+DT	DEHOG	74
		DEMOC	77
C		DEMOC	70
č	DEMOSPAPHY - MODEPN	DEHOG	7/
č		DENOG	70
0	TEXT LT THORNED TO A	JC 106	33
		ULHUG	100
	AROUNDE - NAMARTE VOREAUSTREDERFERDIRFERDIRFEN Dech - Dechintkijehundi Mabech Steche Decha Dechina V	DENOS	101
	- FOR - FERTING - SECURDS - VERTING - SECONDERFORMENT	WE TUG	102
	★FUUTT2 = KFUUTT2 (FUUT2) (UKLFC+PSFPTPPPPFFF)] TU1 DEDM = DEDM(DTA)/2E01644_DD14/280E04_CCECA4_EEATA_DTC)	UEMOG	103
		DEMOG	104
	ULUFT = (VFUUFT2-SJLUFT=VKLAZ*ULUFH)+UI+ULUFH	UEMOG	105
	"MUUHMI = K"JUMI-Y"JUMIT(UKLMI+PSHGH+PPHGTM)*NT	DEMOG	106
	PHGH = PHGH+UIT(=11-12CH172HH=SLSHGH=PHGTH=RHCUPH1)	DENOG	107
	₩₩IJU₩¥2 = <nuij₩2****uu₩2**("ycl#2*#sapn*ppmpta)*ot< td=""><td>DENOG</td><td>108</td></nuij₩2****uu₩2**("ycl#2*#sapn*ppmpta)*ot<>	DENOG	108
	PAPH = PMPH+DI*(CATUPM1-TRLM2*PMPH-SLSMPH-RHPTH-RHOUPM2)	DENOG	109
_	4) CONTINUT	DEMOG	110
C		DEMOG	111

369

.

c	TOTALS - TPADITIONAL AND MODERN	DENOS	112
ċ		05 406	113
•		DEHOG	116
	PFT . FFGT+PFPY+OL)FT	05405	115
	DAT . DECTADADT	06406	115
	TOPOPT + PET+PHT	DEHOG	117
	3PFGT . PF5*/10P0PT	201-105	114
	PPECT . PECT/TOPOPT	DEHOG	119
	PPHGT + PHGT/TOPOPT	DEHOG	120
	1072 T T PHE T 1 7 7 7 7 7 7 7 7 7 9 7 9	DENOG	121
	PFCH + PFPH+0LOFH	76406	122
	DFM = 0FCH++ (FDH+0L)FM	DE HOG	123
	PH4 B P45Hol 4D4	DEMOG	124
	ТСРОРЧ в РЕЧ•РИЧ	DEHOG	125
	PPFGH + PFSH/TOPOPA	06406	125
	PPECH + PECH/TOPOPA	20430	127
	PPAGH + PH3.N/TOPOPH	05406	128
	P4070710710707070	06405	129
	P0000 # 10000+0000	0EH01	133
	VACAPL = PAP+(C227+(PFCT+C212+OFGH)+C225+(PFPT+C212+PFOH)+C224+	DE 10	131
	1 (0LDFT+C212+OLDFH)+C223+ (PHGT+C212+PHGH)+C225+ (PHPT+C212+	2540	132
	2 PAPH) }	DEHOS	133
	TF(T.GT.THON) &UXL12 = AUXL12+(AUX12-AUXL12)+0T/CEL23	DEMOG	134
	AUX12 * (C227+PFGT+^226+PFPT+C226+OLOFT+C228+FFGT+C228+FHPT)*C239/	7E 405	135
	1 TTGLP	7540G	136
	IF(T.29.T900) &UXL12 = AUX12	76435	137
C		20430	135
¢	TOTAL DIGESTIBLE HITPIENTS	DEMOG	139
C	•	26402	160
	TONAT = TONT/TOPOPT	DEM05	141
	TOPEDT + TONT+ VPEDT	05406	142
-	PANT & TOPOT-TOPEDY	05405	163
	PAIT & PANY	06H0S	144
	TF(T.L1.T402)G0 TO 45	05 405	165
	IF(TOP)24.[[.].)50 TO 62	05405	146
	TONAN - TONN/TOPOPH	7E # 06	167
	GO TO 67	05.405	148
	es auden e le	0546	149
	P3 100-04 = 1044-40504	OEH: U	157
	PANN = TOPOPH-TOPE3N	JEMOG	151
	0854 B 0844	05405	152
	45 CONTINUT	DEMOG	133
C		DEMOG	154
Ç	INFIRTILITY AND MASTITIS - PROPORTIONS	06405	155
Ç		7E405	156
	SFINET + FINET/PEPT	DEMOG	157
	7FAAST : TF4457/PF34	20430	139
	FHAST + FFDT+PFHAST+(1,-PFINFT)	DEMOG	139
	FFQT # OFPT-FQST-FINFT	DE 40G	160
	•IFHUT + 723544HINL(2231-0230, (0231-0230) *EXP[-07232*(TONAT-	DEMOS	151
		DEMOG	152
	TF 11. (*, 1703) 60 * 0 \$3	TENOG	163
	JECHENEN, JULI JEN 10 N/	76405	164
	······································	DEMOG	165
	**************************************	17E HOG	166
	NU TO NT AT DETUGN - DENAGN - N	ULHOG	167
	3/ -//3/3 / -/-/3/ 3 /. /	02 406	168
	•		169
			170
	······································		1/1
	6 ENF 1-1 76 36 11 UNAN-46 37777	17,706	1/2

		371		
	្រី១	CONTINUT	DENOG	173
Ç			DEHOS	174
C		ABUTILLOSIS TREATIENT	DEHOG	175
C	•		DENOG	176
			DEHOS	177
		ATAGTT = =====A==X1(1,-C148="TP(-C149=TAGAD), C262)	NE 406	178
		4 1 8 4 4 1 5 4 5 4 5 4 5 4 5 4 5 4 5 4 5 4 5	02406	179
			OT HOG	190
			000	101
			DENOS	183
		CTAR & AWINI (CTART. CTARM, PEPT. PEDM)	7E 40G	145
		TF (OFPH.LC. 2.130 T) 62	DEHOG	145
		PCTARM T TTAIN/CEPM	OEMOG	186
		50 TO 63	05406	187
	- 52	PCTAPH 1 7.	DEMOG	188
	- 53	PCTART + START/PFPT	2E40G	189
		IF (PFGN, LF, C,) 50 TO #5	NENOG	190
			DEHUG	191
			OFHOS	192
	18		DENOG	196
	- 55	PGFTAT + (ATANIT+)T+PGFTAT+(PFGT+8FT+DT))/PFGT	DEHOG	195
C			26430	196
ċ		FOOT AND HOUTH TPEATHENT	DEHOS	197
C			9EH0G	194
		CXPDS + CXPDS+DT+I4P6FT	7E 406	199
		EXPACT V C.	DEMOG	570
		IF(T.SE.TOO) EXPAFT+TACLIE(VAL8, SMALL8, DIFF8, KR, I)	DENOS	231
		ATAFY = ANTNICIPAFT/COSTFT, 10P0PT)	00 405	222
		$\mathbf{A} = \mathbf{A} + $		2/3
		A 1 A 7 1 1 4 A 1 4 7 1 7 A 1 4 7 7 1 A 1 4 7 1 A 1 4 7 1 4 7 1 A 1 4 7 1 4 1 4 1 4 7 1 4 1 4 1 4 1 4 1 4	OFMOG	204
		PATAFT & ATAFTT/TOPOPT	OFHOG	236
		IF (T.GC. TO) PATANT+ (& TAFTT+ATABTT+ (1PATAFT)) /TUPOPT	DENOG	237
C			OEMOG	234
C		ADJUSTMENT OF INTERMEDIATE RATES	DEHOG	229
C			DENOG	210
		CHNGT + 1 (NRLT1+PSFGT+PPFGTT)+NT	DEHOS	211
		RCH4GA = 1 (ORLA1+PSFGH+PPFGTM) *DT	DEMOG	212
		70 153 IF1, KGP3F	111106	213
		RINIFILI) - RINIFILI) - RCANGA	05406	214
	• • •		OFHOG	216
		9C MGT = 1, - (0+L 72+ PSFPT+PPFPT) *0T	DENOG	217
		PCH46H = 1 (D2L #2+0) FFH+PPFFTH) +0T	DEHOG	218
		00 170 1+1, KPPCOF	DEMUG	219
		RINTFTZ(I) = RIHTFTZ(I) *RCHWGT	DENOG	223
		RINTENS(1) · RINTENS(1)+RCHNGH	05402	155
	153	CONTINUE	NEHOG	222
		RCHNGT = 1 (DRLT1+PS4GT+PPHGTT) *DY	DENDG	223
		PCHNG4 1 1.++(04(M1++)NGM+PPMC:4)*01	17E PUG	275
		10 146 1714 DINTHESSISSER NUCT	OF HOC	223
		PINTNHILLS + PINTNHILIPPCNNGH	DENOS	227
		CONTINUE	TE HOG	224
		RCHAGT # 1 (APLT2+PSHPT+PPMPTT) *OT	DENOG	229
		RCHWGH + 1 102L HZ. PSHPH. PPHPTHI OT	DEMOG	230
		00 160 I=1, KP9004	NE HOG	231
		RINTHT2(I) = RINTHT2(I) PRCHNGT	DEHOG	525
		RINTN42(I) = RIN1442(I)+PCH4GM	DENOG	533

152		05406	234
• • •	CALL DIATIOFT, FOUTL, PINTETL, DEROF, TOTEL, DT. KUROF)	76405	235
	CALL STLOT (PEOUPTI, PEOUTZ, SINTETZ, OPRODE, IDTEZ, DT. KPROF)	00430	276
-	CALL STISTIAT, ENSUITE WINTHILL DEROR. IOTHL. OT. KERON)	DEHOS	237
	CALL STLATER OUTLY, ROUTZ, PININTZ, DPRODA, IDTAR, OT, SPRODA)	DENOS	>14
		00#30	219
		DEHOG	240
	744 . (2*)// 12*10*0PH*(TDHEH-TOHEFO))/TOHEFO	DENOG	261
	1000PT1 4 0FGT+F/3T+P1T	CW7	ž
	4FGTT + 544+7FGT/1090PT1	CW7	3
	763TT - 444+5: 77/1900T1	CH7	L.
	PHGTT + 241+PHGT/IOPOPT1	ĠW7	5
	NPTT & PAS+PHPT/TOPOPT1	CH7	6
	3FGT + - 3FGTT	DEMOS	2+6
	28674 + -97677	DEMOS	247
	ристи — -ристт	DEMOG	268
	QHDTH 1 - 24077	DENOS	249
	PPFSTT + PFSTT/PFST	CWA	1
	PPFPTT + PFPT7/PFPT	CWA	2
	IPHGTT = JHGTT/JHGT	CNB	3
	PPHOTT = 74577/5407	CWB	•
	PPFCT4 & PFCTH/PFC4	CMB	5
	₹₹{₽\$\$4,L{.(.)}₽\$\$\$74 • \$.	DOMED	?55
	DEEDIN 1 OFDEN	G 11 B	5
	YF(0F74)00F074 = G.	0E #01	257
	784574 × 74674/8464	CWA	7
	IF (PHGM.LT.1) PDHGTM & D.	DEHOS	259
	DDND14 4 24324704 04	CW4	9
	[F[0474,15,*,10+4014 # 0,	DEMOS	261
	TE (PEGH .GT.*	NC NOS	222
	CALL DILOT (BER, PEDUHI, RINTENI, DGROF, IDTEI, DT, KGROF)	D: H05	263
	CALL DIENT (FEO:1041, PEULAS, PINTERS, DPROOF, IDTES, DT, KPROJE)	7 MOG	26%
	CALL SELOT (4 1H, SHOUML, PINTHAL, OGROA, IDTAL, OT, KGROA)	D. MOG	265
	TALL DELOT (= "DUP"), PHOUME, RINTHME, DPRODH, IDTHE, DT, KPRODH)	DE-ING	266
	60 10 180	DENDG	257
1-1		04,406	269
	2 NATATIAN TO PROVIDE AND A TOTAT	THE HOG	764
1.5		01403	2/3
		JEHOC	2/1
		DEMOS	7/2
		0E #05	27 1
1/5			276
			217
		00 H00	2/0
	*10*0*1*1/ * F04//*10*00*10/01/	05405	277
1.4		DEHU'S	2/3
		OCHO:	
	9 TNTNY2 (1) . P4P4/(000010142)	16 H00	281
. 74	10x134437	05 #05	243
• · •	2001111 B 0004/092014	77#06	7 7 6
		DI. HOC	203
r • ''	· • • • • • •	0, 400	745
č	ATOTA PATES AND RIPTHS - TRADITIONAL	OF MOG	244
ř	and a second and a second second by the second by the second second second second second second second second s	OF HOG	247
•	CRANGE & TABLET (VE.S. SMALLS, DIFFS, KS. PCTART)	OFHOG	74.8
	AFT2 + APT2+AT/ACL1+(8PT-ART2)	OF HOG	246
	THE TAPLE (VALLS SMALLS DIFFS. KS. TOWAT)	36 106	284
	76LT2 + 09T2+(1.+294461)	05405	211
	947 + 90117+FT-7*	00 406	222
	1FT	DE 105	211

SHT . AAT-SET DEHOS 294 CBT = RTTPCTANGT DEHOS 215 C NEHOS 296 С BIRTH RATES AND BIRTHS - HODERN 217 DENOS c DENOS 239 IF(T.LT.THOD)GO TO ZIC -CRANGY = TABLIS(VAL+, SHALL+, OIFF4, K+, PCTABP) RRM2 = TRY2+TT/TL1+(80-R7M2) **PEHOS** 299 DEHOS 133 DEHOS 331 TF (TOPOPH.GT.C.) NRIATANLIE (VALHI, SHALLHI, DIFFHI, KHI, TONAH) DEHOS 332 ADLN2 + 00420(1,--34464) AAM + 42470FF24 DEHOG 333 DEHOS 136 4FH + . 5+144 8H4 + 944-8F4 DENOS 335 DENOG 336 CRH & RDM2+COANGH DEHOC 337 21, CONTINUE DENOS 338 000 309 DENOG JEATH PATES AND DEATHS - TRADITIONAL DEHOG 313 DI HOG 311

 DRAT = TARLIS(VALS, SMALLS, DIFFS, KS, PATAFT)

 PPT1 = PT1+0T/012*(0PGT-0PT1)

 DKT2 = D*T2+DT/011*(0PPT-0RT3)

 DPGT = TARLIE(VAL3, SMALL2, DIFF2, K2, TOMAT)

 DRPT = TARLIE(VAL3, SMALL3, DIFF3, K3, TOMAT)

N2406 312 DENOG 31.3 DEHOS 31.6 DEHOG 315 DEHOS 11.6 DRUTI = DRT1+5P#T DEHOS 317 ORLTZ + DPTZ+DRAT DEHOS 31.6 OFGT + PESTANPLTS 06405 11.3 JEPT + PENT+NPLTZ DEMOG 123 HOLDET + OLDETERALTZ DEMOS 321 DHGT + PHGT+DPLTS DEHOS 122 AMPT + PHPTHAPLT? 02406 353 TOTHST = DEGT+DEPT+DHGT+UMPT+DOLDET DENOS 314 ORT . TOTHST/TOPOPT DENOS 325 C DEMOS 359 Ċ **NEATH RATES AND DEATHS - HODERN** DEMOG 127 C 05405 324 IF(T.LT.THUD)60 TO 263 NEHOG 329 DRAM + TARLIE (VALS, SMALLS, DIFFS, KS, PATAFM) DE HOS 330 0041 + 0841 +01/0712* (086#-0841) 06400 331 0842 + 0942 +01/0113+(0894-0842) DEMOG 332 TELTOPOPH. ST. ... I DRS 1+ TABLIE (VAL 12, SMALL 12, CIFFN2, K42, TOWAN) DEHOG 333 IF (TOPOPH. GT. *.) DRPH= TARLIE (VALH3, SHALLH3, DIFFH3, KH3, TOHAM) DE HOG 334 DELH1 + DRH1+07AH DEHOG 135 UDENS & UDASANS DEHOS 336 NEGH # PESH+DPLH1 DEMOG 137 DEHOG AFPH & PFP4+APLHS 338 DOLOFH + CLOFH+09142 139 UH2# # 8#2#+00F#1 OENDS 360 THAN + PHPH+OPLHZ DEHOG 351 TOTHSH ... DEGE+DEPH+DHGH+DHPH+DOLIEM DENDG 342 IF (TOPOPH.LE.C.)GO TO 258 DEHOG 353 URN . TOTHSM/TOPOPH NZHOG 344 DEHOG 50 TO 263 365 755 OPH = 0. DENOG 356 DEHOG 251 CONTINUE 367 C DEMOG 348 359 SALES - TRADITIONAL DEHOG C DEMOG 358 Ċ DEHOS ¢, OLD FEMALES 351 DENOS ¢, 392 DEHOG np1 = PRAT***CLAST1*PRSIT1 353 PPEST1 = AMIH1(RMXT1, AMAX1(BMHT1, BP1)) DEMOG 35 6

	AD1 - POTST1+C206+PANT/OLOFT	DEMOG	335
	APAT - ANTHICANATI, ANAYLANNTI, APIDI	15 HOS	356
		JENOG	157
		DEHOG	35 9
•	PANI & PANI-SULIMI	06405	137
- C	THREDETIC ANA MARTITE	DENOG	362
÷C	INFERICE AND RESILIES	00 H PO	151
G		06406	152
	4P2 • P441**EL+S12*PKS112	OFHOS	15.1
	PRESTS = AMINI (DMITS, PHAILIDHALS, HECH	CENCG	16.6
	AP2 = FRESTZ+C237+PANT/FINFT	05 405	14.6
	PSINFT & AMINI (AMXTZ, AMAXI (AMNTZ, APZ))	DEMOS	166
	SLINFT + FINFT+PSIN'T	05406	16.7
	PANT . PANT-SLINFT	01400	1.4
	AP3 + PRAT++ELAST3+PRSTT3		140
	PRESTS + AMINI (BMATS, AMAXI (BMNTS, BP3))		377
	AP3 + PREST3+C246+PANT/FMAST		37.0
	PSMAST = ANTN1(AMXT3, AMAX1(AMNT3, AP3))	0240%	3/1
	SLMAST + FIASI*PSHAST	DEMOIS	3/2
	PANT + 74FT-SLMAST	19. HUS	3/3
C		DEMOS	375
C	GROWING MALES	02005	373
С		JEN JG	376
	AP4 & PRAT **ELAST +*PRSIT+	DE 405	377
	PRESTA = AHIHI(RHXTA, AHAXI(RMNTA, 8P4))	05405	37.5
	APL . PRESTLOCZCA PANT/PHGT	36406	379
	PSHGT = AHIHI (AHXTE, AHAXI (AHHTE, APE))	DE HOS	390
	SL SHGT # PHGT+PSHST	2E#06	361
	PANT & PANT-SUSHGT	ne 406	395
C .		95405	393
ć	GROWING FEMALES	9E#0G	394
č		DEMOG	395
•	NPS & PRAT++CLASTS+PKSITS	96M0S	396
	PRESTS : ANTHI (GHITS, AMAXI (BHNTS, 895))	DEMOG	397
	APS & PRESTENCES +PANT/PEGT	06406	388
	PSECT + ANTHICAMITS, AMAXICAMNTS, APSI)	SENOS	119
	SLSFGT + PFGT+PSFGT	DEHOS	398
	PANT + PANT-SLSFGT	DENOG	391
£		DE HOG	392
ř	FERTILE FEMALES	DENOS	393
ž		DEMOS	136
U I	804 - D2470 051 45760 045776	DENOS	305
	ODECTS - ANTHIOMYTE, ANAXI (BUNTE, BPS))	2F 40G	396
	AD4 - 011016-014 +3447/8687	OF HOG	197
	DEFENT - ANTHI /ANYTE, ANAVI/ANNTE, ADELL	OFHOS	197
		DENOG	199
	3675-1 - 7,317-97531 MANT - DANT-CIFCUT	DEHOG	600
•	PAGE FRANCISCIENT	05 806	
C			
2	PRJUGING MALES		
C			403
	<pre>TP7 # P 4# 1 ** { L451/*P(511/</pre>		
	DDEST * # AMINI (BMXI7, AMAXI (BMNT7, BP7))		405
	AP7 = PRLST*+C711**ANT/PMPT	DENDG	• 3 6
	OSHPT : AHIHI(AHKI7, AMAX1(AMNT7, AP7))	DEMOG	437
	SLSHPT + PHP*+PSHPT	NENOG	408
C		DE NOG	40 9
C	MISCELLANEOUS	DENOG	410
Ċ,		DEMOG	411
-	SLSHFT # RHOUTZ*(1(OPLT2*PSHPT*PPHPTT)*DT)	DEMOG	517
	CLSHLT + CLSHPT+SLSHFT	DEHOG	413
	SLSHT + SLSHLT+SLSHGT	DEHOG	616
	SLSCCT + SLINFT+SLAAST -	DENOG	415
SLSPPT + SUSCOTASLECAT 06405 415 PSPPT + SLSPPT/PPPT DE 406 417 FEMSCT + SLSTCT+SOLDFT+C214+SLFERT+C215+SLSFGT SLSFT = SLSFPT+SOLDFT+SLSFGT DEHOG 418 OF H05 619 SLST + SLSFT+SLSHT SLSPTP + SLSPT/TTSLS 20105 420 DENOS 671 SLSPT + (SLSCCT+SOLDFT)+C224+SLSHPT+C225+SLSHFT+SLFEPT+C224+ DEHOG 422 SL&FGT+C227+CL5HGT+C22A DENOS 1 423 C C 424 DENOS SALES - HODEPN IFIT, LT. THONIGS TO 50 DEHOS 425 426 427 TE HOG C C TENOG 0E405 628 OLD FEMALES č 429 GH1 . PRATOPELACHIOPRESINI DENOG 433 PPESHL + ANINI (AMENI, AMARI (BANAL, AME)) DEHOS •31 ANS . PRESH1+C736*PANH/OLDFM NEHOG 432 PSFON = AMINI (AMX41, AMAX1 (AMN41, AM1)) OFHOS 433 SOLDEN = OLDEN+PSEDH PANN = PANN-SOLDEN DEHOG +35 DEHOS C C DEHOG 435 INFERTILE AND MASTITIS DEHOS 437 Ċ DEHOS 438 SHE . PRATOSLASHE PRSTHE 7E 40G 439 PRESHE + ANTHI (BHINE, AMAXI (AMNNE, BHE)) DEHOG 440 ANZ . PRESH2+C2:7+PANH/FINFH COH30 441 PSINFH • ANINI (AMTH2, AMAX1(AMHH2, AM2)) SLINFM • FINFM•PSINFM OE HOS 462 DENOG 443 PANN + PANH-SLINFN DE HOS 444 BHS . PRAT DE HOG 445 РИССИЗ • АНТИL (ВРТЧЗ, АНАХL(ВНИЧЗ, ВНЗ)) АНТ • РОГСЧЗС246*РАНЧ/ЕНАСЧ DENOG 446 DENOG 467 SUNASH - ANINI (ANYAY, ANAXI (ANNAS, AN3)) SLNASH - FNASH+PSHLSH DENOG ---OE HOG 469 PANN = PANH-SEMASH DEHOG 450 DEHOG 451 C DENOG Ċ GROWING MALES 452 20#30 C 453 RNG - PRAT**ELASH4*PESTH4 DENOG 454 455 PRESHS - ANINI (BEX46, ANAXI (MHH6, BH6)) ANG - PRESH662238 PANH/PHGH PSHGH - AMINI (AHX46, AMAXI (AMH46, AM6)) DENOS DEHOG 456 457 DEHOG TLENGH . PHEN-PCHEN DENOG 459 PANN . PANH-SLSHGH DEHOG 459 C DEHOG 468 GROWING FEMALES DEHOG 461 C Ĉ DEHOG 462 DENOG BNS . PRATITELASHSPRSTHS 463 PRESHS + ANTHI (BMX45, AMAXI(BMMM5, BM5)) ANS = PRESHS+C2CS+PANH/PEGH DE NOG 464 DENOG 465 PSFGH & AMINE (AMENS, AMAYE (AMANS, ANS)) DENOS SLSFGH + PFGH+PSF;+ DENOG 467 PANN . PANA-SLSFGA DEHOG 468 DE HOG 469 C FERTILE FEMALES 72 HOG 471 C DEHOG \$71 Ĉ DENOG 472 BH6 . PRAT .. ELASHS . PRSTMS PRESHS . ANTHI (BHING, AMAXL(BING, BH6)) OC HOG 473 ANG - PRFSH6+CZ10+PANH/FERH PSFERH - AMINI(ANX46, ANAX1(AMN46, AM6)) RLFER4 - FERH+PSFERM DENOG 474 DENOG 475 308 90 476

...

	PANA + PANA-SFLEDA	06405	677
c		DENOG	474
	PRODUCING MALES	DEMOG	679
ŕ		DEHITS	693
	8m7 = P04T0+FL45H70PK57H7	DENOG	681
	PHESHE & A STALL ON X47. AMAX1 (PHM47. BH7))	DENOG	482
	ANT . PAFS47+C211+78NH/PHP4	DENOG	693
	0 CMPH - ANTMI (ANTMI, AMANI (ANMN7, AN7))	DENOG	686
		DENOG	
r		DEHOS	496
ř	PTSCELLANEOUS	DENOG	447
ř		DENOS	615
•	\$1\$###_#_\$40;#2*(1,=(0\$L#2*P\$#P## P##}T#)*01)	NENOG	449
		DEHOG	690
		DEHOS	691
		DENOG	492
		DEHOG	643
		DEMOG	636
		DEHOG	695
		DENOG	496
		DENOS	637
	43 FRMCA - SISAIMA AN AFMAC216451 FRMAC216451 SFGM	05406	698
	Fride - Fride State Stat	DEHOG	639
	PERSU B PUTTO PERSUPPE SUPPERS	DENDG	\$10
	51 5 M = 2125 4 4 20 5 1 4 4 2 5 2 4 4 4	05 806	531
		25.405	902
	1313 # 14317137 8. 884 - 181 (871471) 1641 66334 461 640866236461 6466466 66662864	DENOG	511
	27204 = (///////////////////////////////////	OF HOG	515
		00000	RAE
	S) CONTINUE	DENOS	844
C		0000	510
C	TER PRODUCTION	OFMOG	
C		00000	500
		02 100	
	[] MT & {F*RT*(PAIA+1+(], *PAIA+1)*, 82*(], *PAIA+1)*, 10*, 8977*********		710
	1 GTZCZ VMATOTEJEZ (VALO, JI, JI, 4, TUMATA		711
	ANCT = ANT/CITE.2021C FLACINFERTS		716
	IF (MAL. LT. 1) 50 TH 330		713
			719
	04H & (FC3H* (PA11F4*(1,-PA14FH)*,820(1,-PA14FH)*,10*,07))*********************************		717
	2 FA2 J2 * FACH * TE BE XE (VAL6, . 31, . 31,		710
	OPCH = OHAV(CH232*250.*PFLAGH*FEPH)	DEHUG	71/
	TONTINUE	DEHUG	210
C		OE HOG	717
C	EXTRACTION RATIO AND TOTAL SUPPLY	DEHOG	729
C		DEHOG	921
	SUPCTA = SLSCCT+SLSHLT+SOLDFT+C212+(SLSCCH+SLSHLH+SOLDFN)+C213+	DINOG	525
	1 (TOTHST+ TOTHSH)+C214+(SLFERT+C212+SLFERH)+C215+(SLSFBT+	DEHOG	523
	• C212*\$L\$F34) +C216*(SL\$HGT+C212*\$L\$HGH)	02405	524
	SUPBH = SUPCTA/TGL	DEHOG	525
	TOPOPK - TOPUK1 *[#P(C217*T)	DEMOG	526
	SUPS = SUPCTA+TOPJ>K+TABLIE(VAL9+SHALL9+PIFF9+K9+AHOO(T+7+))	DEMOG	527
	TOPOP = TOPOPP+TOPOPK	DENOG	428
	EPS = SUPRITOPOP	DEMOG	529
	CRPT # FERT*UPLT*/TOPOPT-DRT	DEHOS	530
	ERT • (SLSFT+SLSHT)/TOPOPT	DENOG	531
	<i>« Ври в « « Рч• ЧРЦ и ? / Т ОРОРИ-П</i> ЕИ	DEHOG	435
	EFA = (SLSFH+SLS44)/TOPOPH	DENOG	533
	040 # (T0THST+T0THSH)/T0P0PP	DEHOG	534
	5000 · (F521+92L1>+F84+98LH2)/TUPOPR-ORR	MENOG	535
	FRR • (SLST+SLS+)/TOPOP=	DENOG	536
	3PP = (747+84M)/(PF#T+FFPH)	OCHOG	537

-

	511		
c		MENOS	552
C	CATTLE S"CTOR PRINTING	DEHOG	\$53
C		05406	556
	IF(IPRIAT.LT.I)RTTION	76405	555
		DEHOG	566
	PERIANA CIERCE CIERCE CIERCE CIERCE CONTRACTORES CONTRACT	05405	70/
	4 CICNUL CICL ANIMIAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAA	01-06	737
	1 36377143671 BATHTOTA, CI CONT, CI CHIT, CI CHIT, CI CHIT, CI COT, CIOTA, CIIDA, CIIDA,	02405	571
	PT117312, AD1, AD1, AD5, AD5, AD5, AD5, AD5, AD7, DF(CT, DF(CT))	00,400	\$71
	PRINTAL PRESTL. PRESTL. PPESTL. PRESTL. PRESTL PRESTL. PRESTL. PRE	05406	572
	1 PP#FT	DEHOG	573
	PPINT916, 4P1, 4P2, 4P3, 4P4, 4P5, 4P7, TOPOPK, TOPOP	05406	57.4
	PRINT919,PSFOT,PSINFT,PSHAST,PSHGT,PSFGT,PSFEKT,PSHPT,PSFPT	7E HOG	\$75
	PPINT327, ATASTT, CTAST, PGFTST, PCTABT, ATAFT, ATAFFT, ATAFTT, PATAFT,	06400	576
	1 Exn05	DEHOS	\$77
	PRINT926, CRT, EPPT, CRS, PFINFT, PFMAST, PIFNUT, TONAT, PAIT	DENOG	578
	PENT933, NFT, NT, 97, NDLT2, BPT2, CONNET, GNT, GNCT	01 405	7/ 7
	WEINT336, NMAT, DRT1, DRT1, DRY 2, DRLT2, DRGT, DRMT, DRT	05406	700
	IF (T.LT. THOMER, TUNN	- 12 HUG	703
	- REMINAGE FUEL ANT STARS STARS OF FUELOS STARS CONSTRATION FOR STARS	OF HOG	
	• • • • • • • • • • • • • • • • • • •	DENOS	596
	DRTMTAKZ, TOWAN, PAIN, SI SCON, SI SMFM, SI SMLN, SI SFN, SI SMM, SI SPN	DENOS	547
	PPINTA61. REGTL. PFOTL. RHGTL. RHPTL. RTDN. RAA	DEHOG	584
	DRINTAGS, POFGTT, OPFOTT, PPNGTT, PPYPTT, PPFGTH, PPFPTH, PPNGTH, PPNPTH	DEMOG	599
	PRINT466, NH1, NH2, 843, 9H4, 8H5, 9H6, 8H7, PPFCH, PPFCH	DEMOG	593
	PPINTATA, PRESML, PRES	DEMOS	571
	1 ррчрн	DEHOG	59?
	PRINT97+, AH1, AH2, AH3, AH4, AH5, AH6, AH7	02405	593
	PRINTA76, PSFON, PSINFN, PSMASN, PSMGN, PSFGN, PSFERN, PSMPN, PSFPN	75405	594
	PRINT9A?, ATABNT, CTA94, PGFT9H, PCT44H, PFIVEH, PFHLSH, PIFHUH	DEROG	797
			278
	WININGS, DRUT, URAL, URAN, URAN, URAN, ISIS, SIMSC, TOPOP	DENOG	577
	#\$\$11\$79;7**;94*;34;64*;81;81;81;81;7*;35;1***;95;1*****; #\$\$11\$99;7**;94*;34;64*;7*;	OFHOG	633
210	FORMATINGWINHTPUT OF SUBROUTINE OFHOG AT TIME.F8.2/148.9%.	DEHOG	614
	6 64 64 64 64 64 64 64 64 64 64 64 64 64	DENOG	625
	SHOLDET, 7X, WPFLT, AX, WPHGT, 8X, WPHIT, AX, GHTOPOPT/1H0, 9X,	DEMOG	636
	10(511.4.14))	DENOG	617
976	FORMATTINC, 91, ONSLSEGT, 61, 645LINFT, 61, 645LMAST, 61, 6HSLFEPT, 61,	DEHOG	533
	AHSLSFPT, AX, 6HSOLOFT, 6X, SHFEHSCT, 6X, 6HSLSHGT, 6X, 6HSLSHPT,	DEHOG	60 9
1	P 44,6HSLST/140,9X,10(E11.4,17))	DEHOG	510
918	PORMAT(1H3,9X, 6HSLSCCT, SX,6HSLSHFT, 6X,6HSLSHLT, 5X,	DENOG	611
1	SHSLSFT, 72, SHSLSMT, 72, SHSLSPT, 72, 6HSUPCTA, 62,	TENOG	612
1	2 4NSUP8,8X,54SUPRN/1N0,5X,9E12.4)	DENOG	
912	FORMAT(140,)2, 34701, 91, 34802, 91, 34802, 91, 34004, 91, 34004, 91, 34005, 91, 34006,	01406	014
1	44,3H907,211,3HPPF01,71,3HPPF0171H3,01,7164,124,2212,41	05406	017
414	FORMAT(1M7,44,5M742)11,04,04742312,04,04742313,04,04742314,04,	00000	510
		001406	A + A
	с	OFHOG	619
717	9x, 1WAP7, 21x, 6HT0P0PK, 5X, 5HT0P0P/1HC, 8X, 7E12, 4, 12X, 2E12, 4)	DEMOG	524
	FORMAT (1HC . 91, 5HPS 7T . 71, 6HPS 1 HFT. 61. 6HPSHAST. 61. 5HPSHGT. 71.	DEMOG	621
	SHPSFGT, 71, 5HPSFERT, 51, 54PSHPT, 71, 5HPSFPT/1H0, 91, 8(E11. 4,	DEMOG	627
	1x))	DENOG	623
922	FORMAT (1HO, SP, CHATERTT, 6X, SHCTABT, 7X, 6HPGFTBT, 6X, 6HPCTABT, 6X,	DEHOG	624
1	SHATAFT, 7x, 5HATAFOT, 6x, 6HATAFTT, 6x, 6HPATAFT, 6x, 5HEXPOS/	DEMOG	625
2	1H6, AX, 9F12. 4)	OEMOG	626
925	FORMAT (1HO, 9X, 3HEOT, 9X, 6HF RPT, 8X, 3HERS, 9X, 6HPFINFT, 6X, 6HPFMAST,	DEHOG	627
1	67,5HPIFHUT,51,5110HAT,71,6HPAHT/1H6,81,8E12.6)	DEMOG	528

NENDG 93) FORMAT(1HC,9X, 3WAFT,9X, 3W84T,9X, 3HART,9X,5HAPLT2,7X,4H8RT2,9X, 621 1 SHC7ANGT, KX, 3HQT, 4X, MAMCT/1HG, 4X, 4(E11, 4, 1H) 915 FORMAT(1H), 9X, HORST, 9X, 4HART1, 8X, SHORLT1, 7X, WORT2, 5X, SHORLT2, RENOG 530 DEHOG 631 7%, .HOPGT, 94, 4HORPT, 8%, 4HORT /1H), 9%, A(511. 4, 1%)) DEHOG 512 1 45. FORMATCINC, 94, HAFSH, AT, SHFTHEN, 71, SHEMISH, 71, LHFERM, 41, SHPEPH, 81, DEHOG 416 SHOLDEN, TX, . HPECH, AX, SHP 4GH . 8X, SHPHE4 . 8X, 6HTOPOPH/ DEHOG 615 146,98,1:(*11.4.14)) NEHOS 615 **958** FORMAT(1H1,9X,6H5L5FGH,6X,6H5L1HFH,5X,6H5LHASH,6X,5H5LFERH,6X, 1 6H5L5FP1,cX,5H5CDFH,6X,6HFE4SCH,6X,6H5L5HGH,6X,6H5LSHPH, DEMOG 437 0€#0G 539 15405 519 54, 6H5L54/142, 44, 16212.4) DE HOG 952 FORMAT (1H), 91, 3HT DNAP, 71, CHPANN, 81, 6HSLSCCH, 61, 6HSLSHFH, 61, 561 SHSLCHLH, FX, 54 SLSFH, 7X, 54 SLSHN, 74, 4HSLSFH/1HC, AX, 4E12. 4) 05406 661 361 FORMATIINE, 91, SHOFGTT, 71, SHREPTT, 71, SHRYGTT, 71, SHRMPTT, 71, WHRTON, DENOG 562 44, THRAB/1H?, 9X, 6(C11.4, 11)) DEMOG 543 1 05405 953 FORMAT (1HC, 9X, 5HP PESTT, 6X, 5HPPFFTT, 6X, 5HPPHGTT, 6X, 6HPPHPTT, 6X, 5+4 ANPOFATA, UX, 6H PPFPTH, OX, 6HPPMGTH, 6X, 6HPPMPT4/1HJ, 9X, 10436 565 25 405 546 4(E11.4.17)) 955 FORMAT (1H5 , 9X , 3H 141 , 9X , JHR#7 , 9Y , 3H843 , 9X , 3H844 , 9X , JH945 , 9X , JH446 , DEMOS 647 664 91, 34047, 711, 540PFG4, 71, 54PPFC4/14C, 81, 7E12. 4, 121, 2512. 4) DE HOS 1 973 FORMAT (1HS, 31, 6HPRES41, 67, 6HFRESH2, 61, 6HPPLSH3, 61, 6HPRESH4, 61, 25405 569 SHPRISHS, 6X, SHPRESH6, 6Y, OHPPESH7, 18X, SHPPHEN, 7X, SHPPHPH/ DE HOG 651 641 140, 54, 7:12.4, 128, 2:12.4) DEHOS 642 976 FORMAT (1H0, 9X, 3H1 1, 9X, 3HAH2, 9X, 3HAH3, 9X, 3HAH4, 9X, 3HA45, 9X, 3HAM6, DEHDS 653 DENCG 9x, 3HAH7/1H0, 9X, 7(E11.4, 1X)) 976 FODMAT (1H), 94, 5455504, 7X, 64251454, 6X, 64254454, 6X, 5425464, 7X, DEMOG 654 12435 635 SHOSFSH, 71, SHPSFERM, 61, SHPSHPA, 71, SHPSFFM/1H3, 91, DEHDG 656 9(E11.4,1X)) 457 981 FORMATCINS, 91, AMATASHT, 61, SHCTARH, 71, OHPGFTBH, 61, 6HPGTABH, 51, DENDS 201220 SHPFTNFF, SI, SHPFMACH, SI, SHPIFHUM/1HG, AL, 7E12.4) 559 918 FORMATE 1HJ, 9X, 3HBFH, 9X, 3HPHH, 9X, 3HR9H, 9X, 5H89L M2, 7X, 4H3RM2, 8X, 25H0G 559 DEMOG 65: SH"BANG4, 54, 3H JHH, 9X, LHOMCH/1HD, 9X, 8(E11.4,1X)) 392 FORNAT(1H), 94, 647964, 94, 640341, 9X, 540RL41, 7X, 640RP4, 8X, 640247, 8X, 21 HOG 561 DEMOG SHOPLH2, 74, SHDR4, 9X, SHER4, 9X, 4HERPH/1HJ, 8X, 9512.43 56? 1 936 FOPMAT (1H: , 98, 3H32P, 98, 3H0RR, 98, 3HERR, 98, 4 HERPE, 84, 4HETA4, 48 OF HOS 663 DE HOG CHATAF, SY, GHTSLS, SX, SHFE 15C, 7X, 6HTOPOPR/1HC, SX, 9E12. 41 664 1 ENTRY CATSET 25406 669 DEMOS 671 PHGT = 117433". PHPT = 943202. DEMOG 571 25 406 572 PFGT = 114303C. 673 PFPT = 2424306. DEMOG DEMOG 674 0LOFT = 427030. TOPOPT . PEGT+PEPT+OLDET+PHGT+PHPT NE HOG 475 07 406 676 PSFGT # .13 PSFPT = .317 02805 677 P\$461 + .1 NEHOG 675 DEHOG 679 •SHPT # .134 OSFGN + PSFPH + PSHGH + PSHPH + 3. DEMOG 583 SLSFGT + PSFGT+PFGT REMOG 681 SLHAST + .JOS+ PEPT DEHOG 612 SLINFT = 0EH0G 683 SLFERT # . ST 2+PFOT DENOG 684 SUSPPT + SUPERTASUAASTASUINET DEHOG 615 SOLOFT + .3+OLAFT DEMOG 695 SLSFT . SLSFGT+SLSFOT+SOLOFT 647 DEHOS SUSHOT . PENGT PAGT DEMOG 645 SLSHPT . PSHPT+PHPT DEHOG 699 SLSHT . PHONTZ+SLSHGT+SLSHPT DEHOG 693 SLSOT . ISLINFT+SLNAST+SOLOFT) *C224+SLSMPT*C225+SLFERT*C226+ DEMOG 671 SLSFGT ** 227+ SLSHGT *C224 DENOG 692 1 nRGT = ...41 25HOG 593 DENOG 0871 . .: 41 494

	NPLT1 + .15	05405	434
		OF HOG	696
		00000	6977
		01407	4.4.4
		05406	490
		02100	777
		01403	777
		01405	
		02405	112
		DENOG	
		UEHUG	700
	PGF101 • .05	TENOG	735
		DEHOS	736
	EXPART 0 3.	05 110 5	/3/
	EXPTS • 0.	9E=06	/ 38
	PATAFT J	DENDG	734
	PA TADY • C244	17 HUS	/10
	PENAST00126	DENDG	711
	PFINFY + 13	DENDG	712
	PIFNUT = 0.	DENOS	713
	DO 10 I+1,KGROF	DENOG	716
	RINTFT1(I) = PFGT/(OGROF+IOTF1)	DEHOS	715
	RINTFMI(I) = 0.	DENOG	716
10	CONTINUE	DENOG	717
	RFOUT1 = PFGT/0GPOF	OEHOG	718
	NO 12 I=1,KPR09F	DEMOS	719
	RINTFTZ(I) + PFPT/(OPRODF*IOTFZ)	DEMOG	720
	RINTF'42(I) = 0.	OEHOG	721
12	CONTINUE	DEMOG	722
	RFOUTZ = PFPT/0P900	OFMOG	723
	DO 14 I+1, KGRN#	05402	726
	RINTHTI(T) = P4GT/(OGRON*IDTM1)	DEHOG	725
	RINTAM1(I) = 0.	DEHOS	726
16	CONTINUE	DENOG	727
•	RHOUTS - PHGT/DGP04	DEHOG	729
	00 16 IN1-SPRON	DENOG	729
	PINTHT2(I) - PAPT/(APRODA-IOTH2)	DEHOG	730
	RINTHM2(T) = 0.	DEMOG	731
4.6	CONTINUE	OTHOG	732
••	RNOUT2 + PHPT/OPRO24	DEHOG	733
	PEGN . PEPH . OLDEN . PHON . PHPH . J.	nr.HOG	736
	SLSPCH + SLSTPH + SLSHGH + SLSHPH + SOLOPH + 8.	DEHOG	735
	SLSFN & SLSNN & SLAASH & SLINFN & SLFERM # 0.	DENOG	736
	SISCE + 0.	DENOG	737
		OEMOG	739
		DEHOG	739
		DEHING	7.0
		DEHOG	741
		OFHOG	762
		DENOS	763
		DENOG	766
		DENOG	7.5
		DENOG	766
	0KNC - 7**	DENOG	767
		DENOG	7.8
	GUT V V. German – Gerada – Bishim – Dsharm – Dsiar – Dsiar – D.	DENIG	769
	PUPIDA E FLIEDA E FARMAN E FRANKI E FRANKI E VA	05806	76.3
	PATAPA V 1.	OFHOS	78.4
	PROUNT = TRUTH = STOVAL = TRUTH = PHETH = PHETH = PHETH = A	DENOG	762
	REGIT & REVIN & REFIT & REFIN & ROUT & ANDIN & ROUT & ROUT & ANDIN & REFIN & W.	DE MOG	78.1
	PPFGIT = PPFPIT = PPRGIT = PPRFIT = 00	05806	7.7.0 78.L
	PPFGTH & PPFPTH & PPHGTH = PPHPTH = Q.	DE HOC	734
	RF TURN	-12 -706	123

• •

380

.

DEHOG 756

	SUMPOUTINE MORCHD	MODCRD	2
	COMMON /ACCOUN/ DELAT, AMLAT, ACLAT, VLANDT, VLDTXT, TAXCT, AOSUT,	ACCOUN	2
5	DOLAH, ANLAH, ACLAH, VLAHDA, VLDTXH, TAXCH, AGSUN,	ACCOUN	3
	DPLAY, DURU, FARMIC, DOTOUS, DCROT, RCSTDT, EXPLIV, NCFR,	ACCOUN	4
	ALDHI.TEC.TCEC.TRISLI.TLF.TLFP.CPLF.AINT.AINTLA	ACCOUN	5
	CRENT, DIR, C250, C253, C254, LT1, LT2, LT3, CBST0, CST0H,	€≓7	1
	EPCRPJ. MCROPJ. PCROPL. PFCROP.	ACCOUN	7
	EVINC J. YL DCU. YL DCL. YL DFC. DYL DCU. DYLDCL. PYL DFC	ACCOUN	
		ACCOUN	•
	TO A THE TOTAL A TOTAL STATEMENT OF APPT APPT APPTYLS PRIVLE IPPINT	C#5	1
		Č.»9	2
	COMMON VIANDY TOLITOLITALI, TTALUS, TTALUS, TALSFA.	LAND	2
	TREL TURDO, TURDO, TURDO, TURDO, TURDO, TIRONA, TIGURA TELRA	LAND	3
		LAND	4
		LAND	9
	A DEL ADRO ADRO AS CONCET, CPLPT, XDEL APTN ARE	LAND	6
		HODCRD	6
	DIMENSION TRAINING \mathcal{F}	HODCRD	7
	DATA LTI, LT, LTJ, $C_{1,2}$, $C_{2,2}$, C	HODCRO	
		HODCRD	9
	DATA TCJ, TCJ, TCJ, TCJ, TCJ, TCD TA / B.C. LJC, LJC, TCC TOCCOCC	HODCRD	10
	DATA HPTH, ETA / 2/ COV.	N00CR0	11
	TRNSLI = THISLISTICAJA7-BOULE-THAITALSTOLAS	NDDCRO	12
	TRNSL2 # TRNSL2#JT#TBUTI=BOUT27	MODCRD	13
	TRNSL3 • TRNSL3•DT•TBJET2=B00137	RCDCRO	14
	TRAIN1(3) + TRAIN1(3)+A7	HODCRO	15
	CALL POXCIAUXS. GCUTI, THAT SINCOUNT. HUCKING TIL TISUNTAL	HODCED	16
	LF(LT2,LT,1)G0 TO SC	HOOCRO	17
	CALL HOXC(HUUT1,BOUT2,THAIN2,HC)UT2,HOCT2,LT2,SUNTH2/	MODERD	1.
	GC TO 100	MODERO	
50	ROALS + ROALT	HODORD	24
100	CALL HOXC(BOUT2, BOUT3, THAIN3, NCOUN3, NOCT3, LT3, SUNIN3)		21
	AUX5 = (CHH-ARH3-AHIN1()., RLH1/(AHIN1(ARH1, ARH2)+AHH3)))=(CEC-		22
	1 (1 ADIN)		
	ALPH1 = C234+AH1H1(1.+C234, (1.+C234)+EKP(+C236+(PDH+C235)))	HUDCHD	
	CSINPH & CSINPH+RCSTOT	HUDCHD	
	CSIAPH = CSIAPH+RCSTOT	HUDCHD	67
	CSPAPH = CSPAPH+RCSTDT	HOUCHD	
	CSPEGH = CSPEGH+RCSTDT	HODCHO	
	CRSTG # CASTG+RCSTDT	HODCHD	20
	CSTHAP . (1 -CPLF) + CPLPT + CSTAPH	MDDCRO	
	CSPLFG + CPLF+CSPFGH	HDDCHO	
	CETGH - CPLF+CRST5+2,+C253+C254/C250	HODCRO	- 37
	IF (MAL. ED. 1. CR. MAL. ED. 3100 TO 100	HODCRD	25
	CSPLAP + (1.+CPLF)+(1.+CPLPT)+CSPAPH	HODCHO	22
	GO TO 110	HODCRD	34
1 6 8	CS1HNP + (1 + CPLF) + (1 + CPLPT) + CS1 4PH	MODCHD	- 29
4 V Ø 1 1 A	TEC . CSIMAP+CSIMAP+CSPLAP+CSPLFG+CSTON	HODCRO	36
		HODCRD	37
		HODCRO	- 30
		HODCRD	- 39
	CALL GRADULTC'S TCS TCS CROMAXATA ACRDT)	HODCRO	40
	CALL UNATAL CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR	HODCRD	41
	10000 - 107361/6/1 Corne - 1010/11/00/10/10/10/10/10/10/10/10/10/10	HODCRO	42
	UNEUL & AMATLIAUNI - DANG F DEF	HODCRO	43
	ARM1 B CHEDVICKIPTU	HODCRO	44
	ARMZ V NULHA IZ/CICCUMPTIT/	HODCRD	45
	APM3 • (SCFE, APIN)(AKT), ARTAJOICEURPITELTATION	HODCRO	44
	AUX7 . AUX7+LAUX6+AUX73+UTFAUEL	HODCEN	47
	δ ¥7 → 1A X7	KODCRO	44
	ExtR + C297+TPSL	NODCED	44
	IF(EXTH.LE.JOH. NCRDT.LE.8.100 TO 12C	MODERA	
	AS & AMINI(FIZ+EXTA/EXTH, 1,)+(AUX7+AUX7+ARINI(EIJPAUHDT/DUHDT)	400640	24

		1 1.))	MODCRD	21
		00 TO 125	MODCRD	52
	128		MODCRD	58
	125	IF (ANIN3 (ARM3, ARN2) + APH3, LE. 8, 190 TO 138	NODCRD	54
		ALVA & ARM3/CANTNICARMS, ARM2)+ARM3)	HOOCRD	- 55
			MOOCRD	- 56
	1 18		HODCRD	57
	114		MODCRD	50
			MODCRD	51
			NODCRD	60
		PRINTYON, TOUCH IT TOUSES TANKE STARME, ARMS, ARMS, BOUTS, BOUTS, BOUTS	NODCRO	- 41
		PRIMINGS, IRASLI, IRASLI, IRASLI, IRASLI, IRASLI, ACEDIA, ACEDIA, OCEDI	HODCRD	62
			NDDCRD	63
		PRIMITY 17, EX 4, A7, BUF7, BUR7, BU	HODCRO	
			HODCRD	65
			HODCRD	64
	• • •	NETURA Conversional de la destricte Marcha at Time (8.2)	HODCRO	47
	VQQ	FURNAL (SBRZOLIFOL OF SUBJOULTE DO SUB ALTONEL AT ALARMA ANA ANA	HODCRD	
	442	FORWAT (IND, VI, 6HIWASLI) OKT GHENDELT, SK, BHINHSSSSK, SHOULTSTMAR, SK	NODCRO	60
		I	HODCRD	76
		Z VELZ 47	HODCRD	71
	710	PORMAT(INU, VI, SHALPHI, / A, OHICC, VA, MAILECIGA, OHICHIGATA	HCDCRD	72
			HODCED	73
	412	I ORNAT (IND, VI, CHEITH, CA, ENANJIOA CHANDISSER AND CONTANT AND	HODCRO	74
		1	HODCRO	75
	420	FORMAT(ING, VI, 6HCSIAFA, 04, 6HCSIAFA, 6A, 6HCSIAFA, 6HC6HLFO, 6HC	HODCRO	74
			HODCRO	77
_			HODCED	74
č		INITIAL ANCOES FOR CREDIT	HODCRO	7.
č			NODCRO	80
C			HODCRD	
		ENTRY GROSEY	HODCRD	
			HODCRD	Ă.3
		TRNSLZ = 0.	MODERD	84
		TRNSLS .	HODCRO	Ä
		AUX3 = 0.	HODCRO	
		BOUTI - C.	MODERO	
		80012 - 0.	MODIES	
		80UT3 • 0,	MODERO	
		R1 • 0,	HODERD	
		A50 1.	HODERD	
		DO 400 101,6		74
		TRAIN1(I) = 0.	HUULKO	76
	480	CONTINUE	HUPCRD	43
		D0 410 1+1,1C	HODERD	
		TRAINJ(I)	HUDCHD	77
	410	CONTINUE	MONCHD	76
		TRAIN2(1)	HODCHD	
		TRA[H2(2) . 0.	HODCKD	78
		NCOUN1 • 0	RODCRO	
		NCOUNZ = 0	HODCHD	103
		NCOUNS . O	HODCRD	101
		SUMINI = 0.	HODCHO	102
		SUMIN2 • 0.	HODCHO	103
		SUMIN3 • 0.	HODCHO	104
		CRM . 0,	HOJCRD	107
		ARH1 = 0,	HODCRO	100
		ARH2 = 0,	HDDCRD	107
		ARM3 = 0,	MODCAD	104
		RLH1 • 0.	MODCRD	101
		NCFR = 0,	MODCRD	- 110
		AUX6 = 0,	NODCAD	- 911

· · · · ·

	HODCRO	117
AUX/ = V.	HDDCRD	113
	MODERO	114
ACRUTA . U.	HODCRO	115
PDR = 0,	MODCRO	114
CSINPH = 576.	#00C#0	117
CSIAPH = 57A.	MODERO	114
CSPAPH # 66J.		
CSPFQH = 593.		120
CB\$TQ = 0.5		124
CPLF . O.		121
CSTGH = CPLF+CBST1+2.+C233+C234/C25J		122
CSPLFG + CPLF+CSPFGH		123
CSINAP . (1, .CPLF).CPLPI.CSIAPH	400040	144
IF (MAL . FO. 1. OR . MAL . E9.3100 TO 500	MODCHD	10
CSPLAP & (1 CPLF) + (1 UPLPT) + CSPAPH	MODCHD	120
CSIMNP = 0.	MODCED	147
CO TO 510	MODCED	128
CCTMNP = (1 -CPLF)=(1.+CPLPT)+CS1'IPH	MODCRO	129
	MODC#0	120
TEC - CEINNDACEINADACEPLAPACEPLERACESTEN	HOOCRO	121
IEC = C214UnaC214Kbacatery and Fre acter	MODCRO	135
	#0DC#0	133
KETUKM	H00C#0	134
END		
	AUX7 = 0. ACRDT = 0. ACRDT = 0. CSIMPH = 576. CSIMPH = 576. CSIAPH = 576. CSPGH = 59č. CBSTQ = 6.5 CPLF = 0. CSTGH = CPLF+CBST1+2.+C253+C254/C25J CSPLF = 0. CSTGH = CPLF+CBST1+2.+C254/C25J CSPLF = 0. CSIMAP = (1_+CPLF)+CPLPI+CSIAPH IF(MAL,F0.1_0K.MAL,F0.3)00 TO 500 CSPLAP = (1.+CPLF)+(1.+CPLPT)+CSPAPH CSIMNP = 0. CSIMNP = 0. TEC = CSIMNP+CSIMAP+CSPLAP+CSPLFQ+CSTGH TCEC = TEC RETURN END	AUX7 = 0. MODCRD ACRDT = 0. MODCRD ACRDTA = 0. MODCRD VDR = 0. MODCRD CS1MPH = 576. MODCRD CS1APH = 576. MODCRD CS1APH = 576. MODCRD CS1APH = 576. MODCRD CSPAPH = 66J. MODCRD CSPAPH = 66J. MODCRD CSPGH = 59Č. MODCRD CSPTG = 6.5 MODCRD CSPLAP = 0. MODCRD CSIMAP = (1_+CPLF)*CPLPI*CSIAPH MODCRD If (MAL,F0.1_0K, MAL,F0.3)00 TO \$00 MODCRD CSPLAP = (1.*CPLF)*(1.*CPLPT)*CSPAPH MODCRD CSIMNP = 0. MODCRD CSIMNP = 0. MODCRD CSPLAP = 0. MODCRD </td

.

SUBFOUTINE AGACT	AGACC	2
CONFOR 24CC4002 DRLFT,4464T,4CL4T,7L490T,VLDTXT,T48CT,4GSNT,	ACCOUN	2
GOLAN, AVEAN, ACEAN, VEASDA, VEDTYR, TAXCH, AGSUN,	ACCOUN	5
IPLAV, DURU, FARMIC, DITOUS, PCRDT, RCSTAT, EXPLIV, NCFR,	ACCOUN	4
\$ <u>ALPH1, THC, TCEC, THIS, TLF, TLFP, CPLF, HINT, AINT, A</u>	ACCOUN	•
4 CPEET, D14, C25, C253, C254, LT3, LT7, C85TG, C5TGH.	r 🖬 7	1
LACHAT YAR UDAN DAY DE TEL DE COD	ACCOUNT	,
EVERCI, TEOCU, YEDCE, YERFS, DYEDCU, DYENCE, FYEDFC	4000UN	, P
MFAL ILEF	ACCOUN	9
COMPOU /CATTLE/ PEGT.FEWT.PEPT. 24 .T. PEPT. TOPOPT, TOPOPA,	CATTLE	2
1 PFCH, PFPH, PH H, PHP4, TOPOPH, VACAPL, AUXL12.	CATILE	3
SLEFRT, SLSCCT, SDLJFT, SLSMLT, SLSPT, SLSPTP, FEMSCT,	CATTLE	4
1 SLEEH4, SLSCCH, SOLDE4, SLSHLH, SLSPH, SUPA, FEHSCH.	CATTLE	5
A PPFGTT, MPFPTT, PPM3TT, PPNPTT, PPFGTM,	CATTLE	6
4 POPPTA, PONGTH, PONPTA, PESTT, REPT,	CATTLE	7
RUCTT, FROTT, AFGT4, RFPT4, RHGT4, RHPT4 ,	CATYLE	9
7 TTIGT, TDNAT, TONH, TONAP, TONAR, PATADT, 044, CH1 (PFLACT,	CATTLE	າ
1	CATTLE	10
COMMON: /CUNTHE/ T.DT.CUM.IRUN.BEGPRT.PRTCHG.PRTVLS.PHTVL2.IPRINT.	C+5	1
1 MAL, THUU, TOO	C=5	2
COMMON /LAND/ TAL.TTGL.TTGL.TTGL.J.TTGLU2.TGLSF3,	LAND	2
1 [PSL, "L=33, TL=931, "L=932, TL=933, "TGL4, "CL4,	LAND	3
!	C#10	4
\$ x21,R1,W2,PR2,PDR,C1W,RLM,PLM1,PLCRJ,AUX1,AUX3,	14 ML	5
A APRI, ARM2, ARM3, A5, C9, CAT, CPLPT, XDEL, RPTN, GRE	CAND	6
DIMENSION HINDRA(12,4),CAINCA(12,4),CLINCA(12,4),ECADEA(12,4),	AGACC	1
1 EtxCA(17,4),Ev.D*A(12,4),S1NCRA(12,4),TM1NCA(12,4)	AGACC	9
DIMENSION BINCR(12),CAINCR(12),CAPDIP(12),CLINCR(12),ECASER(12),	AGACC	•
1 ELOAN(17),ETCES(12),SINSR(12),THINCR(12),TXCP(12),	AGACC	10
2 VL*ITP(12),FOL4(4),VAL36(4)	AGACC	- 11
DATA C222.C223.C248,C255 / .47.4., .5, .4 /	AGACC	12
JATA C253, C252, C253, C254, C261, C262, C263/1.2, .5, 40., 3, 25, .75, .3/	AGACC	13
JATA C264,0267,0248,0264,0270,0271/.03030001, 5+1. /	AGAÇC	14
JATA C274,C275,C276 /3+.15 /	AGACC	15
DATA C277,C274,C279 / 1., 1., 3.1 /	AGACC	16
DATA PELS,DELA,"EL17,PEL13,DEL14 / 5+5+ /	AGACC	17
DATA DEL19, DEL21, DEL21, DEL22 / 4+5. /	AGACC	18
BATA DIFF36, SMALL36, K36 / 1., 7., 3 /	AGACC	1.
DATA FULM / 110., 170., 190., 170. /	AGACC	20
HATA VAL36 / 4.7, 4.55, 4.6, 4.6 /	AGACC	21
DATA SVAC,SVA.7;SVALH,AGSUBH,STGL4 / 3+1., 2+1. /	AGACC	22
DATA PINT,RINTL'IPH,DIR,RCST / .13, .14, 12, .15, .12 /	AGACC	23
DATA SF*AX,SHTAX,FGLT.TAKLHD / 101., 50., 35., .0042 /	AGACC	24
DATA VLNCU1, VLDCL1, VLDFC1 / 1.11, 1.56, 8.3 /	AGACC	25
DATA DVLDCU.DVLCCL.DVLDFC/ 1.65, 2.9, 10. /	AGACC	26
DATA PHM.EXLMIN / 1., 207000005. /	AGACC	27
DATA SIMCRA / 1., 1., 1.15, 1.45,1.65, 2+1.9, 5+2.,	AGACC	28
2-1., 1.17, 1.7, 2., 2.75, 2.45, 5+2.6,	AGACC	21
2 2 2 1 . , 1 . 5 , 2 . 7 , 3 . 2 , 3 . 9 , 6 . 4 , ,	AGACC	33
2+1., 1.5, 2.7, 3.2, 3.0, 6+4, /	AGACC	- 21
DATA THINCA / 1.1. 1.2, 2+1.25, 3+1.35, 5+1.4,	AGACC	32
1 1.2, 1.22, 1.29, 1.36, 1.43, 1.46, 1.53, 501,57,	AGACC	21
1.3, 1.37, 1.57, 1.9, 2., 2.2, 6-2.22,	AGACC	34
1 1.37, 1.39, 1.6, 1.82, 2.2, 7+2.4 /	AGACC	35
JATA ETACA / 1.1, 1.10, 1.17, 1.21, 1.20, 1.27, 1.28, 501,3,	AGACC	36
1 1.1. 1.25. 1.3. 1.33. 1.35. 1.36. 1.37. 5.1.4.	AGACC	37
7 1.1, 1.35, 1.6, 2., 2.4, 2.55, 6,2.6,	AGACC	38
1.1. 1.75, 1.6, 2., 2.4, 2.55, 6+2.6 /	AGACC	3•
DATA EVLDTA / 3-1., 3-1.2, 6-1.3,	AGACC	40
1 3-1., 3-1-25, 0-1.37,	AGACC	41
/ 3+1., 3+1.3, 0+1.4,	AGACC	42

· · · · · · · · · · · · · · · · · · ·	AGACC	41
	10100	
1., 1.0, 2.3, 2.8, 3.3.	AGACC	
1.1, 7., 7.8, 3., 8.3.4,	AGACC	46
1.1, 1.9, 2.7, 2.7, 8+3,3 /	AGACC	47
DATA CAINCA / 1.17, 1.27, 1.42, 1.42, 3+1.57, 5+1.67,	AGACC	48
1 1.17. 1.32, 2+1.47, 3+1.62, 5+1.77,	AGACC	49
2 1.2.1.47.1.97.2.47.3.47.3.47.4.4.4.27.	AGACC	50
	AGACC	
UATA CLINCA / 2+1., 2+1.23, 3+1.4, 3+1.3,	AGALL	
1 2+1., 2+1.23, 3+1.4, 3+1.3,	AGACC	23
2 1., 1.3, 1.6, 1.6, 2., 7+2.5,	AGACC	54
· 1., 1.2. 1.4. 1.6. 1.9. 7.2.2 /	AGACC	- 55
DATA BINCRA / 0., . (5, .15, C., .15, 2+3., .1, 4+0.,	AGACC	56
1	AGACC	57
	AGACC	. ŠA
	AGACC	- ii
	46466	
	AUALL	
AFTXIC + AFAXIC+(AFIX41+AFIX1C)+D1XDEF72	AUACC	• • •
IF(1,LT.7.)50 TO 15	AGACC	• Z •
PKGH • TABEXE(VAL76,S4ALL36,D1FF36,K36,T)	AGACC	63
*AXCT1 # (P~GT+C222+P4P1)+C223+PKJR	AGACC	
TAXCT2 = FEHSCT+C277+SFTAR+SLSHLT+C278+SHTAX	AGACC	65
15 CONTINUE	AGACC	66
	AGACC	Å7
1 RACID - CZ77 10777	AGACC	
TARCI E TARCIZETARCIZ		
	AUACC	
CTLBLT + CTLBLT+RCSTDT	AGACC	
VETRAC = VETRAC+PCSTDT	AGACC	71
VETRAD + VETRAC+(1.+(PATADT-C244)/(1.+C244))	AGACC	72
CSTANTL + CSTANTL+(CSTANT+CSTANTL)+DT/DEL20	AGACC	73
CSTANT + CT, BLT+YFTPAD	AGACC	74
CINDTI A CINCTIACCINDTEC. HDTLIADT/DELST	AGACC	79
	AGACC	74
OPCLAT & TOPOPT+CSTANI	AGALL	
OPCLAT + TCLR+CLXDT+C207	AGACC	/•
YAT # SLSPT+PAP	AGACC	
Addelet a tada	AGACC	80
AGSUT . AGSJUT+TTGLA	AGACC	8 1
CAPCET . FOLT+TTG: P+C2AB	AGACC	82
	AGACC	
	48400	
ASANI # YZA AINELENE CANEN	AGALL	
VLTXHT • VLTTTV(TTGL=0C271)	AGACC	
ACLAT + OPCLAT+CPCLAT+TAXET+GAPDET+VLOTET	AGACC	
AHLAT . TAT+YHT+ACSUT	AGACC	66
FARILT - ARLAT-ACLAT	AGACC	
VLANDY & AMAX1(SVALT+TTULB, FARILT/RINT)	AGACC	
VINDUT # VIA507/(1+6L2+C269)	AGACC	
	AGACC	
	AGACC	•1
EURI BITENTOPTLALIOLITETTAI		
TCLATL + TC.ATL+(TCLAT+TCLATL)+DT/DEL&V	AUACC	
TCLAT • ACLAT/*TGLH	AGACC	76
TPLAT = EPAP+FSLSPT+(PR4+EG4T)/TTJL4+AGSUBT	AGACC	•7
DHLAT . (TRLAT-TCLATL)+1CD18	AGACC	98
TETT.LT.THOUSED TO 204	AGACC	
CAPT - TI CAC 253-C 244/225	AGACC	104
UNT 1 1677. 620707774675 1728700		
IF ESTORIAL T. CAPISIO IS ADD		111
RPCAPI • C.	AGAUC	115
RLINPS • G.	AGACC	103

•

	úð TO 118	AGACC	184
1.8.8	HPCAPT . (CAP1-STORG) + CHSTG+2./C252	AGACC	115
•••	STORG . STORG-DT+ (CAPT-STORG)/C252	AGACC	170
	BI 1495	AGACC	117
110		AGACC	111
		AGACC	111
		AGACC	110
	ALACTA & ALACTATA AND AND A	AGACC	111
		AGACE	112
		C - 7	•••;
	ALDAN & AMI ICULATE EREDIE	ACACC	
	ALREP + THESE SZETS		
	TAXCH1 = (P+GP+C222+P+PH)+C223+PH iR	AGALL	147
	TAXCH2 + FEHSCH+C277+SFIAX+SLSHLH+C278+SHTAX	AGACU	110
	TAXCH3 = C279+TCP7PH	AGACC	117
	TAXCM B TAXCH1+TATCM2+TATCM3	AGACC	110
	CSTANH & CS*A+H+L^STOT	AGACC	114
	CUNDH = CUNDH+RCSTOT	AGACC	120
	TPCLAH = TCPOPH+CTIN4	AGACC	151
	CSRFGH + CSFFCH+PASIDT	AGACC	125
	CSHARV + CS-14FV++55121	AGACC	123
	OPELNH + TLHCT+CL+CH+(1,-CPLF)+TLFP+CERFGH+TLF+C8HARV+C253+C294	AGACC	:24
	YAN & SLSPHAPAP	AGACC	125
	Чин в Синеран	AGACC	126
	AGSUM	ACALO	127
	CAPDEN A TINCTAELANDALI	ACACT	128
		AGACT	124
		AGACT	139
		454CT	131
		33420	132
	ACLANIA	AGACO	. 33
	ACLAND - CONTRACTOR AND A TATAN BEPATTERAN PHILAPRYANDATER // 1. ANDEL)	ALACC	- 34
		A 5.4 5 5	. 19
		AGACC	1 14
		A 1 A 7 7	
			1.34
	ALTER - THII (ALTER (1430 - LHAR) CERTAI - VELTAI - VELTA	A 6 4 6 6	134
	AFNDHA • AFTY: HX(15HOC\$14PF)	AUALU	14-
	Dn 120 1+1,111	AGACC	141
	ELOAN(1) • *CEC+(1,-*****)/LT1	AGAÇU	146
120	CONTINUF	AGACC	143
	ICEC = 3.+x^EL	AGACC	144
	DO 125 1+1,1CFC	AGACC	145
	ETCEC(1) + TC+C/10EC	AGACC	144
125	CONTINUE	AGACC	147
	TCREC = TCEC+(1,+#PTN)	AGACC	148
	00 130 1+1,271	AGACC	149
	EDBSER(1) . TCRFC+FINTL+1/LT1	AGACC	158
133	CONTINUE	AGACC	151
• • •	1.2 • . 11+. 12	AGACC	152
	1F(LT2.LT.1)6C TC 135	AGACC	153
		AGACC	1 5 4
	00132 1012 112	AGACC	199
	FORSER(1) = FENSER(11)	AGACC	134
132	CONTINUE	AGACC	197
1 14		AGACC	
100		AGACC	
	ited = ites=1		17*
	- Virta		100
	ተምጠና ምር የርዛሮር/ርዘታ 156 - 4 ዓ. የ. የ. የ.	898CC	101
	JO 130 1013,1L3	AGACC	145
		AGACC	: • 3
	FCW2FH(I) • 447VAA+*CNAMPYT	AGACC	104

	387		
138	CONTINUE	AGACC	165
	IF(T.LQ.THON)PRINT979	AGACC	166
	DRLAM = 0.	AGACC	167
	DO 150 I=1, IPH	AGACC	166
	ETXC = TAXCTL+TXCP(1)	AGACC	169
	EVENTA = VETATEVETATE()) ECADEN = EQUITATEVETATE()	AGACC	170
	ECREC = COLITCORTATION FILM	AGACC	172
	EDCLNM = CLMDTL+CLINCR(I)	AGACC	173
	TCLAN . EDHSEH(1)+ETCEC(1)+EOPCLM+EUCLNH+ETXC+EVLDTX+ECADEM	AGACC	174
	TRLAM = EPAP+(SINCR(I)+ESLSPT+BINCR(I)+AUXL12)+THINCR(I)+PRH+EQHT/	AGACC	175
1	TTGI H+EI DAN(1)+AGSUBM	AGACC	176
	URLAM = DRLAM+(TRLAM+TCLAM)/(1.+DIR)++I	AGACC	177
	IF(I,NL,IND))DU ID IDU Udintumo enuclorin etacarin.comun etac evidty.comen.	AGACC	1/0
1	SINCE(1), THINCH(1), ELOAN(1), TRLAN, TOLAH	AGACC	180
150	CONTINUE	AGACC	181
200	CONTINUE	AGACC	182
	DRLAV ={DRLAT=TTGLR+DPLAM={TLMOD+TR5L}}/TGLR	AGACC	183
	ASVLNC = VLAHIDC+C25"	AGACC	184
	VLDTXC = ASVLIC+TAXLND	AGACC	105
	ASVHC B ASVINCTLC	AGACC	100
	TOCON - EDCODULEVINCU	AGACC	107
	CSTHOU = CSTHOUSERCSTDT	AGACC	189
	CSTHCL = CSTHCL+RCSTDT	AGACC	190
	CSTHEC = CSTHEC+RCSTDT	AGACC	191
	CSTHU = CSTHCU+(1,+C274+(YLDCU-YLDCU])/(DYLDCU+YLDCU1))	AGACC	192
	CSTHL = CSTHCL+(1,+C275+(YLDCL=YLUCL1)/(DYLDCL=YLDCL1))	AGACC	193
	GSTHF = CSTHFC+(1,+C276+(YLDFC+YLDFC1)/(DYLDFC+YLDFC1))	AGACC	194
	TCCRUL # TCCRUL+(TCCRU4+CCRUL)+UT/DEL22	AGACC	197
	DCRU = CSIAU AVEIXAC DCRU = CIRCAU-ICCAU I+ICDIR	AGACC	197
	ARCRU . TI CHU+YI DCU+PCRUPU	AGACC	198
	ARCRL . TLCPL+YEDCL+PCRUPL	AGACC	199
	ARFC # TLFC+YLDFC+PFCR0M	AGACC	200
	ACCRU = TEGHU+C5THU	AGACC	201
	ACCRL = TLCHL+CSTHL	AGACC	202
	ACFC = TLFC+CNTHF FADIOC - ADODU-ADOU-ACCRU-ACCRU	AGACC	203
	FARILL B ANTHDOARTHLEAUGRUFAUGRU	AGACC	204
	FARMICA - FARMICANDINFARMIC	AGACC	203
	FAPHIC . FAHICC+FAHIF THVLUTHC	AGACC	207
	VLANDE = AMAX1(VLVDHT+TLC, FARM1C/RINT)	CH2	7
	VLNDHF = VLANDC/TLC	AGACC	209
	GINC = VAT+VHT+VAH+VHM	AGACC	210
	ALPH2 = C261+(C262+C261)+EXP(+C263+C264+AMAX1(GING-EXLMIN, 01))	AGACC	211
	EXPLIV # AMAXICALPHIZ#JINU, EXLMINJ	AGACT	712
	19 [] P [] []	AGACC	213
	PRINTERS IN AV. HERE. TRENJ. TECRIL ATCORESARCRUSACCRUSESTHUS CSTHES.	AGACC	214
1		AGACC	216
-	PRINTPOR, FAMMIC, FAHMICA, VLANDC, ASVLNC, VLDTXC, ARCRL, ACCRL, CSTHL,	AGACC	217
1	(STHCL, ALPH2	AGACC	218
	PRINT912,FAF1CC.FAH1FC,VLUDHC.ASVHC,VLTXHC.ARFC.ACFC.CSTMF.CSTHFC.	AGACC	219
1	FXPLIV	AGACC	220
	PRINT920, DHLAT, THLAT, TCLATL, TCLAT, ESLSPT, EPAP, BOHT, FAHILT	AGACC	221
	PRINTAZA,AHLAT,ACLAT,FAT,THIJAGGUIJCPCLAT,UPLLNTJTARUT,CBPUTT Betutgje (strint'octanti.oti Rit, ufterd, uster, oi NDT, oi NDT, ui runt.	AGACC	772
•	ERIMETEN,COTANI,COTENTESCENDER,TEINAUSTEINAUSEENUISUUNUIESTEANUIS AUVENT,VENTXT	AGACC	224
•	PRINT932. PKNH, TAXCT1. TAXCT2. TAXCT3. TAXCHT. VLNDWT. ABVHT. VLTXHT	AGACC	225
	n e fait name tha an an the ann an the ann ann an ann an ann an ann ann ann a		

	1247 LT THURSHETHER	AGACC	554
	THE TALL AND AN CAST AND AND AND AND AND AND AND AND AND ACLAND. CAPT. PRCAPT. STORG.	AGACC	227
		AGACC	228
	1 HUNDARA HANNA AN SING AN DAMA AND ANA AN DEPAN BEPAN DETOUS AT LMT	AGACC	229
	PINTOSO, VAN, VIN, COMPACTANA COMPANY AND SUARY VIANDAASVLNNA	AGACC	231
	PWINTW62, GPCLAN, IPCLN , ISTAN, ICCNDITION OF COMANY TELE OF THE T	10100	231
	1 TENTER STATES TATES TATES TATES TATES AND TA	AGACC	232
	PRINT760, TAICH, TAICHI, THE CASTANC STANDIC CONDENSES OF THE CONTRACT OF THE CASTANCE STANDICS	AGACC	233
		AGACC	234
	RETURN	AGACC	235
	DUNAT (JANZIUTICT C. JOSCOVINE ANDERUS ANTECOULARS SHTECOUL	AGACC	236
764	JORNAT(INJ, SI, SUDELT, SI SCHOLDER, SI SUCH ST. AMCSTHEU, AX. 4HGINE/	AGACC	237
		AGACC	238
	2 143,41,1,1,2,12,47	AGACC	239
401	S FORMAT(ING, SI, SHIP ARTIC, SE, SHI ARTICL, SE, SHICE SELECCION STATE	AGACC	249
		AGACC	241
	>	AGACC	242
	() WHAT (1) C, VI, 6H) IRIC, 5H, 6H, HIT, 7, 14, 0 C, CHUCTON, 7H, AHCSTHE, AL	AGACC	243
		AGACC	244
• • •	//////////////////////////////////////	ACACC	245
# 20] FUMMAT(1M+, VX, 2000, AT, 77, 2010, AT, AUFONT, AV, AUFADITT/183, AT, AF12, 4)	AGACC	246
	A PALMATALINA AM ALAD AT TE GMACLAT. TY, SUVAT, ON SUVAT. OILSMACS	ATACT	247
	C COMA (1HC, VI,) A C C C C C C C C C C C C C C C C C C	. C	248
	TADMATALWA AN ANGSTANT AF, THOSTANTI AN ANGTI BI TATAANVETRAD.ATA	1.1.1	249
749	COMMALIAND AND AND AND AND AND AND AND AND AND	12422	250
		AGACT	251
	ε · · · · · · · · · · · · · · · · · · ·	AUACO	252
436	C DR A (100, 74, 0) C ON ON ONE (100, 000 T, 100, 000, 0	ACADO	293
		ASACO	254
		ASACC	255
474	AMADIANI'AT ANACLANI.AT ANGLANI.AT ANCAPT.AT.ANAPCAPT.AT.SHSTORG.78.	AGACC	256
		ASACC	257
	а — — — — — — — — — — — — — — — — — — —	AGACC	258
770	A ANALOEP TH ANA LEPALAS. ANDRED BELANSHAL THE THE ASS OF THE AND	AGACC	250
	E COMMATING ON ANOPELAT. SE. ANOPELAT. N. ANORTANN. AR. SHCLHON. 78.	AGACC	200
700	AUCSOFGU AF AUCSMARY, 184, AMVLANDA.AH.AHASVLAMA, AMVLDTEM/	AGACC	261
	5 1h ⁺ . AT. Ab 12 4.121.3812.4)	AGACC	262
	C CONNATIONS OF BUTATON 78. ANTAICHI. AS ANTAICH2. AT ANTAICH3. AT	AGACC	263
	ANTARCTICAR PHEAPDEN. AX. BHVL TATL. AX. ANVL NDHM. 6X, SHASVNM. 7K.	AGACC	244
	2 Auvi Tanu// un. 81.10812.4)	AGACC	245
	E CORMATINA AND THEFE SE SE STRICEC. AN AMEL CLMASK AMEDCLMMASKA	AGACC	244
• * *	ANTYT. 7. ANTYLETE.SE.ANECADEN.SE.ANGINER.AE.ANTHINEP.SE.	AGACC	267
	2 Null nah. AT. ANTAL AN. AT. BUTCLAN1	AGACC	268
		AGACC	269
401	р сулскіх котралет ала та Былра д. «Крл	AGACC	270
•	Ballon - Arthur - Art	AGACC	271
	CFUPS	AGACC	272
		AGACC	273
•	RERTOT A LATTAUCST	AGACC	276
		AGACC	275
		AGACC	274
	689MCL & 1867	AGACC	577
	CETHEC - 191	AGACC	27A
	5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	AGACC	>7
		AGACC	180
		AGACC	244
	TERE TERES		243
	TLDTU - TLDTUI Taanu - 655 (0), 11 DEACORGATATI NO (TLC	10400	741
	ICCHU B CBIHCUPVCANHEVCANDY/HALAN/ICC		143 144
	FARMIC		202
	FARMICA + .	ABACC	200

C		AGACC	287
Ċ	TPANITIONAL LIVESTOCK	AGACC	288
ċ	• • • • • • • • • • • • • • • • • • • •	AGACC	281
•	PKGP • C.	AGACC	290
	TAXCT1 = 0.	AGACC	291
	TAXCT2 # C.	AGACC	292
	TAXCHT = C279+TOPOPT/TTULM	AGACC	243
	TAXCTL + TAXCHT	AGACC	294
	CTUDUT # 14.4	AGACC	295
	VETHAC # 6.11	AGACC	796
	CSTANT + CTIBLT+VFTRAC	AGACC	297
	CSTANTL & CSTANT	AGACC	298
	CLMDT # 35.4	AGACC	299
	CINDTL + CINDT	AGACC	300
	VIDITI - STATIC	AGACC	361
		AGACC	302
		AGACC	303
	TOLAT . TOPOPTA STANT/TTOLROCLADTOCZATOTAXCHTOBOLTOCZEBOVLTXHT	AGACC	304
		AGACC	125
		AGACC	306
		AGACC	307
		AGACC	308
		AGACC	309
~	V(2)(B: •	AGACC	310
		AGACC	111
		AGACC	112
C		AGACC	113
		AGACC	314
		AGACC	119
		AGACC	316
		AGACC	117
	STONG . C.	641	
		AGACC	114
		AGACC	110
	CSRFGH # 372.		320
	CSTAN + 56	AGACC	360
	CLNDA - 150.	AGACC	361
	VLANDY . C.	AGACC	222
	ALDIX + C.	AUALL	363
	AXCH # J.		
	YAM • 0.		364
	тим е ;,	AUALL	367
	ADLAM D C.	AGACC	320
	ACLAM . G.	AUACC	321
	DRLAM # C.	AGACC	360
	TCDIP • C.	AGACC	344
	DO 1027 1+1,1PM	AGACC	220
	ELDAN()) = ç.	AGACC	721
	ETCEC(1) + J.	AGACC	332
	EDUSEP(1) • C.	AGACC	323
	*CD18 • *CD16+1+/(1++514)+•I	AGACC	334
1020	CONTINUE	AGACC	332
	IF (MAL, FO, C) PETUPH	AUACC	220
	DO 106C 1=1.1PH	AGACC	337
	SINCH(1) = SINCHA(1,MAL)	AGACC	220
	THINCP(I) + THINCA(I, MAL)	AGACC	330
	TECP(1) + ETECA(1,MAL)	AGACC	340
	VLTXTP(1) + EVLDTA(1,MAL)	AGACC	341
	CAPOTH(1) + ECAPEA(1.MAL)	AGACC	342
	CAINCH(I) + CAI'CA(I, "AL)	AGACC	343
	CLINCE(I) = CLINCA(I, MAL)	AGAÇC	344
	BINCR(I) + HINCHA(I,MAL)	AGACC	345

1868 CONTINUE Return End	
--------------------------------	--

AGACC	346
AGAÇC	347
AGACC	348

	SUBPOUT THE MODEAT		,
	COMMON /ACCOUNT DRIAT.AFLAT.ACIAT.VLANDT.VLATTT.TAXCT.AGSUT.	ACCOUN	;
	1 DRI AN AN LAN ACI AN YLANDA VI DTXM, TAXCH, AGSUN,	ACCOUN	5
•	2 DRLAY, DCRU, FARMIC, DBTOUS, DCRDT, RCSTDT, EXPLIY, NCFR,	ACCOUN	4
	ALPHS. TEC. TCEC. TRASLS. TLF. TLFP. CPLF. RINT.RINTL.	ACCOUN	5
	4 CREDT, DIR, C250, C253, C254, LT1, LT2, LT3, CBSTG, CSTGH,	C#7	i i
	5 EPCRPJ. PCBOPU. PCROPL. PFCROP.	ACCOUN	7
	EYLDCJ, YLDCU, YLDCL, YLDFC, DYLDCU, DYLDCL, DYLDFC	ACCOUN	
	REAL NCFR	ACCOUN	•
	COMMON /CATTLE/ PFGT,FEWT,PFPT,PMGT,PWPT,TOPOPT,TOPOPR,	CATTLE	5
:	1 PEGH, PEPH, PHOH, PHPH, TOPOPH, VACAPL, AWXL12.	CATTLE	3
i	2 SLFERT, SLSCCT, SOLOFT, SLSHLT, SLSPT, SLSPTP, FGHSCT,	CATTLE	- 4
	\$ SLFERM, XLBCCM, SOLDFM, SLSMLM, SLSPM, BUPB, FEMSCM,	CATTLE	- 5
	● PPFGTT,PPFPTT,PPHGTT,PP+PTT,PPFGTH,	CATTLE	•
	R PPFPTH, MPHGTH, PPHPTH, RFSTT, RFPTT,	CATTLE	7
(A RHCTT, RMPTT, RFGTH, RFPTH, RHQTH, RHPTH,	CATTLE	
	7 TONT, TONAT, TONH, TONMP, TONAH, PATADT, OHM, OHT, PFLAGT,	CATTLE	
1	4 ΥΗΔΤ, ΡΑ, ΡΑΡ, ΡΑΡΡ, ΡΑΑΤ, ΕΧΡΙ, UNEXPL, GT. 02, G214, C244	CATTLE	10
	CUNNON /CONTHLY T.DT.DLH. INUN. BEGERT. PHICHG. PPTALI, PHIALS, IPHIAL	0.45	
1	HAL, TACD, TOO		
	COMMON /LAND/ TGL, TTGL, TTGL, TTGL, TTGL, TTGL, TGL,		
4			Š
		LAND	á
		HODRAT	,
	DATA = 53 = 53 = 53 = 53 = 53 = 53 = 53 = 5	HODRAT	6
	DATA 160, 161, 162, 166, FXMAK / 5 7 11 131. 500. /	HODRAT	•
	GRT . TIGLE/TOPOPT	HODRAT	10
	IF (DRLAV.GE.DCRI)GO TO 200	HODRAT	- 11
	XTLU . TGLUIP	HODRAŤ	12
	GO TO 216	HODRAT	13
200	XTLU = TLCRIP	MODRAT	10
210	LF(T.GT.CT)GC TO 220	MODRAY	17
	CL1 = PLCRU+ARS(DCRU)/(#TLU+(DCRU-DRLAV))	HOURAT	10
	PRINT920, CL1	HUDRAT	1/
920	FORHAT(5+GCL1+,C11,4)	HODRAT	
		MODEAT	2
220	RECRUE CLISTIUS (D. R. DERGERV// ABS(DCRU/	HODRAT	21
520	AUXI = ANINI(HLCHU, 0.)	HODRAT	22
	$AUX = A^{A}X (A C A C A C A C A C A C A C A C A C A $	HODRAT	23
		HODRAT	24
		HODRAT	29
	R2 = H2+(XR1+P2)+FT/XDEL	MODRAT	25
	R1 . R1+(CRH+R1)+FT/XDEL	HODRAT	27
	XR1 • R1•45	MODRAT	20
	PDR . (DRLAH-DRLAT)/ABS(DRLAT)	HODRAT	21
	E3 # E31-E32+FXP(+E33+ETCAM)	HODRAT	20
	PR1 # AMAX1(E7+(1FXP(-E8+(PDR+E9))), 8+)	HODRAY	
	CALL GHAPH(TEC,TE1,TE2,TEF,EXMAX,T,EXT1)	HODRAT	32
	RLMPI = E3+PF1+FXT1	MODRAT	31
	RLMD1 • PR1+TTGL+TLMUD/TGL0	MODRAT	31
	RLHI = RLHPI+RLHDI	MODRAT	
	PR2 & AMAX1(E11+(1,-E7)(-E01+(E41+F)K))) 0, P	MODEAT	31
	CRP & AMAX1(AMJI.1(RLMI) ANIMA(AMM3) ARMGJØARMJJØ U\$J	HODRAT	Š
	GRM 8 (11-00+1N51)/10-0-0	HODRAT	30
•••	GRR B TOLK/TOPOPK	NODRAT	41
300	LUNTINUE TEXTOUIST LE TSUETION	HODRAT	41
	1.1.1.4.1	NODRAT	4
	FR17176681		

	PRINT910, GR4, PLCRU, AUX1, AUX3	HODRAT	43
	IF(T.LT.THOD)RETURN	NDDRAT	- 44
	PRINTOSO, PDR, ES, PRS, EXTS, &LHPI, RLHDI, RLHI, PRZ, CRH	HODRAT	45
	PRINT955, H1, R2, XR1, AL4, GRA, GRA	HODRAŤ	46
	RETURN	HÓDRAŤ	47
900	FORMAT(36H20UTPUT OF SUBROUTINE HODRAT AT TIME.F8.2)	NODRAT	48
918	FORMAT(1H0.9X, 3HGRT, 9X, 9H8LCRU, 7X, 4HAUX1, 8X, 4HAUX3/1H0, 9X, 4(E11.4,	HODRAT	49
	1 1833	HODRAT	58
950	FORMAT(1H0,0X,3HPNR,9K,2H83,10X,3HPR1,0X,4HEXT1,0X,5HRLHP147X,	HODRAT	- 51
	1 5HRLHD1, 7X, 4HRLH1, 8X, 3HPR2, 9X, 3HCRH/IHC, 9X, 9(E11, 9, 1X))	MODRAT	52
955	FORMAT(140.0X,2481.10Y,2482,10X,34X81.0X,3HRLM.0X,3HGRM.0X,3HGRA/	HODRAT	53
	1 1HD. 0X. 6(E11.4.1A))	HODRAT	- 54
	ENTRY HODSET	HODRAT	55
	TOLO + TOL	HÓDRAŤ	56
	RETURN	NDDRAŤ	57
	END	HODRAT	58

SURPOUTINE CRIAC		C	,
COMMON /ACCOUNT	INDE AT AND AT APEAT WEAVET WERTET APRIL.	ACCOUNT	
1		ACCOUL	
▲		ACCOUR	3
	Delva, Dean's Wall, Daigna, DCMDT, MCSIDISERFLIA, HERMA	ACCOUN	
	ALPM1. TEC. TCEC, THISL3, TLF, TLFP, CPLF, MINT, MINTL.	ACCOUN	?
•	C4ELL D14, C520, C523, C524, L11, L12, L13, CB210, C3104,	C # 7	1
?	EPC4PJ, PCROPU, PCROPL, PFCROP,	ACCOUN	7
•	ΕΥΓΡΟΊ,ΤΕΟΟΊ,ΥΓΟΟΓ,ΤΕΟΓΟ,ΟΥΓΟΟΊ,ΟΥΓΟΟΓ,ΟΥΓΟΡΟ	ACCOUN	•
REAL NUFR		ACCOUN	•
COMMON /CATTLE/	PFGT,FEHT,PFPT,PHJT,PHPT,TOPOPT,TOPOPA,	CATTLE	2
1	PFUM, PFPM, PMGM, PMPM, TOPOPM, VACAPL, AUXL12,	CATTLE	3
4	SLFERT, SLSCCT, SOLOFT, SLSHLT, SLSPT, SLSPTP, FEHSCT,	CATTLE	4
3	SLEEK", SLSCCH, SOLDEH, SLSHLH, SLSPH, BUPB, FEHSCH.	CATTLE	9
4	PPrGTT. PPPPTT. PPHATT. PPAPTT. PPrGTH.	CATTLE	
5	PPEPTN, PPNGTN, PPHPTH, RESTT, REPTT,	CATTLE	ž
6	RMCTT.WPPTT.RECTH.REPTH.RHCTH.RMPTH.	CATTLE	
7	TOUT. TOWAT. TOWM. TOWAR TOWAR PATADT. ONN. ONT. PELACT.	CATTLE	
à	WHAT BA . PAD. PADD. DDAT CABL LILCADI . CT205 C514. C344	CATTLE	
COMPON CONTRA	T AT AIM IS IN ACTORT, ACTOM BATHIS, BATHIS TOSTAT	CALLE	•••
CUMPON /CONTAC/	I'DITOONTIKONTREGENITEKICHOTEKIATITEHIAFETTEKIAIT		
1			
COMMON /LANN/	TGL/TTGL/TTGLL, TTGLV1, TTGLV2, TGL3/0,	LAND	
1	TESL, TLMOD, TLMOD1, TLMOD2, TLMOD3, TTGLH, TGLR,	LAND	3
>	TLCPL, TLCRU, TLCRUP, TGLSF, TGLU1P, TLFC, TLC,	LAND	•
3	X91,H1,H2,PR2,PDR,CAN,RLH,RLH1,RLCRU,AUX1,AUX3,	LAND	
4	APH1, APH2, ARH3, A5, C9, GRT, CPLPT, XDEL, HPTH, GRE	LANC	6
DIMENSION VAL510	(5),VA_52(7)	CRTACC	7
DATA 6239.0272.0	273, C281, C282 / .1, 1, 1, 1, 2, 6,, .832 /	CRTACC	8
DATA SHALLS1.DIF	F51+K51 / 4++ 1++ 4 /	CRTACC	•
DATA SHALLSZ. DIF	F52.K52 / 4., 1., 6 /	CRTACE	10
DATA VALS1 / 12	77. 13.5. 13.5. 14.5. 15.76 /	CRTACT	11
DATA VAL 52 / 117	1 . 115.5. 127 147 164 143.5. 474. /	CRTACE	12
	// A798. 18 74 /	CRIACE	
DATA ALISCICHTI	/ .V/CO/ 131/0 / NDAR NDMCR / 134 . 884 . 12 . 984 /	CRIACE	1.5
	adia ' ala 19 1	CRIACE	
UATA POPU / 1541		CRIACC	. 12
DATA CREPH, PEUCH	(, 4141C / 1++ +5+ +1 /	CNO	1
CREMD . CHEM		CHTACC	17
CREP . AMAX1(CRE	P9.CDF8, 0.)	CRIACC	18
CINT . PINTC+CDF		CRTACC	19
CDER + CDER+DT+(CLOAN-CHEPP)	CRIACC	20
CCDT = DUTUUS+(H	(INT-R1);[L)+CDEB+(RINT+RINTC)	CRTACC	21
POP . FCP0+FXP(C	(2#2+1)	CRTACC	22
PERCAP = C2+1+(5	SUPB-EXPL-UNEXPL)/POP	CRTACC	23
ARLSK & ARLAT+AF	LAN+CLOAN	CRTACC	24
ACLSK + ACLAT+AC	LAN+CREP+CINT	CRTACC	25
FARILA . FARILA.	DToFARIL	CRTACC	24
CARTI & ARISK-AC	11 Sx	CRTACC	27
DSREVI A DELEVIA	DT+FARIL+EXP(+DIR+T)	CRTACC	28
5 1 5 1 5 1 5 1 5 1 5 1 5 1 5 1 5 1 5 1	DT. FARMI	CRIACO	20
FIRMI - FIRMISF	2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 -	CRTACE	
PANNI E PANIL-PA	1. ANGM 18 - 1 - 1	CRIACC	30
VLANU B VLANUTAV	(Laging	CRIACC	31
AVCAL + AVCALT+		UNTACC	22
VLDTAX + VLATAT+	VLDIX	CHTACC	22
VLANDH + VLAND/C	TEL8+C272)	CRIACC	34
VLDTXH + VLNTAX/	(TGLR+C273)	CRTACC	39
COMEDX . VERTAH/	VLABD4	CRTACC	36
GOVREVA . GOVREV	A+DT+GGVREV	CRTACC	37
GOVREV . TAXCT+T	ATCH+VLUTAX	CRTACÉ	38
EOPOS . VLAND+VA	CARL-CRE8-D8TOUS	CRTACC	39
ACROTC + PEOCH+F	OPC\$	CRTACC	41
CRDAV # APTHICPS	UCR+EQPUS, ACRDTC)	CRTACC	41
NCEP . AMAXI (FAL	11 + C230+FARMIC-EXPLIV. 1.)	CRTACC	42

		TRADE - INCLUSE CARLESTANTANTANTANTANTAN	CRTACE	43
		IDCRDC - AMARICA ARICACESTANICACETELIA		
		CLOAN & AMINI(CRDAY, YDCRDC)	CHIACC	
		EF(T.LT.4.)90 TG 380	CRYACC	45
		IFIT GT ALLCO TO 200	CRTACC	46
		TY (1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	CRTACE	47
		ETCHE & LEDI IL LACOTODIA CALADIA ANDERA	CRTACE	44
		GO TO 295		
	240	EXCHR & EXCHR+DT+9CHEK	CHIACE	
	295	RCHEX & ALI+EXCHRI+EXP(ALI+(T+8.))	CRYACC	
	100	FOREYA & FOREYADDT-FOREE	CRTACC	- 51
			CRTACC	52
			CATACE	93
			CAT.C.C.	
		WPB = TABLIE(VALS7,SMALLSZ,DIFFSZ,KSZ,T)	CRIACL	
		GO TO 320	CHIACE	22
	310	IF(T,GT,14,)G0 TO 315	CRYACC	20
	-	WPA • WPA1+(1,+WPAR+(T+10,))	CRTACC	
		GO TO 320	CRTACC	58
			CRTACE	
	973		CATACE	
		HPB - HPBC/4, J		
	320	IF(T,EQ.7.) #X5UB • ,15	CHIACE	
		EXMAR = WPB+EXCHR+(1.+,0+EXSUB)+PA	CRIACC	50
		FOREX . WPH-EXPL	CRTACC	• 3
			CHTACC	64
			CRTACE	45
			CRIACO	
		SURS = SURSE+CCDI+VC2DA+V48A4		
		IF(IPPINT.LT.1)HETU#N	CHIACE	0/
		PRINT9CC,T	CRIACO	
		PRINT910, CREP, CINT, CDEB, EAPOS, CRDAV, CLOAN, TOCRDC, NGFR, CCDT	CRTAC	69
		PRINT925. FABIL FARILA.FARNI.FARNIA.DSBEVL.ARLSK.ACLSK.POP.PERCAP	CRTACE	70
		PRINTOTE VALCAR' VIAND, VIDTAX, VIANDA, VIDTXH, COMIDX, BOVREY, COVREYA	CRIACO	71
		DELUTAR FLOR FLORE CLOCK OFFICE CYCHO FORCY FABEYL, CYMAR UBB	CRIACE	12
		LALAAC'S CR2'S CR2'S CR2'S CR2'S CHCH'L CHEV'L CHEV'L CHEVYLENUML HE		
		RETURN	CHIACE	/3
	900	FORMAT(36H2OUTPUT OF SUBROUTINE CRIACE AT TIME, F8.2)	CATACC	74
	910	FORMAT(1HD, 9X, 4HCREP.8X, 4HCINT, 8X, 4HCDE8.8X, 5HEOPOS, 7X, 5HCRCAV, 7X,	CRTACC	- 75
	1	SHELDAN, 74, ANTOCHEC, 64, ANNEFR, 88, 4HEFD7/1H3, 88, 9F52, 43	CRTACC	74
		FADMATING AV ALFADIL TT AMFADILA.AX. AMFADMI. TV AMFADILA.AX.	CRTACE	27
	769.	AUDEREN'AN DER SHIELSE SHIELSE STATE SUDOR ANDEREN AUDEREN	CRAACE	7.
		0HJSHEVL:01, 7HERC3H; /X; 3HACC3K; /X; 3HPUP; VA; 0HPCHC8F1140;	CHIRCO	
		8 8 X, 9 E 1 2 . 4)	CHIACC	/ •
	938	FORMAT(1H0, 0X, 6HVALCAP, 6X, 3HVLAVD, 7X, 6HVLCTAX, 6X, 6HVLAVČH, 6X,	CRYACC	• •
	1	6HVLDTXH,6X,6HCOMLDX,6X,6HJOVREV,6X,7HQOVBEYA/1HB,8X,	CRTACC	61
	2	8F12.4)	CRTACE	82
		FORMATISHO. OT. ANSUDS. BY. SHSUBSA. 7X. SHSUBSF. 7X. SHRCHFY. 7Y. SHFYCHR.	CRTACE	
		THE BUCOBER THE AREA AN BUCHAR THE BUCKAR AN AUGRAPHICAL	CRIACE	
-	2	······································	CHIRCE .	
Č			CRTACE	
Ç		INITIAL AVENEZ LOA TAAPPINEMI VCCOAAINO	CHIACC	
С			CRTACC	
		ENTRY CATSET	CRTACC	
		EXSUR a d.	CRTACE	Â
			CATACO	
			CHIACL	
		TANILA U U.	CHIACC	
		FARMI • •.	CRYACC	72
		FARMIA + 0,	CRTACC	93
		GOVREV . D.	CRTACC	
		GOVREVA . A.	CATACC	
			CATACC	
		996576 7 9, 9469 A A	CHIACC .	
		3U83 - 4,	CHIACC	• • 7
		SUESA e D.	CRTACC	
		FOREX • 0.	CRTACC	•••
		FOREXA = 0.	CRTACE	184
		EXCHE = A.	CRTACC	
				144
			CHIACC	169
		UREF T V.	CHITACC	- 183

•

.

CDEB • 0. CLOAN • 0. DWTOUS • 0. RETURN End

CRTACC	104
CRYACE	119
CRTACC	104
CRTACE	107
CATACE	1.14

RUDROUTINE ARAPHETE, TL. TR. TF. VMARATAGUT)	<u>ARAPH</u>	2
1847-70340.40.12	GRAPH	Ī
1017-11180.40.20	BRAPH	Ŭ,
17(7-72)40.46.3	GRAPH	9
1517-15170.45.4	GRAPH	. i
QUT • 0.	GRAPH	7
AFTURN	GRAPH	
OUT . VHAX+(T+TE)/(T1+TA)	GRAPH	•
RETURN	GRAPH	10
OUT . WHAT	GRAPH	11
RETURN	GRAPH	12
QUT . VHAX+(TF-T)/(TF+T2)	GRAPH	13
RETURN	GRAPH	14
END	GRAPH	15
	BUDROUTINE GRAPH(TE,T1,T2,TF,TH,TH,TH,UT,OUT) IF(T-TE)40,40,15 IF(T-T1)90,60,26 IF(T-TF)70,40,40 OUT = 0. BETURM OUT = VMAX=(T-T5)/(T1-T0) RETURM OUT = VMAX=(TF-T)/(TF-T2) RETURM GND	BUDROUTINE BRAPH(TE,TL,TE,TF,TL,TE,TF,TE,TF,E,TE,TE,TE,TE,TE,TE,TE,TE,TE,TE,TE,TE,T

FUNCTION TABLEFVAL STALL DIFF.K.DUMNY)	TAULIE	2
	TAULTE	3
DUM - ANTHICAMAYICDUMMY-SMALL.S.D).FLOAT(K)+DIF	TAULTE	
	TAULIE	5
	TAULIE	6
TABLE - (1) - (1) - (1) - (1)) - (DUM-FLOAT(1-1)) - DIFF)/DIFF+VAL(1)	TABLIË	7
	TABLIE	
FWD	TABLIE	•

•

· .

FUNCTION TARETEIVAL STALL, DIFF. K. DUNNY)	TABEXE	5
	TAREXE	3
	TABEXE	•
	TABEXE	5
T. BEAL (14) (14) (14) (14) (1) (0) (0) (0) (0) (0) (0) (0) (0) (0) (0	TABEXE	•
TAREAL A TARTITATION AND AND AND AND AND AND AND AND AND AN	TADEXE	7
	TAUEXE	

SUBROUTINE DELDT(4)MR.RUUTR.CROJTR.JEL.IDT.DT.K)	DELOT	2
DIMENSION CROUTR(1)	DELDT	3
DEL1 B DEL PLOATETOTY/EFLOATERSODTY	DELOT	
ROUTE A A.	DELDT	Ś
no 2 le1.1hT	DELDT	•
RIN - DINR/FLOAT(INT)	DELDT	7
nin e sinnyr Gransiyir	DELDT	
	DELDT	•
CROUTRETS & ARCAERINGARES/DELS	DELDT	10
(RUUINII) = ANGUININ-NUUIFEESA	DELDT	11
2 80078 - 80078+660178(f)	DELDT	12
	DELDT	13
ne i vna END	DELDT	14

	SUBBOUTINE DELAY (RINR, KOUTR, CROUTR, DEL. DT.K)	DELAY	2
	DIMENSION CROUTR(1)	DELAY	
	DEL 4 - DEL CIEL DATIN ANTI	DELAY	
	DELL - DEL/ITCONTRACTOR	DELAY	
	NIN FRIM	DELAY	1
	DO 10 1-14K	DELAY	,
	ARC . CROUTH(1)	DELAY	
-	CROUTR(I) . ANC.(MIN-ABC)/DELL	DELAY	
10	RIN . APC	DELAY	18
	ROUTR . CROUTR(K)		
	RETURN		12
	END	DELAY	

BUBROUTINE BOADIBIN*.BOUTE.TRAIN.NOOWNT.NOCY.LT.BUHIN DIMENSION TRAIBIL) BUCOUNT * NCOUNTAL BURIN * SUMIN-BIN* If (NCOUNT.NE.NOCY)BETURN BOUTR * TRAIN(L)/FLOAT(NOCY) DO 10 IFF.LT TRAIN(I-L) * TRAIN(L) TRAIN(I-L) * TRAIN(L) TRAIN(LT) * SUMIN SUMIN * 0. NCOUNT * 0. RETURN END) BCXC BCXC BCXC BCXC BCXC BCXC BCXC BCXC
---	--

:08 :08

.

BIBLIOGRAPHY

BIBLIOGRAPHY

- 1. Abkin, Michael. Policy Making for Economic Development: <u>A System Simulation Model of the Agricultural</u> <u>Economy of Southern Nigeria</u>. Staff Paper Series 72-4. East Lansing: Department of Agricultural Economics, Michigan State University, 1972.
- 2. Anderson, A. W. <u>La Industria de Carne en Colombia</u>. Bogota: Ferrocariles Nacionales de Colombia, 1961.
- 3. Banco de la Republica. <u>Revista del Banco de la Republica</u>. Vol. XL (Jan.-Dec., 1967). Bogota.
- 4. Booz, Allen, and Hamilton, Management Consultants. "Meat Packing Industry Program for Colombia." Report to the U. S. Operations Mission to Colombia, Chicago, August 31, 1961.
- 5. Bowser, Max F. "Prerequisitos y Potencial para la Exportacion de Carne en Colombia en la Decada de 1970." <u>Agricultura Tropical</u>, XXV (Nov., 1969), 675-716.
- 6. Caja de Credito Agrario. <u>El Ganado Vacuno en Colombia</u>. Bogota: Departamento de Investigaciones Economicas, 1970.
- 7. <u>Estimativos Sobre Areas Cultivadas y Produccion</u> <u>Obtenida Para los Principales Cultivos. Años:</u> <u>1965-1969</u>. Carta Agraria, No. 244. Bogota, 1970.
- 8. <u>Calculos de Produccion Agricola de 1958 a 1963</u>. Carta Agraria, No. 144. Bogota, 1964.
- 9. Centro Internacional de Agricultura Tropical. "Aspectos de la Ganaderia Vacuna en las Llanuras del Caribe en Colombia," Cali, 1972. (Mimeographed.)
- 10. Chong, Kwong-Yuan. "A Simulation Policy Analysis of the Western Nigerian Cocoa Industry." Unpublished Ph.D. dissertation, Michigan State University, 1973.
- 11. Comision de Reforma Tributaria. Report of the Commission. Bases Para Una Reforma Tributaria En Colombia. Bogota: Banco Popular, 1969.

- 12. Comite Interamericano de Desarrollo Agricola (CIDA). <u>Tenencia de la Tierra y Desarrollo Socio-Economico</u> <u>del Sector Agricola: Colombia</u>. Washington, D. C.: Pan American Union, 1966.
- 13. Currie, Lauchlin. <u>Programa de Desarrollo Economico del</u> <u>Valle del Magdalena y Norte de Colombia (Informe</u> <u>de Una Mision)</u>. Bogota: Ministerio Obras Publicas, 1960.
- 14. Departamento Administrativo Nacional de Estadistica.
 <u>Directorio Nacional de Explotaciones Agropecuarias</u> (Censo Agropecuario) 1960: Resumen Nacional, Segunda Parte. Bogota, 1964.
- 15. <u>Directorio Nacional de Explotaciones</u> <u>Agropecuarias (Censo Agropecuario) 1960:</u> <u>Departamentos de Atlantico, Bolivar, Cordoba y</u> Magdalena. Bogota, 1964.
- 16. <u>Pronosticos y Estimaciones Agricolas 1971</u>. Bogota, 1971.
- 17. <u>XIII Censo Nacional de Problacion (Julio 15</u> de 1964). Bogota, 1967.
- 18. <u>Encuesta Agropecuaria Nacional 1968</u>. Bogota, 1970.
- 19. Departamento Nacional de Planeacion. "Plan de Desarrollo Economico y Social 1970-1973," Tomo I, Bogota, 1970. (Mimeographed.)
- 20. Federacion Colombiara de Ganaderos. La Rentabilidad Ganadera en el País. Bogota, 1971.
- 21. Forrester, Jay W. <u>Industrial Dynamics</u>. Cambridge, Massachusetts: The Massachusetts Institute of Technology Press, 1961.
- 22. Fromm, Gary. "Implications to and from Economic Theory in Models of Complex Systems." <u>American Journal</u> of Agricultural Economics, 55 (May, 1973), 259-71.
- 23. Garcia Samper, Alfredo. <u>Perspectivas de Colombia en el</u> <u>Mercado Internacional de Carne de Res</u>. Bogota: <u>Centro de Investigaciones para el Desarrollo</u>, Universidad Nacional de Colombia, 1970.
- 24. Gomez, Luis Jair. Aspectos Reproductivos en Rabaños Bovinos Tipo Carne de Colombia. Publicaciones Tecnicas No. 1. Medellin: Fondo Ganadero de Antioquia, 1968.

- 25. Gresford, Guy B. "Systems Approach for Development." <u>IEEE Transactions on Systems, Man, and Cybernetics</u>, <u>SMC-2</u> (July, 1972), 311-18.
- 26. Griffin, Keith B., and Enos, John L. <u>Planning Development</u>. London: Addison-Wesley Publishing Co., 1970.
- 27. Hamilton, H. R., <u>et al.</u> <u>Systems Simulation for Regional</u> <u>Analysis--An Application to River-Basin Planning</u>. Cambridge, Massachusetts: The Massachusetts Institute of Technology Press, 1969.
- 28. Hayenga, M. L., Manetsch, T. J., and Halter, A. N. "Computer Simulation as a Planning Tool in Developing Economies." <u>American Journal of</u> <u>Agricultural Economics</u>, 50 (December, 1968), 1955-59.
- 29. Henning, Robin G. "Ganado y Carne, Colombia: Analysis de la Produccion y Prospectos para Exportacion," Bogota, Ministerio de Agricultura, 1971. (Mimeographed.)
- 30. Hirschman, A. O. Journeys Toward Progress. New York: Twentieth Century Fund, 1963.
- 31. Instituto Colombiano Agropecuario. "Proyecto de Sanidad Animal: Combate de la Fiebre Aftosa y Control de la Brucellosis-Primera Etapa 1971-75," Bogota, Nov., 1970. (Mimeographed.)
- 32. _____. "Silos y Ensilaje." Technical Bulletin No. 8, ______1-16.
- 33. "Analysis Quimico en Base Seca, de Gramineas y Leguminosas Adaptadas a las Condiciones de Colombia." Informacion Preparada para los Programas de Pastos y Forrajes, Nutricion Animal y Suelos, 1968.
- 34. Instituto Colombiano de la Reforma Agraria. "Productividad de la Ganaderia de Cria y Consecuencias Sobre los Programas de Credito Ganadero," Bogota, 1970. (Mimeographed.)
- 35. Instituto de Comercio Exterior. <u>Bases para un Plan</u> <u>Cuatrienal de Exportaciones de Productos</u> <u>Seleccionados, 1971-1974-Resumen</u>. Documento para Comision Mixta de Comercio Exterior, Bogota, 1971.

- 36. Instituto de Fomento Algodonero. <u>Estudios Preliminares</u> <u>de Suelos del Departamento de Cordoba</u>, by L. F. Irusta and E. A. Fortoul. Bogota, 1959.
- 37. <u>Estudios Preliminares de Suelos del Departamento</u> <u>del Magdalena y la Intendencia de la Guajira, by</u> L. F. Irusta and E. A. Fortoul. Bogota, 1957.
- 38. Instituto Geografico "Augustin Codazzi." <u>Levantamiento</u> <u>Agrologico del Departamento del Atlantico</u>. Bogota: Departamento Agrologico, 1960.
- 39. _____. <u>Atlas de Colombia</u>. 2nd ed. Gobota: Litografia Arco, 1969.
- 40. Inter-American Committee for Agricultural Development (CIDA). Inventory of Information Basic to the Planning of Agricultural Development in Latin America--Colombia. Washington, D. C.: Pan American Union, 1964.
- 41. International Bank for Reconstruction and Development. <u>Economic Growth of Colombia: Problems and Prospects</u>. Baltimore: The John Hopkins University Press, 1972.
- 42. "Second Livestock Development Project: Colombia." Appraisal report of a Bank's mission to Colombia, Washington, D. C., September 3, 1969. (Mimeographed.)
- 43. "Use of Simulation in Appriasing a Livestock Breeding/Fattening Project." Economics Department Working Paper No. 56, Washington, D. C., January 1970. (Mimeographed.)
- 44. International Monetary Fund. International Financial Statistics. Vol. 20, (December, 1967). Washington, D. C.
- 45. <u>International Financial Statistics</u>. Vol. 26, (April, 1973). Washington, D. C.
- 46. Johnson, Glenn.. "Alternatives to the Neoclassical Theory of the Firm." American Journal of Agricultural Economics, 54 (May, 1972), 295-303.
- 47. "General, Systems-Science, Simulation Analysis--An Introduct on," Michigan State University, 1973. (Mimeographed.)
- 48. Johnson, Glenn L., <u>et al</u>. "A Simulation Model of the Nigerian Agricultural Economy: Phase I--The Northern Nigerian Beef Industry," Michigan State University, 1968. (Mimeographed.)

- 49. Johnson, Glenn, and Zerby, Lewis. What Economists <u>Do About Values: Case Studies of Their Answers</u> to Questions They Don't Dare Ask. East Lansing: Michigan State Agricultural Experiment Station, 1972.
- 50. Ladipo, Olasupo. "General System Analysis and Simulation Approach: A Preliminary Application to Nigerian Fisheries." Unpublished Ph.D. dissertation, Michigan State University, 1973.
- 51. Lehker, J. N., and Manetsch, T. J. <u>Systems Analysis of</u> <u>Development in Northeast Brazil: The Feasibility</u> <u>of Using Simulation to Evaluate Alternative</u> <u>Systems of Beef Production in Northeast Brazil</u>. Technical Report G-13, Division of Engineering Research, Michigan State University, 1971.
- 52. Llewellyn, Robert W. FORDYN--An Industrial Dynamics Simulator. Raleigh, North C.rolina: By the Author, Box 5353, 1965.
- 53. Manetsch, T. J., <u>et al.</u> <u>A Generalized Simulation</u> <u>Approach to Agricultural Sector Analysis with</u> <u>Special Reference to Nigeria</u>. East Lansing: <u>Michigan State University</u>, 1971.
- 54. Manetsch, T. J., and Park, G. L. <u>System Analysis and</u> <u>Simulation with Applications to Economic and Social</u> <u>Systems--Part II</u>. East Lansing: Department of Electrical Engineering and System Science, Michigan State University, 1973.
- 55. Miller, S. F., and Halter, A. N. "Systems Simulation in a Practical Policy-Making Setting: The Venezuelan Cattle Industry," <u>American Journal</u> of Agricultural Economics, 55 (August, 1973), 420-32.
- 56. Ministerio de Agricultura. <u>Proyecto de ley por la Cual</u> se Introducen Modificaciones a las Leyes 200 de <u>1936, 135 de 1961 y la de 1968, se Crea la</u> Jurisdiccion Agraria y se Establecen Disposiciones sobre Renta Presuntiva. Bogota: Banco Ganadero, 1971.
- 57. _____. "Estudio para la Produccion de Maiz de Exportacion en la Costa Atlantica," Medellin, 1970. (Mimeographed.)

- 58. "Programas Agricolas para 1972," Bogota, 1972. (Mimeographed.)
- 59. ______ Informe del Comite Evaluador de la Reforma Agraria. Bogota: Banco Ganadero, 1971.
 - 60. Oven, Roderick von. "Factors Affecting the Beef Cattle Industry's Return on Capital and Development Possibilities in Selected Areas in South America," Institute of Agricultural Production Economics, University of Goettingen, West Germany, 1968. (Mimeographed.)
 - 61. Riley, Harold M. <u>Beef Production in Colombia</u>. Palmira: Facultad de Agronomia, Universidad Nacional de Colombia, 1962.
 - 62. Rossmiller, G. E., <u>et al. Korean Agricultural Sector</u> <u>Analysis and Recommended Development Strategies</u>, <u>1971-1985</u>. East Lansing: Michigan State University, 1972.
 - 63. Sarmiento, Hector. "Una Decada en la Industria de Ganado Vacuno para Carne en Colombia," Agencia para el Desarrollo Internacional, Bogota, 1972. (Mimeographed.)
 - 64. Secretaria de Agricultura de Antioquia. <u>Sacrificio de</u> <u>Vacas en Antioquia</u>, by Luis Jair Gomez. <u>Publicacion Especial No. 75. Medellin, 1966</u>.
 - 65. Shapiro, Harold T. "Is Verification Possible? The Evaluation of Large Econometric Models." <u>American Journal of Agricultural Economics</u>, 55 (May, 1973), 250-58.
 - 66. United Nations Food and Agriculture Organization. Livestock in Latin America, Status, Problems and Prospects: Colombia, 1962.
 - 67. United States Department of Agriculture. Economic Research Service and Ministry of Agriculture and the Central Planning Agency of Colombia Cooperating. <u>Changes in Agricultural Production and Technology</u> <u>in Colombia</u>, by L. Jay Atkinson. Foreign Agricultural Economic Report No. 52. Washington, D. C.: Government Printing Office, 1969.

- 68. United States Department of State. Agency for International Development. Bureau for Latin America. <u>Colombia Agriculture Sector</u> <u>Analysis</u>, Working Document Series. Washington, D. C., 1970-72.
- 69. <u>Colombia Agriculture Sector Analysis</u>. <u>Agricultural Production Cost Data, Averages by</u> <u>Wage Rate Regions--1970</u>, by S. R. Daines and G. V. Poynor. Working Document No. 1A. Washington, D. C., 1971.
- 70. <u>Colombia Agriculture Sector Analysis</u>. <u>Preliminary Analysis of Costs of Employment</u> <u>Generation in Agriculture</u>, by S. R. Daines. <u>Working Document No. 2A</u>. Washington, D. C., 1971.
- 71. <u>Colombia Agriculture Sector Analysis. Data</u> and Preliminary Analysis of Livestock Production, by D. R. Steen and S. R. Daines. Working Document No. 12. Washington, D. C., 1971.